Combining qualitative and geo-spatial approaches to explore older adults' lived experiences of ageing – *as well as they can* – in place

Presented by

Hannah Grove BA MSc SPHeRE Programme PhD Scholar

A dissertation submitted in partial fulfilment of the requirements

for the degree of

Doctor of Philosophy

in Geography

Faculty of Social Sciences

Maynooth University

July 2022

Supervised by

Professor Gerry Kearns and Associate Professor Ronan Foley

Funded by the Health Research Board (HRB), Ireland (SPHeRE/2013/1)

Head of Department

Dr. Helen Shaw

This page has been left intentionally blank.

Abstract

This thesis explores the lived experience of older people within the Greater Dublin Area of Ireland, to determine how they define and enact ageing well in place. It combines qualitative and spatial methods, including preliminary focus groups (Stage 1), as well as in-depth interviews, mapping exercises, and go-along interviews with older people in two contrasting Study Areas (Stage 2). Results provide participant accounts of the importance of getting out and about, showing how this varies between participants and how it fluctuates over time. Findings reveal why getting out and about is valued by participants, focusing on the importance of connecting with others, and presents both positive and negative experiences of social interaction. Results demonstrate the places or phases where these interactions occur or do not occur, the nature of these interactions and relationships, as well as how older adults develop schedules and routines to maximise opportunities to interact with others. Drawing in depth on four individual lifeworlds with health and mobility challenges, geo-narratives and annotated maps are presented to highlight how getting out and about is navigated and negotiated, based on dynamic personal and environmental contexts, as well as what matters most to that individual for a good quality of life. By integrating and empirically grounding ageing in place and ageing well, this thesis produces a lay and relational conceptual framework of ageing - as well as you can in place, which emphasises the need for pragmatic and subjective definitions of ageing well and recognises the importance of engaging beyond the home to feel and be 'in place' to connect with others. Finally, some recommendations are offered to policymakers, which could support older people not just to age in place, but to age as well as they might want.

This page has been left intentionally blank.

Table of Contents

Abstract	i
Table of Contents	iii
List of Figures	xi
List of Tables	xiii
Acknowledgements	XV
Chapter 1. Introduction	1
1.1 Thesis Overview	1
1.1.1 Thesis Aim and Research Questions	2
1.2 Ireland: A great country (for some) in which to grow old?	4
1.2.1 An Overview of Ageing Policy in Ireland	4
1.2.2 Geographic and Demographic Trends Older People in Ireland	9
1.2.3 Health Characteristics of the Older Adult Population	13
1.2.4 Summary	16
1.3 Thesis Contributions	18
1.4 Conclusion	20
1.4.1 Structure of this thesis	21
Chapter 2. Ageing Well in Place: A Conceptual Framework	25
2.1 Introduction	25
2.2 Ageing in Place	25
2.2.1 Summary	29
2.3 Ageing Well	29
2.3.1 A Capability Approach to Ageing Well	34
2.4 Ageing Well in Place through a Capability Approach lens	36

2.5 Conclusion
Chapter 3. The Importance of Getting Out and About for Older People: A Literature Review
3.1 Community mobility and its importance for older adult wellbeing and quality of life
3.1.1 Mobility and Wellbeing44
3.1.2 Mobility and Independence
3.1.3 The importance of mobility for social interaction and connectedness47
3.2 Places of Importance for Social Interaction
3.2.1 Social Interaction Whilst Travelling
3.2.2 Destinations of Importance for Social Interaction: The Role of 'Third Places'
3.2.3 Nature of Social Interactions and Relationships
3.3 Navigating the dynamic challenges that can be associated with getting out and about
3.4 Conclusion
Chapter 4. Methodology73
4.1 Introduction
4.2 Philosophical Underpinnings of the Thesis73
4.2.1 Ontological and Epistemological Assumptions74
4.2.2 Methodological Approach78
4.3 Study Design and Methods
4.3.1 Stage 1: Exploratory Focus Groups
4.3.2 Stage 2: Main Study
4.4 Sampling, Recruitment and Study Setting
4.4.1 Sampling Strategy and Recruitment Procedures
4.4.2 Study Settings

5.4.2	Engaging with 'meaningful' activities or interactions	32
5.4.3	Summary	34
5.5 Tem	porary fluctuations in getting out and about1	34
5.5.1	Disruptions to health	35
5.5.2	Weather and seasonal variations	38
5.5.3	Summary	39
5.6 Con	clusion14	40
Chapter	6. Getting Out and About to Connect with Others14	45
6.1 Intro	oduction14	45
6.2 Pha	ses of Getting Out and About14	45
6.2.1	Phase 1: Thresholds14	46
6.2.2	Phase 2: Transitory Zones	46
6.2.3	Phase 3: Destinations or Third Places	47
6.3 Hov	v did participants connect to others whilst out and about? 14	49
	v did participants connect to others whilst out and about?	
6.3.1		49
6.3.1 6.4 Pha	Continuing Bridie's Story	49 54
6.3.1 6.4 Phas 6.4.1	Continuing Bridie's Story	49 54 55
6.3.1 6.4 Phas 6.4.1 6.4.2	Continuing Bridie's Story	49 54 55 60
6.3.1 6.4 Phas 6.4.1 6.4.2 6.4.3	Continuing Bridie's Story	 49 54 55 60 61
6.3.1 6.4 Phas 6.4.1 6.4.2 6.4.3 6.5 Phas	Continuing Bridie's Story	 49 54 55 60 61 61
 6.3.1 6.4 Phase 6.4.1 6.4.2 6.4.3 6.5 Phase 6.5.1 	Continuing Bridie's Story 1 se 1: Crossing the Threshold 1 Perceptions of Neighbours 1 Moving beyond Thresholds 1 Summary of Phase 1 1 se 2: Moving Through Public Spaces 1	 49 54 55 60 61 61 63
 6.3.1 6.4 Phase 6.4.1 6.4.2 6.4.3 6.5 Phase 6.5.1 6.5.2 	Continuing Bridie's Story 14 se 1: Crossing the Threshold 1 Perceptions of Neighbours 1 Moving beyond Thresholds 1 Summary of Phase 1 1 se 2: Moving Through Public Spaces 1 Guaranteed Social Interaction 1	 49 54 55 60 61 61 63 65
 6.3.1 6.4 Phase 6.4.1 6.4.2 6.4.3 6.5 Phase 6.5.1 6.5.2 6.5.3 	Continuing Bridie's Story 1 se 1: Crossing the Threshold 1 Perceptions of Neighbours 1 Moving beyond Thresholds 1 Summary of Phase 1 1 se 2: Moving Through Public Spaces 1 Guaranteed Social Interaction 1 Having a "High Profile" 1	 49 54 55 60 61 61 63 65 68
 6.3.1 6.4 Phase 6.4.1 6.4.2 6.4.3 6.5 Phase 6.5.1 6.5.2 6.5.3 6.5.4 	Continuing Bridie's Story 1 se 1: Crossing the Threshold 1 Perceptions of Neighbours 1 Moving beyond Thresholds 1 Summary of Phase 1 1 se 2: Moving Through Public Spaces 1 Guaranteed Social Interaction 1 Having a "High Profile" 1 Non-Human Interactions 1	 49 54 55 60 61 61 63 65 68 69

6.6 Phase 3: At the Destination	
6.6.1 Attending Social Groups	
6.6.2 Attending Mass	
6.6.3 Additional or Extended O	outings and Interactions (Phase 3+)189
6.6.4 Getting Out and About w	ith Purpose192
6.6.5 Using Destinations to Stru	ucture the Week194
6.6.6 Summary of Phase 3	
6.7 Conclusion	
Chapter 7. Navigating ageing - as v	well as you can - in place with health and mobility
challenges	
7.1 Overview of the Chapter	
7.2 Edith	
7.2.1 Navigating personal proje	ects based on existing and changing capacities
7.2.2 Summary	
7.3 Eamon	
7.3.1 The importance of routine	206
7.3.2 Summary	
7.4 Moira	
7.4.1 Challenges with maintain	ing independence209
7.4.2 Summary	
7.5 Dolores	
7.5.1 Trip chaining and negotia	ting finite energy budgets215
7.5.2 Variability across persona	al and environmental factors218
7.5.3 When challenges become	insurmountable: making mistakes and mustering
through	

7.5.4 Summary
7.6 Conclusion
Chapter 8. Discussion and Conclusion
8.1 Introduction
8.2 Ageing – as well as you can – in Place
8.2.1 A pragmatic and subjective definition of health and ageing well
8.2.2 A Capability Approach lens: emphasis on achieving valued functionings
8.2.3 The importance of being and feeling - in place: getting out and about232
8.2.4 Connecting with others: Enacting relational and relative interdependence
8.2.5 Summary
8.3 Recommendations: how can geographers and planners support older people to age – as well as they can – in place?
8.3.1 'Person-in-Environment' Interventions for Ageing Well in Place
8.3.2 Prioritising Health and Wellbeing in Urban Planning and Design
8.4 Reflecting on the Challenges and Limitations of this Research
8.4.1 Carrying out go-along interviews
8.4.2 Sampling and Recruitment
8.4.3 Covid-19
8.5 Conclusion
References
Appendices
Appendix 1. Philosophical and Theoretical Underpinnings of the Thesis
Appendix 2. Stage 1: List of Questions
Appendix 3. Stage 2: Interview, Mapping Exercise and Go-Along Interview Schedule

Appendix 4. Stage 1: Social Group Details
Appendix 5. Stage 1: Invitation Letter
Appendix 6. Stage 1: Information Sheet
Appendix 7. Stage 1: Consent Form
Appendix 8. Stage 2: Social Group Details
Appendix 9. Stage 2: Invitation Letter
Appendix 10. Stage 2: Information Sheet
Appendix 11. Stage 2: List of organisations that did not take part and reasons for
this
Appendix 12. Stage 1: Ethical Approval Letter
Appendix 13. Stage 2: Ethical Approval Letter
Appendix 14. Stage 2: Consent Form (Standard)
Appendix 15. Stage 2: Consent Form (Adapted for Cognitive Impairment)
Appendix 16. Chapter 5 Data Analysis: Thematic Map for Defining a Good Quality of Life
Appendix 17. Chapter 6 Data Analysis: Categories of Social Infrastructure Places
by Participant Case

This page has been left intentionally blank.

List of Figures

Figure 1.1 Map of Greater Dublin Area
Figure 1.1 The Eight Domains of an Age Friendly Community (Source: Age Friendly Ireland, 2021a)
Figure 1.3 Environmental Determinants of Health and Wellbeing (Source: Barton and Grant, 2006, p.2)14
Figure 1.4 Dichotomous Life Pathways (Source: Pretty et al., 2009, p.24)15
Figure 3.1 Restorative City Framework (Source: Roe and McCay, 2021, p.13)53
Figure 4.1 Study Design Diagram80
Figure 4.2 Diagrammatic Representation of Study Area Settings within the Greater Dublin Area
Figure 4.3. Screenshot of NVivo Database
Figure 4.4. Coding Tree showing the variation in participants' ability and 'need' to get out and about (GOA)
Figure 4.5. Screenshot of Initial Coding of Places of Importance
Figure 4.6. Screenshot of Initial Coding Activities of Importance104
Figure 4.7. Flow Diagram of the Development of Data Analysis for Chapter 6 105
Figure 4.8. Coding Tree for Chapter 6106
Figure 5.1 Empirically Grounded Conceptual Model of Ageing – as well as you can – in Place
Figure 6.1 Bridie's Places and Routes of Importance and Map Key153
Figure 6.2 Anne's Concerns within Phase 1
Figure 6.3 Venues for Social Groups177
Figure 6.4 Summary of Participants' Engagement with Mass and Church
Figure 7.1 Edith's Lifeworld and Map Key202
Figure 7.2 Eamon's Lifeworld

Figure 7.3 Moira's Lifeworld	
Figure 7.4 Dolores' Lifeworld	
Figure 7.5 Dolores' Go-Along Route	
Figure 8.1 Ageing – as well as you can – in Place	

List of Tables

Table 1.1 Older Adult Population Change in the Greater Dublin Area 2011-2010	5
(Source: CSO, 2017)	11
Table 4.1 Participant Characteristics	88
Table 4.2. Summary of adaptation of research methods	96
Table 4.3. List of Participant Case IDs and Pseudonyms	98
Table 4.4. Summary of Categories of Social Infrastructure Places	105
Table 6.1 Social Group Details	178

This page has been left intentionally blank.

Acknowledgements

This thesis is the result of so much support and encouragement behind the scenes, without which, it would never have been achieved.

First of all, thank you to my supervisors Professor Gerry Kearns and Dr. Ronan Foley for your support, encouragement, and expertise. Thank you for being patient when I got stuck, for always meeting me where I was at, and for helping me to move forward and continue progressing one step at a time. I always feel clearer and more motivated after our supervisory meetings and for that I am eternally grateful. Thank you to the SPHeRE Programme and the Health Research Board (HRB) for funding this PhD research and making it a possibility. I am extremely grateful for the opportunities and experiences that being part of this network provided. Thank you to Professor Jan Rigby and Dr. Catriona Murphy for your guidance during the formative and data collection stages of this thesis. Thank you to Dr. Ben Meehan for your support with NVivo during the analysis phase and to Catherine Hayes for your help with the coding. Thank you to Professor Mary Gilmartin, Dr. Carlos Bruen, Dr. Darren Clarke, and Dr. Adeline Cooney for your support and feedback. Thank you to members of ROSEnet for the learning experience whilst being a PhD Forum member, particularly to Professor Kieran Walsh for helping me to navigate the world of gerontology when I was new to it.

Those that know me will know that I work best in collaborative and supportive environments. I took every opportunity to write collectively with others during my PhD. I would like to thank everyone that I have written with for your accountability, which helped to keep me focused, build good writing habits, and not feel so isolated. In particular, I would like to thank Aoife G. and Marita for the Achill Writing Retreats and to Catherine for the Tipperary ones. Thank you to members of the Maynooth Geography Writing Accountability Group, the Maynooth Geography Writing Group, the Thriving Researcher Group, the UK Virtual Writing Retreats (VWR) group, and the Maynooth Writing Centre for the regular online writing sessions and on-going support. These were such a lifeline during the pandemic and really helped me to keep going. Thank you to the Geography Department at Maynooth for being such a brilliant and welcoming department, and especially to Neasa for being the all-knowing glue that holds us all together. Thank you for the coffee mornings, the laughter, and for making Ireland feel like a home away from home. A special thank you goes to all of the amazing Geography PhD students but particularly to those in Room 15. This includes Aoife D., Aoife K., Caoilfhionn, Caspar, Catherine, Darren C., Dean, Georgina, Jack, Joe, Lorna, Louise (and baby Faye), Mary B., Martha, Michael, Padraig, Rachel, Sasha & Will. Thank you for your feedback, accountability, and check-ins. But equally, thank you for the chats, walks, lunches, coffees, outings, escape rooms, yoga, games nights, dinners (especially tacos) and drinks in the Roost! Thank you to the wonderful SPHeRE Programme cohort for your on-going support.

Thank you to my amazing friend Leeya for being my 'Sam', as we navigated our respective 'Mount Dooms' - aka - writing up during lockdown. The fact I am writing this hopefully means the eagles will be arriving very soon! Thank you to my brilliant partner Eoin for being with me every step of the way and for all of your support, I simply could not have done this without you. Thank you to Sara for supporting me to navigate writing my thesis in a way that worked for me, and to Mairéad and Mairéad for your guidance. Thank you to my neighbour Emma for the walks and talks around campus. Thank you to my sister Becky for your encouragement and feedback. Thank you to my Mum for not letting me quit when it felt too much. Thank you to my friends and family for your messages of support, they really meant a lot – a special thank you to my neighbour Emma for the Sky'! Thank you to my cat Felix, for your company during lockdown. Even though you were asleep for most of it, I appreciated you being there all the same.

Finally, and most importantly of all, I would like to thank my participants and the social groups that took part in this research, who gave up their time, and without whom, there would be no empirical data and no thesis. Thank you for your warmth, acceptance, good humour, and of course, the tea and the chats. Your experiences kept me company during lockdown and I think back fondly to your interviews and focus groups. I learnt so much from you all, it was a pleasure to meet you, and an honour to hear your stories. I hope I have done them justice throughout.

This thesis is dedicated to my late grandma Esme

- *it was with you in mind that I wrote this.*

Also to my colleagues Matt from the Geography Department

and Siobháin from the SPHeRE Programme,

who for separate tragic reasons,

did not get to complete PhDs of their own

- you can both share this one with me.

This page has been left intentionally blank.

Chapter 1. Introduction

1.1 Thesis Overview

Despite policy initiatives in Ireland emphasising the importance of 'ageing in place', the variation in the everyday realities of older people and the influence this has on their *quality* of experience, has received far less attention. This thesis critically examines the concept of ageing in place, by bringing to it a health geographical understanding of environment, health, and wellbeing. Alongside this, I have incorporated an understanding of 'ageing well' that is subjective and informed by Sen's (1993) Capability Approach, where quality of life is determined by the extent to which individuals can pursue valued 'functionings'.

Informed by existing literature and theories from Health Geography, Healthy Urban Planning, Geographical Gerontology and Environmental Gerontology, I argue that an important indicator of how well someone may be ageing in place, is the ease with which they are able to engage and participate with their broader neighbourhood environment. However, it is also recognised that this is easier for some than others, depending on an individuals' personal and environmental 'fit' or congruence (Lawton and Nahemow, 1976). Where individuals find it more challenging to get out and about, their ability to engage in meaningful interactions and activities may be diminished and their valued routines may have to be curtailed, which can impact their health and wellbeing.

Increasingly, academic literature is recognising the value of 'lay' perspectives and the need to learn from and understand older peoples' everyday experiences of ageing. Within this thesis, I am interested in what matters to older people who live at home, and how they define a good quality of life based on what is of most importance to them. Informed by existing research that highlights the wellbeing benefits that can arise from being outdoors and mobile, I am interested in the importance of 'getting out and about' to older people, which incorporates leaving the home, travelling, attending destinations, as well as the activities and interactions carried out along the way. I am interested in individuals' most valued places, routines, and activities, as well as how easy it is for them to engage with these. This will provide insight into how ageing well in place is both defined and enacted by

1

older people themselves. These findings will be of use to both policymakers and planners, to design places and service provisions that support older people not just to age in place, but to age *well* in place.

1.1.1 Thesis Aim and Research Questions

The overall aim of this thesis is:

to explore older adults' everyday lived experiences to determine how they define and enact ageing well in place.

This thesis has three broad research questions:

- 1. What is of most importance to older people for a good quality of life and to age well in place?
- 2. What places, routes, routines, and interactions outside of the home are most valued by participants?
- 3. What personal and environmental factors influence the ease with which participants can 'get out and about' to engage in meaningful activities and interactions?

To answer these research questions and to explore the subjective and lay understandings of ageing in place and ageing well, I carried out multi-staged and multi-method empirical research in the Greater Dublin Area (see Figure 1.1) that combined qualitative and spatial approaches. Stage 1 included four exploratory focus groups, carried out with thirty-one participants in June 2017 to familiarise myself to the ageing in place experience in Ireland. Stage 2 then focused on individual experiences and involved interviews, 'go-along' interviews and mapping exercises with thirty-four older people between December 2017 and August 2018 in two contrasting Study Areas. Informed by environmental, health and ageing demographic patterns within Ireland (discussed further in Section 1.2), this research was carried out in a suburban and inner-suburban setting. The next section (Section 1.2) of this chapter provides a brief overview of the ageing policy context of Ireland to situate this thesis, before highlighting some of the relevant urban processes, demographic trends and health contexts that were examined for this research. In Section 1.3, I summarise the key findings and identify the empirical, methodological and theoretical contributions of this thesis. I conclude this chapter in Section 1.4 by outlining the structure of the remainder of this thesis.



Figure 1.1 Map of Greater Dublin Area

1.2 Ireland: A great country (for some) in which to grow old?

The Republic of Ireland is a relatively 'young' country by EU standards, with one of the lowest proportions of people aged 65 years across the EU (Department of Health, 2013). However, Ireland's ageing population is growing. According to the latest population figures from April 2021, of a total population of 5,011,500, there were 742,300 people aged 65 and over, an increase of 112,500 (17.9%) since 2016 (CSO, 2021). By 2051, the number of people aged 65 and over is predicted to increase to between 1.53 million and 1.6 million (depending on different migration scenarios) (CSO, 2018). The number of those aged 65 as a proportion of the total population will increase from 1 in 7 to 1 in 4 (Institute of Public Health, 2020). Furthermore, it is predicted that the population aged over 80 will increase substantially, from 147,800 in 2016 to between 535,900 and 549,000 by 2051 (CSO, 2018).

1.2.1 An Overview of Ageing Policy in Ireland

Within this section, I focus on three key topic areas within ageing policy, which are of relevance for this thesis: ageing in place, ageing well, and age friendly environments. I explore two policy agendas in particular: the National Positive Ageing Strategy (NPAS) and Age Friendly Ireland.

National Positive Ageing Strategy

In 2013 a *National Positive Ageing Strategy* (NPAS) was developed by the Department of Health, setting out the vision for ageing and older people in Ireland. The overall Vision Statement for the strategy is that:

Ireland will be a society for all ages that celebrates and prepares properly for individual and population ageing. It will enable and support all ages and older people to enjoy physical and mental health and wellbeing to their full potential. It will promote and respect older people's engagement in economic, social, cultural, community and family life, and foster better solidarity between generations. It will be a society in which the equality, independence, participation, care, self-fulfilment and dignity of older people are pursued at all times (Department of Health, 2013, p.3). The strategy emphasised that later life could provide many opportunities and be a positive time for older people:

later life can and should be a time for active citizenship, for continued contribution and participation in local community affairs, for engaging in the kinds of activities that enhance physical and mental health, and a time for involvement with family, friends, neighbours and the wider community (Department of Health, 2013, p.5).

The strategy promoted a "positive societal approach to population ageing", based on the concept of "intergenerational solidarity" (p.6) and recognised that wider determinants can influence the ability of older people to age well and adopted a life course perspective to examine this. Throughout the strategy there are references to a variety of types or forms of optimal ageing. However, it recognised that whilst there are many different types of ageing well, such as healthy, active, positive, productive and successful, what was common across these concepts is the idea that older people are:

capable of living a self-reliant life, successfully compensating for losses, contributing to the public good, helping themselves and others, as well as striving for positive fulfilment through meaningful engagement (p.6).

The NPAS was underpinned by both the World Health Organization's (2002) *Active Ageing: A Policy Framework* from 2002, which has three key pillars of participation, health and security, as well as the United Nation's (1991) *Principles for Older Persons*, which emphasises the importance of independence, participation, care, self-fulfilment, and dignity. The NPAS identified four national goals which were aligned directly with the *Active Ageing Framework* (p.19):

- National Goal 1: Remove barriers to participation and provide more opportunities for the continued involvement of people as they age in all aspects of cultural, economic and social life in their communities according to their needs, preferences and capacities.
- National Goal 2: Support people as they age to maintain, improve or manage their physical and mental health and wellbeing.
- National Goal 3: Enable people to age with confidence, security and dignity in their own homes and communities for as long as possible.

A fourth national goal, recognised the need for more research about people as they age, to better inform policy (p.19):

• National Goal 4: Support and use research about people as they age to better inform policy responses to population ageing in Ireland.

Within each national goal, several objectives were identified. Many of these related to the broader local environment, recognising that the built environment can be a key determinant of health and wellbeing. Four objectives had strong place-based components, which this thesis speaks to. The first two objectives were linked to National Goal 1, whilst the third and fourth objectives were linked to National Goal 3 (pp.20–21):

- Promote the development of opportunities for engagement and participation of people of all ages in a range of arts, cultural, spiritual, leisure, learning and physical activities in their local communities.
- Enable people as they age 'to get out and about' through the provision of accessible, affordable, and flexible transport systems in both rural and urban areas.
- Support the design and development of age friendly public spaces, transport and buildings.
- Continue to implement An Garda Síochána Older People Strategy and empower people as they age to live free from fear in their own homes, to feel safe and confident outside in their own communities, and support an environment where this sense of security is enhanced.

An Garda Síochána (2010)'s *Older People Strategy* featured prominently in the NPAS, which was developed in response to concerns about the fear of crime amongst older people in Ireland. This recognised the impact perceived fear can have on older people's quality of life by reducing mobility and activity within their communities, leading to the potential isolation and social exclusion of older people. The strategy provided a commitment to older people to support them through four objectives (p.10):

• Develop and maintain effective communication links between Gardaí and older people

- Deliver a timely and effective response by An Garda Síochána for older people
- Increase trust and confidence by lessening the fear of crime amongst older people
- Determine and respond to the needs and expectations of older people on an ongoing basis

A strength of the NPAS was the awareness that ageing in place is more than just being in one's own home but that it includes engaging meaningfully with the broader community and neighbourhood as part of this. It recognises that environmental disparities can impact the experience of ageing in place and prevent older people from being able to engage and participate beyond the home, with subsequent implications for their health and quality of life.

Age Friendly Cities and Communities Programme

Alongside the NPAS, a key international policy initiative which has focused on the community setting in which older people live, is the World Health Organization's (2007a) Age Friendly Environment movement. The Age Friendly Cities and Communities Programme was developed in 2007 and its key strategy and vision was to make the world more age friendly. The programme produced *Global Age-Friendly Cities: A Guide*, which was informed by older people and identified eight domains and created a series of accompanying checklists that could provide more age friendly environments (World Health Organization, 2007b). The eight domains comprised of: 1. Outdoor Spaces and Buildings; 2. Transportation; 3. Housing; 4. Social Participation; 5. Respect and Social Inclusion; 6. Civic Participation and Employment; 7. Communication and Information; 8. Community Support and Health Services (World Health Organization 2007a) (see Figure 1.1 below).

Figure 1.2 The Eight Domains of an Age Friendly Community (Source: Age Friendly Ireland, 2021a)



Age Friendly Ireland

Countries have since adopted this international programme and there is now a global network of Age Friendly Cities and Communities. In many ways, Ireland is leading the way in terms of age friendly policy. In 2014, the *Age Friendly Ireland Programme* was set up. Its overall vision was to "make Ireland a great country in which to grow old", by guiding and supporting communities at various spatial scales and developing age friendly strategies, which would be informed by older people and "enhance their quality of life and participation in Irish life" (Age Friendly Ireland, 2021b). This "commitment to action" was in response to concerns about the need to prepare for the ageing of Ireland's population, recognising the implications of this for public policy, as well as the impact of "environmental, economic and social factors that can play an important role in the overall health and wellbeing of older people" (Age Friendly Ireland, 2021a; 2021b). Louth was the first county and local authority to sign up to the Age Friendly County Programme, and by November 2014, all 31 local authorities had signed up to the *Dublin Declaration on Age Friendly Cities and Communities in Europe*, which committed to "creating an

inclusive, equitable society in which older people can live full, active, valued and healthy lives" (Age Friendly Ireland, 2021c). Ireland was the first country in the world to be fully affiliated with the World Health Organization's *Global Network of Age Friendly Cities and Communities* in 2019 (Age Friendly Ireland, 2020).

In addition to ageing policy, there are some notable urban and demographic trends which are likely to influence the ageing experience of older people, and two trends that are of relevance for the Irish context are now considered. The first is the expansion and development of low-density housing in the peripheral areas of Dublin and the surrounding counties that has occurred since the second half of the twentieth century and which continues to this day. The result of this is that increasing numbers of older people will be living in suburban settings now and in the future. The second trend is the health context of older people in Ireland. Whilst many older people may have good health and mobility and are living longer, there are also many older people with multi-morbidities and health challenges. Research from the UK has shown that disparities in health can follow along health and social gradients, where those with the greatest health challenges have the fewest resources to overcome these challenges (see Hart, 1971; Marmot, 2018; Marmot, 2015). The combination of these demographic and environmental characteristics is likely to have significant implications on the ageing experience for older people in Ireland and as a result this needs to be recognised by and prepared for by policymakers.

1.2.2 Geographic and Demographic Trends of Older People in Ireland

Nearly two-thirds of the Irish population live in urban areas (62.7%) compared to rural areas (37.3%). Much of Irelands' population is concentrated in the east of the country, with approximately 1.9 million people or 40% of the population living within the Greater Dublin Area as of 2016 (CSO, 2017). The Greater Dublin Area is defined as Dublin City and its suburban counties (Fingal, South Dublin and Dún Laoghaire-Rathdown), as well as the surrounding counties of Kildare, Meath and Wicklow which are a mix of both urban and rural areas (see Figure 1.2). Nearly 35% of the over 65 population live within the Greater Dublin Area as of 2016 (see Table 1.1). Furthermore, those counties in Ireland with the fastest growing older adult populations between 2011 to 2016 Census were located in the more suburban or

peripheral counties within the GDA to the east of the country. As shown in Table 1.1, the greatest percentage change of the over 65 population between 2011 and 2016 occurred in Fingal (36.1%), South Dublin (34.1%), Kildare (32.2%), and Meath (27.4%) (CSO, 2017). Dublin City Council, located in the centre of the GDA, experienced the smallest increases in older people aged 65 (8.8%). Owing to the nature of development and demographic patterns within Ireland, this trend is set to continue (CSO, 2017). This means that increasing numbers of older people will be living and ageing within their communities in Ireland in the future and that these communities are likely to be urban rather than rural. In addition, the built form of these areas is increasingly suburban.

Suburbanisation and Urban Sprawl in Ireland

The built form of the Greater Dublin Area is predominantly suburban with a "low density urban form", which is "dispersed and dependent on car-based transport" (Nedovic-Budic et al., 2016, p.159). Ireland has experienced one of the highest rates of urban expansion in Europe, producing urban sprawl (Ahrens and Lyons, 2019). It is estimated that the Dublin area covers twice the land of cities with a similar population, such as Prague, Cologne or Copenhagen (Williams and Shiels, 2002). Several trends have led to the population moving to more suburban peripheral areas, but a significant factor has been the re-housing of inner-city residents during the 1970s in particular to local authority housing in "new towns" (McManus, 2019), as well as high costs of housing in Dublin (Oana et al., 2011; Winston, 2007).

Table 1.1 Older Adult Population Change in the Greater Dublin Area 2011-2016 (Source: CSO, 2017)

County	Population (2011)	Population (2016)	Population Aged 65+ (2011)	Percentage of Population Aged 65+ (2011)	Population Aged 65+ (2016)	Percentage of Population Aged 65+ (2016)	Percentage Change of Population Aged 65+ (2011-2016)
Dublin City	527,612	554,554	66,490	12.60%	72,355	13.05%	8.8%
Fingal	273,991	296,020	19,861	7.25%	27,035	9.13%	36.1%
South Dublin	265,205	278,767	23,053	8.69%	30,925	11.09%	34.1%
Dún Laoghaire-Rathdown	206,261	218,018	29,872	14.48%	34,669	15.90%	16.1%
Kildare	210,312	222,504	16,656	7.92%	22,014	9.89%	32.2%
Meath	184,135	195,044	16,322	8.86%	20,788	10.66%	27.4%
Wicklow	136,640	142,425	15,001	10.98%	18,576	13.04%	23.8%
GDA Total	1,804,156	1,907,332	187,255	10.34%	226,362	11.87%	20.9%
Ireland	4,588,252	4,761,865	535,393	11.67%	637,567	13.39%	19.1%

In addition to low density built form, this region is characterised by less welldeveloped transport networks and faces significant challenges related to historically poor planning and lack of strategic and sustainable planning policies. This is recognised in the *Transport Strategy for the Greater Dublin Area 2016-2035*:

Over the next 20 years, transport infrastructure and services must deal with a historical legacy which saw significant levels of growth and migration of land uses to suburban and peri-urban fringe locations, typically at lower densities and unconnected to existing and planned public transport services and facilities (National Transport Authority, 2015 p.27).

The relationship between urban sprawl and health is complex. For some, moving to suburban neighbourhoods may lead to improvements in health and wellbeing, through gaining more space, greenery and perhaps a garden. However, at a population level, there can be "unintended consequences" associated with urban sprawl (Frumkin et al., 2004, p.221). For example, car dependency can increase traffic congestion, air pollution, which can influence respiratory health and risk of injuries. For the populations that live in these areas without access to a car, there is likely to be greater challenges with travelling. Lower rates of walking and cycling and engaging in more sedentary behaviours can contribute to a variety of poor health outcomes (Ewing et al., 2003). The mental health impacts of urban sprawl can be significant and can influence the ability to develop social capital (see Leyden et al., 2003; Melis et al., 2015), particularly if these environments are stressful and lacking in aesthetically pleasing design. Finally, the consequences of these health impacts may not be felt equally across the population, with certain population groups more vulnerable to the negative effects (Frumkin et al., 2004). Of those, older people, as well as those living in more disadvantaged communities are notable, because they may be less able to compensate for poor urban design, as they may be less mobile and therefore more dependent on their immediate local environments (Buffel et al., 2012; Milton et al., 2015; Yen et al., 2009).

For this reason, low density suburban environments, particularly those characterised by urban sprawl are considered less 'healthy' or 'restorative', owing to a lack of amenities, car dependency, less walkable neighbourhoods, and limited public transport and infrastructure (Frumkin et al., 2004; Roe and McCay, 2021). Fong et

12

al. (2020) have conducted research exploring the lived experiences of older people in suburban communities in Australia and have raised concerns as to whether such environments can meet the needs of the older adult population and argue that it is a "question that policymakers must address urgently" (p.2). As a result, there is a need to consider the impact of suburbanisation and urban sprawl on the ageing population and to learn more about lived experiences and the contextual challenges that older people within these locations may face. In the next section I consider some of the health characteristics of the older adult population and how this varies in Ireland.

1.2.3 Health Characteristics of the Older Adult Population

This thesis is underpinned by several key principles within Health Geography, Urban Planning and Environmental Gerontology. One of the most important theoretical underpinnings is that an individuals' health and wellbeing experience, is in part determined by the broader physical and social environment in which they live. Both the built and natural environment are recognised as determinants of health. In turn, health is likewise a determinant of ageing well. Figure 1.3 presents a socioecological understanding of the determinants of health and highlights the complex ways that an individual and their built environment interact to influence their behaviour and health (Barton and Grant, 2006). This includes the natural and built environment, as well as the types of activities carried out within a particular neighbourhood. The local economy and community can play an important role in health and wellbeing, particularly the social capital and social networks that an individual has. Lifestyle and personal factors can influence the nature of engagement with local environments. Age, gender, and genes also contribute and influence the health and well-being an individual can attain, along with wider structural forces that influence policies, societal and cultural norms, and expectations.

Figure 1.3 Environmental Determinants of Health and Wellbeing (Source: Barton and Grant, 2006, p.2)



Determinants of health can arise over a lifetime and result in extreme variations in health and ageing experiences:

the way we experience and react to places is shaped by the layered accumulation of life experiences in environments over our life span (Rowles, 2018, p.204).

Old age is typically perceived as the latter part of an expected or normal life span. The most used definition is based on chronological age, and is when people would have traditionally retired, at age 65 (Fahey et al., 2007). This is a common threshold for an older person and used within the Census in Ireland. However, The Irish Longitudinal Study on Ageing (TILDA) collects data from people aged 50 and over, showing that there is not always agreement on how to define an older person. Within this thesis I use the term older person or older people, as this has been established and preferred by older people themselves (Age UK, 2019), as opposed to terms such as 'elderly', which can denote images of dependency and vulnerability and negative stereotyping.

Figure 1.4 demonstrates how access or a lack of access to natural environments from a young age can result in different life pathways. These diverging pathways can result in disparities in behaviours and health outcomes, which can widen over a life course. By the time a person reaches 65, they will have different life expectancies and possibilities for a good quality of life. With people living longer, some with and others without chronic conditions, the experience of older adults and their lived realities is becoming increasingly diverse. As a result, it is becoming increasingly difficult to define an older person. For some older people, what can be physically expected from an individual at a certain age is being redefined, but it is important to remember that not everyone is able to attain such optimal levels of health and engage with valued functionings that may be of importance to them. Some older people experience a variety of health challenges and age-related conditions, which may be present before what might be considered a chronologically old age. As a result, sharing the same year of birth does not mean an individual will experience the same abilities, health, outlook, wishes, and desires.



Figure 1.4 Dichotomous Life Pathways (Source: Pretty et al., 2009, p.24)

As a result, there are challenges with defining a population group based on chronological age alone. Usually populations are grouped into chronological subdivisions that have a shared experience or needs, making it either meaningful or useful to do so. However, the older adult sub-group is arguably the most diverse of all age brackets, because of the diversity of experiences that have accumulated over a life time. Many researchers have raised the need to consider the heterogeneity, or diversity of older adults within research (Burns et al., 2012; Cotterell et al., 2018), with Beard and Montawi (2015) arguing that heterogeneity is "one of the hallmarks of ageing" (p.5). Whilst this diagram shows two extreme scenarios and this is just in relation to engagement with green space, it is helpful to demonstrate why older people are so heterogeneous, with many potential life pathways between these two extreme pathways (Pretty et al., 2009). The heterogeneity of older people is therefore partially explained by the "cumulative impact of multiple inequities across life" (Beard and Montawi, 2015, p.6).

Many older people in Ireland can expect to live longer and in better health than previous generations. However, there is also a sizeable and increasing proportion of the older adult population living and ageing with health and mobility challenges, which may impact negatively on their ability to achieve a good quality of life in later years (Department of Health, 2015). Existing data has identified socio-economic health gradients amongst older people in Ireland and research has shown that older people in lower socio-economic groups are at increased risks of chronic conditions and associated disability (Eurostat, 2017; Fahey, 2007; Savva et al., 2011; Sheehan and O'Sullivan, 2020). The culmination of this, is that the experience of being an older adult in Ireland is extremely diverse.

1.2.4 Summary

Both the National Positive Ageing Strategy and Age Friendly Ireland are aspirational policies, setting out visions of what they hope Ireland will become with regards to ageing. A problem with this, however, is that greater attention is placed on ideals and less on the everyday realities of the current older adult population and how this may vary along social and health gradients, as well as influenced by urban development processes such as urban sprawl. Buffel et al. (2012) question the appropriateness of
an objective checklist of age friendliness, as it focuses on an "ideal" age friendly city, instead asking "what are the actual opportunities and constraints in cities for maintaining quality of life as people age?" (p.601). To improve the experiences of older people ageing in place, which both the NPAS and Age Friendly Ireland ultimately intend to do, there is a need to understand the full range of experiences of being an older person, recognising the heterogeneity of this population group.

The establishment of an Age Friendly Cities and Counties Network in Ireland and the subsequent evidence base that arose from this project, was an important step in recognising that where an individual lives influences how they live. It recognises that an individual's environment can serve as a barrier or an enabler within everyday lives and that certain characteristics of the built and social environment are more age friendly than others. However, as I will discuss further in Chapter 2, the World Health Organization (2007a) Age Friendly Environment checklists are built on quite limited and more objective constructs of place, focusing on more tangible components of place and the built environment, rather than the intangible and subjective components, which are often more important to older people themselves (van Hees et al., 2017). As a result, more subjective and heterogenous insight about the older adult experience is missing. This includes a lack of consideration about what is of most importance to older people themselves to age in place and how may they define this. The importance of more subjective experiences of ageing in place and subjective interpretations of age friendly environments is increasingly being recognised within academic literature (Golant, 2015; Lager and van Hoven, 2019; van Hees et al., 2017). The need for this to transfer into policy is likewise acknowledged. Existing research from the UK that has interviewed planning practitioners about *Planning for an Ageing Society* has found that planners are aware of the need for further knowledge about "the composition, aspirations, experiences and requirements of this population, now and into the future" (Hockey et al., 2013, p.538).

A combination of demographic, urban and health trends mean that increasing numbers of older people with a variety of health and mobility needs will be ageing in place with predominantly suburban and car dependent neighbourhoods within Ireland. This could have significant implications for how well older people are able

to realise the policy priority of ageing in place, and yet this has been under-explored within social and planning policy. As Golant (2018) notes:

it is more enjoyable, easier and less costly to grow old in some places than in others (p.190).

Informed by this ageing landscape within Ireland, this thesis addresses several research gaps (Miles, 2017) and, it is hoped, thereby makes several distinct contributions, which are now summarised.

1.3 Thesis Contributions

As mentioned at the beginning of this thesis, there is a policy imperative towards ageing in place and promoting age friendly environments. However, it is acknowledged by the NPAS that we know very little about the everyday lived experiences of older people ageing in place in Ireland. Based on population and environmental trends identified in the previous section, it is reasonable to assume that Ireland is likely to be a great place for some more than others to grow old in and existing research is needed to further explore this. There is a lack of qualitative research that highlights how the interaction of personal and environmental factors can influence the *quality* of this experience. This thesis is therefore responding to these knowledge gaps.

Within this thesis, I focus on older people's experiences of suburban environments, recognising that within Ireland, this is an area where increasing numbers of older people are and will be ageing in place in the future. Informed by existing literature and based on my own experience of working as a Planning Policy Officer in the UK, I am aware that the built forms of suburban environments are likely to provide additional challenges for older people to get out and about and be mobile, especially when they may also experience health and mobility challenges, which increasing numbers of older people within Ireland experience. Exploring this from a Health Geography perspective will allow me to detail the health inequalities and social gradients that influence the ability to age well in place. Again, there is little research within Ireland that has considered how these broader environmental and health contexts may influence the daily lives of older people ageing in place. To address

this knowledge gap, the Study Areas where I conducted my fieldwork were selected with this in mind. For example, Study Area 1 was more suburban in characteristic, with a mix of recently ageing and newly ageing populations with higher level of deprivation and health challenges. Study Area 2 included a mix of inner-suburban and inner-city characteristics and was an area that had a more established older adult population. It was an area that was experiencing urban change, had better transport amenities, more walkable neighbourhoods, and fewer health challenges at a population level.

This thesis examines how ageing well in place is defined by older people themselves and as a result, responds to a practical-knowledge gap, building on existing critiques of ageing in place, age friendly environments and ageing well. This recognises that both professionals and policymakers typically focus on more objective and validated measures of ageing well and age friendly environments characteristics, rather than focusing on the perspective of older people themselves, which may not necessarily align (see Hockey et al., 2013). Lay experiences and forms of knowledge are prioritised within this thesis, as well as the heterogeneity of older adult experience. Consideration has been given to attending to the types of older people and perspectives that may typically be under-represented within age friendly research (Gilroy 2021). To address this population gap, I have tried to capture the perspectives of range of older people who traditionally may be less engaged by researchers, and this is discussed further in Chapter 4.

The overarching theoretical contribution of this thesis is the empirically grounded theoretical model I have developed of 'ageing – *as well as you can* – in place'. This model integrates ideas about ageing well and ageing in place through the Capability Approach lens. A strength of this model is that it recognises the variability of the ageing experience, in terms of changing health, mobility and functioning over time, as well as the subjectivity of what matters most to people as they get out and about. Whilst existing literature has recognised the need for a Capability Approach lens to ageing well (discussed in Chapter 2), my own research grounds these ideas empirically and spatially. This was enabled by the methodological approach taken, by metaphorically and physically meeting older people where they live, using

innovative in-situ methods such as the go-along interview and mapping exercises, and qualitatively mapping their everyday experiences.

Empirically, I have added to existing literature that highlights the importance of getting out and about for wellbeing, showing variations in how this is prioritised and valued by older people and how it changes over time. I have added to the existing literature that highlights the key places of importance for social interaction, including 'third places', responding to Finlay's (2019) call for further research on this. This work confirms much of the existing findings but adds some new ideas about routine and scheduling which may have been overlooked by traditional third place literature, which appear to be of relevance to older people. Finally, this research adds to the literature that has identified the significance of the everyday experiences of older people as they navigate health and mobility challenges. I use annotated maps and 'geo-narratives' to showcase what is of most important to older people and how this is navigated. This empirical research adds to our understanding of the quotidian experience of ageing in place and the reasons that getting out and about is so important for ageing well in place.

1.4 Conclusion

In this chapter, I have provided an overview of this thesis topic, which included the primary research questions and the justification for this research. I have summarised the methodological approach taken, the key findings and the implications of this research, as well as identified the theoretical, methodological, and empirical contributions of this thesis and the research gaps that it seeks to address. I provided an overview of the ageing policy and demographic context of Ireland to provide the backdrop for this thesis, highlighting that policy landscape and key documents related to ageing well in place, including the National Positive Ageing Strategy, as well as the policy drive towards Age Friendly Environments and the desire for Ireland to be "A Great Place to Grow Old". However, I also highlighted health and environmental disparities across the older adult population, which will have implications for the type of experience that may be available to older people as they age in place. To conclude this chapter, I now summarise the structure of this thesis.

1.4.1 Structure of this thesis

In the next chapter (Chapter 2), I present the overall conceptual framework for this thesis: Ageing Well in Place. I begin by reviewing the concepts of ageing in place and ageing well, showing how I consider these within this thesis. I begin by critically examining the concept of ageing in place from a health geography perspective. This recognises that getting out and about and engaging with the broader physical and social environment is highly valued by older people and can influence health and wellbeing both positively and negatively. I then consider the concept of ageing well, showing how it can be differently defined and highlight the growing movement towards lay interpretations of ageing well, as well as the application of a Capability Approach lens as a way of examining what ageing well in place means to older people. In Chapter 3, I present an overview of the literature that explores the importance of community mobility and getting out and about. I begin by summarising the various ways that this has meaning to older people and can influence their health and wellbeing, focusing on the importance of socialising with others as a key motivator for leaving the house. Alongside this, I demonstrate that whilst getting out and about is valued, it can also be challenging for some older people, summarising the main literature that has explored this, as well as some of the ways this is negotiated and navigated by older people.

Owing to the importance of older adult perspectives, perceptions and experiences, combined with a strong place-based component within this research project, a multistage qualitative, spatial and in-situ methoodology will be used. In Chapter 4, I outline the ontological, epistemological, and methodological underpinnings of this thesis, showing how I have used a qualitative and spatial approach to answer my research questions. I outline the study design and research methods used, discuss the sampling and recruitment process and the study settings chosen. Within this thesis I have carried out a flexible, inclusive and care-full approach and I show how I have thought ethically about this research. I also summarise the data analysis steps for each of the empirical chapters.

Chapters 5, 6 and 7 are empirical chapters. Chapter 5 explores the importance of getting out and about to my participants, how this fluctuated, and why this was

connected to how they defined a good quality of life. I conclude this chapter by conceptualising the overarching contribution of this thesis: ageing – *as well as you can* – in place, which I define as having 'good enough' health and mobility, to be able to get out and about to participate and engage in meaningful activities or interactions. In Chapter 6, I will demonstrate how participants' wellbeing is influenced both positively and negatively by getting out and about, focusing on a component that was most valued by my participants: socialising and connecting with others. Here, I summarise participants' *experiences* of getting out and about, applying a framework developed by Gardner (2011) to show how social interactions occurred within different places or 'phases'. I consider the nature of these interactions (including whether they were positive or negative), why they were important, how planned and frequent they were, and the types of routines and schedules developed by older people to ensure that they happen.

A common thread throughout all the empirical work is that getting out and about was easier for some participants than others and I demonstrate how this varied across my entire participant sample. For example, in Chapter 6, I demonstrate how the need and ability to get out varied between participants, as well as for the same participant at different times, depending on seasons, weather conditions and times of day. I also provide insight into how some participants avoided certain forms of getting out and about and the reasons for this. Within Chapter 7, I focus in-depth on four participants with differing health and mobility challenges and show their experiences of ageing – as well as they can - in place and how they navigate and negotiate getting out and about in ways that are meaningful to them. Each individual's 'lifeworld' is described as a geo-narrative with related annotated maps. I provide insight into the patterns, routines and decision-making processes that happen gradually over time and highlight the sheer determination to keep connecting with what matters most to them.

In Chapter 8, I situate my findings in relation to existing literature and consider the implications of what I have found. I present and further develop the theoretical model I have developed for this thesis: ageing -as well as you can - in place. I provide an overview of four parts of this model and situate these components in relation to existing literature. I reflect on some of the challenges and limitations of

this research, including a reflection on the Covid-19 pandemic and outline two key recommendations that geographers and planners could implement to support older people to age - *as well as they can* - in place. I now turn to my review of relevant literature and conceptual thinking for this thesis.

This page has been left intentionally blank.

Chapter 2. Ageing Well in Place: A Conceptual Framework

2.1 Introduction

As demonstrated in Chapter 1, Ireland has developed several policy responses to its demographic ageing context, which focus on supporting older adults to age in place through promoting age friendly environments and ageing well. In this chapter I more critically examine these concepts drawing from the relevant academic literature and provide a theoretical and conceptual framework for this thesis. In Section 2.2, I summarise the well-established literature that discusses 'ageing in place' and show how I have been informed by more geographical understandings of place. To do this I draw on perspectives from Geographical and Environmental Gerontology, as well as Health Geography sub-disciplines. I highlight literature that demonstrates the importance of physically being in and feeling in place whilst ageing in place. This involves older people leaving the home and engaging with the broader neighbourhood and community. In Section 2.3, I consider the existing literature on the topic of 'ageing well', highlighting how it can be differently defined, by older people and practitioners, and why this is important to recognise. I then introduce literature which examines ageing well through a Capability Approach lens. In Section 2.4, I integrate these two concepts to consider ageing well in place and how I conceptualise this, informed by the health geographies of ageing. Finally, I conclude this chapter in Section 2.5.

2.2 Ageing in Place

'Ageing in place' is typically defined as staying in your own home or community as you age, or "growing older without having to move home" (Phillips et al., 2010, p.17; Wiles et al., 2012). Traditionally, policies supporting older people to age in place have focused on providing adaptations and support services within the home (Wiles et al., 2012), to allow people to continue ageing within their homes for as long as possible. Such policies stem from concerns about costs of institutional care on the one hand, and assertations and assumptions that ageing in place is what older people themselves desire and prefer (Wiles et al., 2012). More recently, scholars within Environmental Gerontology and Geographical Gerontology disciplines have

engaged critically with the concept of ageing in place (Lager, 2015). This has included more geographical understandings of place being employed (Finlay and Finn, 2020; Finlay et al., 2019; Finlay, 2018; Lager, 2015; Phillips et al., 2010; Wiles et al., 2012). Such an approach recognises that whilst home is an important component of ageing in place, there is a need to think beyond the home to consider the broader neighbourhood and community as important factors in the overall experience (Hillcoat-Nalletamby and Ogg, 2013; Lager and van Hoven, 2019; Pani-Harreman et al., 2020; Peace et al., 2011; Phillips et al., 2010; Wiles et al., 2012). Some researchers have called for a change in term from ageing in place to ageing in community to recognise the broader than home and relational elements of ageing in place, as well as the need for adults to be meaningfully involved within their communities for their health and wellbeing (Black et al., 2010; Provencher et al., 2014; Thomas and Blanchard, 2009).

The importance of incorporating a broader interpretation of *place* within ageing in place has been reinforced within a recent scoping review of thirty four studies. Pani-Harreman et al. (2020) identified five key themes related to ageing in place as a concept, of which *place* was the most important theme. Throughout the studies, place could be used to describe home, the home environment, or wider neighbourhood and community. Whilst this included physical or tangible aspects of place, such as physical environment characteristics, it also included more intangible, emotional and experiential aspects, such as place attachment and belonging (Pani-Harreman et al., 2020; van Hees et al., 2017). Pani-Harreman et al. (2020) emphasise that ageing in place is about "not only staying in one's home", but also includes "remaining in a stable and known environment where people feel that they belong" (p.25). This involves feeling a sense of "*being in place*" (as opposed to "*being out of place*"), which Rowles (2018) argues is the "essence of well-being in later life" (Rowles, 2018, p.202 and p.208).

Research has shown that older people themselves define ageing in place more broadly than the home environment. Wiles et al. (2012) explored what ageing in place meant to older adults in New Zealand through a series of focus groups and interviews. Results showed that ageing in place was valued by older people but that whilst it was a term common amongst service providers and policy makers, it was not familiar to most older people, highlighting that its definition is not always "fixed" or clear (Wiles et al., 2012). However, "staying in one's home or community" was perceived as beneficial in several ways, including through providing a sense of attachment to communities and social connection (p.360). It was linked to a sense of security and familiarity, with home seen as a "refuge" and community perceived as a "resource" and form of support (Wiles et al., 2012, p.361). Finally, ageing in place ensured a continued sense of identity and this was linked to feeling independent and autonomous, highlighting the importance of being able to make choices about *how* to age in place. As a result, this research confirmed that engaging with the broader than home environment is very important to older people.

Although research has demonstrated benefits to ageing in place, a "blanket" political and policy prioritisation of ageing in place has been criticised by many (Phillips et al., 2010) and caution has been raised about viewing ageing in place "as a 'one stop' solution to later-life aspirations and needs" (Hillcoat-Nalletamby and Ogg, 2013, p.1780). Critiques focus on the *assumption* often placed on ageing in place about it automatically being desirable, without considering the diversity of the population who are ageing in place, how these needs may change over time, and the dynamic environmental circumstances that individuals are ageing in (Phillips et al., 2010). Whilst policies that assume older people prefer to age in place, draws from geographical concepts and theories such as place attachment and familiarity (Wiles et al., 2012), they lack consideration of individuals' diverse lived experience and realities. For example, the experience an older person may have as they age in place, and how positive this proves to be, depends not only on how attached they are to their home-place, but also how well this environment suits their shifting needs and abilities. It will depend on an individual's person-environment congruence (Lawton and Nahemow, 1973) and will be influenced by how well they can adapt to, and cope with changing circumstances over time (Peace et al., 2011).

Such a view of ageing in place recognises that it is "not a continuous, uniform experience or solution, but will vary in its 'do-ability' depending upon evolving lifecourse needs" (Hillcoat-Nalletamby and Ogg, 2013, p.1788). Existing research has demonstrated the importance of thinking about ageing in place relationally, recognising that "older adults' experiences are viewed as an outcome of the complex and dynamic interplay of self, others, place and time" (Lager et al., 2015, p.3).

Ageing in place therefore needs to be recognised as the complex and dynamic process that it is, where older adults are continually navigating and negotiating their relationship with and to people and places (Andrews et al., 2007; Hopkins and Pain, 2007; Lager et al., 2013; Skinner et al., 2014; Wiles et al., 2012; Ziegler et al., 2012).

Research has shown that ageing in place is not always a desirable, ideal or an easy option for older people, particularly when ageing "in the margins" (Finlay et al., 2018), and ageing in "unsuitable" (Severinsen et al., 2016), or "difficult" places (Scharf et al., 2007). Finlay's (2018) work exploring the lived experiences of older adults in Minneapolis, Minnesota in the US, adopted a broad perspective on the experience of ageing in place and demonstrated how place attachment can be unattainable for many. Furthermore, it has shown that staying in one's home can involve risk and hazards, as well as social isolation due to lack of supportive infrastructure and crime. Reaching these 'tipping points', participants had a lack of options to change their situation, leading to exclusion, where they were "striving to age well 'in place", but unable to do so (Finlay et al., 2018, p.768).

Owing to differing person-environmental contexts and the heterogeneity of older adult experiences, it is important to recognise that there may be situations where individuals feel "out of place" (Brittain et al., 2010) or "unfamiliar" (Phillips et al., 2013) whilst ageing in place. This could be through personal factors such as mobility or cognitive impairments or through a variety of physical and social environment factors, as well as including urban transformation and change (see Lager et al., 2013; Brittain et al., 2010). Phillipson (2007) argues that globalisation has the potential to create more extreme inequalities amongst older adults, whereby the "elected" have more agency, knowledge, and control to improve their existing environment, or to move to more age friendly environments. On the other hand, there are then the "excluded", typically in more deprived communities, who may be living in areas with fewer services and provisions. They are less able to influence their situation, are more negatively impacted and disempowered by their immediate physical and social environment and may lack the financial resources to move to an alternative location which may be more supportive. Phillipson (2007) highlights the importance of considering how macro-level forces serve to influence and change the everyday experiences of older adults within their neighbourhoods. Considering the experiences of more disadvantaged older adults is therefore an important step to "address

growing inequities and reimagine ageing 'in place' along more inclusive and socially just lines" (Finlay et al., 2018, p.781).

2.2.1 Summary

Building on existing research that critically engages with the term ageing in place, this thesis applies a geographical interpretation. Whilst the importance of the home environment has been identified by many as an important component of the experience of ageing in place (see Sixsmith and Sixsmith, 1991; Stones and Gullifer, 2016 for example), within this thesis I focus predominantly on the broader local physical and social environment, neighbourhood, and wider community in which a person lives. I incorporate both the physical experience and the emotional feeling of being *in place*. Whilst policies that assume older people prefer to age in place, stem from geographical concepts and theories such as place attachment and familiarity (Wiles et al., 2012), they can also lack consideration of individuals' diverse lived experience and realities (Golant, 2018). Existing research has highlighted the need for further research that considers how older adults are ageing in place, including the quality of experience, which is lacking within an Irish context. A key component of this is how *well* individuals can age in place. To consider this, it is first necessary to clarify how ageing well can and should be defined.

2.3 Ageing Well

Alongside ageing in place policies and research, are parallel debates relating to 'ageing well'. Several terms relate to ageing well, which I use here as the umbrella term to consider the varied types of ageing well. These include successful ageing, positive ageing, productive ageing, active ageing, and healthy ageing. Whilst various desirable forms of ageing are referred to within gerontology literature and policy, successful ageing remains dominant. The wide variation in types of optimal forms of ageing reflects the lack of consensus and complexity as to what a desirable form of ageing is. Within this section, I consider how existing literature defines ageing well and why this is important.

The term successful ageing is multi-dimensional (Martin et al., 2015; Phelan et al., 2004; Rowe and Kahn, 1997), and several definitions are presented within the

literature. Definitions are typically categorised within the literature into biomedical and psychosocial definitions. Biomedical approaches emphasise the importance of optimal functioning and engagement (Phelan and Larson, 2002) and is seen as a "better than normal" status (von Faber et al., 2001, p.2694), in that an individual goes beyond "usual" ageing" (Tate et al., 2003, p.737). An example of a dominant biomedical model of successful ageing is that of Rowe and Kahn (1997), who defines successful ageing as the achievement of the following (p.439): avoidance of disease and disability; maintenance of high physical and cognitive functioning; and sustained engagement in social and productive activities.

Psychosocial approaches on the other hand, recognise more subjective components of successful ageing. In this instance, successful ageing is viewed as more of a mental state or outlook (Glass, 2003), as well as a lifelong process of adaptation and adjustment (Tate et al., 2003; Baltes and Baltes, 1990), as opposed to an objective standard or criteria. Bowling and Dieppe (2005) highlight the importance of concepts such as life satisfaction, social participation, and functioning, as well as psychological resources (p.1549). Such psychological resources for successful ageing include: a positive outlook and self-worth; self-efficacy or sense of control over one's life; autonomy and independence; and finally, effective coping and adaptive strategies in the face of changing circumstances (p.1549). Phelan and Larson (2002) conducted a review of the definitions of successful ageing and identified seven major elements that are typically included in successful ageing studies. These include: life satisfaction; longevity; freedom from disability; mastery or growth; active engagement with life; high or independent functioning; and positive adaptation (p.1307). Some of these studies focused on predominantly biomedical definitions, whilst others looked at more psychosocial components.

Traditionally, it has been clinicians that have referred to objective definitions and lay people or older adults themselves that have preferred more subjective interpretations (Glass, 2003). A major criticism of earlier research on successful ageing was the lack of consideration of how older adults themselves viewed successful ageing (von Faber et al., 2001). In recognition of this, researchers are increasingly arguing for the need to pay greater attention to the voice(s) of the older adult population (Tate et al., 2003), and for a more 'lay' or patient-centred definition of successful ageing that

would be of greater relevance and validity to those who are ageing (Phelan and Larson, 2002; Bowling and Dieppe, 2005).

Phelan et al. (2004) sought to identify which aspects or attributes of successful ageing were most important to older adults and found that many of these were strongly related to health. These included: remaining in good health until close to death; being able to take care of themselves until close to time of death; and remaining free of chronic disease (p.213). This highlights the overlap between ageing well and health. Subjective components of successful ageing were also valued. These included: being able to act according to their own inner standard and value; being able to cope with the challenges of later years; being able to meet all their needs and some of their wants; being able to make choices about things that affect how they age, such as diet, exercise, and smoking (p.213). Many of the attributes identified were consistent with maintaining a sense of self, resilience, and coping, being able to get by but also to flourish, and have sufficient choice. A key finding of Phelan et al. (2004) was that quality of life was perceived to be more important than quantity, emphasising the importance of quality of life as a core component of ageing well.

An important difference between the biomedical (more objective) and psychosocial (more subjective) aspects of ageing well, is the degree of success that older adults can attain, with objective definitions tending to have much stricter criteria than subjective definitions. The number of people that meet objective successful ageing criteria and measurements is typically far lower, however the number of people that *think* they are successfully ageing is typically higher. For example, in a study looking at older adults aged 85 and over, only 10% met successful ageing criteria pertaining to optimal levels of functioning and wellbeing, yet nearly half of the participants reported an 'optimal' state of wellbeing. This was despite identifying physical limitations (von Faber et al., 2001). In qualitative interviews within the same study, successful ageing was perceived by older adults rating wellbeing and social functioning more highly than physical and psycho-cognitive functioning (von Faber et al., 2001, p.2699).

In a review of successful ageing definitions, Jeste et al. (2010, p.80) identified that only a small percentage of older adults were successfully ageing according to objective definitions based on absence of disease (15%) and freedom from disability (38%). However, a far higher percentage believed they were ageing successfully according to more psychosocial domains, such as: active engagement with life (74%); positive adaptation (81%); life satisfaction (84%); self-rated successful ageing (90%); and independent living (94%). With increasing policy and societal pressure targeted at older people to adopt and follow a successful ageing lifestyle, we are witnessing "societal demands to age well and positively" (Breheny and Stephens, 2010, p.41). However, depending on whether this is defined objectively or subjectively, will influence the number of those that can achieve this. If defined objectively, then far fewer individuals will meet the criteria.

The policy response to encourage successful, positive, and productive ageing aligns with neoliberal responses to health promotion, which put the responsibility of achieving good health on individuals, without considering the variability in people's capacity to achieve it, whether due to individual circumstances, or wider structural, societal, or environmental factors. Focusing on individuals and lifestyle choices in influencing a healthy older age, runs the risk of making older people who cannot attain it feel like failures, instead of recognising the "complexities of life choices and chances and the impact they may have on the later life circumstances of older people" (Breheny and Stephens, 2010, p.46).

A positive ageing discourse benefits older people who can take advantage of opportunities to engage beyond retirement and who want to. However, those with fewer resources, who may have chronic conditions and reduced mobility, are "additionally burdened by the demands to age positively, rather than supported by an expectation of care as they age" (Breheny and Stephens, 2010, p.46). Individuals that cannot objectively age positively, actively, or successfully are "excluded from participating in an acceptable way of life", but not from the "imperative to age well" (Breheny and Stephens, 2010, p.42). Breheny and Stephens (2010) argue that the only option available to those restricted by poor health and material resources to demonstrate agency and attempt to age well, is to focus on more subjective aspects of successful ageing, through a "determination to have a 'positive attitude' toward their situation" (p.44). However, the problem with this, is that underlying structural

factors that created inequalities to begin with are not addressed (Breheny and Stephens, 2010). Stephens et al. (2015) warn about the dangers of idealising older adults as a healthy, active, and homogeneous group that are willing and able to continue to contribute towards society. Such an approach ignores the diversity within the older adult population and the structural inequalities that may exist between older adults, which can influence their ability to age well (Stephens et al., 2015).

Parallel debates are taking place about how to define concepts such as health, quality of life, and wellbeing, which as I have already shown, are strongly interrelated with ageing well. Such debates revolve around how to define these concepts and whether more idealistic, medicalised, consistent, and objective measures should be used, versus more realistic, subjective, lay, and flexible definitions. Within the health field, the traditional definition of health developed by the World Health Organization (1946) as "a state of complete physical, mental and social wellbeing and not merely the absence of disease or infirmity" is increasingly being criticised as "no longer fit for purpose" (Huber et al., 2011, p.1). This is largely in response to the rising prevalence of chronic conditions and ageing populations, which means that perfection is beyond many. Instead, an alternative definition of health has been presented by Huber et al. (2011).

Rather than focusing on an idealised and unattainable version of health, Huber et al. (2011) define health as "the ability to adapt and self manage in the face of social, physical and emotional challenges" (p.1). The important distinction between the traditional definition of health and the more recent definition of positive health is that you can attain and achieve positive health in a less than perfect state. What matters is not whether you have an underlying condition or disease, but whether you are able to cope and manage this disease and whether you have the skills to adapt your circumstances in such a way that you are able to lead as normal a life as possible and that you perceive yourself to be healthy. This connects with Antonovsky's ideas about an individuals' sense of coherence (how an individual sees their world and how they fit within this) and ideas about *salutogenesis*, which is a more enabling and proactive view of health, which considers what creates health, rather than what causes disease or illness (Antonovsky, 1987). More recently, this work has been applied to consider healthy ageing in place (see Walsh, 2014).

This work also aligns with Baltes and Baltes' (1990) psychosocial model of selective optimization with compensation (SOC), where ageing involves three components. When an individual is faced with physical or cognitive challenges or restrictions, they will *select* and focus their attention to those areas within their life that are of the greatest priority to them. In addition, individuals will also *optimise* by continuing to engage in those behaviours that enhance their physical or cognitive capacities. Finally, individuals may *compensate*, or negotiate, by using certain psychological or technological strategies. As a result, the older person engages in a process of adaptation and negotiation, using their agency and resources to maximise their desired outcomes and avoids negative outcomes. Ageing well in this instance is their ability to navigate this successfully.

2.3.1 A Capability Approach to Ageing Well

Owing to the challenges of defining ageing well, quality of life, health and wellbeing described above, some researchers have begun to employ more novel ways to conceptualise these elusive terms. One increasingly common way of doing this is to define ageing well using the Capability Approach.¹ Originally developed by Amartya Sen, the Capability Approach is "based on a view of living as a combination of various 'doings and beings', with quality of life to be assessed in terms of the capability to achieve valuable functionings" (Sen, 1993, p.32). There are two overarching principles. The first is that having the "freedom to achieve wellbeing is of primary moral importance" and secondly, well-being should be defined in relation to people's capabilities and functionings (Robeyns and Byskov, 2021). Functionings are 'beings' and 'doings', the "various states of human beings and activities that a person undertakes", while capabilities are an individuals' "real freedoms or opportunities to achieve functionings" (Robeyns and Byskov, 2021). It is the individual who must specify which capabilities are most valuable. However, whether someone can achieve a functioning depends on "personal, social political, and environmental conditions" or "conversion factors" (Robeyns and Byskov, 2021).

¹ This is often used interchangeably with Capabilities Approach but in this thesis, I use the term Capability.

Developing a Rawlsian critique of the founding principle of the World Health Organization (1946), Daniels (1985, p.28) insists that expectations should be constrained by "normal species functioning", and that the implied right to health care is limited to a fair share of medical resources. The Capability Approach to human rights suggests that people have a fundamental right to the various bases of human flourishing, but in each case, the standard is socially and contextually specific. Thus, among other capabilities, the right to life is to one of "normal length", bodily health includes the right to be "adequately nourished, to have adequate shelter", and the capacity for "sense, imagination, and thought" should be "cultivated by an adequate education" (Nussbaum, 2003, pp.41–2). Notions of 'normal' and 'adequate' have both individual and societal frames of reference. For the purposes of this thesis, it can be expressed as an interest in understanding how and whether individuals can age as well as they might want given their limitations.

More recently, Stephens (2016) has applied a Capability Approach to types of ageing well. Rather than putting the responsibility on older people to age successfully with no consideration of their ability to do so, this approach prioritises supporting older people to "achieve valued functionings" based on their existing capabilities, and to understand their specific needs in "actual circumstances" (Stephens, 2016, p.7). This aligns with more psychosocial descriptions of both health and ageing well, rather than biomedical and objective standards. Stephens et al. (2015) conceptualised and defined health in their research as being able to carry on doing the things that are valued by older adults, which makes it more accessible to all and means that it can be attained with physical decline. They argue it is a "more nuanced version of health", in that is also takes into consideration the "role of social structure, unequal incomes, spatial contexts and social provisions" (pp.728–9). Considering ageing well through a Capability Approach lens, can help to identify the wider social and physical environmental influences on health, as well as the wider societal contexts within which older adults are situated and affected by. Such an approach would ask questions such as: "What are the environmental and social conditions which support these particular capabilities?" and "how are these capabilities valued by older people in a particular context?" (Stephens, 2016, p.6). Stephens (2016) has also been informed by Baltes and Baltes (1990) model of SOC and has combined this with a Capability Approach, recognising that the ability to navigate this is influenced by the

"social circumstances of people as they age" (p.6) and the need to understand "the actual goals of older people, and their capability to select, optimise and compensate to achieve those goals" (Stephens, 2016, p.6).

Meijering et al. (2019, p.232) ask for more research on "how older adults can live a meaningful life in the context of the impairments they experience" and argue that a Capability Approach offers a useful lens to explore this. With a focus upon how individuals value for themselves the ways they can realise their capabilities in meaningful activities, i.e., their various "functionings", the Capability Approach provides a way of "valuing and giving voice to people's own conceptions of what matters" (Gopinath, 2018, p.258). Such an approach would lead to a "shift in focus towards the capabilities that support older adults to achieve independence as their valued functioning, rather than on how 'successfully' they age" (Meijering et al., p.251).

Ageing policies that use terms such as productive, successful, and so on have a particular theoretical underpinning that stems from more biomedical and objective definitions of quality of life. As a result, they typically have less relevance to what matters most to older people themselves. Within this thesis I am interested in exploring ageing well subjectively and based on what older adults themselves feel is of most importance to them. The Capability Approach is a useful way to combine ideas around ageing well and ageing in place through a lay person lens. There is a small but growing literature that has explored the valued functionings that are of most importance to older people themselves and this is now considered.

2.4 Ageing Well in Place through a Capability Approach lens

Broadly, health geographies of ageing are concerned with understanding how place can impact older people's health (Wiles, 2018). As a sub-discipline, this has moved and evolved over time from more biomedical understandings of health as being the absence of disease, to embracing "more holistic socio-ecological understandings of ageing and health in social, physical and symbolic contexts" (Wiles, 2018, p.31). As a result, this discipline is well situated to integrate the concepts of ageing well and ageing in place, to consider "how environments are supporting quality of life for older people" (Gilroy, 2006, p.343). This recognises the complex factors that influence the overall experience of ageing in place, including the role of broader structural forces (such as inequalities), which can influence the experience of, and ability to age in place in a beneficial way. In sum, individuals realise valued activities and meaningful engagements to an extent that is strongly influenced by context:

One of the key determinants of the capabilities of older people, and whether they can achieve the things that are meaningful to them, is the environment in which they live (Beard and Montawi, 2015, p.5).

There are several empirical studies that have examined the lived experience of older adults through a Capability Approach lens and have broadly summarised the valued functionings of older people. In semi-structured interviews with 145 older adults aged 63–93 years and living independently in both urban and rural locations in New Zealand, Stephens et al., (2015) asked participants to talk about their daily needs and practices, including the resources that they currently lack or would like, the characteristics of their community and physical environment, how this influenced their ability to participate socially, as well as their ability to manage on their income. Six broad domains of functioning were valued by older adults: Physical Comfort; Social Integration; Contribution; Security; Autonomy; and Enjoyment (pp.720–4).

With regards to *Physical Comfort*, everyday goods, such as clothing, housing and health services were identified. Using the example of food, participants claimed that having choices provided them with agency and enhanced their wellbeing. In terms of *Social Integration*, opportunities to engage with friends and family, to attend activities and events were highly valued. Participants felt that this helped them to feel a part of everyday life and not miss out, however some acknowledged limitations in being able to carry out such activities due to transport or financial restrictions, which led them to feel socially isolated. Participants identified the importance of continuing to make a *Contribution*, whether that was giving time or money to family, friends, or the wider community through volunteering. This was seen as a source of satisfaction for many and helped them to feel needed. Often participants put their family members' needs before their own.

Both financial and personal *Security* were identified. Having the resources to provide for the rest of your life was seen as very important, along with daily security, such as

knowing that you could cope with any event. Fear of the unknown and burglars was common amongst older people living alone and there was a recognition that health conditions and physical frailty led to feelings of decreased safety. Health care security was also valued, along with having access to medical care. *Autonomy* was highly valued amongst participants, in particular the ability to make one's own decisions, about what to buy, how to spend one's own time and where to live. Living independently was highly valued even though it contributed to social isolation. A sense of freedom and wellbeing was expressed by individuals in economic terms, i.e., having enough money to do what you want. Many people raised concerns about losing the ability to get around, either by private car or public transport and the impact this would have on them achieving independence and retaining control over their lives.

Finally, *Enjoyment* was identified as a valued functioning. These pleasures included everyday activities, as well as special outings or treats. Having things to look forward to and enjoying rewards was especially important. However, participants often had to adjust to continue doing the things they enjoy as their health declined to bring about happiness and maintain identity. Those with fewer resources tended to emphasise "small pleasures" (p.724), which may not necessarily be health-promotive (e.g., gambling or eating chocolate), but they did contribute to that individual's overall happiness.

Applying a Capability Approach to the quality of life of older people within a UK context, Gilroy (2006) identified several domains that were important to older people themselves, as well as how environments can contribute towards supporting a good quality of life for older people. The domains identified included: health; an adequate income; mobility; a safe neighbourhood; a comfortable and secure home; and social relations and support (pp.346–53). Whilst home was mentioned as an important domain, engaging safely with the broader than home environment was significant. Valued functionings such as being mobile and engaging socially were vital components of this:

Factors that tend to produce a good old age are: a secure home, a supportive neighbourhood, the ability to get out and about, a strong social network, health and income that allow participation in social life, the capacity to make a

contribution to the life of the community, and the ability to access information and activities, including opportunities for learning (Gilroy, 2008, pp.149–152).

Gilroy (2008) notes that there is strong "interdependence" between these domains but that the "qualities of local environment are most fundamental", as place is "the arena in which other elements of a quality old age may be achieved or eroded" (p.152). Gilroy (2006) demonstrates how these vital components of quality of life can often be "compromised by poor policy provision" (p.343) and identifies the need to broaden the focus of ageing policy "from the intensive needs of the frail" and instead consider the ways that older people can be "supported to live lives characterised by independence and well-being" (Gilroy, 2006, p.354). This is also reinforced by Phillips (2018), who notes that "to date, planners and designers have done little to explicitly enhance and support the well-being of older people" (p.68). Meanwhile, Burton et al. (2011) argues that for "ageing in place to work well, housing and neighbourhood environments need to facilitate older people's independence and wellbeing" (p.2).

There is ... increasing recognition of the need to design and plan for ageing populations and communities across the globe in terms of designing suitable, sustainable environments providing the opportunities to age well in place, to retain independence and to be mobile and socially connected (Phillips, 2018, p.68).

In the Netherlands, Meijering et al. (2019) carried out in-depth interviews with older people living both independently and within sheltered housing to explore the various capabilities of importance to older people. They identified three key capabilities that contributed to the achievement of being independent which Meijering et al., (2019) argue is the overarching goal for older people. These included: 1) to be comfortable at home and in the neighbourhood, 2) to enjoy a fulfilling social life (including maintaining reciprocal social relationships) and 3) to be mobile (p.240). Results highlight the ways that capabilities are shaped by both contextual and individual factors, which influence the functioning of being independent and what that looks like to an individual. As a result, it provides a more "nuanced view on how older adults themselves define capabilities that are crucial to their independence" (p.247), as well as how they draw on resources, conversion factors and agency to negotiate this. Capabilities, therefore, are about being able to do something, whereas

functionings are about what an individual actually does, which are weighed up by factoring in the resources or conversion factors they may have. There are then many different pathways to independence with great variation in the specific ways that this is defined and enacted by an individual.

Most recently, Bigonnesse and Chaudhury (2021) have proposed a conceptual framework exploring ageing in place through a Capability Approach lens, emphasising that ageing in place is influenced by the following five components: place integration; place attachment; independence; mobility; and social participation (p.64) and that these components are in turn influenced by the following four factors: individual characteristics; the accessibility of the built environment; proximity of services and amenities; and finally, the development and maintenance of meaningful social connections (p.64). Bigonnesse and Chaudhury (2021) define ageing in place as an "ongoing dynamic process of balance between the demands and resources of the environment and the individual capacities" (p.69).

2.5 Conclusion

This chapter has presented the conceptual framework for this thesis. I began with a critique of the concept of ageing in place, and informed by existing literature, have adopted a geographical interpretation within this thesis, which emphasises the importance of not just ageing at home, but also remaining connected with the broader community and neighbourhood, to maintain a sense of being in place. Recognising that this is easier for some places than others, increasingly the *quality* of the ageing in place experience and how wellbeing is influenced, is regarded as a key concern for academic research.

Within this thesis, I align with literature that explores ageing well through a subjective lens, as opposed to more objective interpretations. This is particularly important when recognising that older adults within Ireland are ageing in place with a variety of health and mobility challenges. I adopt a lay interpretation of ageing well, recognising that only an older person themselves can define what this means and owing to the heterogeneity of older people, this is likely to vary. Within this thesis, I recognise that lay interpretations of ageing well can often conflate ideas of quality of life, wellbeing, health and ageing well. My interpretation of ageing well

therefore incorporates all these elements and I often use these terms interchangeably. Within this thesis I examine the concept of ageing well through a Capability Approach (CA) lens. This is a way of looking at quality of life and wellbeing, which emphasises the importance of being able to carry out valued functionings, i.e., the beings and doings of importance to an individual.

The importance of being and feeling in place in a way that specifically captures the importance of moving beyond the home to engage with a broader neighbourhood environment, has been under-developed within quality of life and ageing well research. However, work within the sub-discipline health geographies of ageing has moved towards an emphasis on ageing *well* in place on the one hand and the importance of *place* for ageing well on the other. This conceptual framework integrates and explicitly links these concepts through a Capability Approach lens. Ageing well in place, examined through this lens, is about having the capability to engage with the broader than home environment and to be able to do this safely and comfortably to enact the specific components that lead to a sense of independence for an individual. It is about carrying out those valued functionings that are of most meaning to an individual and often, these can only happen by engaging with the broader than home environment.

Exploring ageing well in place through a Capability Approach lens, reveals the inherent spatiality of these beings and doings. In most instances, valued functionings, as well as capabilities, happen at a particular time and place (also argued by Robeyns, 2020). Furthermore, many of the valued functionings of importance to people, require older people to leave their homes to engage and participate within their broader neighbourhood and communities. As a result, *place* matters. Existing research has identified many reasons for this and cross-cutting strands across this work include mobility, independence, and social participation or connection. In the next chapter, I consider the existing literature that explores *why* these themes are so important to older people.

This page has been left intentionally blank.

Chapter 3. The Importance of Getting Out and About for Older People: A Literature Review

Within this chapter, I explore the existing literature that identifies why moving and engaging beyond the home is valued by older people. In Section 3.1, I explore the literature on community mobility or 'getting out and about' as a lay term of this, outlining the main ways that it can contribute to older adult wellbeing both as an extrinsic activity (to get to somewhere) and intrinsically (in and of itself) (Graham et al., 2020). I then focus on the possibilities for social interaction whilst out and about, which existing literature suggests is one of the strongest motivators to leave the house for older people. In Section 3.2, I summarise literature that specifies the key places where social interaction occurs beyond the home, as well as the types of social interactions and relationships with others that are most valued. Recognising that getting out and about can be easier for some than others, in Section 3.3 I explore literature that has identified the challenges associated with this, particularly focusing on literature that looks at the dynamic and varied nature of these challenges, which can often be overlooked within policy. I then summarise literature that has explored how these challenges are navigated and negotiated by older people, before concluding the chapter in Section 3.4.

3.1 Community mobility and its importance for older adult wellbeing and quality of life

Existing literature suggests that humans have a fundamental and inherent *need* to leave their homes, expressing the "limits" to the comfort and security a home environment can provide (Tse and Linsey, 2005, p.137). Instead, escaping from the home environment and the tasks within it, on occasion, appears to be vital for quality of life, health, and wellbeing in old age (Gardner, 2014; Olsson et al., 2013). Getting away from the home environment can provide a distraction from the challenges or "troubles" of home life, can help to obtain perspective, and provide a more positive outlook upon return (Beard et al., 2009, p.231). There is a wealth of research demonstrating that the home environment can be a place where negative feelings and emotions such as boredom, frustration, worry, and distress can manifest for older people (Sixsmith and Sixsmith, 1991; Tse and Linsey, 2005). This is particularly the

case when physically or metaphorically "trapped" (Tse and Linsey, 2005; Olsson et al., 2013), "stuck" (Franke et al., 2018; Ziegler and Schwanen, 2011; Coleman and Kearns, 2015; Rogers, 2017), or "imprisoned" (Gardner, 2014; Olsson et al., 2013; Sixsmith and Sixsmith, 1991) within the home (Davidson et al., 1993). Participants have described how leaving the house can offer a change of scenery and fresh air, as opposed to the stale "sameness" at home (Gardner, 2014, p.1252).

3.1.1 Mobility and Wellbeing

Mobility is in general, defined as "the ability to move oneself (e.g., by walking, by using assistive devices, or by using transportation) within community environments that expand from one's home, to the neighborhood, and to regions beyond" (Webber et al., 2010, p.443). Like Gardner (2011), Webber et al. (2010), distinguish between what they describe as different "mobility zones", defined as "concentric areas of expanding locations", beginning within the home and expanding outwards (p.446) (see also Buttimer, 1976). Mobility is commonly measured by defining a person's activity space, or life-space (Baker et al., 2003), which is the "spatial area a person purposefully moves through in daily life" (Tsai et al., 2015, p.e368). There is a growing literature that has examined the activity spaces of older people (see for example, Hirsch et al., 2016; Hirsch et al., 2014; Perchoux et al., 2019; Tsai et al., 2015; 2016). Whilst this is useful for capturing the "extent of travel", it does not tend to capture the experience of travel and how this can influence the quality of life and wellbeing of an individual (Stalvey et al., 1999, p.472). Subjective experiences of mobility have tended to be been overlooked within mobility research (Curl and Musselwhite, 2018), although increasingly the importance of this is being recognised and incorporated (see Franke 2018; Meijering and Weitkamp, 2016; Milton, 2015; Sturge et al., 2021; van Hoven and Meijering, 2019).

Mobility is often used synonymously with the word travel but within this thesis I draw from Metz's (2000) conceptualisation of mobility, which includes both travel but also the additional and often unanticipated benefits to wellbeing and quality of life which are experienced as a result. Metz (2000, p.150) identifies five components of mobility that are important to older people. This includes the traditional view of mobility, which is *"Travel to achieve access to desired people and places"*. In

addition to this, Metz (2000) highlights the importance of the "*Psychological benefits of movement – of "getting out and about*", however they also recognise that these benefits may be "offset by feelings of vulnerability" if this becomes too challenging or risky. Mobility provides "*Exercise benefits*", the extent of which will depend on the mode of travel taken. It also provides opportunities for "*Involvement in the local community*", allowing older people to obtain benefits from their "informal local support networks". Finally, Metz (2000) recognises the value of "*Potential travel*", so that when an individual knows that a trip could be made, there are benefits even if not undertaken. Places therefore do not even need to be actively visited to be meaningful to individuals (see Coleman and Kearns, 2015).

Existing research has therefore shown that mobility, getting outdoors, or getting out and about, is about far more than just travelling from A to B, it is in fact a vital component of wellbeing and quality of life to older people (Alves and Sugiyama, 2006; Carp, 1988; Curl and Musselwhite, 2018; Gabriel and Bowling, 2004; Gilroy, 2006; Goins et al., 2015; Holland et al., 2005; Metz, 2000; Mollenkopf et al., 2011; Musselwhite, 2017; Nordbakke and Schwanen, 2014; Parkhurst et al., 2014; Schwanen and Ziegler, 2011; Sugiyama and Ward-Thompson, 2007; van Hoven and Meijering, 2019; Walsh, 2014; Webber et al., 2010; Ziegler and Schwanen, 2011). Mollenkopf et al. (2011) capture the various ways that mobility provides meaning to older people. They argue that mobility is an "emotional experience" and is something "essential for life itself", identifying physical movement as a "basic human need" (p.789). In addition, mobility allows for engagement in natural environment settings, which can provide additional wellbeing benefits, as well as allow social needs to be met, recognising older adults' "desire for social integration and participation" (p.789). Like Metz (2000), they identify the importance of the possibility of movement and having this available as an "expression of personal autonomy and freedom". Mobility can also be a "stimulating" activity but also a "diversion". Overall, they argue that being able to move is a "reflective expression of the life force one still has" (p.789).

This thesis has been informed by Gardner's (2014) definition of "community mobility" as "independent outdoor locomotion", which includes participation in a range of activities that are both "required for daily living", as well as for "personal

well being" (p.1249). Such a definition recognises that the journey from A to B is a "meaningful lived space in itself" and holds value to older people (see also van Hoven and Meijering, 2019). In a thematic synthesis of qualitative research that explored older people's everyday travel experiences, Graham et al. (2020) highlighted that mobility was valued by older people for both extrinsic reasons such as reaching a specific destination, as well as intrinsic reasons, where travelling was valued and perceived as beneficial to wellbeing in and of itself. Furthermore, Musselwhite (2017) has identified the importance of "discretionary travel", travel for its own sake to older people and how this is important for overall health and wellbeing. Van Hoven and Meijering (2019, p.1) argue that what might appear as more "mundane" or "everyday" forms of mobility, such as short and regular trips in urban neighbourhoods, may be overlooked within urban planning, but play an important role in the overall wellbeing of older adults and contribute to place attachment over time.

The number of outdoor personal projects older adults participate in is positively associated with self-rated quality of life (Curl et al., 2016). However, the nature of these projects is important. For example, a negative relationship was observed between participation in utilitarian projects (i.e. chores) and quality of life. Curl's (2016) findings demonstrate that not all forms of getting outdoors are equally beneficial for wellbeing. If individuals *have* to get out, particularly in more unsupportive environments, this could have adverse impacts on quality of life

3.1.2 Mobility and Independence

A core wellbeing benefit that results from mobility is that it enables older people to be independent (Graham et al., 2020; Schwanen et al., 2012; Schwanen and Ziegler, 2011). Schwanen et al. (2012) conceptualise independent mobility as:

older adults' ability to move fluidly through geographical space; their ability to do things at different sites in geographical space and thereby be socially connected, participate in civil society, and enact desired identities (p.1321).

As highlighted in Chapter 2, independence is the overarching valued functioning of older people and is the portal through which all other valued functionings are

accessed through (Meijering et al., 2019). As a result, mobility is intrinsically linked to older peoples sense of identity and getting out is therefore a way of enacting and preserving that identity (Gardner, 2014; Kellaher et al., 2004; Peace et al., 2005; Schwanen et al., 2012). In a metasynthesis of qualitative studies on older adults' perception of mobility, mobility was found to be an "integral part of sense of self and feeling whole" (Goins et al., 2015, p.935). Recognising the contribution that older people can provide both to others and to their communities (Wiles and Jayasinha, 2013), Croucher et al., (2020) have demonstrated that mobility is important to provide support and can be an "act of care" (Croucher et al., 2020, p.1789). Whilst the ability to get outside is crucial for older adults' sense of independence, research has also shown that independence is a "complex and fuzzy notion" (Schwanen et al., 2012, p.1313), and is typically defined broadly and fluidly by older people, based on existing contexts and abilities that change over time (Allam, 2015). Schwanen et al. (2012) found that participants in their research described dependency as relying on other people for lifts and provoked feelings of being a burden, whereas being independent meant being able to drive, walk, use buses or taxis.

3.1.3 The importance of mobility for social interaction and connectedness

A key reason that mobility (intrinsic or extrinsic) is so highly valued by older people, is because it is important for social interaction and connectedness (Graham et al., 2020). Existing research exploring the lived experiences of older people within their local physical and social environments using 'go-along' methods, has shown that socialising with other people is one of the strongest motivators for leaving the house and getting outdoors (Carroll et al., 2020). For many people, the home can be be a place of loneliness and isolation, particularly if individuals live alone and have few visitors (Gardner, 2014). Often, to overcome a sense of loneliness or social isolation, older people *need* to leave the home. Several studies have shown that older people leave the home to counteract "the loneliness and tedium of home life", which in turn helped them to feel connected to the "wider social world" (Graham et al., 2020, p.864).

Social participation and engagement are crucial components of the overall quality of life of older people (Bowling and Gabriel, 2007; Victor et al., 2004). This is unsurprising, as existing literature has also shown that individuals who have fewer social relationships, are socially isolated, who perceived themselves to be lonely, and who live alone, are at greater risk of premature mortality (Glass et al., 1999; Holt-Lunstad et al., 2015). Furthermore, these risk factors are comparable to well-established mortality risk factors such as obesity and smoking (Holt-Lunstad et al., 2015; Holt-Lunstad et al., 2010). I distinguish here between loneliness as a perceived state and social isolation as a measurable amount of social contact (Holt-Lunstad et al., 2015). Someone may be objectively socially isolated but not feel lonely and vice versa. Equally, someone living alone may or may not feel lonely, depending on the extent that they feel socially connected (Holt-Lunstad et al., 2010).

In ethnographic research with older people aged 75 and over who lived alone in Toronto, Canada, Gardner (2014) observed the "push of loneliness and the pull of "others"" (p.1252), that encouraged older people to leave their homes. When asked why individuals leave the home, the most common response was the "desire to interact with others" (Gardner 2014, p.1252). Gardner's (2014) work highlights the important role of social engagement in community mobility and how getting out and about to interact with others was of great importance to older people themselves. Community mobility is therefore vital for feeling socially connected, which Morgan et al. (2019, p.1126) argues is a "multi-level concept" including interpersonal relationships, as well as more broadly the neighbourhood and wider society.

Connecting back to the need for relational understandings of ageing in place identified in Chapter 2, owing to the importance of the social environment, mobility also needs to be thought about relationally (Ziegler and Schwanen, 2011). This connects to Duff's (2011) work on enabling places, recognising the social, affective, and material resources that can support individuals, as well as the relational nature of agency. In particular, social participation needs to be viewed as "spatially and temporally embedded everyday relational practices" (Ziegler, 2012, p.1297), which are influenced by the design of the built environment owing to the role they play in the "construction" of social relationships. (Phillips, 2018, p.74). Certain places along the way and at destinations appear to be of particular significance for older people to engage with others whilst they are out and about, and these are now considered.

3.2 Places of Importance for Social Interaction

The importance of social connectedness and social participation to older people, raises the importance of considering the types of physical environment characteristics or 'Social Infrastructure', that can provide the "stage on which networds of social interaction take place" (Buttimer, 1976, p.285). Social Infrastructure is defined by Klinenberg (2018) as any place where people congregate, engage or interact in some way:

Social infrastructure is not "social capital"–a concept commonly used to measure people's relationships and interpersonal networks–but the physical conditions that determine whether social capital develops (Klinenberg, 2018, p.5).

Gardner (2011) offers a useful way of exploring this through the identification of the "social places of public life" (p.264) and key "sites of significance" for older people (p.268). This involves three categorisations of places: thresholds, transitory zones, and third places. Thresholds are "hybrid" and "semi-public" areas, building on the work of Peace et al. (2005), who recognise the importance of the "intervening space between accommodation and street life" (p.202). Examples of thresholds would typically be a garden (Peace et al., 2005), but also porches, patios, or balconies (Gardner, 2011). These places provide "easy and readily available opportunities for social interaction", typically involving neighbours (Gardner, 2011, p.266). Existing research has shown that gardens are of great importance to older people for social interaction (see Bhatti, 2006; Bhatti and Church, 2001; Burton et al., 2015; Milligan et al., 2004). For older people less able to get out, thresholds can be of particular significance, for example by enabling them to continue to engage more passively with nature (Orr et al., 2016). These "micro-mobilities" can provide ways for older people to remain connected with the "vitality" of a neighbourhood, as well as help to confirm their "presence and status" (Lord et al., 2011, p.58). The second category is transitory zones, which are not destinations, but instead the "places we pass through during the course of daily public life", often travelling to get to destinations (p.267).

These include footpaths, building lobbies, a train or underground station, a seat on a bus, or a queue at a supermarket or bank. I now consider the importance of social interaction whilst travelling.

3.2.1 Social Interaction Whilst Travelling

While travelling is important to reach destinations for social interaction (discussed later), travelling itself can also be an important "site" and there is a need to recognise *"the social worlds of travel"* (Graham et al., 2020, p.861), with certain modes of more importance for this than others. Existing literature has shown that social exclusion of older people is multi-dimensional and influenced by several inter-connected domains, including: neighbourhood and community; services, amenities and mobility; social relations; material and financial resources; socio-cultural aspects; and civic participation (Walsh et al., 2017, p.92). Within this, the neighbourhood context and the availability of accessible transport vital sub-domains and can support older people to remain connected and socially engaged within their communities (Sutton and Hill, 2010; Walsh et al., 2017). I now consider the key modes of travel: walking, cycling, public transport and driving and how social interaction can vary amongst these.

Walking is an activity that is highly valued amongst many older people (Franke et al., 2013; Graham et al. 2020; Lager et al., 2019). Walking can be beneficial to health and wellbeing through benefits associated with physical activity and exercise (Graham et al., 2020). Depending on where the walking takes place and the quality of this experience, it may also provide therapeutic benefits (see Gatrell, 2013), for example when walking in or near blue, green or white spaces, and in areas of higher aesthetic quality (Day, 2008; Finlay et al., 2015; Roe and McCay, 2021). Furthermore, therapeutic benefits of walking and being outdoors have been found specifically for people with dementia (see Brittain et al., 2010; Cedervall et al., 2015; Odzakovic et al., 2020; Olsson et al., 2013). Engaging with nature is highly valued by older people as a restorative and therapeutic activity, and can be an important motivator to leave the house for a change of scenery and be an important destination in its own right (see Finlay et al., 2015; Milligan et al., 2004; Gardner 2014; Graham et al., 2020). Whilst some older people may prefer to engage with nature to escape,

find solitude, peace, or tranquillity (Orr, 2016), research on green spaces has shown that they can also be important sites for social interaction and developing social networks (Finlay et al., 2015; Gardner, 2011; 2014; Home et al., 2012; Kemperman and Timmermans, 2014; Kweon et al., 1998; Sugiyama et al., 2009; van Eck and Pijpers, 2017). However, research has shown that role of green spaces for social interaction for older people is strongly influenced by perceived safety (Hong et al., 2018; Kemperman and Timmermans, 2014; Sugiyama et al., 2009).

Walking as an activity is important for providing opportunities to socialise with others (Day, 2008; Franke et al., 2013; Lager et al., 2019; van Eck and Pijpers, 2017; van Cauwenberg, 2012; Walsh, 2014) and research in Galway in Ireland has shown that more walkable neighbourhoods can lead to the development of social capital for all ages (Leyden, 2003). Van Eck and Pijpers (2017) explored the daily walking routines of older people in an urban park in the Netherlands using participant observation and walking interviews. They found that going for a walk is a strong motivator for leaving the house for those older people who are in the routine of doing so, highlighting the power of these walking habits and routines to older people. In addition, the layer of social relationships as an important part of this routine was revealed, showing how the "recurring encounters between familiar strangers" are "full of significance" (p.166). Such walking patterns provide opportunities for and can sustain "an atmosphere of fellowship that encourages people to notice, and care for, each other", as well as notice when someone may be absent (p.166).

A study in Copenhagen exploring the importance of neighbourhood open spaces surrounding older people's homes, including local parks and village greens, has shown that walking was more prevalent in areas with adequate shade, seating, landscaping, and where footpaths are in a good condition (Schmidt et al., 2019). Social interaction, however in this instance, was negatively associated with walking, which emphasises that older people in the study often prefer to sit and talk in parks, therefore raising the importance of benches. Other studies have shown that benches can "become like porches" (Ottoni et al., 2016, p.33) in how they foster social interaction, highlighting the need for walkable local environments to include resting places (see also Gardner, 2014). The quality of the local walking environment, however, is a key factor in the levels of social interaction. Van Cauwenberg (2012), for example, highlights that to promote walking for transport, neighbourhood environments should have:

good access to shops and services, well-maintained walking facilities, aesthetically appealing places, streets with little traffic and places for social interaction. In addition, the neighborhood environment should evoke feelings of familiarity and safety from crime (p.1).

This raises the importance of the context in which walking occurs, which can influence the quality of experience, as shown by Curl and Mason (2019). For example, in areas less well served by public transport, individuals may *have* to walk rather than *choose* to walk as an enjoyable activity, which can provide a very different experience and impact either positively or negatively on wellbeing.

Another place that can serve as both an important site for travelling (particularly walking), and serve as a destination, is the high street. Existing research has shown that "street socialising (Day, 2008, p.308) is very important to older people (McDonald et al., 2021; Phillips et al., 2021). In general, research has shown that having destinations and amenities within close proximity to the home can encourage opportunities for social interaction, as well as increase the walking patterns of older people (see Nathan et al., 2012). Day (2008) highlights the way that exercise, walking, nature, and social interaction are interrelated, and has identified five qualities of the local outdoor physical environment that can either support or challenge health. Day (2008) argues that a "healthy outdoor environment" for older people is one that is (pp.304–309):

- clean and free from pollution
- peaceful and quiet
- facilitates physical exercise predominantly through walking
- supports social interaction
- is emotionally uplifting e.g., through pleasing or inspiring views or aesthetics.

Many of these factors are components included within the *Restorative City* framework, which has been informed by theories of restorative environment and the
concept of salutogenesis and is designed to support practitioners to implement what Roe and McCay (2021) call "restorative urbanism" (p.1). As shown in Figure 3.1, the framework identifies seven inter-connected pillars of a restorative city: green, blue, sensory, neighbourly, active, playable, and inclusive and highlights the complex ways that these factors can influence mental health and wellbeing (Roe and McCay, 2021).





In Ireland, results from the Healthy and Positive Ageing Initiative (HaPAI) survey showed that 66% or two-thirds of participants aged 55 and over walked in their local area for health or fitness. Meanwhile one-third either did not walk (29%) or were physically unable to due to mobility challenges (5%) (Gibney, 2018). In the Irish National Travel Survey from 2019, 70.8% of males aged 65–74 walked weekly and of these, 46.4% walked at least 5 times per week (CSO, 2019). This reduced to 39.8% for men aged over 75, whilst 52.2% never take a journey by foot. Similar rates are shown for women aged 65–74, with 69.5% walking weekly, and 41% of walking 5 times a week. However, a much higher percentage of women aged over 75

walked weekly at 55.5%, with 27.9% walking at least five times a week, whilst 33% never walked. The average length of journey taken by foot for men aged 65–74 was 2.6km, compared with 4.3km for women of the same age. For men aged 75 and over 2.6km compared with 2.3km for women. Walking for most older people in Ireland is therefore a daily or weekly practice, although this does decline with age.

Existing literature has shown that cycling can be beneficial for social interaction (see Garrard et al, 2012). However, Ireland does not have the rates of cycling observed in other countries, with just 15% people aged over 18 choosing to cycle as a mode of transport (CSO, 2019). In TILDA (The Irish Longitudinal Study on Ageing), which is a nationally representative cohort study of older adults aged 50 and over, just 6% of people aged 50 and over travelled by either bicycle or motorbike (Donoghue et al., 2017). These low figures reflect the "underdeveloped" cycling infrastructure within Dublin and Ireland (Conway et al., 2019).

Owing to the "public" and "shared" nature of public transport, existing research has shown this mode of travel can be important for social interactions, although this depended on perceptions of the safety of engaging with strangers. As a result, social interaction was not always guaranteed and was often fleeting (Currie and Stanley, 2008). Research in Northern Ireland exploring the experiences of older people using a rural transport community bus service, has shown how it is highly valued for "helping escape isolation, maintaining autonomy, and providing an informal space for relationship building and accessing local news" (Hagan, 2019, p.2519). Initiatives such as free public transport travel, which is the case in Ireland for those aged 66 and over, can support older people to be more mobile, but only if they are able to avail of this (Shannon, 2012). Existing research has shown the importance of concessionary or free bus travel for social interaction amongst older people (Green et al., 2012) and how it is associated with reduced loneliness and depressive symptoms (see Reinhard et al., 2018). However, studies have also shown that many older people face barriers using public transport, owing to unreliable bus services, lack of help with getting on and off the bus, as well as racial discrimination for older adult migrants (Jones et al., 2013; Morgan et al., 2019).

Driving as a mode of travel can lead to reduced opportunities to socialise with others, but this depends on who is in the car and whether individuals are receiving or giving lifts. Being able to drive can provide more opportunities to socialise with others in areas that could be inaccessible otherwise (Ziegler, 2011). Existing research has demonstrated the negative impacts on wellbeing associated with having to give up driving (e.g. Davey et al., 2007; Goins et al., 2015; Mezuk and Rebok, 2008; Zeitler and Buys, 2014) and relying on lifts, as this can lead to individuals feeling like a burden (Schwanen et al., 2012). Literature suggests that older people are "increasingly (auto)mobile" (Böcker et al., 2017, p.831) and this is certainly the case in the Irish context. According to findings from TILDA, almost 90% of people aged 50 and over travel predominantly by car and of these just over 72% drove themselves (Donoghue et al., 2019). Meanwhile, one in ten rely on public transport and this increases to one quarter of people living in Dublin, reflecting the increased availability of services compared to more rural locations. 14% of people aged over 50 use taxis as a form of travel (Donoghue et al., 2017). TILDA research has explored the relationships between transport modes and driving status, with measures of depressive symptons, quality of life, loneliness, monthly involvement in social leisure activities or volunteering, and social network strength (Donoghue et al., 2019). Results found that individuals whose main mode of travel was driving themselves or travelling by public transport reported greater participation in social activities or volunteering, compared to those who relied on a lift from others (Donoghue et al., 2019).

3.2.2 Destinations of Importance for Social Interaction: The Role of 'Third Places'

The final category Gardner (2011) identifies as being an important site for social interaction is *third places*. Originally developed by sociologist Ray Oldenburg, these places are not home (first place), or work (second place), but are "informal public gathering places" (Oldenburg, 1997, p.6). They include places such as bars, main streets, pubs, post offices, cafes or coffee shops, bookstores, and hairdressers (Oldenburg, 1989). In his influential book "*The Great Good Place*", (Oldenburg, 1989) identifies seven key distinguishing features of third places: neutral ground; social levellers; conversation is the main activity; accessible and accommodating;

visited by regulars; a low profile; a playful mood; and a home away from home (p.22–42). Whilst Oldenburg (1989) recognised that third places were important for older people, especially those who are retired, he did not focus specifically on them within his work. However, Gardner (2011) has updated and applied the work of Oldenburg to explore the key third places of importance to older people. These include public parks and certain types of local businesses, including diners, bakeries, barbershops and small grocery stores or local shops. Gardner (2011) argues that smaller "single purpose shops" can be more "accessible" and "comfortable" for older people to use than larger supermarkets (p.66).

Alongside Gardner (2011), there is additional research that has explored the importance of a range of (third) places for older people for the purposes of social interaction, and have shown how they can contribute to wellbeing in later life (Alidoust, et al., 2019; Cheang, 2002; Fong et al., 2002; Hutchinson and Gallant, 2016; McDonald et al., 2021; Morgan et al., 2019). Alidoust et al. (2019) observed older adults aged 65 and over as they participated in third places and carried out semi-structured interviews with over 50 individuals. The field work was carried out in suburban areas in Australia with different built forms, including more conventional suburban developments, as well as master planned areas including agesegregated developments. Results highlighted the important role of third places in providing opportunities to engage with others in the surrounding community, and how this contributed to both social health and wellbeing. Several groups of third places were identified as being important to older people (pp.1470–3). This included mixed-use places such as shopping areas, banks, post offices, and cafes and restaurants. Often these were located together, for example, on a high street and this was valued for convenience and accessibility and to reduce the number of trips needed. Another group of third places were more local and included churches, shopping centres, public libraries, and clubs. Participation in age-segregated clubs and social groups were identified as important and led to older people having a high number of social ties. Finally, common areas and leisure centres in master planned communities were identified, although the frequency of use of these facilities was mixed.

An important consideration of the use of these third places was how accessible they were, and this strongly influenced whether older people could participate. In the more suburban built forms of Alidoust's et al (2019) study, many participants would have found it challenging to travel to these third places without a car and raised concerned about what they would do if no longer able to drive in the future. Participants described how they would avoid travelling in some areas that were busier or less familiar. Many barriers with using public transport were also identified. This was particularly the case for individuals living in cul-de-sacs where the bus stop was located a "considerable distance" from their homes and the frequency was unreliable (Alidoust et al., 2019, p.1466). This suggests that older people living in more suburban locations and no longer able to drive, or have never driven, may struggle to attend third places, and engage in social activities, which could lead to restricted social lives, with implications for quality of life and wellbeing.

The Importance of Social Groups to Older People

Whilst Oldenburg (1989) emphasised the unstructured and informal nature of third places, Gardner (2011), like Alidoust et al. (2019), noted that more formal community organisations and institutions, such as community centres, older adult specific activity centres, and local churches, can "operate" as third places (p.266), applying a far less rigid criteria compared with Oldenburg's original work. These destinations are of relevance to older people and are typically attended on a regular basis and at a specific time, either for all ages or specifically for older people.

Several studies have explored older people's experiences of these social groups or clubs (see Alidoust et al., 2019; Fong, 2020; Gardner, 2011; Hutchinson and Gallant, 2016; Morgan et al., 2019; O'Brien et al., 2004; Tse and Linsey, 2005; Wiles et al., 2009; Ziegler et al., 2012). Whilst several benefits were found, not all experiences were positive and this largely depended on the quality of the space and experiences offered, as well as how on well it aligned with the interests and matched the abilities of attendees. Fong et al. (2020) carried out focus groups with 31 older people who attended the same bridge club in a suburban location in Australia. Results identified the various ways that this club was important to older people, by providing purpose,

a "reason to get out of bed" (p.14), a "permanent place to go", a "stable routine in their daily lives" (p.11), a shared identity, and long-lasting social supports, even when individuals were no longer able to attend. Participants described a "ripple effect" of social interaction (p.18), where attending this group led to additional opportunities for interactions outside of the club. Fong et al., (2020) also described some of the challenges with the club, including practical challenges with using the space due to it not being large enough. Participants described a shared desire to improve the facilities through fundraising and the space provided a base for "collective action", as participants expressed frustration and felt that the provision of community spaces had been "overlooked" by politicians as it was a more affluent neighbourhood (p.16).

Hutchinson and Gallant (2016) carried out community-based participatory research with members of a 'senior centre' in Canada. Whilst results showed that the centre provided opportunities for meaningful engagement, sociability, and providing participants with a role, the opening hours were deemed restrictive and there were limited opportunities to take leadership and be involved in the decision-making of activities. Hutchinson and Gallant (2016) questioned whether senior centres could truly operate as third places owing to the age-segregated nature of them. Another study from Canada evaluated the impact of a community-engaged arts programme on the wellbeing of older people, which involved weekly workshops over a period of three years (Phinney et al., 2014). Results demonstrated that older people improved their perceived health, experience of pain, and sense of community. During group interviews, multiple benefits were identified (pp.340–2). The first was that attending the workshops provided structure and discipline, as well as motivation to "rediscover and sustain a healthy lifestyle through increased activity and interaction" (p.340). Secondly, the programme was helpful for *facilitating coping* and helped participants to "better manage the physical and emotional challenges they faced", as well as provided a "distraction from the discomforts of everyday life" (p.340). Participants enjoyed that the programme provided a "sense of being challenged", requiring hard work (p.341) and through the creative process, allowed participants to "develop and explore their identity", bringing out one's artistic side (p.341). The workshop was important for *promoting social involvement*, providing opportunities for participants to "build a community and create strong social ties", both within the

group and outside the group (p.341). Finally, the programme helped participants to feel like "valued members of society" and that they were *making a contribution* (p.342).

Within an Irish context, using case studies of two towns in a suburban county within the Greater Dublin Area, McDonald et al. (2021) identified places of importance for social interaction of older people and demonstrated how older adults engage socially through daily activities whilst out and about. This included whilst out walking and shopping, as well as through attending organised groups and Mass. In addition, Gallagher (2012) conducted research in Ireland exploring the social and communal life of older people, carrying out ethnographic research, observation work, and interview-based surveys older adults in a suburban area of Dublin city, as well as a rural village in the north-west of Ireland in Co. Donegal. Results confirmed that older people in Ireland had a broad and varying range of leisure interests and engaged in a variety of forms of social engagement. Approximately half of the larger sample in suburban Dublin were members of some sort of social group, which provided opportunities for "meaningful engagement" (p.91). However, results identified that there were some participants that did not attend these types of groups for varying reasons. Some specified that they were not for them, e.g. "I am not a joiner" or that they were "a home bird" (p.91). Others claimed they were shy and needed a bit of encouragement to attend, whilst some did not like groups that were just for older people, preferring activities that included a broader range of ages (Gallagher, 2012).

Gallagher (2012) identified several important social and communal settings, including "sports and social clubs, churches, voluntary groups, day centres, pubs, special interest groups and community events" (Gallagher et al., 2012, p.91). Attending church and Mass was also identified as being of importance to older people. Results emphasise that many older people in Ireland "are connected within vibrant informal ties of neighbouring, church and community settings" and that these are a "rich landscape of relatedness consisting of multidimensional relationships based on kinship and friendship" (p.98). Gallagher (2012) argues that policymakers and service providers need to recognise the "holistic nature of well-being and the enormous value added to people's lives by neighbourhood social groups and clubs" (p.99). To put this into context, TILDA results demonstrate among the 90% of the older Irish population who identified as being a member of a religious group (of which 90% were Catholic), 67% of older people aged 65–74 attended a religious service at least once per week and this increased to 76% for those aged 75 and over. Religion was identified as being more important to women than men (Timonen et al., 2011). Meanwhile, results from the HaPAI Survey reveal that 48% of people aged 55 and over participate in community activities at least once a month.

Existing literature has highlighted the importance of paying attention to the daily rhythms and routines of older people, as this can influence the experience of ageing in place (Franke, 2013; Lager et al., 2016). Informed by the work of Lefebvre (2014) on Rhythmanalysis, Lager et al. (2016) explored the daily rhythms and routines of older people in the Netherlands using go-along interviews. Results revealed a "slowing down" of activities in many instances for participants, along with a "shrinking life world" (p.1571). Alongside this, there was evidence of participants participating in regular activities and routines as "anchors", to provide structure to their days. This included walking, shopping, cleaning, and participating in club activities, "punctuating time" to make life more eventful and to ensure that a perception of "keeping busy" was maintained, the "preferred rhythm" of most participants in the study (p.1574). Results highlighted some "discontinuities" between generations, as well as with older people themselves and their expectations based on their younger selves (p.1574). This led to older people feeling "out of synchronicity in time and place" (p.1575). They were also influenced by seasonal rhythms which influenced their daily experiences. Furthermore, research elsewhere has highlighted that mobility patterns can vary across the day, with many older people less likely to travel at night (Scharf et al., 2001; Smith, 2009).

Overall, travelling to a destination can provide additional motivation to get out, by providing "structure and focus to the trip" and "enhance the opportunities to socially interact with others" (Graham et al., 2020, p.856). Furthermore, the availability of local destinations to visit can enhance motivation and experience of travel and provide additional opportunities to socially interact with others (Graham et al., 2020). I now consider the nature of relationships and interactions that can happen

whilst engaging in the different sites or phases of getting out and about identified by Gardner (2011).

3.2.3 Nature of Social Interactions and Relationships

Klinenberg (2018) argues that different types of social infrastructure "play different roles" within a community or neighbourhood and in turn foster different forms of "social ties":

Some places, such as libraries, YMCAs, and schools, provide space for recurring interaction, often programmes, and tend to encourage more durable relationships. Others, such as playgrounds and street markets, tend to support looser connection – but of course, these ties can, and sometimes do, grow more substantial if the interactions become more frequent or the parties establish a deeper bond (Klinenberg, 2018, pp.17–18).

Gardner (2011) identifies three types of relationships that older adults typically experience whilst out and about. These include: relationships of proximity (for example with neighbours), relationships of service (for example with cab drivers, wait staff and 'sales clerks', and finally, relationships of chance (typically with strangers) (pp.267–8). Gardner (2011) demonstrates how these relationships combine and culminate to create an "informal, neighborhood social network", or what she describes as a "natural neighborhood network" (p.267), defined as:

a web of informal relationships and interactions that enhance well being and shape the everyday social world of older adults aging in place (p.268).

Gardner (2011) considers the different types of interactions that occur within third places and transitory zones and argues both tend to encourage more "natural relationships and interactions":

Natural in this context means they are not 'forced' or 'formal' (i.e., they were not paid service staff, volunteers from support agencies, or healthcare professionals), nor are they 'familial'. Instead, these interactions are more universally shared (e.g., across age groups), often spontaneous, informal, everyday encounters and relationships with non-family members (p.267). Furthermore, she argues that these networks are situated within, and complement a layer of additional forms of informal and formal support systems:

Natural neighborhood networks do not replace or negate the importance of informal systems of family and friends, or formal support systems provided by public and private agencies and services. They complement them (p.269).

In a systematic literature review of the social needs of older people Bruggencate et al. (2017), found a diverse array of social needs. Expectations of social needs and neighborliness can vary significantly between older people (Yen et al., 2012). Both close relationships, as well as peripheral relationships, are important to older people and contribute to a sense of social connectedness and safety. Existing research has highlighted the role of third places and social groups as being important for the development of peripheral relationships. For example, Alidoust et al. (2019) identified that the nature of social ties within the bridge club were predominantly weak, where there may be some degree of acquaintance, or absent, where an individual may merely nod to a stranger. In some instances, they were stronger and had led to friendships and deeper ties. In the mixed-use third places, participants would often meet people they already had strong ties with, which were developed through participation in other activities.

The importance of informal, or weak ties to older people has been shown elsewhere, for example Franke et al., (2019) has highlighted the importance of "superficial contacts" to older people, which are the "everyday encounters people have with others they do not know while out" (p.1654). However, existing research has shown instances where opportunities for this have decreased, particularly in areas experiencing change or where there are fewer reasons to engage with others (see Ziegler, 2012; Day, 2008). Day (2008) compared the experiences of older people living either in inner city or suburban locations and found fewer opportunities to walk and visit local services in the suburban locations, and thus fewer opportunities to engage with others. Ziegler (2012) has highlighted that thresholds and streets can be important places for social interaction. However, presenting two in-depth case studies of working-class older women, Ziegler (2012) showed how social interaction had reduced in these places over time, predominantly due to the presence of cars.

Instead, the women had to rely on participating in social clubs for interaction with others.

Research has shown that older peoples' relationships with others can change over time and across the lifecourse. Some older people may pro-actively choose to reduce their ties "in accordance with one's age-specific needs" and this may contribute to improved wellbeing (Lang, 2001, p.321). Owing to the diverse social needs of older people, it is important from a Capability Approach perspective, that the social environment can match expectations or desired functionings, so that an individual can engage to the level that they themselves seek or desire. Some older adults are content with more limited encounters, whilst others may prefer solitude at varying times (Cattell et al., 2008) and that this is likely to be influenced by their situation at home. For example, research by Ziegler and Schwanen (2011) emphasised the importance of finding "a balance between social contact outside the house and solitude at home" and having the choice about "when to interact with others" (Ziegler and Schwanen, 2011, p.771). This research highlights that valued functionings, along with the social environments in which older adults engage with, are fluid and diverse. Wiles et al., (2009) have developed the concept of "social space" to capture this, recognising older peoples' elastic physical, imaginative, emotional and symbolic experiences of and connections to people and place across time and in scope" (p.664).

3.3 Navigating the dynamic challenges that can be associated with getting out and about

Whilst leaving the home has the potential for a variety of wellbeing benefits, depending on an individuals' personal and environmental context, it can also present challenges, fear, and risk. When an individual deems these to be too great, it may be altogether avoided (Peace et al., 2011). This highlights that whilst community mobility is usually valued by older people, the process is "complex, dynamic and often difficult" (Gardner, 2014, p.1249). Existing research has shown that as people age, their activity space may shrink and the importance of the immediate local environment becomes more apparent (Moeyersons et al., 2022; Schwanen and Páez, 2010). However, the relationship between activity space and the ability to be mobile

is complex, as certain modes of travel can lead to greater activity spaces than others (driving versus walking for example). The relationship between life-space, social connectedness and wellbeing is also complex. For example, a qualitative GIS approach to older adult mobility has shown that the way 'local environments' are defined by older people can "stretch" beyond administrative and neighbourhood boundaries and that conceptualisations of neighbourhood can change seasonally and over the lifecourse. Furthermore, mobility tends to be associated more with social factors than places (Milton, 2015). Nonetheless, research has shown that as lifespace declines, so too does quality of life (Rantakokko et al., 2016) and that barriers to both mobility and outdoor social participation are associated with both increased loneliness and isolation (Rantakokko et al., 2014), as well as a reduction in social connectedness (Morgan et al., 2019). For those individuals experiencing reduced mobility and loneliness, they report a sense of a loosening "grip on the world" and describe a "literally and figuratively shrinking world" (Moeyersons et al., 2022, p.1).

A reduction in mobility does not necessarily reduce social connectedness, reduced wellbeing and increased social isolation, although research has shown it does "increase the necessity for leveraging pre-existing social capital" (p.1133). This can lead to feelings of burden. As a result, the experience of reduced mobility is influenced by the extent of an individual's existing social network and secondly, the willingness of the older person to draw on this. As a result, the size of an older persons' activity space does not always equate with reductions in wellbeing, and there is a need to consider the on-going experience of getting out and about. The size of an activity space does not appear to be as important for wellbeing, as what is available to an individual within it, the quality of this experience, or how well a transition or change is managed and coped with.

There is a wealth of existing literature that has identified features or characteristics of the local environment which can be barriers for older people getting out, particularly those older people whose mobility may be constrained (see Brookfield et al., 2017; Gardner, 2014; Gilroy, 2006; Lavery et al., 1996; Newton et al., 2010; World Health Organization, 2007). As highlighted in Chapter 2, many studies have described the ageing in place experiences of older people who are living in environments that are not conducive to ageing well. Both the design and

maintenance of the external physical environment can influence the ability of older people to get out and about (Newton et al., 2010), as well as a variety of material, social and psychological factors (Holland et al., 2005). During in-depth interviews including structured questionnaires and photo elicitation methods with 200 older people aged 65 and over in the UK, older adults identified several built environment characteristics that influenced their decisions about getting out and their sense of safety. Approximately half of the participants identified having mobility, vision, or hearing difficulties, which limited their daily activities. Just over one-third used some form of mobility aid, and one-fifth had either stumbled or fallen outside within the six months prior to the interview. Salient features included the need for wellmaintained, firm, flat and wide pavements (including steps and ramps); safe pedestrian crossings; adequate seating to provide places to rest between home and local facilities; well-maintained greenery to improve the walking experience; bus shelters with seating; public toilets; and simple, easily visible, and understandable signage (pp.26–8).

Alongside a growing awareness of the need to plan for and design age friendly environments, and increasing understanding about environmental characteristics that may serve as barriers to older people, it is increasingly recognised that "an 'optimal' or 'ideal' environment for 'ageing well' is complex and multifaceted" (Phillips, 2018, p.68). Approaches that focus on more objective and universal environmental standards, characteristics or barriers have been criticised for a failure to recognise the heterogeneity of the older adult population and their experiences, and that what may serve as a barrier for one individual, may not for another (Phillips, 2018). The use of universal standards, therefore ignore context and difference (Buffel et al., 2018; Phillips, 2018; Handler 2014). Phillips (2018, p.70) raises an important question as part of this critique: "What kind of older people are age-friendly communities trying to reach?". This creates challenges, because the needs of older people are so different (Lavery, et al., 1996). Focusing soley on the needs of one group may serve to reinforce the exclusion of another, unless this diversity is recognised (Phillips, 2018). A key challenge for policymakers, planners and service providers, therefore, is how to plan for the varying needs and wishes of older people in a way that captures diversity of experience, recognises dynamic processes that influence both personal and environmental characteristics, as well as what is valued by the older person

themselves, whilst also recognising that existing inequalities that make it easier for some to age well than others?

Examining mobility from a Capability Approach lens, mobility can be defined as "the ability to choose where and when to travel and which activities to participate in outside the home in everyday life" (Nordbakke, 2013, p.166). Nordbakke (2013) shows that the opportunity to be mobile is not "fixed", instead it is "managed, shaped and directed by the individual" (p.166). Individual resources and contexts at a given moment in time, provide the opportunity to be mobile, as well as the strategies available to overcome barriers, which can then shape overall mobility patterns. Ryan and Wretstrand (2019) have applied a Capability Approach framework to explore the link between modes of travel available to older people and opportunities to participate in everday activities. Results highlight that a combination of health challenges and a lack of public transport infrastructure, particularly lack of services, can lead to participants being unable to have "the possibility to participate" in activities that were valued (Ryan and Wretstrand, 2019, p.107). However, research has also shown that older people can be highly strategic decision makers when determining how to interact with their broader than home environment (Holland et al., 2005).

Components that tend not to be incorporated into age friendly environment design are the more intangible qualities that are often of importance to older people themselves, as well as the dynamic nature of the person-environment relationship. Existing research has shown that health professionals and older adults themselves differ in the identification of places of importance for ageing in place. Health professionals are more likely to report more objective characteristics such as access to amenities, mobility, and meeting places, which can support older people to live independently. Older people, however, place more emphasis on their "specific lived experience and attachment to specific, intangible and memory-laden public places" (van Hees et al., 2017, p.11). Often these valued components can only be captured through more novel and creative methodologies (Handler, 2014) and this is discussed more in Chapter 4. The complexity of personal and environmental factors and how they interact to influence the experience, capability, and frequency of an individual to get out and about, as well as how this varies over time, shows that a "one size fits

all" approach to age-friendly environments is simply not possible (Hammond and Saunders, 2021, p.10).

Examples of dynamic environmental factors include weather conditions, as well as changes to the social and physical environments. Several studies have identified that certain weather and seasonal conditions can be a barrier to getting out and about, including the impact of rain on walking (Franke et al., 2013). Gardner (2014) showed that weather was a "constant consideration" for participants, and this was particularly the case during winter months (p.1252). This is reinforced by Hjorthol (2013), who showed differences in older adult mobility patterns between the summer and winter in Norway, and that winter weather conditions can produce additional risks to older people, particularly when using pavements. In a study in the US, Finlay (2018) explored the experiences of white spaces and found that both snowfall and icy weather conditions impacted the physical and psychological health of older people when they were unable to get out. However, this research also showed some of the benefits that can be experienced due to this weather, including the beautiful scenery, satisfaction that comes with shovelling snow, as well as strategies to navigate this, including the need to "walk like a penguin" (p.81).

Existing research has shown that neighbourhoods experiencing significant change through urban renewal or demographic composition, can influence the amount of social interaction available for older people and lead to them feeling dis-connected (Lager et al., 2013; Morgan et al., 2019). Furthermore, visiting unfamiliar local environments, whether due to environmental change or visiting new places can cause older people additional challenges (Phillips et al., 2013), and this can be particularly problematic for people with dementia (Brittain 2020, p.281). Research that has explored the experiences of people with dementia has highlighted the increased risk of getting lost, and that often this is more of a concern for carers than for the individuals themselves (McCabe and Innes, 2013). Some strategies to manage this include staying within more familiar areas, as well as relying on social networks for support. This is particularly the case in smaller neighbourhoods where people are known. Another adaptation mentioned by people with dementia is attending events for shorter periods of time (Sturge et al. 2021), allowing individuals to still benefit from getting out of the house. This highlights that changing mobility can often

"involve difficult decisions that were complex, emotional, and often influenced by others" (Goins et al., 2015, p.938).

Gardner (2014) highlights the remarkable determination of older people to continue getting out and about, despite facing numerous challenges, and identifies several adaptations older people made as they navigated this, recognising that:

Participants' ability and willingness to journey into their neighbourhoods were challenged by a myriad of individual and environmental factors that changed from one day to the next (Gardner, 2014, p.1249).

Examples of adaptations include (p.1251): being flexible and changing plans if needed, adapting travel routes, and only availing of those that are perceived as safer, even if this takes longer, as well reconsidering or consolidating errands to reduce the number of trips taken. In some instances, participants physically prepared to go out by having a nap beforehand. Other considerations included choosing between more comfortable clothing and more weather appropriate clothing. In a study with 34 urban and rural older adults aged 70 and over, Rudman and Durdle (2009) explored the lived experience of older people with low vision. Results showed that a core aspect of their experience was "living with a pervasive sense of fear regarding one's body and way of being" (p.106). Results revealed that participants "continually gauged risks associated with mobility" and "engaged in risk avoidance and management strategies" (Rudman and Durdle, 2009, p.106). These perceived and physical risks often led to restricted community mobility, which in turn led to a reduced ability to participate in a variety of physical and social activities (Rudman and Durdle, 2009).

Another strategy available to older people with declining physical functioning is the use of mobility assistance devices. However, the use of assistive devices can bring mixed feelings, with participants describing both benefits including improved function, reduced main, stability, and psychological security, as well as barriers including embarrassment and a fear that this would lead to dependence (Goins et al., 2015, p.938). Often there was resistance to the use of mobility devices and their use would be delayed and only used as a "last resort". In some instances, individuals felt that their lives had improved, because of "renewed access" to environments that had

previously been unavailable to them and due to increased opportunity for more "supported social interaction" (Goins et al., 2015, p.938). For older people experiencing health challenges, ageing in place in suburban environments can lead to restricted mobility, with individuals having to prioritise travel closer to home. Lord et al., (2011) revealed how driving as a mode of travel for these individuals can be "the last chance for getting around independently" but is also adapted by older people in this situation where only those routes that are perceved as safer are selected, for example, those where disabled parking might be guaranteed (p.58). Lord et al. (2011) argue that this can lead to a more "fragmentation" (p.59) of someone's activity space, where parts of the city are "forgotten" (p.58).

For some individuals, the last response to increased mobility challenges is to disengage. Wiles (2003) demonstrates how some older people become more disinterested in getting out, even when it was previously enjoyed. Told from the perspective of a daughter who described how she tried to motivate her older adult parents, she revealed how they were no longer interested in getting out. Based on findings elsewhere, this may because it had become too difficult, was too much effort, they didn't want to ask for help, or had reluctantly accepted that it was no longer possible. For those older adults who become homebound, it is therefore likely that they have to give up leaving the house because it is perceived as too risky or because they are no longer physically unable to, rather than no longer desiring it. Whilst homebound older adults are at greater risk of depressive symptoms (Choi and McDougall, 2007) and health challenges (Cheng, et al., 2020), coping strategies and resources such as social support and engagement in frequent exercise, can help buffer these negative effects.

3.4 Conclusion

Within this chapter, I have considered literature that explores the importance of community mobility or getting out and about for wellbeing. Reviewing this literature, it becomes apparent that a vital component of the overall experience of ageing in place, stems from the experiences of the broader than home environment:

aging well *in place* goes beyond being able to live independently in one's own home; it also encompasses feeling socially connected with one's local community (Fong et al., 2020, p.2).

Existing literature has shown that community mobility is vital for older adult wellbeing and quality of life for a variety of overlapping reasons, of which opportunities to connect with others and socially interact appears to be primary motivator for leaving the home. Whilst getting out and about has the potential to provide a range of wellbeing benefits to older people, the quality of this experience will be influenced by the ease with which an individual can get out, as well as the opportunities that getting out provide. Research has demonstrated that getting out and about can be challenging and difficult in certain contexts and for certain older adults, and that this is continually negotiated between older people and their environments. Even with considerable barriers to getting out, participants still chose to "maintain their community mobility" and "rarely elected to stay home" (Gardner, 2014, p.1252). As Holland et al., 2005 comment:

People in general want and need to get out and about. Older people are no exception: shopping, working, socializing, using services, and giving and receiving support. Beyond the practical, most older people also relish the pleasures and challenges of life beyond the home and strive to maintain for as long as possible their independence and ability to get out and about. (Holland et al., 2015, p.49).

If we can better understand what is of most importance to people and how easy it is to attain these wishes, we may be better able to support older people to achieve those activities of most importance for a good quality of life. Recognising the impact that not engaging beyond the home can have on quality of life, supporting older people to extend this process for as long as possible is vital. Having the capability to be mobile and get out and about are therefore vital components of ageing *well* in place. Furthermore, challenges with getting out and about may be an indication that an individual may be finding it difficult to access and engage with their most valued functionings and may not be ageing in place as well as they could or might wish.

This research will explore what ageing well in place means to older people themselves and how they both define and enact it during their everyday lives. I am interested in the lived experience of mobility for older adults, including those valued functionings that happen beyond the home, including the places, routes, routines, and interactions of importance to older people. I am interested in seeing how this is navigated by older people in Ireland, focusing on suburban settings, which appear to include many features which may constitute an unsupportive environment. To explore ageing well in place from the perspective of older people, will require spatial, subjective, and qualitative approaches, and with this in mind I turn now to the methodological approach taken for this thesis.

This page has been left intentionally blank.

Chapter 4. Methodology

4.1 Introduction

My overall aim for this thesis is to explore how well older people are ageing in place. I am interested in how easy and important it is for older people to get out and about to engage with meaningful activities, interactions, routines, and places, recognising that this is likely to have significant implications on their overall quality of life. As I am interested in individual experiences beyond the home space of older people, the most logical way to explore this is to apply a combination of both qualitative and spatial research methods. Within this chapter, I outline how I designed and carried out my fieldwork. I begin by outlining the overarching ontological, epistemological, and methodological underpinnings of this thesis (Section 4.2). From this, I demonstrate how applying a qualitative and geo-spatial approach aligned with the philosophical underpinnings of this thesis. In Section 4.3, I outline the study design and key methods used for the empirical work presented in later chapters. In Section 4.4, I discuss and reflect on the sampling and recruitment processes, as well as how I determined the study settings for this research. A vital component of this thesis was carrying out a flexible, inclusive, and care-full approach and in Section 4.5, I reflect on the ethical considerations. In Section 4.6, I identify the data processing steps to manage my data and in Section 4.7, I summarise the approaches taken to analyse my data and how I adapted my approach for different empirical chapters and research questions, before concluding in Section 4.8.

4.2 Philosophical Underpinnings of the Thesis

Building on the theoretical principles established in Chapters 1, 2 and 3, I now summarise the ontological, epistemological, and methodological underpinnings, which informed the study design and methods used for this project. A summary of these can be found in Appendix 1.

4.2.1 Ontological and Epistemological Assumptions

Ontology is the study of being and the nature of reality or the social world (Bryman, 2016; Denzin, 2008, p.31; Denzin and Lincoln, 2018). Closely related to ontology, epistemology is defined as the theory of knowledge and how social worlds should be studied by the researcher (Bryman, 2016; Denzin and Lincoln, 2018). Objectivism is an ontological stance, where social phenomena have a reality that is external to individuals that experience them (Bryman, 2016). This stems from more positivist epistemological approaches, where there is one objective truth that is independent of social actors. Constructionism or constructivism, on the other hand, is associated with interpretivist epistemological stances, and asserts that social phenomena and forms of knowledge are created by social actors through their perceptions, actions, and experiences (Bryman, 2016).

My worldview is pragmatic in nature, where depending on the research question in mind, it may be necessary to "blur" traditional paradigm dualities and incorporate objective and/or subjective approaches (Bryman, 2016). I accept the existence of both objective and subjective forms of truth, a perspective common within more applied and mixed methods forms of research (Tashakkori and Teddlie, 1998). Adopting a pragmatic stance, I recognise the existence of an objective or single "real world" but that in addition to this, I acknowledge that individuals can have a unique interpretation of this world, depending on their experiences and that "knowledge is produced through experience" (Elwood, 2010, p.99; Morgan, 2007). Within this research I considered "what difference it makes" to believe one form of truth over another (Morgan, 2007, p.68).

Whilst I accept the existence of both subjective and objective forms of knowledge, within this thesis I focus on *subjective* experiences. In my application of pragmatism, I lean towards a constructionist or relativist ontology and interpretivist or subjectivist epistemology. In doing so, I recognise that there are "multiple realities" and both the researcher and researched "co-create understanding" using naturalistic methodological approaches (Denzin and Lincoln, 2018, p.20). This means that to understand the "complex social world", I need to see it "from the point of view of those who operate within it" (Goodson and Phillimore, 2004, p.36). Within this

thesis, I recognise the diversity of experience and interpretations and that no two people will experience a place in the same way. As a result, I follow an interpretivist approach, which values and incorporates the "meaning that social actors give to events and behaviour" (Bowling, 2014, p.433). From a health geography perspective, this recognises that:

meanings (of health, illness, experience) are *constructed* out of the interactions (which may be conversations or encounters) that we have with each other in everyday life (Gatrell and Elliott, 2015, pp.37–8).

There are two main ways that I have applied this stance throughout my thesis. The first is how I think about health and wellbeing, including concepts such as quality of life and ageing well. The second is how I think about place. Taking the example of defining a good quality of life, I recognise that quality of life can be defined objectively. An example of this would be CASP-19, used in The Irish Longitudinal Study of Ageing (TILDA), which defines quality of life through a needs model and identifies four key domains: Control, Autonomy, Self-realisation, and Pleasure (McGee et al., 2011). Defining quality of life in this way allows researchers to carry out assessments with large numbers of older people and allows for comparison across multiple datasets. A criticism of more objective quality of life definitions, however, is that they do not provide space to consider how an individual may define a good quality of life and what components may be more or less important to them. Within this thesis, I have adopted Sen's (1993) Capability Approach to well-being to explore the quality of life of my participants. This allows me to consider subjective definitions that focus on valued beings and doings that are of most importance to individuals. Whilst I recognise the existence of both objective and subjective ways of defining a good quality of life, in this thesis, I place greater importance on subjective forms of knowledge within this thesis, where the voice of "ordinary people", or lay people, have equal "status or validity" as that of a health professional (Gatrell and Elliott, 2015, p.38).

Turning now to place, within this thesis I recognise that there are multiple environmental layers of both objective and subjective reality. Whilst positivist approaches can be used to record or identify objective characteristics about an individual or their neighbourhood context, they are less suited to capture what that

means to the individual, for example their subjective (or psychosocial) experience. For example, in an older person's local environment, there may be certain places or characteristics that can be quantified and the existence of these is not questioned. An example could be a transport network or whether certain places or facilities are physically present. However, alongside this layer of objective reality, there are layers of subjective experience and interpretation. I recognise that just because a place is physically present, does not mean it will be used or experienced; instead as researchers we need to understand the "true" environmental exposure through activity spaces (Perchoux et al., 2013). Furthermore, to understand more fully "the social and spatial processes involved in shaping health-related behaviors and outcomes", we need to understand how people experience, interpret, and use their surroundings, rather than just focusing on what is merely present (Gatrell and Elliott, 2015, p.38). Within this thesis, I also value and incorporate "experiential knowledge" (Elwood, 2006), such as layers of perception and emotion, such as how an individual feels when they engage with a particular place. These perceptions are socially constructed and influenced by past experiences, as well as what is important to that individual, where "meanings of place are created through practice" (Cresswell, 1996, p.17).

The existence of both subjective and objective experiences of place has long been recognised within humanistic geography and the social sciences. For example, in *The Production of Space*, Lefebvre (1991) distinguishes between conceived, perceived, and representational space, recognising that place has physical, social, and experiential components (Pierce and Martin, 2015). Meanwhile, Tuan (1977) distinguishes between place and space, where place has layers or meaning, experience, perception, and attachment, whereas space is a more abstract notion about the relations between activities at different places. Considering older adults' subjective experience of place, also necessitates thinking about time, of which there are both objective and perceived time (Dodgshon, 2008) and everyday rhythms (Lager et al., 2016; Lefebvre et al., 2004).

A social interactionist perspective will tell part of this story, an individual's experience, and agency. However, there are often structural factors which can influence an individual's agency and their ability to carry out activities or

interactions that are most valued and important to them. Such an approach does not consider how unequal opportunities and individual circumstances can unfairly limit the ability to achieve health or a concept such as successful ageing. To recognise this, the thesis takes a structurationist stance. This recognises the role and the importance of the individual (their agency), but also recognises wider social and political forces (structure) that may influence the ability of individuals to achieve good health or to age well. Most notably, I recognise the role of health inequalities that can accumulate and produce a lifetime of advantage or disadvantage (Gilroy, 2021) A structurationist approach considers both these approaches and comes to a middle ground, in that it recognises the "duality of structure and human agency" (Gatrell and Elliott, 2015, p.50), where everyday life is "located within broader relations and distributions of power that play out unevenly within the particularities of time and place" (Dyck and McLaren, 2004, p.513).

I value and focus on subjective experiences within this thesis, as I believe they are underrepresented within both Healthy Urban Planning and Health Geography. Within this thesis, I place an emphasis on the importance of lay forms of knowledge. This stems from my background as a planner in practice and from my interest in community engagement and participatory planning practices. In terms of what difference it makes, there are existing power imbalances, where objective and quantitative approaches are often prioritised over qualitative, lay, and subjective experiences (Elwood, 2006). Furthermore, research has shown that more "official" versions of reality and sources of data, do not always align with "lived' reality" (Knigge and Cope, 2009, p.104; Richard et al., 2005). A reliance on objective approaches alone risks excluding and undervaluing human centred experiences, which will have implications on the type and quality of places that we produce and value. This thesis therefore emphasises subjective and experiential knowledge to redress this power imbalance. As I will show in the next section, such an epistemological and ontological stance requires the use of a qualitative and spatial methodological approach.

4.2.2 Methodological Approach

Owing to the emphasis on the *subjective* lived experiences of my participants within this research project, I have adopted a qualitative methodological approach. I distinguish here between qualitative research that can refer to a particular qualitative *method*, which are techniques or approaches to data collection or analysis (commonly known as 'small q' research), and a qualitative research *methodology*, which is more than a method but a wider framework or paradigm in which to conduct qualitative research (known as "Big Q" research) (Braun and Clarke, 2013; Goodson and Phillimore, 2004; Kidder and Fine, 1987).

The qualitative research carried out within this thesis is Big Q research, in that I have applied a methodology built on the aligned epistemological and ontological underpinnings outlined in the previous sections. Using a qualitative inquiry, I place an emphasis on "understanding the world from the perspective of its participants" and value the importance of both interactions and interpretations of social life (Goodson and Phillimore, 2004, p.4). Within this thesis, I am concerned with participants' "subjective understanding of social reality" (Limb and Dwyer, 2001, p.6), recognising that qualitative research can provide "new insights and understanding about individual and social complexity" (Saldaña et al., 2011, p.4). Research within Health Geography and Gerontology is increasingly adopting qualitative methodologies to understand the social world of groups and the meaning(s) individuals place on this (Bowling, 2014; Fenton et al., 2016; Goodson and Phillimore, 2004, Phoenix, 2018).

Human and health geographers recognise the value of qualitative research and the need to access and understand the subjective and individual experience within particular places or settings where these meanings or interpretations develop (De Lyser et al., 2010; Limb and Dwyer, 2001). This requires researchers to recognise the:

complexity of everyday reality, the multitude of influences that shape lived experience, and the importance of the spatial contexts of human interaction (De Lyser et al., 2010, p.7). To do this, it is important to think both qualitatively and spatially. In addition to traditional qualitative approaches, health geographers are increasingly adopting more innovative and diverse methods and methodologies as part of this (Fenton et al., 2016). This includes the use of more geo-spatial and in-situ methods (Hand et al., 2017; Bell et al., 2015; Foley et al., 2020; van Hees et al., 2017). These approaches allow researchers to better understand "situated" meaning and experiences and to "interpret qualitative expressions of spatial knowledge" (Bell et al., 2015; De Lyser et al., 2010; Elwood, 2010; Fenton et al., 2016, p.4). As a result, I adopt a methodological approach that combines both qualitative and spatial or in-situ approaches within this thesis.

I am using spatial approaches in a qualitative (Big Q) way using multiple-methods, as opposed to mixed-methods research which combine both qualitative and quantitative methodologies. In doing so, I draw from existing Qualitative GIS literature, which recognises the qualitative potential of GIS and mapping. Qualitative GIS stems from critical and feminist GIS perspectives, which recognise that GIS is no longer tied to its originating epistemology (Cope and Elwood, 2009). Instead, GIS or spatial approaches (including technologies and applications) are not fixed but follow along more pragmatic lines depending on who is using it and how (Pavlovskaya, 2017; Cope and Elwood, 2009; Kwan, 2002). Through the work of Qualitative GIS, traditional GIS approaches have been challenged at an epistemological level and have been reimagined for qualitative research (Pavlovskaya, 2017). Although this research was informed by Qualitative GIS approaches, I prefer to use the term "geo-spatial" to reflect the methodological approach more accurately. This is because Qualitative GIS would traditionally involve using a mixed method approach including the use of more quantitative forms of data and the use of GIS databases to represent qualitative data (Cope and Elwood, 2009). Instead, my research was multi-qualitative methods and was geo-spatial in that "the participant [was] actively involved in collecting data related to locations or places" (Hand et al., 2017, p.e50). In short, GIS, geo-spatial approaches and mapping-based research can incorporate subjective representations, experiences, perceptions, and meanings; therefore, it can be qualitative.

4.3 Study Design and Methods

I now turn to the study design and data collection methods used for this thesis. I have applied a multi-staged and multi-method study design combining qualitative and geo-spatial methods. Data collection consisted of two stages and the use of several in-situ methods alongside more traditional qualitative methods such as interviews and focus groups. Figure 4.1 summarises the overall study design of this thesis.



Figure 4.1 Study Design Diagram

4.3.1 Stage 1: Exploratory Focus Groups

Stage 1 involved carrying out four exploratory focus groups with older people as a "preliminary data gathering" exercise (Barrett and Kirk, 2000, p.622). The aim of these focus groups was to acquaint myself to the Irish ageing experience. As I had only recently moved to Ireland from the UK in September 2015, it was important for me to obtain insight from older people in Ireland before carrying out the main research project, to ensure that I was not making any assumptions about this population group. Four focus groups were carried out with older adults (n=31) in June 2017 and lasted between 43 minutes and 1 hour 19 minutes. The purpose of these focus groups was to identify features of their local neighbourhoods that were important to them, to identify challenges older adults face in Ireland, and to consider what a supportive or unsupportive environment might look like for certain older adult sub-groups. I asked my participants about techniques for recruiting harder to reach population groups and for ideas about where to carry out my main study. For a full list of questions asked in the focus groups see Appendix 2.

The focus groups were helpful in that they allowed me to familiarise and explore this topic. The group dynamic provided a stimulating discussion and insight into the Irish context of ageing in place (Bowling, 2014). However, a key limitation of these focus groups was that they did not allow for in-depth examination of what was of most important to individuals. Furthermore, the group dynamic and camaraderie meant that it was difficult for individuals to disagree. The aim of Stage 2 was therefore solidified as a result, to focus on individuals' lived experience and to capture more in-depth information on what mattered most to older adults as they engaged with their local environments.

4.3.2 Stage 2: Main Study

Fieldwork for the main study was carried out between December 2017 and August 2018 and included two distinct steps for data collection. The involved a semistructured interview, including a mapping exercise. The follow-on step was a goalong interview, and these are now each discussed in turn below. There were several methods that I could have used which combine geo-spatial and qualitative approaches, to obtain insight into the lived experience of older people beyond the home. Common examples within literature include using photovoice (for example van Hees et al., 2017; Mahmood et al., 2012), using mapping exercises (see Carpiano, 2009), and using go-along interviews or mobile methods (see Carroll et al., 2020; Finlay and Bowman, 2017; Gardner, 2011; Lager et al., 2019; Lager et al., 2015; van Cauwenberg et al., 2018; van Cauwenberg et al., 2012). However, I chose to carry out mapping exercises and go along interviews rather than use photovoice methods. Whilst I could also have asked participants to take photographs and it would have provided valuable data, I felt that this would be asking too much from my participants in addition to them carrying out a mapping exercise and go-along interviews, as it would have necessitated multiple visits and interviews. However, where appropriate and when participants agreed, I did take photographs of some of the places they showed me and the barriers that they experienced.

Another increasingly common method within mobility studies is the use of mixed methods approaches using accelerometer and GPS monitors and/or travel diaries to capture an individuals' activity space. This involves collecting information about how physically active an individual is, as well as the places they have visited over a set time (for examples of this work see Bell et al., 2015; Christensen et al., 2011; Mennis et al., 2013; Zenk et al., 2011). I decided not to use GPS or accelerometer data to capture an individuals' activity space, predominantly because I did not have access to this type of equipment. Typically, this type of research is carried out within larger teams with greater resources. One of the main benefits of using GPS data over travel diaries or other methods asking participants to identify places they have visited, is that is reduces the need for accurate participant recall, which studies have shown can be an issue (Milton et al., 2015). I considered the use of travel diaries but felt that go-along interviews and mapping exercises within an interview were more appropriate for my research questions and epistemological and ontological stance. Furthermore, I did not feel that participant recall was a significant problem given my research focus. This is because I was not trying to capture a complete picture of my participants' activity space including the more "mundane" trips that individuals may take for granted and overlook (van Hoven and Meijering, 2019, p.1). Instead, I was interested in capturing the places that were *most* important for their quality of life and those places that held most meaning to them. I felt that these were less forgettable.

Step 1: Interview and Mapping Exercise

For Step 1, a semi-structured interview including a mapping exercise was carried out. Interviews were chosen, because they are a useful and common method for exploring how experiences of ageing are shaped by the places that older people live in (Hand et al., 2017). During the interview, participants were asked how they interacted with their local environment, how important it was for them to get out and about, whether they felt limited by their health, and how they would define a good quality of life. For a full interview schedule, see Appendix 3.

Mapping exercises occurred during the interview. Participants were given an A3 map of their local neighbourhood. Prior to the interview, I printed several different maps of the study area and beyond, at different scales. Participants chose the map that was most appropriate for them and that reflected their activity space. In some instances, this involved using two maps at differing scales. Participants were asked to identify the places of most importance to them and explain their significance. A place of importance was not necessarily positively described, and this was often the case with green spaces. Places that were perceived negatively helped to identify more subjective barriers to getting out and about. Participants were also invited to draw regular routes they might take. All participants that took part in the interviews completed the mapping exercise in some form. Participants were encouraged to complete the mapping exercise themselves where possible, unless they were unwilling to, or physically unable to. In five instances, the researcher marked the routes and places whilst the participant described them, and in one instance a carer did. Common reasons that the mapping exercise was challenging for older participants, was because they were "not good with maps", they struggled to see the maps, or they struggled to draw on the maps. Some were also initially reluctant to draw on the maps because they didn't want to "ruin" them, but I reassured them it was fine to do so.

Mapping exercises complemented the interviews, because they provided a talking point "triggering thoughts and reactions" and helped to "set the stage" for the goalong interview which was carried out afterwards (Carpiano, 2009, p.270). Furthermore, it provided a useful resource and piece of data for analysing and

integrating the spatial and qualitative data later. A total of thirty-four participants took part in twenty-four interviews and mapping exercises.² The duration of the combined interview and mapping exercises ranged from 22 minutes in length to just over 2 hours. On average (both mean and median), interviews were approximately 1 hour and 8 minutes.

Step 2: Go-Along Interviews

Although interviews and mapping exercises offer possibilities of thinking spatially, the setting remains detached from the places mapped (Carpiano, 2009). The go-along method is a type of in-depth qualitative interview carried out in context (Kusenbach, 2003) and "on the move" (Finlay and Bowman, 2017). Go-along or "mobile interviews" are therefore in-situ and immersive and can provide additional insight beyond the traditional interview (Bell et al., 2015; Finlay and Bowman, 2017; Foley et al., 2020). Mobile interviews elucidate person-place interactions (Carpiano, 2009) and are increasingly used in research on ageing and environment to explore the everyday experiences of ageing (see Finlay and Bowman, 2017; Gardner, 2011; Lager et al., 2019; van Cauwenberg et al., 2018; van Cauwenberg et al., 2012).

Go-along interviews were optional, and participants were encouraged to only do what was within their capability, and the researcher kept pace (Foley et al., 2020). Participants chose where to go and what to show based on the following instructions: "*I would like you to show me some of the areas in your local environment that are important to you, or where there may be some issues or barriers that make everyday life more difficult*". There were no structured questions during the go-along interviews and participants acted like "tour guides", being the experts in their local area and somewhat redressing the traditional power imbalances that exist between researcher and participant (Carpiano, 2009, p.267; Finlay and Bowman, 2017). The route of the go-along interview was captured using the GPS software app *Endomondo*. Settings in Endomondo were modified to ensure that data captured was private. Fieldnotes were taken immediately after each interview and

² Some of these interviews were joint or group interviews – these were tailored to suit the wishes of the participant.

go-along interview to make a note of the places of interest identified during the interview, along with any observations I had during the fieldwork.

Using these methods, I was able to obtain an insight of individuals' subjectively meaningful "lifeworlds", rather than an objective map of everywhere they had visited in a set time frame. This allowed me to identify the "culturally defined spatiotemporal setting[s] or horizon[s] of everyday life" (Buttimer, 1976, p.277), recognising lived experience as the "orchestration of various time-space rhythms" of meaning to an individual (p.289). This approach allowed participants to consider places that were not currently visited but held meaning to them and to explore the reasons behind this. As with any of these qualitative and geo-spatial methods used, the data collected offers a snapshot of participants' worlds. However, it will always be a partial representation of their lived reality, influenced, indeed, by my role in eliciting it (Cope and Elwood, 2009).

4.4 Sampling, Recruitment and Study Setting

4.4.1 Sampling Strategy and Recruitment Procedures

Stage 1: Exploratory Focus Groups

For Stage 1, an opportunistic or convenience sample of older people living in their own homes was chosen. It was purposive in that I deliberately sought older people who attended community groups. To see more details about the social groups, including the types of activities that were carried out there, see Appendix 4. I contacted several community groups in the two research areas using information from newspapers and websites. When I had contacted the groups and they expressed an interest in taking part, I spent several weeks prior to the focus groups attending and participating in the activities to build rapport with the attendees. Prior to the focus groups, I provided those in attendance at the group with an Invitation Letter (see Appendix 5), along with an Information Sheet (Appendix 6), and a Consent Form (Appendix 7). We then arranged focus groups for a particular date, and those in attendance on the day who wanted to participate took part. For Stage 2, a purposive sample of older people aged 70+ living in their own homes was sought. Of this "maximum variation" was pursued where possible. This is a type of purposeful sampling where the aim is to find heterogeneity (Teddlie and Yu, 2007). This form of sampling was sought to try to obtain a diverse perspective of what it meant to age well in place for older people with differing interests, as well as contrasting environmental and personal circumstances. This was not carried out in a systematic way, instead I asked myself throughout: who is likely to find it more challenging to participate in research; whose perspectives might be missing and how can I broaden my recruitment to capture a wider range of experiences? (Gilroy, 2021).

Participants were recruited through local gatekeepers and like Stage 1, this was primarily through a range of local community centres and social groups. I chose to recruit participants this way for two main reasons. The first was that the community groups and organisers served as gatekeepers in accessing older people and meant that I did not have to knock on older peoples' doors as a stranger. As a lone female researcher, I felt uncomfortable doing this, and I was conscious that older people may not open their door to someone they did not know (Russell et al., 1998). The second reason was that community groups offered a convenient and ready-made group from which I could then recruit multiple participants at once. Community groups were identified using local newspapers, internet web searches and word of mouth.

A limitation of recruiting participants through social and community groups, was that it resulted in the recruitment of individuals that were able access these groups. It was therefore likely to have excluded perspectives from those older people that were unable or unwilling to attend for a variety of reasons. Acknowledging this, a particular effort was made to recruit "harder to reach" participants who experienced health or mobility challenges, recognising that their experiences of ageing in place are often less well known or understood (Gilroy, 2021; Hockey et al., 2013). Within ageing research and from my experience engaging with local communities as a Planning Officer, it is commonly the more active retired and engaged older people that are recruited, usually because they have the fewest barriers to participating (Gilroy, 2021). As a result, I sought to recruit participants not just from *Active*

Retirement social groups but to also recruit from groups specifically designed to support older people who may find getting out and about more challenging. This included an Alzheimer's Day Centre, as well as a social group specifically designed for more socially isolated older people. For a summary table of all social groups that I recruited from during Stage 2, see Appendix 8.

When I met with a particular social group, I handed out Invitation Letters (see Appendix 9) and an Information Sheet (see Appendix 10). I placed A3 Invitation Letters in libraries and within the community centres and health centres. Interviews were conducted either in participants' homes or at the community groups that they had been recruited from, depending on individual preferences. One interview was carried out in a coffee shop. One of my participants was not technically recruited by a community group and was a more opportunistic recruit. I had been told by other participants about an assisted living complex in Study Area 2 that had a community centre within it, and I went there with the hope of recruiting some participants. Whilst there I spoke to one of the residents in the garden. I told him about my project, and he volunteered to take part. Whilst he lived in the assisted living complex, he didn't attend the community centre.

Table 4.1 provides an overview of participant characteristics recruited, including their age, gender, how they were recruited, their use of any mobility assistance devices, and whether they identified any health challenges that may limited their activities.

Characteristics	Details
Gender:	Male (n=6)
	Female (n=28)
Age Range (at time of interview and if known):	65–69 (n=1) 70–74 (n=12) 75–79 (n=4) 80–84 (n=4) 85–89 (n=2) 90–94 (n=1) Unknown (n=10)
Recruited through:	Local Community Centre / Social Group (n=30) Public Health Nurse (n=1) Dementia Day Care Centre (n=2) Home (Assisted Living) (n=1)
Use of mobility assistance:	Yes – walker (n=4) Yes – mobility scooter (n=1) No (n=29)
Health challenges identified in interview which influenced their ability to leave the house:	 Yes (n=18) This included the following (some participants identified multiple): Chronic Obstructive Pulmonary Disease (COPD) (n=3) (one participant used an oxygen tank) Alzheimer's Disease (n=2) Parkinson's Disease (n=1) Bowel Conditions (n=2) Visual Impairments (n=2) Vertigo / Dizziness (n=2) Joint-related mobility challenges or ongoing impact from previously broken bones (n=5) General reduction in activities / restricted mobility identified but no specific health conditions mentioned (n=1) No (n=6)
	Not identified during interview (n=10)
In addition to recruiting participants from social groups, in the main study I broadened my search to try to capture harder to reach groups, particularly those older people who may have additional health challenges, who may find it more difficult to leave the home and could even be homebound. This included recruiting through Public Health Nurses, organisations such as Alone, 'Meals on Wheels' services³ and Traveller community groups. My attempts to recruit a more diverse older adult sample yielded limited success. Just one participant across both study areas was recruited through a Public Health Nurse. Whilst I met with a gatekeeper at the Traveller community group and they expressed interest in the research, they felt that it was not possible to participate at that time.

I have included details of the organisations or individuals that I contacted but where I was unsuccessful in recruiting participants in Appendix 11. I believe it is important to be transparent about this, because contacting harder to reach groups is not without its challenges and they are harder to reach for a reason. Furthermore, the fact that I was unsuccessful provides valuable learning for future research and to ensure that we continue to carry out research with a wide range of older people. This includes recognising whose perspectives we may not be hearing from and learning from some of the challenges to recruiting these groups. It is important to note for context that Traveller groups in the Republic of Ireland face significant health challenges. On average, their life expectancy is on average 8 years lower they are also more likely to face severe limitations because of health challenges (29% compared to national average of 17%). This is particularly the case for males (36%) compared to females (23%) (European Union Agency for Fundamental Rights, 2020).

Sample Size

A social interactionist perspective will typically study much smaller numbers in geographically smaller areas, such as small communities or neighbourhoods (Gatrell and Elliott, 2015). As is common in qualitative research, where there is in-depth and rich data which can be time consuming to analyse, my sample size was relatively

³ This is a service which delivers meals to individuals at home, usually because individuals are no longer able to cook or purchase meals themselves. Typically this service is for homebound older people.

small, and I aimed to capture perspectives of approximately 15 participants in each study area. I did not seek to obtain data saturation as it is often criticised amongst qualitative researchers (Braun and Clarke, 2013). Theoretically, if you subscribe to ontologies about the subjectivity of experience within quality of life and environment, data saturation will never be reached because everyone will experience their world in a different and unique way. When I applied for ethical approval, I had planned to carry out fieldwork in a third study area and to examine a more coastal suburban setting too. However, I decided not to carry out research in this area. This was partly due to time constraints, but also due to a shift in focus from area comparisons to lived experience approach, which required more in-depth analysis of data at an individual level.

4.4.2 Study Settings

For Stage 1, as this was an opportunistic sample, I chose the local university town where I studied and a nearby town to focus my enquiries. At that point, I had not finalised the locations of Stage 2, although I had some ideas. For the main study, I chose study areas that combined demographic and environmental characteristics I identified in Chapter 1 as being important within this thesis. I wanted to explore an area that was newly ageing within a suburban setting and with higher rates of health challenges. To help inform where I should carry out my research, I carried out Exploratory Spatial Data Analysis. This involved examining population data, such as age profiles by looking at Census 2011 and 2016 data. I also explored socioeconomic characteristics from the HP Pobal Deprivation Index using All-Ireland Research Observatory Mapping and Data Visualisations (AIRO, 2016). Finally, I examined health mortality data from the Health Inequalities 2006-2011 Interactive Atlas (Centre for Health Geoinformatics, 2016; Rigby et al., 2017). This helped me to identify Study Area 1, where these factors combined. At a population level therefore, Study Area 1 has higher levels of deprivation and health challenges, compared with Study Area 2. Study Area 1 is also more newly ageing compared with Study Area 2.

To protect the identity of participants in this study, I have decided to not reveal the names of my study areas within this thesis and in published work. This is for both

methodological and ethical reasons. Within Chapter 7, I describe in-depth several participants everyday experiences and reveal their health and mobility challenges. I felt that combining this information with Study Area information could lead to the identification of my participants, particularly in a country with a relatively small population such as Ireland. Instead, I provide a summary of some of the key characteristics of the areas for background context.

Study Area 1

Study Area 1 is a large working class new town located in outer suburban Dublin. Originally a village, it has experienced rapid suburbanisation and population growth since the 1970s, when many people moved to the area from inner-city Dublin. Many of these individuals have now reached or are reaching retirement age. The centre of Study Area 1 includes the former village and has a main high street. From here, there are good public transport links into the centre of Dublin, and this takes between 50 and 60 minutes. A variety of cultural and social assets and amenities have been built in recent years including a shopping centre, a theatre, and a library. Moving away from the centre, the study area becomes much more suburban in nature. These areas are more residential with fewer amenities, although some do have smaller shopping parades, health centres and community centres, including bus routes to the centre of Study Area 1. The more recently built neighbourhoods in Study Area 1 are the most geographically isolated and furthest from the centre of Study Area 1. They are also the most disadvantaged and where the number of residents that are older has grown dramatically quite recently. The outer suburban areas are the most car dependent and from them, it would take at least 20–30 minutes to walk to the centre of Study Area 1.

Study Area 2 is technically an inner-city location due to its proximity to Dublin city centre. However, much of the urban form is more suburban in nature and so I have classified it as inner-suburban overall. The density of Dublin is lower than most European cities. It is comprised of four distinct neighbourhoods within it. Of the four neighbourhoods within Study Area 2, one is an inner-city location working class neighbourhood, two are inner-suburban working-class neighbourhoods, and one is an inner-suburban more affluent neighbourhood. All areas are close to the coast. This

area has experienced significant urban change, gentrification, and high-rise office development. There are good public transport routes to the city centre, which takes between 10–20 minutes depending on the neighbourhood. It would be possible to walk to the city centre from between 10–50 minutes. Figure 4.2 diagrammatically represents the Study Areas and how they are located within the Greater Dublin Area.

Figure 4.2 Diagrammatic Representation of Study Area Settings within the Greater Dublin Area



4.5 Ethical Considerations: a flexible, inclusive, and care-full research approach

Ethical approval for Stage 1 was granted on 22nd February 2017 (Reference: SRESC-2017-017) (see Appendix 12 for Stage 1 ethical approval letter) and fieldwork was carried out in June 2017. Ethical approval for Stage 2 was awarded in November 2017 from the Social Research Ethics Subcommittee at Maynooth University (Reference: SRESC-2017-092) (see Appendix 13 for Stage 2 ethical approval letter). Fieldwork was carried out prior to Covid-19 in June 2017 for Stage 1 and from December 2017 to September 2018 for Stage 2. This research involved contact with more vulnerable sub-populations of older people, such as those with cognitive impairments or frailty. However, I also did not want to exclude them from participating in the research but instead provide the necessary accommodations to enable them to take part. To mitigate risks to participants, they were supported in a "care-full" way in all stages of the research to ensure they were enabled to participate (Foley et al., 2020). For example, two participants had Alzheimer's disease and were recruited through a dementia day-care centre.

In this instance, consent forms were adapted to make them more visual (see Appendix 14 for standard consent form and Appendix 15 for adapted consent form). Preliminary meetings were carried out and carers, family members and the participant themselves provided consent (Hubbard et al., 2003). As a result of discussion with carers, who were present during data collection, interviews were shortened to reduce demands on participants, but still focused on activities and places of importance, as well as how they defined a good quality of life.

Although I did not recruit any homebound participants, I did capture a broad range of older adult perspectives. Many participants were supported to participate in this research with varying health and mobility challenges. I believe that the flexible, inclusive, and care-full approach I have taken throughout this research has led to insights from participants who may have been prevented from participating without the adjustments that I made. It also meant that it more accurately reflected their existing experiences. An example of this is when I contacted a social group in Study Area 2, the group leader and attendees told me that they would prefer to carry out the interview and mapping exercises together. Throughout the interview, I learnt that this group was specifically for older adults who were more socially isolated, and many participants did not feel confident enough to attend a larger social group and community centre nearby. Participants were much more comfortable doing the research together and I anticipated that many would have said no if I had asked them to take part individually. Carrying out a group interview was much more challenging from my perspective and meant that I was not always able to identify participants during the transcribing of this. However, adapting my research to be more flexible of participants' needs meant that I obtained access to harder to reach participants that I would likely not have recruited otherwise. Table 4.2 provides an overview of the variations in the interviews, mapping exercises, and how they were adapted to suit participant needs, abilities and wishes.

Interview	Format:
Total (n=34)	 Individual (n=18) Accompanied by family member (n=1) Joint (2 instances where 2 participants took part at the same time) (n=4) Joint (2 participants also accompanied by carer) (n=2) Group (one instance involving 9 participants) (n=9)
Mapping Exercise Total (n=34)	 <u>Format:</u> Participant completed this themselves (n=28)
10tal (II–34)	 Carer completed this (n=1) Researcher completed this (n=5)
Go-along Interviews Total (n=20)	 <u>Type of Go-Along interview:</u> Driving (researcher drove) (n=6) Driving (participant drove) (n=1) Cycling (both researcher and participant cycled) (n=1) Mobility scooter (n=1) Walking (no assistance) (n= 8) Walking (with use of walker) (n=2) Visualisation at home (n=1)
	 Accompanied with carer (n=2) Accompanied with family member (n=1) Joint go-along interview (n=2) Distance: Between 300m (walk) and 37km (drive) Duration: Between 7 minutes and 3 hours*

Table 4.2. Summary of adaptation of research methods

*This three-hour interview was unusually long and was a cycling interview with a former ultra-triathlete and his dog.

Another flexible adaptation that I made was in relation to age of participants. Whilst most participants were aged over 70, in one neighbourhood within Study Area 1 a participant aged 66 was included. Interestingly, this participant lived in the

neighbourhood with the fewest environmental supports, in an area that was newly aged, and they also identified significant health and mobility constraints. This contrasts with participants in Study Area 2, some of whom were aged in their 80s and far more active. This reinforces the need to recognise chronological age does not always reflect abilities, particularly when taking into consideration a lifecourse perspective of inequalities and wider determinants of health. Another ethical consideration during the go-along interview was whether to reveal that we were doing a go-along interview if we engaged with other people. I kept the audio device hidden from view in my pocket during go-along interviews to not draw attention to the participant and to make them feel more comfortable whilst out and about. This was particularly important when participants revealed fears about walking in their local environment during the interview, which did happen in some of the study areas (discussed further in Chapter 6). Participants decided if they wanted to reveal they were conducting a go-along interview when interacting with bystanders. The conversation data with bystanders was not used, as they did not provide consent to take part, but the interaction itself was noted.

4.6 Data Processing and Organisation

Focus groups, interviews and go-along interviews were audio-recorded and transcribed verbatim. Transcribing was partially completed by a professional transcribing service and partially by me. Interview transcripts, the completed map from the mapping exercise, fieldnotes, and any photographs taken during the go-along interview were imported into NVivo 12. Data from Endomondo was exported shortly after data collection to both table and spatial forms (as shapefiles) and images of the route and then imported. All imported data was assigned to a particular case ID. This referred to a particular focus group for Phase 1, or participant for the main study. Participants were given pseudonyms (see Table 4.3 for participant case IDs and the corresponding pseudonym).

Stud	ly Area 1	Stu	dy Area 2									
Case ID SA1-001 SA1-002 SA1-003 SA1-004 SA1-005 SA1-006 SA1-007 SA1-008 SA1-009 SA1-010	Pseudonym	Case ID	Pseudonym									
SA1-001	Jennifer	SA2-001	June									
SA1-002	Moira	SA2-002 Michelle					Moira SA2-002 Mich			Moira SA2-002 Mic	SA2-002 Mich	Michelle
SA1-003	Anne	SA2-003	Mairead									
SA1-004	Edith	SA2-004	Dolores									
SA1-005	Bridie	SA2-005	Shauna									
SA1-006	Nuala	SA2-006	Áine									
SA1-007	James	SA2-007	David									
SA1-008	Noelle	SA2-008	Brenda									
SA1-009	Niamh*	SA2-009	Margaret*									
SA1-010	Méabh*	SA2-010	Nessa*									
SA1-011	Darragh	SA2-011	Bríd**									
SA1-012	Emer	SA2-012	Sinead**									
SA1-013	Michael	SA2-013	Clodagh**									
SA1-014	Eamon*	SA2-014	Sarah**									
SA1-015	Louise*	SA2-015	Cara**									
* = Joint Interview	VS	SA2-016	Francis**									
** = Group Interv	iew	SA2-017	Anita**									
		SA2-018	Breeda**									
		SA2-019	Jack									

I chose to use NVivo as a CAQDAS (Computer Assisted Qualitative Database Analysis Software). The use of CAQDAS allowed me to a convenient way to manage my data, where I could arrange and organise multiple forms of data, code, query, and analyse my data in novel ways, as well as keep a clear audit trail of the analysis I had conducted (Bazeley, 2007; Silver and Lewins, 2014; van Hoven and Poelman, 2003). Data was stored on an encrypted laptop. Figure 4.3 provides a screenshot of my NVivo Database, with participant ID (cases are to the left, an example of the types of data recorded for each participant in the middle, and an example of a photograph taken to the right).

Figure 4.3. Screenshot of NVivo Database



I also had some hard copies of data such as the maps produced my participants in the mapping exercises, as well as my notes. These were labelled and kept in a folder by participant ID. Consent forms were kept in a locked drawer.

4.7 Data Analysis

To analyse my data, I used a range of spatial and qualitative techniques. My overarching goal within the data analysis stage was to make sense of what my participants had told me and to identify interesting qualitative and spatial patterns within their daily lives. Whilst there are many different approaches to qualitative analysis, a common thread is that they "seek to make sense of the data produced through categorisation and connection" (Kitchin and Tate, 2013, p.229). Qualitative data analysis is fundamentally about "searching for patterns" (Braun and Clarke, 2013, p.224). Continuing with the flexible approach I have taken throughout my thesis, I chose to use the thematic analysis approach developed by Braun and Clarke (2006) and more recently updated as "reflexive thematic analysis" (see Braun and Clarke 2019), to guide my analysis of the qualitative data. Thematic analysis provides a flexible structure to qualitative data analysis, which involves categorising text first into codes, and then grouping these codes into themes (or categories) (Braun and Clarke, 2013; Bryman, 2016). I chose thematic analysis because much of

this work was exploratory, and I intended to use different approaches to coding and categorising depending on the research question I was answering. I did not want to be bound by a particular analytical approach; instead, I wanted to be flexible and intuitive to respond to what my participants were telling me. Typically, thematic analysis involves 5 key overlapping phases, beginning by familiarising oneself with the data (Phase 1), generating initial codes (Phase 2), searching for themes (Phase 3), reviewing themes (Phase 4), defining and naming themes (Phase 5) and writing up a report (Phase 6) (Braun and Clarke, 2006, p.87). Where I carried out more traditional thematic analysis approaches, I followed these phases. Throughout the analysis, I carried out multiple rounds of coding, categorising, and writing.

During Phase 1, I transcribed and organised the data (as described above), carrying out repeated readings. I took notes, drew diagrams, and wrote annotated memos during this phase to generate ideas and make sense of my data. The most important and essential part of this analysis was Phase 2, where I completed extensive rounds of coding for different purposes, depending on the chapter I was working on. A key component of most qualitative research is coding, which are the building blocks of analysis from which it is then possible to find patterns and meaning within the data (Saldaña et al., 2011). As is common within qualitative research, I began the analysis with a broader research question, which were then refined over time, as I found something I wanted to explore further that could only have been revealed through coding and familiarisation of the data. When I had an idea of a topic area that I wanted to explore in more detail, I would carry out deep dives of the data. This is where I coded the entire dataset to produce an extracted data set focusing on a particular topic area (e.g., how participants defined a good quality of life). I included data from both Stage 1 and Stage 2 within these data sets. Sometimes this involved looking at the relevant question asked during the interview, but also looking across the entire dataset, in case it was discussed outside of this question. Here I also made use of text search queries within NVivo, to identify words that may be connected, and this provided an extra layer of accountability to ensure that I did not miss anything important. This phase of coding typically generated more broad-brush coding of entire paragraphs or sections. It was also deductive and primarily about "reducing" the dataset to make it more manageable to work with (Saldaña et al., 2011).

Once I had an extracted data set of a particular topic area, I printed this out. Although I used NVivo as a database, I preferred to use more of a hybrid approach when it came to analysis, where I printed and inductively coded smaller extracts of data by hand (Maher et al., 2018). Coding of interview transcripts was carried out with another PhD student on a sample of four interview transcripts, to discuss levels agreement about the codes identified, as well as my interpretations. In Chapter 5, the thematic analysis I carried out was inductive, where I was seeking to generate empirically grounded concepts (Braun and Clarke, 2013). For Chapter 6, I undertook a thematic analysis which drew from content analysis, where I was trying to code and categorise places and activities and was interested in how many participants identified these. I framed my categories by a particular theory and so this was more deductive. In Chapter 7, my analysis followed a narrative (or geo-narrative) approach. For this analysis, instead of cutting and combining everything said about a topic area, I would look at everything an individual said and produce a narrative of this (Saldaña et al., 2011). I now provide more details for the approach taken for each empirical chapter.

4.7.1 Data Analysis for Chapter 5. Conceptualising Ageing Well in Place: Getting out and about for a good quality of life

Within Chapter 5, I carried out a thematic analysis to explore how my participants defined a good quality of life and to examine how important getting out and about was to them. I had two key extracted data sets for this: everything my participants said about a good quality of life and everything they said about getting out and about (both positive and negative). Within the initial coding phase (Phase 2) for Chapter 5, I produced 68 codes that referred to how my participants defined a good quality of life and 21 codes that related to the importance of getting out and about. The next step was to generate themes (Phase 3) and then review themes (Phase 4). I did this by hand initially and then inputted this back into NVivo.

I categorised my codes relating to quality of life into four themes. I produced a series of diagrams at this stage to think through how the codes related to each other. Some of these were drawn by hand and others were produced using Miro (see Appendix 16 for an example). These themes inform my conceptual contribution of ageing well in

place, which is presented at the beginning of Chapter 5 and referred to throughout the remainder of this thesis. However, these themes only told part of the story – what mattered for ageing well; it was missing the spatial element. Alongside this piece of coding, I identified four themes related to getting out and about and these have been used to structure the sections of Chapter 5. Through writing and revisiting codes (Phase 5 and 6), I realised the importance of getting out and about for a good quality of life, how participants varied in the amount that they *needed* to get out, and how this was intrinsically linked to their quality of life through their health, mobility, and independence. Initially my codes and themes related to getting out and about and quality of life were distinct, but during this phase I began to integrate these and focus on those aspects of quality of life that were related to getting out and about. The final themes solidified as I wrote the empirical chapter, recognising the iterative nature of qualitative data analysis and thematic analysis. Figure 4.4 provides a coding tree of the final themes that I used to structure my chapter.

Figure 4.4. Coding Tree showing the variation in participants' ability and 'need' to get out and about (GOA)



4.7.2 Data Analysis for Chapter 6. Getting Out and About to Connect with Others

For Chapter 6, I had a different set of guiding research questions and coding was adapted as a result. Initially, I coded all places mentioned by my participants to create a database of places. There were over 640 codes for this phase. I have shown aggregated summaries of these in the screenshots (Figure 4.5) to not reveal the locations.

🔺 N	ame	Files	References ∇
₽ Р	laces of Importance	47	2923
±. (Study Area 2	25	1115
<u>ب</u>	Study Area 1	36	1113
<u>ب</u>	Outside Study Areas	43	604
	Previously Lived	25	46
÷٠ (Home	19	28
··· (Hospitals (not specific)	8	12
	Nursing Homes (not specific)	1	2
	Car	1	1

Figure 4.5. Screenshot of Initial Coding of Places of Importance

For each place, I could easily find out how many participants identified this place, how many times they referenced the place. I could also retrieve data about everything that was said about each place. This allowed me to consider places in turn and compare what my participants said about these places. By carrying out cross-matrix analyses, I crossed individual cases (participants) by places to see the number of places mentioned by each participant. I identified the coordinates of these places wherever possible so that they were spatially referenced. In addition to places of importance, I coded activities that were mentioned by my participants, so that I could get a sense of what participants did in the places of importance to them. There were 222 codes generated for activities during the initial coding phase (see Figure 4.6 for most highly referenced activities).

Name	Files		References
Activities, Interests and Interactions of Importance when Out	& About	46	158
Social Interaction		44	26
		11	14
Having a Laugh		3	4
Sit and Talk		1	1
Attending Social Groups or Clubs		36	238
Walking		39	193
Dog Walking		10	18
- Shopping		32	7
Charity Shopping		2	2
Craft-based activities		18	62
Knitting		12	25
Art		10	17
- Crochet		8	14
Weaving		1	2
Taking Trips Away		21	39
Day Trips or Days Out		16	37
Travelling		12	20
Attending Mass		20	34
Going into Town		14	3:
Going for Lunch or a Coffee		14	33
Volunteering or Fundraising		14	24

Figure 4.6. Screenshot of Initial Coding Activities of Importance

As with Chapter 5, the scope of Chapter 6 refined and narrowed over time. The most highly referenced activity was social interaction, and I began to explore the places where social interaction occurred (or did not occur) whilst out and about, focusing on places that might be considered "Social Infrastructure" (Klinenberg, 2018). Within Chapter 6, I do not focus on all places of importance to my participants, but instead those places identified as important for social interaction. As I had this initial coding already completed, I was able to extract all the codes for social interaction as a data set and carry out inductive coding for this, as I had done for Chapter 5. I also produced a refined list of Social Infrastructure places and categorised these into similar groups of places. For a summary table of these categories, the number of places within these, as well as the total number of references see Table 4.4. For a more detailed table, which breaks this down by Study Area and case (participant), see Appendix 17. Carrying out this type of analysis allowed me to consider what was it about certain places that made them useful to older people for social interaction. Focusing on social interaction and Social Infrastructure as a topic area for this chapter led me to examine Gardner's (2011) work on the "sites of significance" for social interaction (p.268). I applied Gardner's (2011) categories of place (thresholds, transitory zones and third places) as a framework to think about the importance of

these places and used this to structure Chapter 6. In Figure 4.7 I have provided a flow diagram to summarise the development of this thinking and analysis over time.

	Category of Places	Code Number of Places in Each Category			Number of References									
			SA1	SA2	OSA	All	Rank	SA1	SA2	OSA	All	Rank	All	Rank
1	Main Community Centres	MCC	5	4	0	9	9	135	77	0	212	3	298	3
2	Other Community Centres	OCC	10	5	3	18	5	53	23	10	86	5	290	
3	Shops	SHO	15	12	18	45	1	171	40	90	301	2	301	2
4	Post Offices	POF	4	4	0	8	10	14	9	0	23	11	23	10
5	Banks & Credit Unions	BCU	1	2	2	5	11	4	3	3	10	12	10	11
6	Hairdressers	HAI	3	1	0	4	12	3	1	0	4	13	4	12
7	Health-Related	HRE	10	3	4	17	6	58	5	17	80	6	80	5
8	Green & Blue Spaces	GBS	8	16	9	3	2	74	207	72	353	1	361	1
9	Trip Destinations	TRI	0	0	21	21	4	0	0	61	61	8	61	7
10	Eating & Drinking Places	EDP	8	11	4	23	3	24	19	5	48	9	48	8
11	Cultural Buildings	CUB	4	2	10	16	7	22	4	39	65	7	65	6
12	Churches	CHU	8	4	2	14	8	45	25	19	89	4	89	4
13	Libraries	LIB	2	1	0	3	13	13	13	0	26	10	26	9
14	Miscellaneous - Recycling Point	MIS	0	1	0	1	14	0	2	0	2	14	2	13
		Totals =	78	66	73	187		616	428	316	1360		1368	

Table 4.4. Summary of Categories of Social Infrastructure Places

Figure 4.7. Flow Diagram of the Development of Data Analysis for Chapter 6



In addition to identifying the places and activities that participants valued whilst out and about, I wanted to explore the temporality and rhythms of this, recognising that all activities happen in a particular time and place. I was able to explore the temporal analysis aspect of these places and activities by using a weekly routine template, which I completed for participants to map out what they had told me during the interview. This was something I added later in the analysis and if I were to conduct the research again, I would ask participants to complete this themselves or complete it with them. Nonetheless, I found this a useful exercise within the analysis, as I began to see patterns of routine and activity that I might otherwise have missed. This led me to consider the planned and scheduled nature of social interaction within each of Gardner's (2011) different place categories within Chapter 6. Figure 4.8 below shows a thematic coding tree, where I show the key themes identified within each of the three categories of places for social interaction.



Figure 4.8. Coding Tree for Chapter 6

As is common within qualitative research, which is more of an iterative process, the scope of this research narrowed over time (Agee, 2009). Initially the scope of this chapter was to focus on RQ2: What places, routes, routines, and interactions outside of the home are most valued by participants? However, to narrow down the focus of inquiry for this chapter, I revised my research questions over time and identified some additional sub-research questions. This included:

- RQ2.1: Where did participants connect to others whilst out and about?
- RQ2.2: What types of social interaction occurred within these different places and who were they with?
- RQ2.3: How important were these interactions to participants and why?
- RQ2.4: To what extent were the engagements with these people within these meeting places planned and how frequent were they?

4.7.3 Data Analysis for Chapter 7. Ageing - as well as you can - in place with health and mobility challenges

Chapter 7 builds on the findings of Chapters 5 and 6 but instead of looking across and comparing the experiences and perspectives of all participants, I focus in-depth on four individuals with differing health and mobility challenges. The coding and analysis carried out for the previous chapters assisted with the data analysis for this chapter, but I did not carry out a specific form of thematic analysis for this. Instead, the extracted data set was at an individual or case level and I looked at this using a narrative approach. Findings within this chapter were analysed and presented as a "geo-narrative" (Bell et al., 2015; Bell et al., 2017; Kwan and Ding, 2008), that is as spatially referenced narratives or experiences (Yuan 2020), which typically involves the use of "visual or cartographic means (maps, GIS, and GPS) to generate rich interpretive accounts of people's experiences of certain aspects of the urban environment" (Kwan, 2021). In this instance, I was interested in narrating and representing individuals' geographically grounded lifeworlds (Seamon, 2018). This included an individuals' meaningful activity space, including their everyday experience, their activities, and their social contacts that happen whilst out and about, as well as barriers and risks as they navigate this with health and mobility challenges.

One of the most powerful tools in geography is the ability to visualise data using maps. Qualitative GIS and mapping can therefore hold "ontological power" (p.6), where "counter-mapping" non-dominant or missing perspectives can "construct new imaginaries of place and space that can contribute to inclusive citizenship" (Pavlovskaya, 2017, p.9). The perspectives of individuals with health and mobility challenges are an under-represented population of older adults. Whilst I have used

databases to collect and hold both spatial and qualitative data, I have not integrated or analysed the data in ways that are more common with Qualitative GIS approaches which have used computational technologies (using approaches such as Matthews et al., 2005; Knigge and Cope, 2006; Kwan and Ding, 2008 for example). Instead, I have integrated the qualitative and spatial components using annotated mapping (such as Bell et al., 2015). However, a key difference is that I have carried out mapping at an individual level through person-centred mapping, representing their lifeworld, rather than in a collective way (as would be more common in participatory approaches involving communities). This fitted in with what I was trying to show through the maps, which was how my participants conceptualised a good quality of life for them, as well as how they navigated ageing in place based on their existing personal and environmental contexts.

The geo-narrative had two parts to it: it had the story of the experience focusing on the research questions and the accompanying annotated maps. From the interviews, I learnt what mattered most to each participant, and this was clarified with participants during the interview, mapping exercise and go-along interviews. For each participant I examined all research questions, by first considering what was most important to them for a good quality of life and to age well (RQ1), as well as the places, routes, routines, and interactions that were most valued (RQ2). I then considered their person-environment fit, exploring the factors that made it easier or more challenging to get out and about in ways that are meaningful to them (RQ3).

To produce the maps, I took notes by hand of each participant and the places they identified. I then combined the results from the interview, mapping exercise and goalong interview and produced my own annotated map, as well as a weekly routine log. To produce the maps digitally, I produced base maps in ArcMap 10.2.2 using Open Streetview layers but I modified them, so they did not have labels of streets or buildings. I kept some aspects of the built form of the neighbourhood but removed street patterns so that the local environments were not identifiable. I also rotated the map so that locations were less easily identifiable. Only approximate locations of home addresses were provided for finalised maps, and I created buffers to obscure their exact address and to show Euclidean distances from participants' homes. I then exported these base maps, before digitising the annotated parts relevant to the individual participant. I produced these in Microsoft Publisher, because I found it more user friendly to carry out annotations in this software compared with ArcMap 10.2.2. I made several decisions about the style of the maps (detailed in the map key within Chapter 7), to add qualitative dimensions of meaning and emotion to them. This included using stars for places of importance, where the larger the star the greater the importance to the participant. I also used colour coding to show where the places had positive meaning to the participant (green), mixed feelings (amber), negative feelings (red), and finally, where they were valued but the participant was no longer able to access them (grey).

4.8 Conclusion

To summarise, this thesis has followed a pragmatic, flexible, care-full, and inclusive methodological approach to explore the lived experiences of older adults and how they conceptualise and navigate ageing well in place. I have combined qualitative and spatial research methods and analysis to explore this topic and to identify patterns in experience, behaviour and meaning. Now that I have outlined what I did for this research project, the next three chapters present the empirical findings for this thesis.

This page has been left intentionally blank.

Chapter 5. Exploring the importance of getting out and about for a good quality of life

I don't want to be the granny sitting in the corner with a shawl around her shoulders and a blanket around her knees, I want to be able to get up and go. (Edith)

5.1 Introduction

As demonstrated in Chapter 3, leaving the home can provide a range of wellbeing benefits. As a result, older people usually have some want or need to leave their house to get out and about (Holland et al., 2005). Within this first empirical chapter, I set the scene for the entirety of this thesis' empirical work, by examining the importance of getting out and about to my participants and how this relates to their quality of life and ability to age well in place. To begin, I introduce my empirically grounded conceptual model of ageing – *as well as you can* – in place (see Figure 5.1), which I continue to use throughout the remainder of this thesis. Within this model, I argue that a key component of a good quality of life in old age is being able to get out and about, to engage in activities and interactions of most importance to an individual. The four themes important to my participants for a good quality of life and included in this model are:

- 1. Having 'good enough' health, wellbeing and mobility
- 2. Having 'adequate' levels of safety, security and support
- 3. Engaging in 'meaningful' activities and interactions
- 4. Being 'relatively' independent

All the words in apostrophes emphasise that they are subjectively defined. As I will demonstrate throughout this chapter, each of these four inter-connected themes is related to getting out and about in some way. The figure below illustrates the ways that each of these themes relate to each other and how they contribute to a good quality of life and ageing well in place.

5.1.1 Integrating Quality of Life and Getting Out and About

Figure 5.1 Empirically Grounded Conceptual Model of Ageing – as well as you can – in Place



Having 'good enough' health, mobility and wellbeing is connected to more subjective interpretations of types of ageing well, discussed in Chapter 2. However I have amended this to 'ageing as well as you can', to reflect the more realistic views of health that participants had. As I will show later on in this chapter, 'having your health' was very important to my participants for a good quality of life and was one of the most highly referenced codes. However, when this was examined further, the way that participants defined health revealed that health was highly valued, because it *enabled* participants to get out and about. Furthermore, for some participants with health and mobility challenges, they were still *supported* to get out and about and felt safe and secure doing so, which is why Themes 1 and 2 are so inter-connected. As I will demonstrate in Chapters 6, which explores how getting out and about can be important for social interaction and Chapter 7, which focuses in-depth on four participant lifeworlds, getting out and about was much easier for some than others. The nature of the risks faced by individuals, as well as how these risks impacted the ability to get out, varied considerably. In this way, participants had to negotiate and navigate their existing capacities within the environmental context they found themselves, to age as well as they could. *Having 'good enough' health, mobility and wellbeing,* combined with *Having 'adequate' levels of safety security and support,* were therefore necessary in order to get out and about. *Engaging in 'meaningful' activities or interactions* and *Being 'relatively' independent* are connected to being in place, or getting out and about directly. In order to participate and engage in 'meaningful' activities or interactions, getting out and about in some form was usually necessary. Furthermore the way participants conceptualised *Being 'relatively' independent* was connected to getting out and about, because this was the physical enactment or embodiment of being independent.

I now dedicate the remainder of this chapter to participants' accounts of how important it was to them to get out and about. Whilst getting out and about was consistently valued amongst participants, there was considerable variation in the way participants answered the question: *How important is it for you to get out and about?* Broadly, I was able to group my participants into three groups. The first group included those participants that spoke urgently and strongly about getting out and about and described it as something they *must* do and these experiences are summarised in Section 5.2. This group typically spent a greater amount of time out and about. I begin this section with a vignette from one participant in particular that I use as a case throughout Chapters 5 and 6, who demonstrates the importance of getting out and about for a good quality of life particularly well. The second group included those participants that were reassured by knowing they can get out, but did not necessarily have to be out all of the time and these are described in Section 5.3. Finally, the third group included participants that were more limited in their ability to get out, but still valued getting out *some of the time* and these are outlined in Section 5.4.

In addition to variations in the importance of getting out and about between individuals, there were considerable individual variations or fluctuations in how often they got out. These variations reflected different personal circumstances such as health circumstances, as well as differing environmental contexts, including seasonal and weather variations or time of day. Section 5.5 of this chapter considers some of these variations. Throughout all sections I have woven participant accounts of the importance of getting out and about with the themes for a good quality of life, highlighting the inter-connectedness of these two concepts.

5.2 "I have to get out"

5.2.1 Introducing Bridie

Bridie, aged 72 at the time of her interview, had lived in the same house in her neighbourhood within Study Area 1 for the past 43 years, and was originally from an inner-suburb of Dublin. Bridie had lived alone in her house for the past fourteen years, as her husband had to move into a nursing home; she has been widowed for the past five years. Like many participants, Bridie conceptualised a good quality of life as having her health and being able to get out. For Bridie, having health was defined as being as free from pain as possible and being able to get out and walk:

Well health, once you have your health and you are able to get out, and free from pain as possible that you can get out and walk. I think it is what you make of it yourself really and truly. (Bridie)

When I asked Bridie specifically how important it was for her to get out of the house, she replied:

It is very important, definitely very important. After the last week [unprecedented snow weather event], I really sympathise with anyone who can't get out. If I couldn't get out of the house, I don't know what I would do, because it is important to get out to meet people - for your own sanity - get out and meet people. I enjoy it. (Bridie)

During data collection in Study Area 1, an exceptional snow weather event took place in Ireland from Wednesday 28th February to Sunday 4th March 2018. It was described by Met Éireann as one of the "most significant snowfall events of recent years" (Met Éireann, n.d.), with an estimated 40cm of snow falling in some parts of the country overnight on Thursday 1st March. This unprecedented snowfall in Ireland was caused by the combination of two weather systems, a cold air mass from Siberia, informally known as the *Beast from the East*, and a depression named *Storm Emma*. As a result of infrequent occurrences of heavy snowstorms in Ireland, events such as these cause more disruption than would be the case in countries where snowfall is more frequent and planned for. Status red weather warnings were issued and people were advised to stay indoors throughout the event. Public transport ceased and the country came to a standstill (Met Éireann, 2019).

Bridie discussed how important family and friends were to her, and how they were also very important to her for a good quality of life. She commented that her grandchildren helped her health wise, because they kept her active. Bridie has cared for others for most of her life. Before her husband died, she spent a great deal of her time either travelling to the nursing home or visiting him. When he died, she explained that initially, she didn't know how to fill her time:

I found I was at a loss for the time, I just didn't know what to do. (Bridie)

Yes, I had so much time. Where [before] I had no time to spare... but I did find when he passed, I couldn't get used to having so much time then... I was able to go to all my clubs and that. But it was grand then after a while, you'd get used to it. (Bridie)

She described that "after a while" she adapted and began attending various social groups and clubs because she was suddenly able to, and gradually she began to get used to this change in her life. One of the reasons that getting out and about was so important to Bridie, was because it provided the opportunity to meet other people, as well as fill some of the time that she suddenly had available to her when her husband died. Bridie's overarching valued functioning could be described as connecting with others whilst getting out and about. In addition, it was important to her not to spend too much time at home on her own, because this made her feel as though she had too much time on her hands. I return to Bridie in more detail in Chapter 6, when I consider her daily routines, places and activities of importance.

5.2.2 The importance of getting out and about for mental health and wellbeing

Bridie was someone that had a very strong motivation to get out and about. It was very important to her for her mental health, wellbeing and her quality of life and she admitted that she would not know what to do if she couldn't. Other participants also described how they *needed* to get out and spoke about the detrimental impact it would have on their physical and mental health if they couldn't. Shauna felt very strongly about the importance of getting out so that she could talk with others. She stated during her interview that she would "crack up" if she couldn't get out and would need to be "committed" and how getting out was her "therapy":

100% yes, if I couldn't get outside the house, they could commit me. I told the daughter that. (Shauna)

I think my therapy, everyone is different, keep out and meeting people. It is better than a tonic. I am never sitting. (Shauna)

In Margaret and Nessa's joint interview, they discussed how if they had to stay within their house "for any length" of time, they would "nearly go mad":

But also, it could be the difference of living longer because I personally, and Nessa I think is the same, if we are in the house for any length, through no fault of our own, we nearly go mad. (Margaret)

Getting out and about was also identified as a must for David:

No, I *have* to get out... I wake up every single day, can't wait to get out and much to the annoyance of my missus. (David)

Whilst David enjoyed interacting with others whilst out and about (shown more in Chapter 6), his primary motivation for getting out was different to the previous participants. He felt that getting out and about was important for his mental health, but emphasised different pathways to this. In particular, it was being physically active and carrying out regular cycling routines, as well as connecting to nature, that provided him with health and wellbeing benefits. David was quite unusual amongst my participants and was by far the most active, as he was a former ultra-triathlete. David's routines beyond the home are described in more detail in Chapter 6, but included cycling long distances with his dog. Here I summarise why David felt so strongly about getting out in the first instance. David felt that getting out and about was "key" to his mental health. In particular, he felt that his routines ensured that he never felt lonely or depressed:

The key to good health in terms of mental health and all is being out and about, and I never look on myself, I could never see myself being lonely. (David)

During his interview, David discussed the prevalence of depression that he observed at a societal level. He speculated whether people would feel depressed if they engaged and connected more with nature, something he particularly valued. David provided examples of activities that he felt would be beneficial to people, which were activities he obtained "pure enjoyment" from. This included watching waves during a storm, seeing cygnets on a nearby river, and feeding two robins outside his house that he had named "Cheeky" and "Nervous":

There is too much depression. How many people really suffer from depression? How many people will get rid of feeling depressed if they went out for a walk in a storm and watched the waves beating up over a wall 25 foot high? Exhilarating stuff... And getting out... Like I go down [name of nearby river] every day, there's five beautiful little cygnets are growing at a great rate down there, and all that kind of stuff. It does your heart good. We have two little robins that fly into the hall here, we feed them with cheese. My friend's kids come down especially to see Cheeky and Nervous. And they whistle from 2am. They are fantastic. It is pure enjoyment here at times with just nature. (David)

5.2.3 Concerns about getting "stuck" at home

For some participants, it was clearer to obtain a sense of what was *not* a good quality of life, rather than what was. Edith, Shauna and Bridie all expressed concerns that about spending too long at home. For these participants, there appeared to be an unnacceptable line that would constitute a poor quality of life. This included not being able to get out, not moving and not interacting in any way, to the extent that you would effectively "vegetate". In this way, *not* having a good quality of life was conceptualised as having a completely passive existence both physically and mentally and having no meaningful activities or interactions to look forward to:

I don't want to become a vegetable sitting in the corner. Life is for living and for whatever bit I have left I want to live it. (Edith)

You either sit in the chair at home and just veg out, or you get out. (Shauna)

I know there are some people who don't like going out and stay in all the time, but I think you would just vegetate if you didn't get out. (Bridie)

This has been identified as a common concern among older people within lay person definitions of quality of life (see Aberg et al., 2005, Borglin et al., 2005, Bowling and Gabriel, 2007, Holland et al., 2005). Connected to fears about spending too long at home, several participants raised concerns about being "stuck" in their homes and not being able to leave. When I asked Jack how important it was to get out, he responded:

Oh it's, yeah, yeah, I would hate to be stuck in all the time. (Jack)

Getting "stuck" included both physically being unable to leave or mentally unable to, where the perceived barriers become too difficult, or when an individual gets out of the habit of doing so. In the group interview, there was a discussion about people who might be stuck in the house for various reasons, and how it could be very difficult to overcome this:

I think once you get out it becomes a habit and then you can't get stuck in the house. (Clodagh)

James described how he would feel it he could not get out and felt as though he would effectively be imprisoned:

Oh, that is very important, you don't want to imprison me, you don't want to put me in jail in the... now, at the same time, I probably mellowed a bit in recent years. In that, I can sit in a lot longer and go on the laptop and do something or read a book or something like that, but I still like to get out, fairly often. (James)

What is interesting here is that James also discussed how he has "mellowed" over time, suggesting that his expectations about getting out, and the perceived need to be out has changed as he has aged. Being a prisoner within your own home was also identified in Focus Group 2: Yes, I think it is important people get out though, don't make a prisoner of their home, and you can be a prisoner to your own home. And if you start tidying and cleaning, it is endless and nobody sees it, only yourself. (Participant from Focus Group 2)

When Michelle described her weekly routine, she talked about how she was not "locked in", because she could still get out and about:

I don't stay in the house too long, weekends, maybe a bit, but my sister comes down on Sunday mornings and she has a bit of lunch and then she goes off home. Because she is 84 and she drives, she lives out in Shankill, but she comes on Sundays so I am not, like you would say, locked in, no. (Michelle)

The strong emotive language used within these accounts, to describe being at home for too long, was particularly striking. Older people feeling as though they were imprisoned, has been identified in research for individuals living in both nursing homes (Falk et al., 2013) and at home (Smith, 2012). Furthermore, within research exploring community mobility and the important role of social engagement in maintaining identity for older people, Gardner (2014) found that the home environment was "frequently referred to as a prison from which escape was necessary" (p.1252). Existing literature has described the "bodily imprisonment" experienced when difficulties arise with carrying out desired everyday tasks (Falk et al., 2013, p.1005) and particular challenges have been identified for individuals with health conditions such as dementia (Heggestad et al., 2013) and Parkinson's disease (Bramley and Eatough, 2005).

For many participants, the length of time spent at home influenced whether being at home was perceived positively or negatively. For example, in the quotation above, Margaret stated that if she stayed in the house and didn't get out for "any length of time", as in an extended period of time, she would "nearly go mad". Participants described frames of reference that they felt were acceptable amounts of time that they could *cope* with being at home. However, if this was exceeded, it went from being something enjoyable, to something that had to be endured and their mental health deteriorated. In a joint interview with Niamh and Méabh, they both expressed concern that their days would be "endless" if they stayed indoors all the time:

Méabh: I'd go mad always sitting... endless days like this, just sitting.

Niamh: They would be endless.

Méabh and Niamh demonstrated how they would perceive time differently if they were at home, because they would not have as many different activities to carry out and would spend more time just sitting. Again, Méabh referenced how she would "go mad" if this was the case. This example highlights variations in perceptions of time and the importance of recognising subjective perceptions of time, or "experiential time" when considering why getting out and about is so valued (Dodgshon, 2008). This example shows how this is linked to the types of activities carried out by individuals, in particular, how certain activities or occupations typically carried out at home or beyond the home can positively or negatively influence perceptions of time. A common account by participants was that if they stayed in all day, they often felt that the day was too long, but if they left the house, even for a short period, it helped to break up the day. This was the case for Nuala. Nuala described instances when she might be in for the whole day, her perception of time would change. This could be during winter time or when her daughter that she lives with is out, as she usually goes out with company:

I do like to get out, if it is only for an hour. It breaks the day. Sometime in the winter days I probably wouldn't go out, my daughter may be out, and I wouldn't go out if it is raining or what have you and the day is endless. (Nuala)

June described how when she stayed at home all day, it became a "very long day":

When you stay in all day it is a very long day. I did recently over the holiday weekend I stayed in all day and that night you are saying, God I have been in this house all day. Not that there is anything wrong with the house, but it is better to get out for an hour, even walking the dog gets you out and you meet people. (June)

Both participants felt that getting out for an hour was enough to help minimise the day feeling too long. Shauna described how she preferred to go out in the afternoon, because it "shortens the nights". Going out in the afternoon was then a strategic action by her to ensure that the length of time inside the house was not too long and that going out at a particular time helped with this. I return to the topic of routines and structuring the day by getting out and about in Chapter 6, but these examples

highlight that there was an acceptable level of time to be outside for participants, in order to make being inside more easily coped with and enjoyable.

The amount of time needed out of the house varied amongst participants. Bríd, the leader of the community centre that took part in the group interview, recognised the importance and benefit of being involved in a structured activity once a week and explained what a big deal it was for some of her members to achieve this:

So it is very important to be able to get out and to join a club or become involved in something even if it is only once a week. (Bríd)

The participants that took part in this group interview described the purpose of their social group, which was to help older people within the locality that were not yet ready to attend the bigger social groups for older people in their area. It was deliberately a much smaller social group, designed to support people that had become isolated to re-engage with their communities:

We are providing an outlet for people that are isolated and in isolated areas and don't really want to go to big centres, are not ready to go to big centres. (Bríd)

For older people who have not been used to leaving the house regularly, attending an organised activity once a week would be quite an achievement, particularly when they may be rebuilding their confidence about leaving the house.

Variations in perception of time, depending on whether participants were at home or beyond their homes, can be partially explained by literature on "flow state", a concept originally developed by Csikszentmihalyi (1975), defined as the positive "experiential state that occurs as one approaches optimal engagement with a task" (Payne et al., 2011, p.738). In these examples, getting out and about was perceived to make time move more quickly, which is an important component of being in flow state, where an individual loses track of time (Csikszentmihalyi, 1975). When time was mentioned by participants, moving quickly was preferred, as has been found in literature exploring the rhythms and time-spaces of older people (Lager et al., 2016). Faster perceptions of time are also associated with better psychological functioning, improved wellbeing amongst older people, particularly with regards to sense of purpose and control (Baum et al., 1984). That is not to say that being at home cannot also produce flow state; indeed one study found that participants were more likely to enter flow state at home, rather than beyond the home and found gendered components of this (Heo et al., 2010). What matters though, is the quality of the experience, the type of activities carried out and how occupied the person is. It is also important to recognise that the feelings associated with getting out and about, can extend beyond the event itself. In particular, getting out and about can influence how participants feel when they come home, as well as how they feel about home.

5.2.4 *Getting out and about is more important when you live alone or are widowed*

An important factor that influenced how important it was for people to leave the home, as well as how long they left the home for, was whether or not they lived alone. Many of the participants that felt very strongly about the importance of getting out either lived alone, were widowed, or both. Arguably, they valued getting out and about even more because of this. Michelle lived alone and reflected that because of this, she spent a greater amount of time out of the house:

I spend quite a bit outside; I don't stay in the house because I live on my own. (Michelle)

Brenda lived alone in sheltered housing but spent much of her time visiting family members throughout the week. She described how she was "willing to go anywhere" to have company and to enjoy herself. Living alone gave her extra motivation to get out and about, even though she found getting out challenging due to health issues:

I am willing to go anywhere just to enjoy myself, company. Because it is very lonely when you are on your own, so you need company. (Brenda)

From the most recent Irish census (2016), among those aged 65 and over, 12.1% of males were widowed, compared to 35.0% of females (CSO, 2021a), while 21.3% of males lived alone, compared to 31.5% of women (CSO, 2021b). It is well documented in literature on ageing and social isolation that living alone increases with age and that women are more likely to experience living alone in Ireland (Kamiya and Sofroniou, 2011). This is partly due to the increased likelihood of being widowed. In general, living alone (Wenger et al., 1996; Finlay and Kobayashi, 2018;

Kearns and Tannahill, 2015; Klinenberg, 2016) and widowhood (Utz et al., 2014) are associated with a greater risk of loneliness. Bridie, in particular, showed remarkable resilience and agency in developing strategies for managing loneliness after widowhood and these are discussed further in Chapter 6. This has been found elsewhere, where widowhood initially led to an intense experience of loneliness, but like Bridie, some individuals develop strategies to cope with this transition (Davies et al., 2016). This included getting out and about to connect with others, developing routines and keeping busy (Bennett et al., 2005; Davies et al., 2016). This finding highlights the importance of the social environment in supporting individuals to manage this transition (Zebhauser et al., 2015; Pinquart and Sorensen, 2001).

5.2.5 Summary

This first group of participants felt very strongly about the need to get out and as shown by Bridie, it was an important component of how a good quality of life was defined. As shown throughout this section, getting out and about was perceived to have a variety of benefits to mental health and wellbeing. These benefits arose from connecting to others, engaging with nature and being physically active, as has been found elsewhere within the literature (Alves and Sugiyama, 2006; Sugiyama and Ward-Thompson, 2007). However, these findings also highlight that the relative importance of each of these varied between participants. Participants raised concerns about the deterioration of their mental health if they were unable to get out. In particular, they feared getting "stuck", "vegetating" at home and effectively being "imprisoned" in their homes. An important factor in how positively being at home was perceived was the length of time they were at home versus out and about. Participants demonstrated good awareness as to an acceptable length of time for them and if this was exceeded, they would feel more negatively about being at home. This was partially because their perceptions of time changed and appeared to slow down.

Getting out and about was particularly important for participants who lived alone and were widowed and this was due to the need to connect with others outside of the home. I return to the topic of getting out and about to connect with others in more detail in Chapter 6. These findings have important implications for the Covid-19

123

pandemic, when older people were told to 'cocoon' at home and were unable to participate in valued activities beyond the home. However, as this fieldwork was not carried out during Covid-19, I do not discuss this within the findings chapters, but instead consider what these findings mean for the global pandemic in Chapter 8.

5.3 "I know I can get out"

For participants such as Mairead, it was a comfort to know that she *could* get out, but she didn't feel the need to physically get out all the time. Mairead described herself as a "stay at home kind of a person", and identified a number of activities at home that she enjoyed, including reading. She explained that getting out and about "doesn't bother" her, because she knew she could:

It is just that is the way I am, it doesn't bother me [getting out]. I know I can go out. (Mairead)

Mairead's response suggests that there could be wellbeing benefits simply from knowing you can get out, even if you do not actually get out. This has been found within therapeutic landscapes literature, where more passive experiences of nature and green spaces, such as viewing these from a window, can still produce (some) restorative and therapeutic benefits (Elsadek et al., 2020; Orr et al., 2016; Pearson and Craig, 2014; Taylor et al., 2002; Ulrich, 1984). This has implications for mobility research, particularly literature that focuses on measuring and defining individuals' activity spaces, which explores where people physically travel to and from. A limitation of these approaches is that they do not capture imaginative mobilities and the potential wellbeing that comes from the potential, possibility and availability of getting out and about (Curl and Musselwhite, 2018; Metz, 2000; Parkhurst et al., 2014; Ziegler and Schwanen, 2011).

Mairead was someone that got out most days and had routines that were very important to her (discussed more in Chapter 6). Whilst Mairead appeared to show disinterest in getting out, she mentioned several times during her interview that she "should" get out more to walk and that she often felt stiff because she was not as active as she could be. What was not clear from Mairead's account, was whether she appeared to be disinterested in getting out because she was genuinely not interested,
or whether she downplayed its importance because she had begun to find it more challenging and it was easier for her to stay at home.

Variations in the need to get out reflected the activities, interests and reasons participants had to get out for, as well as the habits, routines, narratives, identities and beliefs about themselves, that had built over a lifetime. For example, although Mairead did get out, she saw herself as a bit of a homebody. Mairead's example suggests that there may be gendered components to expectations of getting out and about, particularly for those generations where women were expected to stay at home. This has been found elsewhere with regards to driving patterns but not with regard to expectations about getting out (Davey, 2007).

Another way that participants responded to the question about getting out and about, was by highlighting the importance of having the choice and freedom to be able to do so. Michael described how much he valued having the choice either to get out or stay at home and how retirement gave him the freedom to do what he wanted:

More than I have the right to do when I want to, that I am able to do it, whether I want to, I like that choice, and if I don't want to do it... the one thing I realised about retirement was that I could do what I wanted when I wanted and that is what I have been doing. But it is important to be able to get out of the house. But it is also equally important that I can stay in the house when I want. (Michael)

Here Michael acknowledged both the importance of getting out but also the importance of staying in if he chooses. This was emphasised by Darragh, who explained the important distinction he made between *having* to do something and *wanting* to do something:

I don't feel the need to have to get out the house, you know? If something requires me to get... I have a policy in life, don't do something 'cause you have to do it, do it because you want to. (Darragh)

Darragh did not feel that he *had* to get out and about, instead he *chose* when to engage in activities that he valued. Throughout a typical week, Darragh had many instances where he wanted to get out and did get out, to the extent that he needed a diary to keep track of everything:

I'm occupied ... one hundred percent of the day, meeting people like yourself, Hannah ... getting involved in different projects. I have to say I have a diary, and I need a diary, because of all the activities I'm involved in. (Darragh)

This highlights the importance of what participants do when they are out and about and how this can influence whether the experience contributes to wellbeing. In particular, this will depend on how important it is to them, and how much they *want to* carry out activities, versus *have to*.

Emer described how being at home as an activity in itself was something that she valued on occasion. She talked about the comfort of a "lazy day" at home:

Yes, sometimes I would like a day in. If I looked out and saw it was lashing down raining and I would feel, oh gosh this is a great day for staying in and doing as little as possible or maybe catching up on something you have to do. But a lazy day, yeah, I like that too, definitely. (Emer)

As an activity carried out once in a while, having a "lazy day" in the house was a very positive and comforting experience for Emer, but this was something that was chosen in response to a particular weather condition. However, it would likely be a very different reaction if it was the only option available to her and she had no choice in the matter.

5.3.1 Having your health to 'be' independent and 'do' independence

Having choice and control about when to get out, as well as physically being able to do so, was an important factor for participants and was influenced by both personal and environmental characteristics. I return to this idea in subsequent empirical chapters, but in this section I outline the ways that health, independence and getting out and about interacted to influence the quality of life of this group of participants. For many of them, getting out and about was the physical embodiment or enactment of independence and helped them to feel good about themselves. This was discussed at several points during Focus Group 2:

It is important to do things on your own because you feel stronger for it, rather than depending on people to go places. (Participant from Focus Group 2)

It is important when you get to our stage in life that each individual person is able to do their own banking, pay bills, things like that are very important as well. (Participant from Focus Group 2)

As discussed in Chapter 3, existing research has demonstrated the vital and overarching role independence, freedom and autonomy play in the quality of life of older adults (Gilroy, 2006; Borglin et al., 2005; Meijering et al., 2019). Literature has also identified important overlaps between mobility, freedom and independence (Curl and Musselwhite, 2018), where mobility is seen as "an expression of personal autonomy and freedom" (Mollenkopf et al., 2011, p.788) and the act of moving independently brings about wellbeing benefits (Ziegler and Schwanen, 2011; Schwanen et al., 2012; Schwanen and Ziegler, 2011). This is often discussed in relation to older people driving and the negative impacts experienced when an individual is forced to give up driving, particularly on identity and sense of self (Alidoust et al., 2019; Davey, 2007; Ziegler and Schwanen, 2011), but also related to carrying out everyday activities outside of the home (Franke et al., 2019). Accomplishing daily tasks and being mobile has been shown to influence wellbeing in a variety of ways, but important factors include the sense of accomplishment, fulfillment and confidence that comes from achieving this (Curl and Musselwhite, 2018; Ziegler and Schwanen, 2011).

Another important component of a good quality of life identified by participants, which was very inter-related with independence, was health. This was one of the highest referenced codes for quality of life (see Appendix 16):

To me good health is very important, to be able to get up and do things every day. (Nessa)

One of the reasons that "having your health", as it was commonly described by participants, was so important to participants was that it was seen as a form of wealth. The phrase "health is wealth" was mentioned several times by participants. Having health was valued so highly because it enabled participants to 'be' independent but also 'do' independence, i.e., to be able to get out and to engage with people and places that were important to them. This was illustrated by both Áine and Nessa:

Health is very important, most important because if you haven't got your health, you can't do and be independent. And my independence is very important to me. I don't want to have to rely on anybody, but I would like to think and know that I can and that is important to everybody. That you can ask for assistance if you need it but don't become a burden on anybody, try and always keep your independence around you and strive to be of good character. (Áine)

Well, the fact that I am able to get up every morning, look after myself, I clean my own house. I don't depend on anybody for anything, but they are there if I need them but thank God, I am very independent and that is how I would define it [a good quality of life] and I have the good health to be independent. (Nessa)

Áine and Nessa's descriptions of a good quality of life highlight the delicate balance of being independent versus dependent and not wanting to be a burden, as well as the importance of reciprocal relationships, which have been identified in existing research (Borglin et al., 2005).

5.3.2 Summary

For this group of participants, wellbeing arose from knowing that getting out and about was an option for them. This highlights the importance of independence and having the choice and freedom to be able to get out in the first instance and the important distinction between having to get out and wanting to get out. This section has demonstrated how participants enact their independence through getting out and about and how this is both influenced by, and influences their health. However, looking at all participants, whilst both independence and health were commonly valued by participants, they were interpreted and defined differently across participants and this was usually related to existing abilities or contexts. In this final group of participants, I consider some of the variation in how participants defined health and independence, developing the conceptual framework of ageing – *as well as you can* – in place, and emphasising the importance of being 'relatively' independent and having 'good enough' health.

5.4 "I like to get out a little"

While getting out and about in some form was consistently valued, the ability to get out, as well as how often individuals got, out varied considerably across participants. Aged 89, Nuala was one of my oldest participants. Her description of the amount she got out was quite different to most participants, in that she stated she liked to get out "a little":

I like to get out a little. I used to go out to visit but unfortunately most of my friends are gone as well, all I can do is visit their graves, which is the sad part about it. But I used to go out quite a lot. (Nuala)

Nuala often described getting out in the past tense during her interview, recognising that it used to be something she did more of. This reflected her changing capacity to get out but also the opportunities or reasons she had to get out (this is discussed more in Chapter 6). Nuala appeared to have accepted getting out less frequently and had changed her expectations as a result. Nuala acknowledged that she did not get out very often, she did still like to get out for short periods of time, identifying a specific amount of time that she preferred to get out for, which was an hour (see Nuala's quotation in previous section). Nuala's example shows how both the extent older people get out can vary over time but also how expectations about getting out can change. Whilst it may be less than previously had been the case, some form of getting out and about was still desirable.

5.4.1 Having 'good enough' health to be 'relatively' independent

Nuala referred to the importance of being independent during her interview. Like her desire (and ability) to get out, the level of independence she desired (and required) had also shifted. Nuala recognised that she needed more help than she used to but still liked to have "a little bit of independence":

Nuala: I just think I get up in the morning and I am able to do my own things, a little bit of independence. Now a lot of things I need to have help with I know, but I think independence.

Hannah (Interviewer): [Independence] is very important to you?

Nuala: Yes, I like to be. Now a lot of things I have to get help with, but I like being independent.

For Nuala, some support was required to ensure that she could still get out and participate. Nuala had support from her niece, who accompanied her at social groups and was present during her interview and go-along interview. This support from her family members ensured that she was "never really stuck" at home. Nuala appeared to be willing to accept this help, because she was still able to have some independence, was able to stay in her own home and was still able to get out.

Nuala's example demonstrates that participants' conceptualisations of independence was not uniform, recognising the "fluid" and "fuzzy" ways that it can be defined by older people (Allam, 2015; Schwanen et al., 2012, p.1313). I return to this idea later in Chapter 8. Nuala's account also highlights the importance of having the opportunity to identify what independence means to her and the capability to choose which aspects of independence she may be willing to re-negotiate over time. By having this freedom, losing some independence may be less damaging to an individual. To recognise that some revisions may be necessary for individuals, I revised Theme 3 related to independence to being *relatively* independent.

Moving on to health, whilst health or "having your health" was commonly identified as being important, it was not as easily or uniformly defined by participants. Participants often had frames of reference that they felt constituted having good health, which would not necessarily align with a definition of complete health, such as that defined by the World Health Organization (1946, p.1). Instead, participant definitions of health varied and in most instances, reflected what the participant themselves was able to attain, ensuring they could frame themselves as having good health in some way. This is closer to Norman Daniels' (1985, p.28) description of a right to health consistent with "normal species functioning", which recognises that this will vary with age.

Returning to Bridie, introduced at the beginning of this chapter, having her health meant being relatively free from pain, as well as being able to do things such as get out of the house and go for a walk. These were criterion and activities that she was able to achieve whilst living with Chronic Obstructive Pulmonary Disease (COPD). Adapting her definition of good health, meant that she could self-identify as having her health. Niamh considered herself lucky to "have [her] health" because she did not have "sickness", was not taking any medication, and was able to keep busy:

I'm at this state now I'm going wherever I can. I'm lucky, I'm not on any medication and I'm not, I don't have sickness, thank God, you know, while I have my health, I'm busy. (Niamh)

Darragh, meanwhile, explained that he was told by his doctor that he had great health. When I asked him whether he was ever restricted in his activity due to his health, he described his experience of being a bus mechanic and how he had always found ways to adapt and compensate, such as "to increase the leverage" when using a spanner to loosen wheel nuts, when he was not as strong as some of his other colleagues. As a natural problem solver, he did not feel limited by his health but instead made small adjustments, asking "is there a better way of doing this?". Whilst he did take some medication, it was only for preventative purposes and because it was preventative, he felt he had good health:

So, yes, my life is very positive, thanks be to God, I have great health, my doctor will say to me, [participant's name] keep doing what you're doing, it's working for you. OK, the only medication I'm on is cholesterol tablets, not because I have cholesterol, but... because of my age. It's a protective, it's a preventative medicine, rather than some ... you know? (Darragh)

Participants using more flexible or "pragmatic" definitions of health over time has been identified elsewhere in the work of Gilroy (2006), who found that participants chose accounts of health that aligned more with subjective wellbeing, rather than objective health and emphasised the importance of engaging in enjoyable activities. Furthermore, this finding aligns with Huber et al. (2011), who argue that the traditional World Health Organization's definition of health is no longer "fit for purpose", owing to the prevalence of chronic health conditions. Instead, they have proposed an alternative definition of health, which is the "ability to adapt and self manage in the face of social, physical, and emotional challenges" (p.1). This definition emphasises the subjectivity of defining health and how this may need to be adjusted over time, and places a greater emphasis on wellbeing and quality of life (Borglin et al., 2005).

5.4.2 Engaging with 'meaningful' activities or interactions

Participants tended to align more with alternative or pragmatic definitions of health, particularly those who had existing health and mobility challenges. Most participants with health and mobility challenges still felt that they had a good quality of life. This was in large part, because they were still able to get out and about and find meaning in their lives (Theme 3). Whilst optimal health and mobility were desired and preferred, reductions in either were more readily accepted and coped with if participants were still able to participate in meaningful activities, many of which occurred beyond the home:

I used to do set dancing, but I don't do it anymore 'cause I wouldn't be able ... it's my age as well as my ... I won't call it my illness, but like, my age... I do have a kind of problem with my hips ... I do have a bit of a problem with my bones, like, you know, I've arthritis and that, but ... it's not stopping me from getting out, and I'm quite happy what I'm doing. (Niamh)

Some re-negotiations were sometimes necessary, as shown by Niamh, where certain activities were no longer possible, but participants still needed to get out and about in some form.

Anne was the only participant to state a direct "No" to the question of whether getting out and about was important to her. She described how she was quite happy to stay in, and was not a person that had to be out all the time. When I asked her how important it was for her to get outside the house, she responded:

No, not really, no... I'm quite happy in my own, you know, my own little... I wouldn't be one of these people that had to be out all the time. Now in the night, I'm like a vampire... going to bingo... I like to get out and walk, but I wouldn't have to be out, out, all the time, do you know what I'm saying? I can sit in my home, I enjoy sitting in my home. (Anne)

Anne's response was interesting, because she began by saying how getting out and about was not that important to her. However, she then clarified that she was not someone that had to be out *all the time*. Yet, when I examined her valued routes and routines through the mapping exercise, she was someone that was out quite frequently, regularly walking relatively long distances, shopping, attending social groups and playing bingo. These accounts did not appear match up, in that she stated it wasn't that important to her and yet her actions contradicted this. To Anne, someone that valued getting out and about spent more time than she did out of the home. However, because she enjoyed being at home, it is possible that she did not recognise the importance of getting out in smaller amounts and may even have taken it for granted.

Anne described how she would spend "the best part of the day in the house". In this instance, she differentiated between people who were out of their house for extended periods of time, whereas she enjoyed being at home. Unusually amongst the women in my study, she was often out at night playing bingo. Many of the activities that were important to her were home-based. When reflecting on how she would define a good quality of life, being able to carry out home-based activities was very important to her:

I like decorating and doing things... so far so good, I can still do my bit of painting and do things. And the day that I can't do that I'd really and truly find it very, very hard. That's one thing that would have a... I'd know then that... I'm getting old. I know I'm old, but you know what I mean? If it started interfering with me that way, that would be a big, big deal of my life. 'Cause I like to just get up and do things and not hang around. (Anne)

One of the activities she valued most was decorating and doing home improvements. These were activities that necessitated being at home more, however, they still required some engagement with outside the home to obtain relevant materials. Anne differentiated between an older person who was still able to do things of importance to them, with an older person unable to do things that they valued. If at a point in the future, Anne was no longer able to carry out these valued activities, she felt that a line would have been crossed, where she would know that she was "getting old". She would also find this very difficult to accept.

Anne's account aligns with a definition of quality of life that is about being able to do the things that are most important to her, whether that is inside the home or outside of the home. This fits with a capability approach to a good quality of life and ageing well – achieving valued functionings, or life's beings and doings (Gilroy, 2006; Meijering et al., 2019). In Anne's instance, this involved carrying out activities

in the home but also getting out and about and engaging beyond the home in some form. Whilst Anne identified a number of activities and hobbies that were important to her, some were rated more highly than others and would have resulted in more negative consequences if they were no longer possible. This highlights the subjectivity of both a quality of life and valued functionings.

5.4.3 Summary

Health was important to individuals to an extent, because it enabled people to get up, be mobile, engage with meaningful activities and get out and about. For those that had challenges, was still possible to have a good quality of life, as long as they can still do those things that are of *most* importance to them. Health challenges were more readily accepted and coped with by participants if their challenges did not prevent individuals from getting out and they had suitable alternatives to replace more vigorous activities that they may no longer be able to carry out. In this way health amongst participants did not have to be perfect to have a good quality of life, it just had to be 'good enough' to be able to participate and engage in meaningful ways. This was the motivation behind the modification of ageing well to "ageing as well you can" and this conceptualisation is further developed in Chapter 8.

5.5 Temporary fluctuations in getting out and about

So far within chapter, I have focused on highlighting the variation in the extent that participants felt they *needed* to get out and about. In this section, I consider the ways that individuals themselves fluctuated with regards to getting out. Several participants have already shown that expectations about getting out changed over time, for example, James and Nuala. I now discuss two ways that participants demonstrated their appreciation of getting out and about and its importance to them, by reflecting on a time in the past when they had been unable to get out and were confined to their homes. The first was due to disruptions to health and the second was due to seasonal and weather variations.

5.5.1 Disruptions to health

One way that participants had been confined was due to temporary disruptions in their health, as was the case for Shauna, David and Michelle. Shauna and David had both broken their legs in the past, which meant that they needed a crutch to walk and were limited in their mobility for a period of time; two years and a month respectively. Shauna and David differed considerably in their recovery and how this influenced their independence. For Shauna, she admitted that there were two "bad years" but she now had home help to assist with cleaning which was a big help to her. David's account was particularly striking, because of how he pushed himself to recover more quickly. This was due to the importance of cycling to him:

I broke my leg two years' ago, my femur, three places, and in a month I was on my bike doing a time trial up on the quays, a month ahead of schedule. And the doctor [was] screaming at me to give up cycling. If I had listened to them, I'd have been nobody. The bicycles and the trike puts me in touch with everything. I am completely independent. I get my groceries in my rucksack, and I exercise the dog 10 miles a day, he needs big mileage. Not for him, me walking around on the lead with him, he runs, and he is part of my life. I have had dogs all my life, but we were told we were getting too old for a dog. No way, he has given me back, I never lost it anyway, independence and keeping an eye out to what is going on. (David)

David's account is revealing when he stated that the doctor told him to give up cycling, but if he had done that he would have been "nobody" because it was his cycling that kept him connected. He spoke of the cycling routine he carried out with his dog and reflected on his decision to get another dog. He started to describe how the dog gave him back his independence, before quickly revising this to state that he "never lost it anyway". This example provides an insight into the sheer determination and motivation David had to get out. This was unusual amongst my participants, even for those that highly valued it. David's answer to the question of whether he ever felt limited by his health was particularly striking and highlighted well the heterogeneity of the older adult experience. His answer was categorised as yes in Table 4.1 in Chapter 4, and yet as a former triathlete, he queried the fairness of such a question:

Well, I can never run a 3 hour 15 minute marathon, which I have done. I can no longer finish an Iron Man, which is a 2½ mile swim, 114 mile cycle and the marathon. I am no longer capable of doing them in 11½ hours. So that is not a fair question to ask a 75 year old guy who eats his heart out when he looks up at the mountains to say, I have run them a million times and I can name every one of them from one end to the other. And I challenge Wicklow men to run the mountains I run in Wicklow. So I am badly curtailed... But looking at myself again 75, I cover 10 miles a day at the very least with the dog and then I am on my other bike then. I can go into town and back in 25 minutes, especially to get bread. (David)

Looking at general trends for older people he would not seem particularly "curtailed" for a 75 year old, but according to his own previous standards he felt he was. This highlights the "discontinuity" between his past self and current self and how he was coming to terms with this (Lager et al., 2016, p.1574).

Michelle described how she nearly lost her sight as a result of being Type 1 diabetic. As a result of medical treatment and injections, much of her sight had been restored and she was very grateful that she was able to read again:

I can see, I am reading my books, the thing I love mostly. (Michelle)

She had recently joined a book club and spoke very enthusiastically about this, as well as how lucky she felt that she could still get out:

But it is great, I am lucky, and I can go out. But yeah, I get out, I definitely do, I am always doing something, I am always busy. (Michelle)

Whilst some of her sight had been restored, she still struggled with her sight, particularly when out walking. She was able to get out, but admitted she had not been walking as much recently, particularly at night. Shauna, David and Michelle were all participants that *had* to get out and possibly this was partially because they had experienced a time when they couldn't and appreciated it all the more when they could as a result.

As well as describing direct experiences, other participants such as Noelle described the experiences of friends or family who had not been able to get out and how difficult it was for them. By comparing her situation with theirs, she was able to reflect on why getting out and about was so important to her. Noelle discussed how her husband had been unable to walk for some time after he had a heart attack and how he missed getting out to his community centre:

I know all he longed for was to get back on his feet. He lost his walk at one stage, all he longed for was to get back on his feet to go [to attend his local community centre] ... it was the first thing that he went to when he got back on his feet again... I mean he goes every day since, so it's so good for him to get out of the house. (Noelle)

When I asked Noelle how important it was for her to get out, she described a friend who was unable to leave the house, because of caring responsibilities to her husband. Initially, she had to travel to the hospital every day and now she recently she had to stay at home to look after him:

Oh, very important to anybody, I think. I think it's just not me, it's everybody ... I see my friend now that used to do the bingo with me. And ... my heart goes out to her, because I know what she must be going through because she loved coming to bingo to meet the other women ... But this woman doesn't get to bingo, she doesn't get out anywhere ... must be I'd say a good five months every day going to the hospital, and he came home last week and now she's kind of confined to the house, she can't leave him ... She lives on her own with him so she can't leave him ... I kind of find, how it must be to her, that she doesn't get ... And ... with somebody like that, sick as well, you're not inclined to have friends coming in, because her whole time is taken up with him as well ... That to me is ... why it's so important to be able to go out. (Noelle)

Noelle mentioned how her daughters made sure that she still attended her favourite activity of bingo whilst her husband was in hospital and that her sister would come up to stay with her, and she admitted "that's how important these things are to you". In this instance, Noelle's friend was hindered by caring responsibilities but Noelle herself, through respite and the care received from her daughters, was enabled to continue attending her social group. This emphasises the relational aspects of being able to be mobile and how it can also be a caring act (Croucher et al., 2020). These findings also highlight that caring responsibilities for others can potentially impede attendance in social groups, which may contribute to the social isolation of informal older carers (see Greenwood et al., 2019).

5.5.2 Weather and seasonal variations

Another way that several participants reflected on a short period of confinement to their homes, was during the unprecedented snow event that Bridie referred to at the beginning of this chapter. Several interviews with participants took place immediately after this event. For Bridie and Edith, there was still snow on the ground. During this weather event, many of my participants had been housebound for several days and some reflected on the experience during their interviews. This was mainly those that had been interviewed immediately after the event, but there were some mentions of it amongst those interviewed several months later. Bridie spoke about the "cabin fever" she had experienced during this time:

Hannah (Interviewer): How have you found the last few days?

Bridie: Cabin fever, certainly cabin fever. It was desperate, but we got through. The neighbours were great, calling in making sure I had everything, they were very good. Everything was fine, thank God.

Bridie recognised how easy it could be to "get into a rut" and how this was particularly the case during the winter months:

It is very easy to get into a rut and it is very hard then to break the habit to get out, especially during the winter, it is very hard when you are sitting here to get a coat on and walk down, very, very hard. But once you get down there you enjoy it, and you are glad then when you come back that you have done it. So it is very easy to not go out. (Bridie)

She described a neighbour that struggled with getting out and how hard she found it to "break the habit" and get out:

And God love her she died on her own. She only died there in December, and she never done it, she was a lonely, lonely woman. She just wouldn't break the habit of [not] going out. (Bridie)

She spoke about how she tried to encourage her to attend social groups but she would not attend. She reflected how she would "hate" for that to happen to her:

I would hate that to happen to me, I really would. That is how I spend the time, I would hate to be stuck in the house, it is important for me to get out, it definitely is. (Bridie)

Brenda was interviewed in July 2018, approximately 5 months after the event but described how she had struggled during the previous "bad winter" and how lonely she had felt:

Sure, the house, the bad winter there, holy God there was many a time I had to miss the knitting and all, it was so lonely because everybody's door was closed. (Brenda)

James, described how he had coped reasonably well during the snow and was able to adjust in the short term, but revealed the relief he and his wife felt when the snow had cleared enough to be able to walk to the shops:

James: I managed alright during the snowy days. When it came to the point where it was nearly good enough to go out for a walk or go down to the shops, I was glad. Even herself [James' wife] was the same. She said "I haven't been out in one ah..." ...we went down to the shops even when there was nothing in it.

Hannah (Interviewer): Just for the walk? [laughs].

James: Just for the walk yea. So, so it is important to get out, but I can adjust myself if need be, but I wouldn't like to be confined to it yea.

James' account contrasted with Bridie's and Brenda's in that he "managed alright" during the days that he couldn't get out. He felt that he was able to "adjust" himself if needed but that he wouldn't like to be "confined" all the time. Brenda and Bridie found the experience much more challenging, due to the strength with which they usually *needed* to get out, owing to them both living alone and being widowed, and the subsequent need to get out to engage with other people.

5.5.3 Summary

This final section provides accounts of temporary disruptions to getting out and about, as a result of health challenges, as well as seasonal and weather variations. Weather and seasons have been identified as important environmental factors which influences the extent and ease with which older people get out and about (see Böcker et al., 2017; Finlay, 2018; Franke et al., 2013; Hjorthol, 2013; Holland et al., 2005; Ludwig, 1997; Tanner, 2010). Existing research has also highlighted the "seasonal rhythms" that occur amongst older people (Lager et al., 2016, p.1573) and how for some this can lead to a "dependence on the weather" (Lager, 2015, p.104). What was notable about these accounts was not that fluctuations existed, but the variation in the extent that participants felt that health challenges, weather and seasons influenced their ability to get out. This highlights the heterogeneity within the older adult experience. These examples demonstrate that for some participants who have either experienced a time when getting out and about was more challenging, or known someone else that had, were more appreciative of being able to get out when they could. As a result, getting out and about is often more highly valued when it is threatened, or at least its importance is most acknowledged when at risk. Experiences such as these highlight how difficult many participants found spending several days at home, as well as variations in how adaptable they were to these changing circumstances. These findings have implications for the recent Covid-19 lockdowns and pandemic, which I will discuss further in Chapter 8.

5.6 Conclusion

This chapter has set the foundations for the remainder of the empirical work. Within this chapter, I have not explored in detail the ways that participants got out and about to engage in meaningful activities and interactions (Theme 3). In Chapter 6, I provide an overview of this and focus in detail on how participants got out and about to connect with others. I have begun to demonstrate how getting out and about can be easier for some of my participants than others and how this is linked to their complex and dynamic person-environment interactions, as well as having adequate levels of safety, security and support (Theme 2). I explore this theme in more detail within both Chapters 6 and 7.

There were marked differences with regards to how getting out and about was valued and how much participants felt they *needed* to get out. Some participants commented that it was vital to their wellbeing and had a strong urge to get out; for others it was nice to know they could but stated that they didn't have to get out all the time. Others, particularly those who were more constrained, liked to get out a little. Some participants took getting out and about for granted and those that had experienced a time when they were homebound in the past, valued it more highly, because they had experienced life without it. Getting out and about is an everyday, lay term and is often assumed to be uniformly important to older people Whilst there is a wealth of literature that examines the importance of older people getting outdoors, leaving the home, or being mobile, as demonstrated in Chapter 3, there is a lack of research that has looked specifically at the importance of getting out and about as a concept and why this is important to older people themselves. However, this research has highlighted some of the subtleties in terms of its meaning to participants.

This chapter has provided an overview of how important getting out and about was to my participants. To illustrate the variation between individuals I grouped participants into those that *had* to get out, those that *knew they could* get out and those that were more restricted but liked to get out *a little*. For those that had to get out, I began with Bridie, whose account highlighted how getting out and about was vital for her quality of life. Within this section, I emphasised the ways that getting out and about was important for mental health in particular. I also identified concerns that participants rose about getting stuck at home and showed that getting out was particularly important for participants living alone and who are widowed. This research has shown that the *amount* that participants get out does not necessarily align with its *importance* to an individual.

For those that knew they could get out, this provided comfort and wellbeing benefits because they were still able to attain it. This section highlighted the importance of freedom and choice in how to get out, as well as the important role that health plays in enabling the independence to get out. The final group valued getting out in some form, but due to varying preferences and in some instances reduced abilities, they liked to get out a little. This group had reduced expectations and had to make some adjustments, for example, being relatively independent or defining health more pragmatically or alternatively. Yet, because they were still able to get out in some form to engage in meaningful activities, they were able to insist that they had a good quality of life. For example, Anne would know that she was old when she was no longer able to do what mattered most to her. This aligns with a capability approach to ageing well, emphasising the valued functionings of older people (Gilroy, 2006; Meijering et al., 2019). This is the theoretical lens for the conceptual model of ageing – *as well as you can* – in place, introduced at the beginning of this chapter. Participants had very clear ideas about what was a good quality of life for them and what was not. Being able to get out and about for a variety of reasons was at the heart of this.

Mobility contributed to participants' quality of life in many ways. Having the choice and option to get out helped to ensure that individuals do not feel trapped at home. Many participants raised concerns about getting stuck and feared being "imprisoned" in their own homes, which has also been found elsewhere (see Gardner, 2014; Olsson et al., 2013; Sixsmith and Sixsmith, 1991). For many participants, getting out and about was the physical enactment of independence and achieving this enhanced wellbeing. This was predominantly due to the freedom of being able to move and travel in desired ways. These findings connect with literature that has examined the wellbeing benefits of out of home mobility. For example, Mollenkopf et al. (2011) recognise that mobility is an emotional and affective experience, which can create a sense of independence and that independence is an "expression of personal autonomy and freedom" (p.788). The empirical findings within this thesis confirm existing research that has emphasised the importance of mobility (and getting out and about more generally) for wellbeing, with a particular emphasis on the complexity and inter-connectedness of mobility, independence and wellbeing (Schwanen and Ziegler, 2011). There were considerable differences in how much getting out and about was valued, as well as in the extent to which individuals wanted and were able to get out. For many of my participants, just knowing that they could get out was enough for them, they didn't have to then enact this. This highlights the importance of imagined and potential mobility (Metz, 2000). If we were to only look at how often or the extent that participants got out as a measure of its value, such as through activity space data, its importance to older people would be under-valued.

There has been a growth in research examining the activity spaces of older people, to understand where they travel and how (see Hirsch et al., 2016; Hirsch et al., 2014; Perchoux et al., 2019; Tsai et al., 2015; 2016). However, as highlighted in Chapter 3,

142

a limitation of this work is it is unable to capture the meaning or value placed on this travel, as well as the perceived wellbeing benefits. Responding to this, literature has begun to incorporate more lived experiences into mobility research (for example, Franke 2018; Meijering and Weitkamp, 2016; Milton, 2015; Sturge et al., 2021; van Hoven and Meijering, 2019). Furthermore, activity space research does not allow for the importance of potential travel to be recognised (Metz, 2000). This has implications, because only looking at where people go only tells part of the story and risks undervaluing the importance of getting out and about to older people. As this research has shown, the amount a person gets out does not necessarily coincide with its importance to them. Indeed, for individuals who struggle to get out, a trip to the main street has significantly more value and importance than someone who can do this regularly and who is uninhibited; and who may take this for granted as a result.

This page has been left intentionally blank.

Chapter 6. Getting Out and About to Connect with Others

But you come around here [to older adult social group], you have a laugh, you have a joke, you go out on the street, and you meet somebody. And that is why community is so important, it is better than a tablet. The doctor can prescribe all the medication he wants but if you haven't got that blanket around you, nothing will do you any good. That is my look at it. (Unidentified Participant from Group Interview)

6.1 Introduction

As demonstrated in Chapter 5, many participants valued getting out and about because of the opportunities it provided to connect with others. Chapter 5 introduced the empirically grounded conceptual model for this thesis: ageing – *as well as you can* – in place, defined as being able to get out and about, in order to participate and engage in meaningful activities and interactions. This chapter develops Chapter 5 further, by showing *how* getting out and about and connecting with others can contribute (positively and negatively) to ageing – *as well as you can* – in place. Throughout this chapter I demonstrate the various ways that participants *enact* their desire to connect with others by attending and being seen in a variety of places throughout their daily lives. These places and spaces then provide the settings for various types of social interaction to occur.

6.2 Phases of Getting Out and About

Within this chapter, I have grouped some of the places important for social interaction into different phases of getting out and about. Grouping places in this way has helped to draw out some of the differences within and between them with regards to how participants interacted with others. The phases I have chosen, have been informed by the literature from Chapter 3, including literature on social infrastructure (Klinenberg, 2018), third places (Oldenburg, 1989; Oldenburg, 1997; Oldenburg and Brissett, 1982), as well as Gardner's (2011, p.268) "sites of significance" for older people and categorisations of third places, thresholds, and transitory zones.

6.2.1 Phase 1: Thresholds

Gardner (2011) argues that thresholds provide "easy and readily available opportunities for social interaction", typically involving neighbours (p.266). The initial purpose of being in a threshold could be to do some sort of garden-related activity, to be in nature, to obtain some fresh air, or it could specifically be to have an interaction of some kind. The interactions are typically more informal and spontaneous, but they may be anticipated, especially if past experiences had shown it is likely to happen. Within this chapter, I focus mainly on interactions within gardens for this phase, as these were most frequently discussed by my participants. I also include experiences of being on a road located in very close proximity to a participants' house. This is not quite a threshold (Phase 1), as it is beyond the garden and technically within a public space, but it was not deemed quite Phase 2 (discussed below) either, because it did not involve travelling as an activity. Being "on the road" was the phrase my participants used for a particular kind of experience within this public space, which happened close to the house and therefore served as an extension of the threshold.⁴

6.2.2 Phase 2: Transitory Zones

This phase of getting out and about is partly captured by third place literature and Gardner's (2011) transitory zones. Within this phase I differentiate between the different purposes behind this movement. As explained within Chapter 3, Gardner (2011) describes transitory zones not as "destinations" but places we "pass through" (p.267). However, by specifying that transitory zones are only passed through, implies that they would not be chosen to visit unless moved through. However, as shown in Chapter 3, this is not always the case, as quite often individuals may choose to move just for movement's sake, without a destination in mind. By thinking of these spaces as transitory zones, rather than actively chosen, risks diminishing the importance of these acts and the value placed on being in these spaces.

⁴ I distinguish here between chatting on the road outside the immediate vicinity of a home (Phase 1) and chatting on a road or street when out and about, either walking in and of itself, or to or from a destination, which I would consider to be Phase 2.

and of itself, for example going for a cycle or a walk (2a) and moving to get to a particular place or destination (2b), for example getting a bus to the supermarket. Interactions within this phase could be spontaneous or pre-planned, with friends or neighbours of strangers. This recognises both intrinsic and extrinsic benefits of travel (Graham et al., 2020).

6.2.3 Phase 3: Destinations or Third Places

I focus within this chapter on public places beyond the home identified by my participants as important for social interaction, rather than other private dwellings, such as the houses of friends of families. These include the various places where social groups take place, which are predominantly community centres, as well as churches, predominantly to attend Mass. Some destinations were important because they were specifically visited for the purpose of social interaction, whilst others created opportunities for interaction alongside or during an activity, but where socialising was not the primary reason for attending (adventitious interactions). Other destinations were important because they provided the reason to travel and therefore created opportunities to interact with people along the way during Phase 2.

Scheduled and Frequently Attended Destinations

Destinations typically included places that would traditionally be considered third places. However, the features of third places have not been rigidly applied here for several reasons. Oldenburg (1989) traditionally conceptualises third places as informal places, where activities are not scheduled or organised and where membership is not required. However, my analysis showed social groups as the most highly referenced category of places for social interaction, and these were typically engaged with in a frequent and scheduled way. Participants' frequency of attendance varied. Some were regularly attended and part of a daily or weekly routine, whilst others were visited more irregularly. Additionally, some of these engagements were pre-planned and scheduled, whilst others were more spontaneous. Adhering strictly to the traditional view of third places would exclude many participants' important places. Furthermore, findings suggest that the nature of these engagements being frequent and scheduled, and in turn part of a planned daily or weekly routine, was

very important. In some instances, participants attended multiple destinations or meeting places whilst getting out and about. This is a common travel pattern known as "trip chaining", which is where people, typically women, carry out "several small interconnected trips" in one go (Criado-Perez, 2019, p.30). I believe that the scheduled nature and trip-chaining patterns associated with attending destinations is something that has been overlooked within third place and social infrastructure literature and highlighting it is one of the key contributions of this chapter.

To begin the empirical findings, I continue with Bridie from Study Area 1, cited in Chapter 5 when her overarching functioning, to get out and about to connect with others, was identified. Bridie was one of the participants whose place-based functionings were clearest and best demonstrated the interactions within the different phases of getting out and about. For this reason, I have used Bridie as a case from which I then compare the broader sample's experiences in Section 6.3. For the remainder of the chapter, I provide participant experiences in each of the phases of getting out and about. Phase 1 in Section 6.4, focusing on interactions within gardens and in front of the house. Phase 2 is considered in Section 6.5, showing how for some participants moving throughout their neighbourhoods provided a guarantee of social interaction due to their "high profiles". Non-human interactions and interactions with strangers, and how participants varied in terms of how comfortable they were doing this are examined, before considering some of the variations in walking routines and how this influenced participants' visibility in Phase 2. Phase 3 in Section 6.6 focuses on two destinations: attending Social Groups and attending Mass. Some of the additional outings and interactions that occur due to attending Social Groups in the first instance are also shown. Within this section, I demonstrate how attending scheduled activities such as these, provided additional purpose and structure for participants to get out, before concluding in Section 6.7.

For each phase, the key places of importance for social interaction identified by participants, including the number of references that a category of place was mentioned. Consideration is given to the motivations, purpose, and activities carried out within these different places (RQ2.1). Examples and experiences of the different types of social interactions that occur and who they were with (RQ2.2) are presented, before providing insight into how positive or negative these interactions were and

how important they were to participants (RQ2.3). Finally, I examine how planned or unplanned, frequent or irregular, spontaneous or expected these interactions were (RQ2.4).

6.3 How did participants connect to others whilst out and about?

6.3.1 Continuing Bridie's Story

Throughout Bridie's interview, it was apparent that she was very busy and socially engaged. She proved to be one of the most socially engaged people I met during my field work. She admitted she loved talking to and visiting people because she was "a real chatterbox". Like many participants, Bridie attended several different activities throughout a typical week, where she engaged with others. During her interview, she identified places, activities, interactions, routes, and routines that were important to her when she was out and about. One of the main activities that Bridie carried out throughout her week was attending Mass at a local church first thing in the morning, which she attended most days:

I go to Mass as often as I can, which is almost every morning. (Bridie)

In addition to attending Mass, Bridie attended different social groups in several community centres throughout the week. The main community centre she attended was her local community centre, located a short walk from her house. Here she attended (and I recruited her from) a craft social group that met twice a week. She was also President of a Ladies Club that met there once a week. In a different community centre, she attended an exercise class at least once a week and attended church meetings regularly. Bridie took part in additional trips because of her initial engagement with these social groups. She explained that once a month Bridie's Ladies Club would take the evening off and do something different and go for a day trip or evening outing:

We go out with the Ladies Club one Tuesday every month, we take the Tuesday night off and we go to the cinema, or we go for a meal, or we just go for a drink, or we might take the whole day off and go down the country somewhere because we all have free travel, it doesn't cost us anything. So different scenery, different four walls as I do say. (Bridie)

149

Bridie had another group that she travelled to Dublin with once a month to attend a market and have lunch:

There is this group of women that I go into [a Dublin market] every third Saturday of the month, and we have our dinner out and go around the market and that. I am out a good lot, so it is very important to have your friends and everything, very important. (Bridie)

Bridie attended additional events at her local community centre when they were on, including a monthly pub quiz. She was also involved with a music group with her daughter in a local church and regularly helped with that.

On days when Bridie did not have a community group scheduled in the mornings, she typically carried out the same routine and walking route during this time. Bridie mapped this route during her mapping exercise, and we drove some of it during her go-along interview (see Figure 6.1 below). This route involved walking to the town centre of Study Area 1 immediately after attending Mass. She would then go to the supermarket, walk back, and have her lunch. Other places she might visit on this route included a second-hand shop and her local credit union. On Fridays, she visited the local post office to collect her pension. Below is an extract from her mapping exercise where she described this route and how she met people along the way:

This is where I go for my walk, I come out of mass and I come down [street name] and the by-pass and there is the [name of small shopping centre], and I would go down that way and down to here. [Name of supermarket] would be down here. I would go to there and then walk back up to the bridge, cross the bridge and back home. It doesn't look much on that. But there are shops along here. It takes me a good while, again I meet people, I'd be walking with people up here and I'd be yapping with them. And then when I cross the by-pass there's shops here, there is a second-hand shop and I love going into the second-hand shop. And then I would go to the credit union, which is here, and then I would go to [name of supermarket] and I'd come back up from [name of supermarket] and go into [name of second supermarket] before I would come back home. (Bridie)

Bridie reflected during her interview that not a day went by where she did not talk to someone and sometimes, she would not even have to leave her garden for this to

happen. She mentioned that if she was in her front garden, she often talked to someone as they walked by:

I would talk to anyone [laughs], I would talk to anybody. There is not a day [that] goes by that I don't speak to somebody because all I have to do is walk down to the gate some days and you will meet someone going up or down the road. You can see yourself they are going by all the time. And all I have to do is stand out there [in her garden]. Sometimes during the summer, I would go out and cut the grass and I would time it, it would be half an hour before something comes on the telly or whatever and I could be out there for four hours talking. (Bridie)

As a result of Bridie's attendance in a variety of social groups over time, and because she has lived in her neighbourhood for many years, Bridie knew a lot of people and felt very comfortable in her surroundings:

I like the fact that people will never pass you by without talking to you [emphasis added]. You could walk up the road, which is only a couple of hundred yards, and it could take you an hour, it could take you two hours, you'd be talking to people so much and that is what I like the friendliness and you feel so comfortable in your environment. I feel comfortable anyway. I don't feel an outsider in it, I feel very comfortable actually, I love living around here. Everyone is just so nice, we are very lucky in that way. (Bridie)

Figure 6.1 overleaf is a map of the key places and routes that Bridie identified throughout her interview, mapping exercise and go-along, complete with a map key. I have labelled the main routes that she travelled within her neighbourhood (Phase 2), as well as the key destinations (Phase 3) that she visited. I have differentiated between scheduled and frequent activities that she attended such as social groups and attending Mass, and other destinations such as shops, post offices and health-related buildings. This was because they were either less frequently attended or not scheduled. I have identified some of the challenges that she identified during her routes, labelled as orange warning triangles. One of these was a bridge close to her house that she had to climb and cross to get over a busy dual carriageway, in order to get to the centre of Study Area 1 where the shops are located. She pointed it out on the map and explained how "That bridge there gets me". Although she enjoyed walking, she found it difficult to climb the stairs because she had COPD and

struggled with her breathing. The second challenge identified was the "long road" that she had to walk to get to the post office. She did not like to do this route because it felt "never ending" and "monotonous":

Bridie: That is a good walk. This road is monotonous, from the time I moved in here I hated it, I hate walking it, but I have to do it.

Hannah (Interviewer): It is not a very pleasant walk?

Bridie: No. I don't mind walking down but walking back up it, I hate it, it is never ending and ours is the last turn. But look, you do it and that is it.

To summarise, Bridie was a very socially engaged individual, who interacted with others in each of the three Phases. I now consider my broader participant experiences across each of the Phases.



Figure 6.1 Bridie's Places and Routes of Importance and Map Key

Мар Кеу

\star	Scheduled and Frequent Destinations (Social Groups and Mass)
\star	Other Destinations (including Shopping Centres, Charity Shops, Supermarkets,
	Health-Related Places, Post Offices and Credit Unions)
0	Bus Stop
\land	Challenge identified along walking route
	Regular Walking Route after attending Mass
•••••	Longer Walking Route to collect pension at Post Office (weekly)
_	Route of Go-along Interview
	Approximate location of home
	Approximate 500m buffer from home

6.4 Phase 1: Crossing the Threshold

Within this section, I discuss the importance of gardens, as well as being in front of the house and on the road for social interaction. Most participants had a front garden and many identified gardens as important to them (32 references). Gardens were discussed in three of the four focus groups and by twelve participants in the main study. Of these, two participants discussed or pointed out other people's gardens during their go-along interviews, whilst one participant described her daughter's garden. The remaining nine participants referred to their own gardens. In terms of the activities carried out within gardens, gardening was the most common activity (16 references). Two of the focus groups and eight participants in the main study mentioned gardening. Shauna (from SA2) described gardening as a particular hobby of hers, although in recent years she was finding her garden difficult to keep on top of, due to a leg injury (discussed in Chapter 5). Three participants mentioned sitting in their gardens or reading as an activity. Several participants specifically mentioned social interaction taking place within gardens or in front of the house (9 references):

If you were out doing the garden, they would stand talking to you if they were walking. (Participant from Focus Group 2)

It is nice to meet neighbours on the road and have time to talk. (Participant from Focus Group 4)

Neighbours would be out in the back garden still; particularly with next door on this side we still have a low wall. And there would be a regular conversation. (James)

Now I would see the family there if they were in the garden and I'd be standing up there talking over. That is nearly gone, having a chat over the wall. (Shauna)

Interactions typically involved a participant being in the garden and then either chatting with a neighbour in their garden, facilitated by having "low walls" in their back gardens, or chatting with someone walking past their front garden. Many other studies have identified gardens as important sites for social interaction (see Bhatti, 2006; Bhatti and Church, 2001; Burton et al., 2015; Milligan et al., 2004), as well as talking on the road (see Ziegler, 2012).

The use of gardens for social interaction was captured particularly well by Bridie's example, when she reflected if she were to be in her front garden, she would often get talking to someone who was walking by, particularly during the summer. Bridie described seasonality to her being in thresholds such as gardens, which in turn meant there was seasonality to the frequency of interactions within them. This was also described by other participants:

In the summer if you are out in the garden, somebody passing by, and you'd have a little chat. (Nuala)

I would talk a bit more particularly with the summer, okay with winter you mightn't spend as long outside, and you wouldn't hang around as much. (James)

I witnessed first-hand one of these interactions during a joint go-along interview with Margaret and Nessa from Study Area 2. During this, we walked from their local community centre, where that they had just attended bingo (and where they were recruited from), back to their respective houses. This go-along interview took place on a sunny early evening in August. At one point, we passed by a house with people sitting in their front gardens. Margaret and Nessa proceeded to chat with them for several minutes. Afterwards Margaret said to me "you see, on a nice day, everyone's sitting out". This demonstrates the important role of front gardens for socialising, as they can provide a space to look out and connect with the world from the comfort and safety of one's own private space. The interactions in Phase 1 typically involved spontaneous, light-hearted interactions with neighbours or acquaintances. For some participants, like Bridie, they were anticipated because they happened regularly and because participants felt comfortable being outside in their gardens or on the road. The frequency of interactions within this Phase, and indeed, whether they happened at all, was therefore strongly influenced by how participants perceived their neighbours and neighbourhoods.

6.4.1 Perceptions of Neighbours

A key topic of conversation during two of the Focus Groups was the importance of good relationships with neighbours. During the Focus Groups, participants reflected on whether they felt they had good relationships with their neighbours. One participant described the particular nature of these relationships and why they were so important:

They are essential, having a relationship with them and knowing your comings and goings and looking out for you really without you saying anything re: the house and all of that. So the friendship of neighbours without them coming in for tea non-stop, it is just respect. (Participant from Focus Group 4)

Interestingly, this participant talked about a particular type of relationship or friendship with neighbours, one which was not too invasive or based on *too many* interactions, but instead one of mutual respect and looking out for one another. Several of the participants from the main study also described interactions with their neighbours. Áine from Study Area 2 shared an interaction she had with her neighbour's grandchildren and within this, gave examples of neighbourly gestures her and her neighbours carried out for each other, such as taking out the bins:

They [next door neighbours] have a little tent in that garden. The kiddies play... The other morning, it was a wild day out of the blue, I was coming out my door and as I came to the door, I knew there was something at the door, and [it was] their tent! The wind had picked it up, and they came over and it was outside my door. I said, "Oh! I have a new spare room!" [Laughter] [I would] also take in her bins. That's what we do. [Name of next-door neighbour] next door might put out the bins the night before for us all, and then if they're not around and they're empty, someone else will take them in and put them in. It's just a little neighbourly thing. (Áine)

Eamon from Study Area 1 described interactions with his neighbours over time and amongst different generations. He described how everyone moved into his estate at the same time. People had since had children and these children had stayed in the same area. He described an instance many years ago where he told a child off for kicking a ball at a wall because it hit his car. He then explained how the person that used to kick the ball at the wall is now grown up. A few weeks ago, his son was out the front of the house kicking a ball against the wall and the father was now telling him off. Eamon described how he made a comment about this to the father, and they laughed about it. For many participants, neighbourly interactions such as these were common. However, some participants also raised concerns about a decline in these types of interactions. Participants within Focus Group 2 discussed how they felt that these interactions had diminished over time:

Very little [interaction] on my road now, all young people really on our road and there are electric gates now, you wouldn't see... I could go for a walk for a month, and I mightn't meet or see one neighbour. (Participant from Focus Group 2)

Another participant talked about how the interaction that used to be there "years ago" has "all gone" (Participant from Focus Group 2). Whilst some participants commented on a decrease in neighbourly interactions, there was also a noticeable *absence* of these types of interactions mentioned during interviews, or observed during go-along interviews, compared with interactions in Phase 2 and Phase 3.

In some instances, participants raised concerns or fears and identified negative interactions with others within this Phase, which provided insight into how they perceived some of their neighbours. Anne showed me two places very close to her home during her go-along interview in Study Area 1. The first was her back garden, a place that was very important to her. She had views of the Dublin Mountains in her garden but what she wanted to show me was a new housing development immediately beyond her wall that she was concerned about. Anne was "distressed" about who would be moving into the houses and whether they would climb over her wall. She was likely to lose her view as a result of this development, which she had for over 40 years. She explained to me that it was not the loss of her view that she was concerned about, but her sense of safety. Anne identified that there used to be a lane behind her garden which was a place for anti-social behaviour. The community campaigned to remove this lane in the 1980s. However, because of this development, a new lane was going to be built, adding to her concerns. She reflected:

I was a young woman at that stage, now I'm 70 and they're bringing this on us. (Anne)

The second place that Anne showed me was a temporary alleyway close to her house, built whilst the development was taking place. Before this alleyway it was a shortcut to get to her friend's house (Jennifer). During the go-along she took me to this lane to show how unpleasant it was. As we left her house to go to this lane, some of her neighbours were out by the front of her house by their car. Anne had raised concerns about these neighbours during the interview, alluding to them engaging in criminal activity. During the go-along, I felt uncomfortable when we were outside the front of the house, as the neighbours watched what we were doing, which was intimidating. I made sure that I hid the audio device, because I was concerned for her safety and didn't want them to suspect that she had reported them, or that I was someone official. This example contrasts sharply with the positive interactions earlier and show how for some, interactions with others were far more enjoyable and safer than for others. Figure 6.2 below shows Anne's concerns about her immediate local environment. Within this figure, I show Anne's house and garden, the proposed alleyway, and the disrupted route to Jennifer's house through the temporary lane.



Figure 6.2 Anne's Concerns within Phase 1⁵

Jennifer lived in the same estate as Anne, and they were friends who both attended the social group that they were recruited from at the local community centre. Jennifer

⁵ An account of Anne's experiences, along with Figure 6.2 was included in a chapter of a book (see Drilling et al., 2021), which arose from the COST Action CA15122 Reducing Old-Age Social Exclusion project, where I was a PhD Forum member and a member of the Spatial Exclusion research group.

talked about how she used to chat with neighbours on the road when she first moved into the estate in the 1970s, but not anymore. She explained that many of her neighbours moved out because a grant in the 1980s incentivised people to leave their social housing and subsequently bought houses elsewhere. She described how she had one "good neighbour" in her immediate vicinity. During Jennifer's go-along interview, as we got into the car, she greeted and chatted to this "good neighbour". She described another instance during her interview of talking to a less well-known neighbour outside her house, who gave her a lift once to her social group.

Both Anne and Jennifer had mixed experiences of neighbourly interactions, largely because of concerns about new people moving into their area. Unlike Bridie, who had positive past experiences, experience had shown them that the people moving in were not always friendly and in some instances, were feared. They did mention positive experiences with neighbours, yet there appeared to be a persistent underlying concern about who was going to move into their area. There was a horizontal cross-section of neighbours, who were perceived as "good" because they were known, either because they had moved in at the same time, were a similar age, or they had moved in later, but a relationship had since developed. These perceptions of neighbours in turn influenced the types, or lack of interactions that happened in these places.

For Jack, who lived in Study Area 2 within an inner-city neighbourhood and whose house was located on a main street into the city centre, it was not so much a concern for his immediate neighbours but frustration with strangers who had stolen items from his garden in the past:

In my front garden now, as I say I am into flowers and my missus she's into gnomes and stuff like that... we have been robbed more times than anything else you know? And so have a lot of other people around the area you know? ... mostly gnomes and little things in the garden, angels or something like that... They go missing... [It would be] be drunken drunkards coming home late at night... Now a fellow around the corner from me, he heard a commotion there one evening, it was about three o clock in the morning and the fellows ... these were well to do fellows... and they were climbing into the garden and taking out plant pots, ornaments, the whole lot just for devilment. (Jack)

159

This example shows that the location of an individual's house may influence the character of any engagement with strangers.

6.4.2 Moving beyond Thresholds

Whilst participants identified examples of interacting with others in thresholds, there were many examples of this not occurring. As a result, there appeared to be a limit as to the type of interactions possible in this Phase. If connecting to others was valued by individuals, relying on social engagement in thresholds alone would therefore not be enough. Participants would need to move beyond the threshold to places in Phase 2 or 3, in order to create additional opportunities to connect with others more regularly and frequently. This was demonstrated by Niamh, who admitted that whilst she liked her gardening, she also liked "mixing" with others, suggesting that this wouldn't be possible unless she went beyond her threshold:

Oh, very important I would say. Well, for me, I like the house and I like gardening, but I do like mixing. (Niamh)

For some participants, Phase 1 did not feel like it was truly "out and about". For example, Brenda, who lived in sheltered accommodation, described that on Wednesdays she might sit in her communal garden but that she still considered this to be "in":

Brenda: Wednesday I do nothing, the place is cleaned, I knit. If it is fine, I might sit out there [in her communal garden] knitting.

Hannah (Interviewer): So you stay here on a Wednesday?

Brenda: I am in on a Wednesday.

Another participant from Focus Group 2 saw Phase 1 as very much an extension of the house, but on a day that she had her social group to attend, she was "going out" and this was a different experience:

I could be in the house all day and I would be very happy there in the garden and all that, but Monday is a day when I am going, and I am going out. (Participant from Focus Group 2)
6.4.3 Summary of Phase 1

Thresholds, in particular gardens and being near the house, were important places for social interaction to many participants. Some mentioned enjoying gardening specifically and others preferred activities such as sitting relaxing or reading. The nature of these interactions was relatively unplanned and spontaneous, although as shown with Bridie, not wholly unexpected. Participants described positive social interactions within gardens, as well as talking on the road in front of their house. These interactions varied seasonally and more likely during the summer when people were outdoors more often. The extent of interactions taking place in thresholds was strongly linked to perceptions of neighbours within the locality. In some instances, participants commented on the changing nature of the populations living in their areas, and how this influenced their perceptions of safety, as well as the nature of interactions that happen. Several participants raised concerns that these interactions have been lost over time, some did not mention them at all, and several participants reported concerns about neighbours or strangers within these places. As a result, participants were sometimes spatially excluded from interactions within these places.

6.5 Phase 2: Moving Through Public Spaces

By cross-referencing both places or importance and activities of importance with social interaction, I identified the key places and forms of movement or travel that resulted in social interaction with others.⁶ The key places that my participants described as being important for social interaction whilst on the move included streets and roads (5 references by 5 participants), village centres (1 reference by 1 participant), and local parks (4 references by 3 participants). Shops (5 references by 4 participants) were also frequently mentioned as a place or destination that they were travelling to or from when these types of interactions occurred. Shops have been identified as important for social interaction within existing literature (see Gardner, 2011; Lager et al., 2015; Mahmood et al., 2012). The form of movement, travel and activities most associated with social interaction was walking, including

⁶ Note: this is not the number of times each activity of place is mentioned, but specifically how they were mentioned in relation to social interaction.

with mobility assistance devices such as walkers as well as walking dogs (9 references by 6 participants), using or waiting for public transport (5 references by 3 participants), giving lifts (1 reference by 1 participant) and cycling (1 reference by 1 participant).⁷ These findings coincide with existing research that has shown these modes of travel are important for social interaction (see 3.2.1 in Chapter 3), as well as literature on the role of dogs as "catalysts" for social interaction (McNicholas and Collis, 2000, p.61).

During interviews, participants often talked about their neighbours and the types of interactions they may have with them. However, during the go-along interview I witnessed this first-hand. Go-along or mobile interviews were therefore particularly illuminating for this phase, allowing observation of when interactions with others occurred in person. Interactions typically happened with neighbours or members from the community who varied in terms of acquaintance, as well as strangers in some instances. If an interaction was with someone known, a more in-depth conversation might be had, or they might ask after someone or their family. The type of interactions that occurred was strongly linked to how well the person was known and how well they knew other people within their community. Common interactions described by participants included greeting someone, including saying hello (6 references), giving someone a smile (1 reference), talking about the weather, or commenting on the day itself (2 references). If one person was driving and the other person walking, beeping a horn might be a way of greeting them. This was mentioned by Louise from Study Area 1. Unusually amongst my participants from Study Area 1, Louise had lived there all her life. Throughout her interview she described how she enjoyed "walking the roads" and that people were always "beeping the horn" at her. She explained with great amusement that this was just what people originally from Study Area 1 do, stating "this is the habit".

There were several instances throughout go-along interviews where participants talked to neighbours or acquaintances. During Nuala's go along interview, we walked for approximately 15 minutes to her public transport stop, accompanied by a family member. During this time she greeted three individuals, one was a neighbour

⁷ For a reminder of participant mobility patterns, see Table 4.1

who lived on the same road as her, one was a passer-by that she didn't know but just said hello to and another was her next door neighbour after arriving back at her house. She reflected during her interview:

If I am coming up the road, I was struggling probably yesterday evening coming up the road and one of my neighbours from up there, "I see you are doing okay, you are still able to get up the hill". (Nuala)

In another go-along interview, Brenda chatted to a neighbour by the gate of her assisted living complex on the way out and with people outside the community centre. In Darragh's interview, we walked to his local church and he described conversations he had previously had with some of the men who were homeless and who slept there. On the way back to the interview start point, a car horn beeped and Darragh saw that his daughter had just parked up, which led to introductions.

6.5.1 Guaranteed Social Interaction

As demonstrated in Chapter 5, connecting to others whilst out and about was highly valued and for some, this meant talking to others whilst moving through public spaces:

If I go out you would meet them going through that passageway, I would meet them in the park. It is every day and I think that is the secret of where you are living if it is feasible to talk to people. (Shauna)

Michelle, in particular, saw it as an important component of how she defined a good quality of life:

To be able to get out and meet with people and mix with other people and say hello to everybody, even people you don't know. And sometimes I would say hello to people, and they'd say, now I wonder who that was [laughs]. But it is lovely. I was speaking with a man the other day, never met him in my life, and we had a great chat, an elderly gentleman. And yeah, it is good to meet people and to talk to people and to smile at people. I would say I am very contented. (Michelle)

Four participants (two in each of the Study Areas) commented that one of the best things about where they lived was the guarantee of these types of interaction and the fact that people in their community would never "pass them by" (7 references). Bridie talked about this during her interview and that it was one of the reasons why she liked her neighbourhood and felt comfortable in it:

There is *always* [emphasis added] someone you'd meet around at the shops or in the queue and you get into conversation with, I always talk to someone everywhere I go [laughs]. (Bridie)

Louise mentioned this when discussing her rambling and walking through the neighbourhood and how people beeped their horns at her. She explains how "they never pass you" because they were "reared" in Study Area 1. There were many instances of participants stating that there would "always" be someone to talk to. Shauna from Study Area 2 was particularly vocal about this. She reflected that in her neighbourhood it is a nice community to go out and walk in and that one of the best things about her neighbourhood, was that she would "always get a hello".

That is the beauty of [Neighbourhood in Study Area 2], *no one passes each other* [emphasis added], you *always* [emphasis added] get a hello. That is the beauty of going out in the daytime. (Shauna)

...it is a nice community if you go out. Just say you are taking a walk out and you meet someone, you may not know them well, but one always says hello, that is the beauty. (Shauna)

You always meet somebody. You see you always meet someone for a chat, sometimes you have a chat, sometimes you don't have a chat. (Shauna)

She admitted that sometimes she would chat for longer and sometimes she wouldn't, but there was always a guarantee of a hello as a minimum exchange, and she took much comfort from this. Mairead also expressed this guarantee of "always" meeting someone, which meant that she never felt lonely:

And like here you will always go out and you will always see someone, you will always have a chat with someone, you are never lonely... if [it's] someone that you haven't seen for a while... you might be just going to say "Hello, how are you?", and go off, and then you'd stop... and then, you know, [you would think] I enjoyed that chat. You are never that busy that you can't stop. (Mairead)

Those participants that felt this way about their neighbourhoods, did so because past experience had created norms of behaviour whilst out and about. This was linked to length of time spent in their neighbourhoods, resulting in familiarity and comfort.

6.5.2 Having a "High Profile"

As discussed in Chapter 4, most participants in Study Area 2 had been born in their neighbourhoods and their families had lived there for several generations, whilst in Study Area 1 most people had moved to their neighbourhoods during the 1970s when the houses were built. Most participants had therefore spent a large proportion of their lives within their neighbourhoods, and many were very well known as a result. Margaret revealed in her joint interview with Nessa that her grandchildren would no longer go with her to the village centre within her neighbourhood (including a local shopping parade) because she was always talking to people:

Margaret: Let's just say my grandchildren stopped coming to the village with me at an early age.

Nessa: You see Margaret was born and reared in the village and she knows everyone, and everyone knows Margaret.

Margaret: And my mother, God rest her, when she died, at the funeral my father said, "I know now why the dinner was never on the table". Because in those days the dinner was on the table in the middle of the day, the poor man got such a shock there were that many people at the funeral. But Nessa is the same, excuse me, it is not just me [laughs].

During Margaret and Nessa's go-along interview, which was the only joint go-along interview conducted, I got a sense of how they would usually chat about the news on their usual walk home together, but also how they would talk to others along the way. Many times during the go-along interview, they exchanged news about people within the neighbourhood. There were three interactions that I observed. I have mentioned one already when they chatted to people sitting out in their garden, but in another example, we observed a lady cleaning the street with a broom across the road. Margaret commented that she was "probably just doing it for a chat if she sees someone she knows passing"; Margaret then shouted across the road to greet her and laughed, saying:

There's another lady who, who'd chat all day! [shouts to lady across the road sweeping the road] I'm looking at you! [laughs]. (Margaret)

We also met another participant (Áine), who Margaret and Nessa knew from one of the exercise classes they attended. During the go-along they had a brief interaction about when it was due to start back after the summer break. Margaret and Nessa explained during their go-along, that if they were in a hurry, they had to "close [their] eyes". For some participants, like Margaret, having this reputation and being so well known was an important part of their identity. This was demonstrated by Bridie, when she described that her daughter and grandson commented that she knew everyone, and it took her a long time to go to the shops:

Every time I go down to the shop, I meet ... as my daughter used to tell me I am the only one they know that it takes three minutes to walk to the shop and an hour and a half to walk back. So I have a great many friends. As my grandson said to me when he was only five, he said, "Nan, do you know everyone in [Study Area 1]?" [Laughs]. I don't, but I do know a lot of people. (Bridie)

Being well known and knowing others made interactions in Phase 2 more likely and contributed to participants feeling "comfortable" in their surroundings and neighbourhood. This was articulated by Bridie at the beginning of this chapter when she said that she doesn't feel like an "outsider" in her neighbourhood. This demonstrates her place attachment and how this is developed through the social "insideness" that she feels (Rowles, 1983, p.299).

David was another participant with a "high profile" in his neighbourhood. He was the participant in my study who was most visible in Phase 2. Although David had stopped training competitively as a triathlete two years previously, he still carried out a series of daily routines and training regimes, which involved cycling around his local neighbourhood with his dog. He explained these to me during his interview and we enacted some of this during his go-along interview, which was a 3-hour cycle with his dog:

David: I have a high profile in that sense, I meet lots of people every day.

Hannah (Interviewer): You are very visible on the street because you are out so much?

David: Yes I am visible and not just here, in [adjacent neighbourhood] you meet lots of people you know because it is a village.

One of the reasons for David's high profile was because he trained many of the younger people in his neighbourhood to compete in triathlons. During his go-along interview, we met one of these children (now an adult). I was introduced to them, and we chatted for several minutes, the topic mostly focused on bikes. In his interview, he described notable people that he enjoyed talking to, whose brief exchanges "make his day":

I meet an old man, [name of acquaintance 1], I love to meet this man, he is 93 years of age, and we only meet at the supermarket now and then and he gives me a handshake that would crush the average hand. And I grip him and say, "[name of acquaintance 1] how are you?" And we only know one another by our first names. And when I meet that man he makes my day, if I have five minutes. And sometimes if I didn't see him for a month I am saying, did he pass on? He is a wonderful old man. And this is the other thing about it, there was a woman, [name of acquaintance 2], she was a nurse, I used to meet her every day walking her little dog [name of acquaintance 2's dog], she was 95. She had steel cobalt eyes and a lovely personality, and just to meet her for five minutes every day, this is what used to make my day. (David)

Not everyone valued these types of interactions. Michael admitted that he would avoid interactions in these places because he didn't feel comfortable conducting "small talk". However, he would still say hello to people:

Hannah (Interviewer): And if you were walking along a street would you stop and chat with strangers or ...?

Michael: I would try not to. If I felt that there was something interesting I wanted to talk about or if I knew somebody had a problem ... I would certainly do it, but I generally just keep going and just say hello to people. There is a ring road here and it is a known place for people stopping and talking and I just keep going. It sounds very anti-social, but I am not great for small talk.

It is therefore important to remember that not all participants valued or felt comfortable interacting with others, recognising the diverse social needs of older people (Bruggencate et al., 2017).

6.5.3 Non-Human Interactions

Another form of interaction observed during go-along interviews and which described during interviews, was interaction with non-humans, in particular with dogs. June from Study Area 2, described how she regularly walked her dog and interacted with more people because of this. She described how she would sometimes know the name of the dogs she meets regularly but not the owners:

You meet more people with dogs, and you get to know the name of the dogs and you don't know the name of the people. You say, "ah here is Toby, here is Rocky". And you don't know the owner's name and you can't say, "what is your name?" It wouldn't be nice. But you know the dogs' names. [Laughs]. (June)

David also described how if he went to the shop without his dog, she would be missed and people would ask where she was:

Oh the dog, [name of dog] went everywhere with me. My dogs, if I walked out there now to the shops someone would say, where is the dog? (David)

During another go-along interview with Dolores (who we meet more fully in the next chapter), we walked for approximately 20 minutes along a local shopping parade. During this time Dolores was greeted by several passers-by, including a neighbour that she had a good relationship with. After she said hello to them, she then told me a story about this neighbour and their family. At another point during the go-along interview, Dolores lit up when she noticed a dog. She greeted the dog and the owner warmly, and they had a brief conversation about the dog. She speculated what breed the dog was and planned to look it up on her tablet when she got home. Dolores kept dogs all her life but was no longer able to look after one due to her health challenges. Existing literature has shown that dogs can be "catalysts" for social interaction and that wellbeing benefits can stem from this (McNicholas and Collis, 2000). In this instance, Dolores demonstrated how the wellbeing benefit of this interaction lasted beyond the event and reinforces how interactions such as these

may be particularly important for older adult wellbeing, especially when no longer able to keep pets of their own.

6.5.4 Interacting with Strangers

Not all of the interactions that I observed during go-along interviews were with people who were already known. I also experienced people greeting strangers during go-along interviews. During Michelle's go-along interview, she said hello to six people on different occasions on an hour long walk around her neighbourhood. The first time this happened, I asked her if she knew the person she was saying hello to and she admitted that she didn't know them. The third time this happened, she turned to me and said "And don't ask me if I know him because I don't" and laughed. There were also some notable instances when participants did not say hello to people. During Margaret and Nessa's go-along some teenagers walked past but they did not say hello to them. This highlights a need to consider the intergenerational nature of social interaction and encounters (also argued by Yarker, 2021), particularly when research elsewhere has shown "discontinuities" with younger populations (Lager et al., 2016, p.1574).

During the main interviews participants were asked how comfortable they were talking to strangers in their neighbourhood. Some claimed they would talk to everyone. Emer describes that she "would talk to anyone", while Louise reveals "I talk to everybody". Edith explained, "Well, I'd say hello to most people, whether I knew them or I didn't know them [both laugh] I would say good morning". This was interesting because both Emer and Edith provided examples where they adapted their behaviour or felt intimidated by others whilst out walking. Emer described how she only walked through certain parts of a local park where she could be seen and avoided areas in the centre of the park where there were a lot of bushes. She mentioned that a man had been killed in the park recently. Meanwhile, Edith described how there were groups of men who drove on motorbikes around her neighbourhood and local field and if she walked on a footpath they might come up behind her, which she described as "quite intimidating".

Other participants were openly far more guarded or cautious about interacting with strangers and explained that there were certain types of people and certain places that

provoked anxiety about talking to strangers. This was connected to experiences and perceptions of neighbours, as discussed in the previous phase. For example, Noelle from Study Area 1 explained that whether she would talk to strangers "depends" on the situation. She provided an example where she would be happy to talk to strangers; this was women of a similar age to her on public transport. Noelle told me about an incident where her bag had been taken and how after that she became more fearful walking to or from places, particularly at night:

One day I was going to work, and my bag was snatched, and ever after that I gone ... didn't like walking ... I just, got more fear into me. (Noelle)

Meanwhile, Nuala described how she talked to certain people while using public transport and that she perceived as safe to talk to:

If I see somebody maybe with children, I might pass some remark. Or in the bus maybe if you are sitting beside somebody, top of the weather, we are always talking about the weather anyway. (Nuala)

Anne appeared to be quite conflicted about interactions with others when she was out. As explained in the previous section, she differentiated between people she knew in her neighbourhood, which were typically the people she moved in with and "new people" that had moved in more recently and she felt less comfortable around. For those people she had known a long time, they would often ask after her mother who was aged 92. She provided an example of having conversations with people in the main shopping centre of Study Area 1. She described how she likes to give everyone a chance and that generally she was quite friendly, until she saw something that gave her cause for concern:

Anne: I'm quite friendly with people, like, 'til you see maybe something and then... [Pauses]

Hannah (Interviewer): Puts your guard up?

Anne: Yeah, yeah. You like to give everyone a chance.

She described how she would be cautious around certain people when she was walking and that groups of people in particular would make her feel uncomfortable and that she would be "on the alert": I think the way, that in general like, the way this, you have to be very cautious when you're walking out, when you see groups coming towards you, you know. I think it's the same everywhere, not only here, you know? You just have to be on the alert (Anne).

During Margaret and Nessa's interview I asked them whether they were comfortable talking to strangers in their neighbourhood, but their response was far more positive, although they did comment that if someone looked "shifty" they wouldn't talk to them:

Margaret: We do, all the time [laughs] and if we see somebody who looks a bit bothered, we would go up and say, "Are you okay?"

Nessa: If they are looking at maps and they look lost.

Margaret: Yes we do talk to strangers.

Hannah (Interviewer): There would be no one that you wouldn't feel comfortable talking to?

Margaret: No. Well, if they looked a bit shifty, we wouldn't talk to them, we are not that daft.

However, they did not clarify who might fit into this category. As was the case for Phase 1, some participants commented on changes to the social fabric of their communities, which resulted in them knowing fewer people than they did in the past. This had implications for how comfortable they felt interacting with others in Phase 2, as well as Phase 1. Some participants felt there were more strangers than there used to be, with new people moving to the area. This was the case in both Study Area 1 and 2. June from Study Area 2 reflected that she used to know everybody, but now she doesn't. She explained that she still tried to say hello to everyone and commented that most people will say hello if she said hello to them:

Now there would be a lot of people you don't know because of all the high-rise blocks and all. I mean years ago I used to know everybody, but you wouldn't know everybody because there is a lot of strangers. But you know, most of them will say hello if you say hello to them. You try to say hello to people and make them feel welcome if you can. (June) Michelle commented during her interview about how people have come and gone and moved to the area. Her experience was that some of the people who had moved to the area didn't want to converse and that they lived very private lives. She found this sad because she really valued those interactions when she was out and about:

I know lots of people around because I have lived here all my life ... There is a lot of people I find that are moving into the area and they really don't want to know you, they don't want to converse with people. I would never push myself on anybody, I would say hello to everybody, but they live very private lives, and they just don't want to ... It is not the same at all, it is very sad really. (Michelle)

For some participants, fear of others in certain places was so great that they avoided certain areas. One of the clearest examples of this was Moira, who we meet in more detail in the next chapter, and I talk about her experience then.

6.5.5 Variation in Walking Routines

Participants varied in terms of how frequently they moved through public spaces. Participants that both valued the interactions and movement within it and were able to engage comfortably and safely, spent a large proportion of their day in this Phase. This resulted in them being much more visible to begin with, making interactions more likely. David and Áine strongly valued movement in and of itself and had established walking and cycling routines and habits, to the extent that it was very automatic to them. Both David and Áine had developed these patterns over a lifetime and both had been connected with groups that helped them to develop these habits. For David, this was his triathlon group. Although he was no longer part of a group, he still carried out training by himself and was disciplined in his routine. As a result, he was out "all times during the day":

And as soon as I get myself together the dog and myself are out. (David)

First of all, I start every morning by working out the important things that needs to be done and I do them early morning and then the rest of the day is down to reading and that. I will be out all over the place, but I am also a home bird. (David)

That is the routine, I take him [David's dog] out five times a day. (David)

Áine, on the other hand, developed a love of being outdoors through a walking group she joined when she was 16. Aged 83, she still regularly walked and hiked and had a routine every morning where she did four laps of her local park. She walked into the city centre to do her shopping on what she described as a "regular" basis. She described how important walking with others was for keeping her going on hikes:

As you are walking you are talking to friends and you keep going and that is great. (Áine)

Louise was also someone that highly valued walking or "rambling" as she described it. She described this as a way of life and how she was always out rambling and talking to people. Louise was recruited from a social group for people with dementia and so it was not clear to what extent she still achieved these valued activities and interactions, but it was nonetheless something that was very important to her.

Bridie, Anne, Michelle, Jennifer, and Eamon were participants that made efforts to walk and referred to specific walking routines. For Bridie, this involved going to the shops after Mass when she didn't have a community group to attend. For Anne, this was walking to the centre of Study Area 1 to do her weekly shop. For Jennifer, who had COPD and used an oxygen tank, it was a short daily walk to her local shop to collect her telebingo and for Eamon it was the same routine but through a local park to collect his newspaper (discussed more in Chapter 7). Margaret and Nessa made a conscious effort to "walk everywhere if [they] can". They estimated that they would walk every day for an hour, with breaks in-between. I was particularly struck by their sense of adventure and curiosity about discovering new places. They described to me a trip they made when the new LUAS line opened in Dublin:

One day we went on the LUAS to [last stop on the LUAS line] to see what was there and there was nothing there, so we just turned around and came back again. (Margaret)

Like Mairead described in Chapter 5, some participants discussed walking in reference to it being an activity that they "should" do more of. For example, James describes how occasionally he might "deliberately go out for a walk" but recognised that he "probably should do it a bit more often". When I ask Emer how often she walked in her neighbourhood, she replied:

Not enough. I am supposed to have half an hour every day. I am having physiotherapy for my hips, but I haven't been. I didn't get the time lately to do it, but I have to get it. (Emer)

Michael was another participant that mentioned he didn't walk enough:

If it is once a week, I would be lucky. It should be more. (Michael)

He described that when he looked after his grandchild he would go for a walk with her in the pram, as it gave him an "excuse to go walking":

I probably don't do as much walking as I should do. One of the things about minding a grandchild is I brought her out in the pram, that gives me an excuse to go walking. (Michael)

In Michael's interview, he explained that he had a lot of pain in his knees that prevented him from walking more and he found walking up hill particularly challenging. Some participants described how they no longer walked as much as they used to or identified health or mobility challenges that made walking difficult and was now seen more as a past tense activity:

I don't go walking any way much anymore really. (Nuala)

I like to walk but I haven't been walking as much in latter times, especially at night, because I can't see. (Michelle)

I used to love walking. (Dolores)

I used to go around the park, there is a friend of mine, she would be at Mass and the two of us would go around the park. I haven't done that in a while, the wintertime I stop doing it. I went around yesterday or the day before, no it was yesterday I went around, my legs were aching, but I need the exercise. (Mairead)

Yet many participants also recognised how important it was for them to do it and were aware that it would benefit their health and mobility if they did:

Well not an awful lot, once this week, but I need to walk, I need to keep myself because I feel if I don't walk, I feel stiffness setting in. (Mairead) Whilst for many participants being in Phase 2 was valued, both for social interaction and in and of itself, the amount it was valued, varied. There were large differences in the amount of time spent moving and travelling, which in turn influenced how much interaction with others occurred as a result.

6.5.6 Summary of Phase 2

In this section I have shown how participant experiences in Phase 2 were very mixed. Some participants felt very strongly about the importance of interacting with others on the streets and roads and talked about it a lot, others only mentioned it briefly. Many of my participants didn't mention it all. Some actively disliked it, or avoided it, or were fearful of these types of interactions. For some, walking and moving was highly valued and so were the interactions that occurred along the way. For others, both movement and interactions were far less likely and in some instances feared and risky. The amount that people moved through their local environments was influenced by how able participants were to move and travel in the first instance, how easy it was for them to get out and whether they had developed routines doing this. Another factor was how safe participants felt moving and travelling through their local environments and the availability of places to walk. This was linked to their mobility and risks associated with the movement, but also the risks associated with who they met along the way. This resulted in large variations in how participants engaged with Phase 2 in and of itself, as well as how likely positive interactions were.

As with Phase 1, if connecting to others is particularly valued by individuals, relying on social engagement whilst moving or travelling through public spaces may also not be enough, particularly if the ability to do this is reduced and there is fear about using these spaces. Participants in these situations would need additional opportunities to engage with others in safe places that are accessible to them and which also create the "excuse" to get out and engage in the first instance. For this reason, destinations are particularly important meeting places and I now turn to these.

175

6.6 Phase 3: At the Destination

As mentioned when I defined Phase 3 in the previous section, the places most commonly mentioned by participants for social interaction were those places where social groups were attended, i.e. 'Social Group Places'. This category of places was the most commonly referred to by my participants and for this reason I focus on this category predominantly within this Phase. The second most frequently referenced place or destination for social interaction was Churches, typically through attending Mass and so I also explore this place within this phase, although not in as much detail.

6.6.1 Attending Social Groups

Hannah (Interviewer): Do you think there are good opportunities to socialise in [name of FG2 location]?

Female Participant: If you want them. But it is *useful* [emphasis added] being part of a group of some sort.

In an earlier round of categorisation of places, I identified a total of twenty-seven different *Community Centres*, nine of which were *Main* in that participants were recruited from these places, while eighteen were *Other* places (see Appendix 17). There were a number of different types of place or venues that these social groups occurred, some of which did not take place in community centres. As shown in Figure 6.3, the most common venues were local community centres, or some form of community centre. Social groups also took place pubs, art centres, shopping centres, schools, gyms, leisure centres, global offices, and churches. In some instances, social groups were based in outdoor locations related to the activities that were being carried out. This included a fishing club and a walking and hiking group. To reflect this, in Figure 6.1, I have provided a summary of *Social Group Places*, which has slightly different totals to Appendix 17 and includes those places where social groups were carried out, rather than the particular category of place of community centre.

Figure 6.3 Venues for Social Groups



Some of these places were venues for just one group, while others held multiple groups. Participants connected to others through these social groups alongside the activities they carried out whilst attending. Participants attended a variety of different social groups with different activities and purposes. In Table 6.1, I provide an overview of the social groups and activities that participants carried out at each of the places identified, including the type of place, the social groups and main activities carried out at each group, as well as the number of participants that were recruited from these social groups. In addition, I identify the number of social groups happening at each place, where they were located (either within Study Area 1, Study Area 2, our outside both of the Study Areas), as well as the number of references within the data and the number of participants attending each group.

No. of Activities or Study No. of Total No. of No. of Participants Type of Place Social Groups & Main Activities Code Social Groups References Attending SG Area References Identified FRSG-SA1-01 Community Centre Active Retirement Group *(n=3); Bingo 2 SA1 24 3 FRSG-SA1-02 Community Centre Women's Crochet Group *(n=2); Retired Men's Group with various activities throughout the week *(n=2); Computer Classes; Art Classe 4+ SA1 57 4 135 FRSG-SA1-03 Local GAA Club Active Retirement Group with various activities throughout the week *(n=5) 1+ SA1 37 5 FRSG-SA1-04 Resource Centre Public Speaking Group *(n=1) SA1 13 1 1 FRSG-SA1-05 Day Care Centre for PwD Dementia Social Group with various activities throughout the week *(n=2) SA1 4 1+ 2 FRSG-SA1-06 Contact Centre Church Meetings 1 SA1 3 1 FRSG-SA1-07 Community Centre **Computer Classes** SA1 2 1 6 FRSG-SA1-08 Shopping Centre Knitting Group SA1 1 5 1 FRSG-SA1-09 Pub Bingo 1 SA1 8 2 FRSG-SA1-10 Resource Centre Friendship Club 1 SA1 2 1 SA1 55 FRSG-SA1-11 Pub Bingo 1 4 2 FRSG-SA1-12 Arts Centre COPD Prevention Class; Writing Group; Choir SA1 3 10 2 FRSG-SA1-13 Leisure Centre Attending Exercise Classes 1 SA1 6 2 FRSG-SA1-14 Church Musical Group SA1 7 1 1 FRSG-SA1-15 School Venue for Men's Group from FRSG-SA1-02: Bowls, Boulé & 'Go For Life' Games 1 SA1 3 1 FRSG-SA1-16 Sports & Social Club Active Retirement Group 1 SA1 1 1 FRSG-SA2-01 Community Hall Active Retirement Group *(n=5) 1 SA2 34 5 SA2 16 FRSG-SA2-02 Community Centre Bingo *(n=3); Film Club; Scrabble; Tai Chi 4 3 77 FRSG-SA2-03 Senior Citizen's Centre Older Adult Social Group with various activities throughout the week *(n=8) SA2 11 1+ 8 FRSG-SA2-04 Resource Centre Older Adult Social Groups with various activities throughout the week; [Recruited due to volunteering *(n=1)] 1+ SA2 16 5 FRSG-SA2-05 Social Centre Bingo 1 SA2 1 1 Knitting Group FRSG-SA2-06 Health Centre SA2 1 3 1 FRSG-SA2-07 Leisure Centre 19 Badminton Class; Swimming SA2 2 2 1 FRSG-SA2-08 Global Company Offices Older Adult Special Events including Zumba 1 SA2 11 7 FRSG-SA2-09 Pub Book Club SA2 2 1 1 FRSG-OSA-01 Various Locations Hiking & Walking Group OSA 3 1 1 FRSG-OSA-02 Harbour **Fishing Club** 1 OSA 1 2 FRSG-OSA-03 Gym Exercise Classes OSA 2 1 1 FRSG-OSA-04 Church Friendship Club 1 OSA 1 20 1 FRSG-OSA-05 Gym Weightlifting OSA 5 1 1 FRSG-OSA-06 Cinema Film Club 1 OSA 2 1 FRSG-OSA-07 Pitch & Putt Club Venue for Men's Group Activity from FRSG-SA1-02: Pitch & Putt OSA 1 6 1 Total = 306 Colour Coding: Colour Coding: Key: * = social group where participants were recruited from 0-9 References 0-4 Participants + = instances where attending one social group may result in a number of activities throughout the week 10-24 References 5-9 Participants 10-14 Participants 25-49 References

Table 6.1 Social Group Details

Some of these social groups focused on one particular activity, while others had multiple activities within them. The frequency of attendance and how regularly this happened also varied. Most groups occurred once a week, some had multiple opportunities to attend activities throughout the week and some were more sporadic in nature. Participants attended on average two social group places (median). All but one participant (David) attended at least one social group on a regular basis, although as previously mentioned, he was a former member of a triathlon group. Ten participants attended one social group place, thirteen participants attended two social group places. Seven participants attended three social group places and four participants (including Bridie), attended four different social group places.⁸

The Meeting Place

Social groups served an important role in bringing people together, creating additional opportunities to meet and mix with other people. Through thematic analysis exploring what participants said about their social groups and why they were important, "meeting people" was mentioned a total of nine times. There were two instances of participants mentioning how important it was to "mix" with others and how social groups provided this opportunity.

Whilst the social group provided the opportunity and place for socialising generally, participants commonly described the typical forms of interaction that took place. Some of these were similar to those interactions identified in Phase 2, including saying hello to someone (6 references), asking after one another (2 references). Additional interactions identified in social groups were associated with more meaningful interactions, including being given a hug (1 reference), sharing worries or concerns (2 references), or offering someone a listening ear (1 references). Laughter or having a laugh was commonly mentioned (5 references), as well as sitting and "having a chat" (14 references), usually with a cup of tea, which was a very common feature across all the social groups I attended. The emphasis on joy and pleasure is a reminder that social groups related to physical activity need to

⁸ Note: Because some places held multiple social groups and some social groups had multiple activities during the week (see Table 6.1), this does not specify engagement throughout the week. I raise this here because it is important to recognise that attending a social group can result in differing levels or opportunities for engagement or attendance throughout the week and varying access to several types of activities.

ensure that these are not neglected (see Phoenix and Orr, 2014). Participants spoke about finding shared or common experiences with people within the groups, particularly if the group was about a particular topic or area of interest. Social groups were important places for exchanging the local news:

It is a great place for us because some of us age better than others and they can't get very far but going into the [social group] it is good because you get all the information that is happening in the area. (Áine)

They were also places for reminiscing and talking about "years ago", as mentioned by two participants from the group interview from an older adult social group:

Unidentified Participant 1: And we might talk about years ago, which we always end up doing.

Unidentified Participant 2: We always end up doing that yeah, years ago.

Staying Connected

Social groups provided opportunities to re-connect with people who might have been acquaintances from the past, but for whom participants have since lost touch with. An example of this was provided by Noelle who explained that she would already have known the women that attend her social group from when her children were at school:

I would know most of the women that comes there as well from the schools years ago. (Noelle)

When Noelle described her local area, she stated that one of the good things about where she lived was that she knew a lot of people. She credited this with the fact that she has always been "involved" in the local GAA club where she now attended her social group. Noelle referred to certain places where she would be "seen" and where she might bump into people and how her social group was an important place where this happened, alongside Mass, or in the local shop:

You see, I wouldn't be really, in the environment five or ten years ago, because I was working at that stage. And, looking back now, er... working, you don't have the same... you don't have involvement with people outside your job. I mean, people that I knew from going to the schools, when I'm retired, and

started going to the shops again, I met people and said, "I thought you moved years ago!", because I was thirteen years [working]... And people kind of say "I thought you moved years ago", 'cause the only other place I would be seen really, would be either in this club, or at Mass. (Noelle)

Niamh mentioned that was very important to her to be "involved with different things" that were happening in her local area. Darragh exclaimed during his interview that it was "amazing when you're involved with people, how you get to know other people". Shauna was someone that highly valued talking to others whilst travelling through her neighbourhood, but also discussed in her interview that because of where she lived and the usual routes she took, she had "lost touch" with some of the people from the centre of the "village". She found that her social group offered the chance to "keep in touch":

I find it very, I won't say relaxing, it is a good chance of keeping in with the community. I wouldn't know many living down this end, they are [name of Shauna's neighbourhood], so if I am leaving my home at the moment, I head up to [adjacent neighbourhood]. So I go this way across the park, rather than the village, so I lost touch with the folks in the village. So the [name of social group] gives you that chance to keep in touch and it is a lovely gathering. (Shauna)

Shauna described how because of certain forms of travel patterns, she was no longer walking through the village, which meant she wasn't seen and therefore didn't run into people within Phase 2 and stay connected to people in the same way.

The people that were met at these social groups varied by depth of acquaintance; social groups were therefore meeting places to connect with new people, as well as meet people who are already known. For one participant in Focus Group 3, they specified that they liked to meet "new" people through social groups, while two other participants, including Anne who specified that it was a place to meet their existing friends:

Ah yeah, it's great like, being in the club, and all your friends that I have up here, like, they're in that club, and I go on holidays with them, before [we] even went to the club. (Anne)

For those participants who were meeting with friends, such as Anne above, these friends might have even encouraged them to attend in the first place. This was the case for Michelle, who was finding that she didn't "go out that much":

My friend was trying to encourage me to go because I didn't go out that much and she was trying to get me mixing with other people, [name of friend], and eventually I went down, and I absolutely love it. I love the exercise, it is good for you and especially with my condition [Type 1 Diabetes] it is really good. They are lovely people, all the women, you get to know them, and it is great. (Michelle)

One participant during Focus Group 2 described how it was important to have at least one or two "contacts", who would then encourage them to either go to social groups or meet in other ways, such as going for a coffee or dancing:

You need to have at least one or two contacts and they would encourage you to go. We go dancing on a Saturday night and sometimes during the week as well and it is just a social thing, you just drop in. (Participant from Focus Group 2)

"School gates" appeared to be an important place where connections were made in the past and in some instances these had sustained over time and developed into friendships. This was the case for Margaret and Nessa, who were neighbours and attended multiple social groups together throughout the week:

Margaret: And then we met at the school gates.

Nessa: Margaret came back [both Margaret and Nessa had spent several years living in the UK before moving back to Ireland] around the same time. We didn't know each other, my son was in the same class so that is how they got to know each other.

Margaret: I think that is how most women make friends, at the school gates.

Having someone to go with to social groups or other outings was one way to make attendance more likely, because it provided additional encouragement, accountability and support to attend in the first instance.

Deepening Connections

Whilst some participants may or may not have known each other to begin with, by attending the same social group frequently, over time these interactions resulted in "getting to know" people. This was the highest referenced code across all themes related to why social groups were important (10 references and 7 participants from main study). Nuala, who was aged 89 and attended an social group weekly, explained why she enjoyed attending her group. She admitted during the interview that whilst she did not engage with activities as much as she used to, she still attended the social group that she was recruited from with her niece. She explained that she enjoyed it because she has attended for a long time and she had got to know the people there. Michelle, when speaking about her social group, mentioned how great it was that she has got to know the women that attended. A participant in Focus Group 4 described that as a result of attending one of her social groups on a regular basis, she knew approximately fifteen additional people "by their first name". Another participant from the same Focus Group described that in social groups there were "regulars" that you then got to know.

Getting to know people created a "bond". Bríd and Anita, who attended the social group for isolated older people and took part in the group interview, described how people who initially attended might feel nervous, but they gave it a try. After a while they felt at home:

You make a bond when you come in [to the centre]. People come in here, they might be a bit nervous, they come along and try it and before you know where you are they are at home. (Bríd)

It takes people who are lonely out of their own homes, and it helps them to communicate with everybody else. (Anita)

Sometimes a bond made at a social group led to new friendships, as was the case for Mairead:

If there was something bothering you [name of Michelle, who also attends the centre] is a great friend, I would ring up and if you were missing, she would ring up, where are you, are you all right? You have good friends around. (Mairead)

These bonds helped participants remain connected to others through the social group, but also throughout the week once they had left the social group, demonstrating a "ripple effect" of third places (as described by Fong et al., 2020, p.1). This was because people would look out for each other and if they did not attend one week, they would be missed and someone would check that they were OK. This is consistent with findings from Alidoust et al. (2019) and the nature of social relationships and ties found in third places amongst older people. One participant from the group interview described how she knew there were people around from the social group if needed, which comforted to her:

You mightn't see anyone until next Monday but on Monday I will come here and see them and then I might be lucky, and I might see them two or three weeks, but I know they are around. (Unidentified Participant from Group Interview)

Jack described that because of his involvement volunteering at a social group, he met a lot of people at the centre and as a result he was not "stuck for anything":

I am meeting a lot of people here... I'm not stuck for anything. (Jack)

Owing to the fact that participants were recruited through social groups, most attended them. Michael was unusual amongst the participants in that he only attended social groups that were a mix of ages and expressed a dislike and avoidance of groups that were age-segregated. Michael felt that quality of life was about "what suits the individual" and he was someone who highly valued "self-education". He felt that a lot of people his age had decided to "retire mentally as well as physically". He had received several requests within his community centre to join clubs specifically for his own age but had resisted, stating:

I don't want to be with people my age, no disrespect, because there is a mental thing that is going on, that it is kind of like people waiting to pass away, if you know what I mean. Rather than challenging themselves, improving their skills, learning. Like, a lot of people, if you talk to them about learning something new, they will tell you, "I am too old for that". People just want to sit around and have a chat and maybe do as little as possible. There is nothing wrong with that, but I think there is kind of an acceptance when you get to a certain age that you fall into a certain trap and people accept it. (Michael)

Michael felt that there was increasing opportunities now for people to do new things and challenge themselves "right to the end". He saw value in "having a go" and trying something new, even if it was difficult and felt that older adult specific groups did not offer him adequate opportunities to do this. He felt that they were places that could be quite negative, and instead he wanted to surround himself with what he perceived as more positive experiences. He reflected that Men's Sheds were a better social group than many, because they "build and create things", however, because he had spent his whole life doing this, he was not interested in doing that in retirement. Occasionally, when he attended workshops, they included people of just his age group, but the focus of the group was usually on writing or other creative tasks, rather than it being an old age space. He saw this as an important distinction.

Michael's reflections raise some interesting points about the role of social groups and the nature of activities within them. Yarker (2021) has argued for the need to consider the intergenerational dimensions of social infrastructure and it appears that more critical interrogations of age-segregated groups are needed. Michael demonstrates that age-specific social groups are not always perceived positively by older people themselves, and do not see themselves as fitting in or belonging in these spaces. This is due to the nature of the interactions, as well as the activities being carried out. This finding aligns with existing research that has shown that not all older people value social groups (Gallagher, 2012) and that there is a need to provide a wider range of creative activities to appeal to wider audiences, including more creative programmes such as those identified by Phinney et al. (2014) in Chapter 3.

6.6.2 Attending Mass

In addition to social groups, attending Mass at church was identified as an important place for connecting to others by some participants. Churches were mentioned or referenced a total of ninety-six times and within this, fourteen distinct religious buildings were coded as places of importance. There were thirty-seven references to attending Mass as a specific activity by fifteen participants.

How frequently did participants attend Mass?

Of the participants that took part in full one to one in-depth interviews and mapping exercises, a total of ten participants revealed that they attended Mass at least weekly and twenty attended Mass in some form. Returning to Bridie, as shown at the beginning of the chapter, one of her main activities that she carried out throughout her week was attending Mass at a local church first thing in the morning. In addition to Bridie, Edith, Mairead, Dolores, Áine and Nessa also attended Mass most days or daily. Noelle, James, June, and Margaret attended Mass weekly, typically on weekends. Another nine participants (Niamh, Méabh, Darragh, Emer, Seamus, Shauna, Brenda, Bríd and Anita) mentioned that they attended Mass, but did not specify how frequently they attended. Jennifer mentioned that she used to attend Mass more regularly, but due to health constraints she only attended more important events, such as funerals.

Darragh, Emer and Bríd were actively involved with the church and the Mass proceedings. Darragh regularly did readings during Mass, while Emer was a sacristan at a hospice,⁹ and Bríd was a Eucharistic Minister.¹⁰ Attending Mass was not the only activity that participants carried out in churches or religious buildings. Four participants mentioned additional activities that they also carried out, including Bridie, who attended two social groups associated with or located in two different churches. James volunteered with a charity organisation based in a church building that delivered food to those in need, and Emer volunteered at a hospice. Darragh spoke of attending retreats that happened at his local church. Shauna and Darragh mentioned that they also liked to visit church when Mass was not on, to sit and contemplate. This gave them tranquillity and inner peace:

I go to Mass wherever I am or sometimes I pop in if I need tranquillity when there is nothing being said. (Shauna)

⁹ Someone who prepares a church for prayer and worship.

¹⁰ Someone who assists the priest in offering communion (consecrated bread and wine) to Mass attendees.

I would simply take a walk up to [local church], sit there for a few minutes, think of many of the people that have passed on, say a few prayers ... It gives me inner peace as well. It's a great sense of peace. (Darragh)

Nessa explained that it was a space for her to contemplate and be grateful for what she had:

And I am very thankful for that [that she has good enough health to be independence], thank God. One of the reasons I go to Mass every morning is just to thank God that I am as independent as I am. But your friends and your family are in the background, and I think that gives you a bit of a boost as well, a bit more confidence. (Nessa)

Twelve participants did not mention churches or Mass as either a place or activity that was important to them. That did not necessarily mean that they did not attend, but it wasn't important enough to mention compared to other activities. Half of these participants took part in the group interview, where there were fewer opportunities to identify places than in the individual interviews. David and Michael were the only participants who specified that they were non-religious. Michael confirmed that he did not attend church, and how he struggled with religion as a concept. He spoke about how he recognised the important role the church has had in the past, being at the "heart of the community", but expressed that this was declining and raised concerns that there hasn't been anything to replace it. James stated that although he attended Mass weekly, it did not necessarily mean that he agreed with everything it stood for. Jack mentioned during his interview that he had both witnessed and experienced clerical abuse as a child. He did not specify whether he attended Mass or church, although it was heavily implied, when he stated how he "would never go to see the Pope or anything like that". Figure 6.4 provides a summary graph of the key trends of attending Mass across all participants in the Main Study and how this varied.¹¹ It shows the frequency that participants attended Mass, how many times a

¹¹ The number of references are slightly different in Appendix 16 (89 references) compared Figure 6.4 (95 references) because there was one instance where 'Church' was generically coded (7 references) but could not be attributed to a particular place. I do not include this in Appendix 17 but do in Figure 6.4. In Figure 6.4 I also removed one reference to a church where a participant confirmed that this was not their church during a go-along interview.

church or Mass was referenced during the interview and whether this was discussed positively or negatively.



Figure 6.4 Summary of Participants' Engagement with Mass and Church

For those that attended Mass, why was it important?

For those that attended Mass daily, it was important for several reasons, and again this is best demonstrated using Bridie as an example. During the interview, Bridie described that religion itself was very important to her and she enjoyed attending Mass, but that there was more to Mass than just the event and ceremony itself. It was one of the key places where she developed her friendships and social connections within her neighbourhood:

My religion is very important to me, the church isn't too far away as you know, and I like going to mass... I just love to go to mass and there is a great community there, you will always get chatting to somebody, make friends from around there. (Bridie)

Later in the interview, Bridie referred to someone in her neighbourhood that was finding it difficult to leave her house. Bridie recommended to her that she should attend Mass, not because she wanted to "preach religion" to her, but because it was the place where she had made all her friends over time:

Now I don't preach religion, because I like my religion, but I don't preach it, but I used to say to her [someone she knew that wasn't getting out of the house and felt lonely], "All you have to do is go around to Mass some Sunday and say hello to this person and the next week another little word and another little word". That is how I made all the friends. (Bridie)

In addition to Mass being the place where social interaction takes place, it was a reason for Bridie to be up in the morning and to get out of the house, and an important component of her morning routine. Edith and Mairead also commented on the importance of this routine and how it motivated them to leave the house:

Get my breakfast, go to Mass, come home, and go back out to anything that's going on. (Edith)

Usually, my day would be I go over to Mass, I get up at eight and potter around having my breakfast, tidy around, have my shower and I might be ready at ten, I am slower than I was, and if I wasn't going out, I'd be slower again. So it kind of motivates you. And I come back. I used to go around the park, there is a friend of mine, she would be at Mass and the two of us would go around the park. (Mairead)

6.6.3 Additional or Extended Outings and Interactions (Phase 3+)

Attending social groups and Mass were useful for attending an activity and socialising at the destination, but another way that they were useful was through providing additional opportunities to engage with others. This took place either before, immediately after through "trip chaining" (Criado-Perez, 2019, p.30), or at a later date. Bridie provided an example of this earlier on in the chapter, where she attended Mass first thing in the morning (Primary Destination) and then carried out movement (Phase 2) throughout her neighbourhood, where she might then go to the shops (2nd Destination) and to the post office (3rd Destination). Mairead provided an example where she went for a walk around the park with a friend of hers after attending Mass. What was interesting about this form of trip chaining was that the first destination visited was the one that was scheduled and frequently attended, so it was already an established habit. This was then followed by a walking route that was

not as formally scheduled but was a routine carried out with some degree of regularity, although for Mairead, she did refer to this in the past tense. I return to the concept of trip chaining in more detail in Chapter 7, focusing on Dolores. However, these findings suggests that trip chaining is particularly important for older people whose mobility is more restricted and who may be unable to get out multiple times per day. This adaptation strategy has also been found by Gardner (2014).

Noelle described during her interview that as a result of attending her knitting group, she became friendly with one of the other women. She then arranged to meet her for a coffee before the activity. She explained that one of the benefits of attending social groups was that she can then "get out another day":

I ... got very friendly with a woman from [name of another town], and I meet her beforehand... we go into a coffee shop for a cup of tea beforehand... actually can get out another day by going to these things [social groups] as well, because you could make friends and you could go for a cup of coffee. (Noelle)

A common component of several of the social groups, particularly those that were older-adult focused and organised through *Active Retirement Ireland*, was that being a member resulted in occasional special outings, day trips or trips away that were organised through the social group itself. Seventeen participants spoke about attending either day trips or trips away through their social groups (61 references and twenty-one distinct places). Day trips or outings included city-centre attractions, cinema, theatres or day trips to particular places, typically coastal locations. Jack carried out a number of fishing trips through his social group to a variety of different locations. In Study Area 2, four participants mentioned that they had been invited to the offices of a global company located within their neighbourhood and that this had been organised through their community groups. This office group then provided special outings and activities for them.

Trips away mainly involved long weekends to other parts of Ireland. Again, coastal locations and large cities were popular. Áine mentioned travelling abroad with her hiking group. Noelle mentioned during her interview that outings were so popular in her social group that there was now a waiting list to join the group. This underlines that there can be a "ripple effect" of both social interaction and additional outings

and challenges the narrative that older adults mobility declines with age. In some respects, then older adult mobility might grow and develop in unexpected ways that extend far beyond the immediate local environment.

Darragh and James both attend the same men's group in Study Area 1. What was unusual about this group was the broad range of activities provided throughout the week. Darragh and James identified multiple additional places that they attended through the men's group, depending on their interests, in addition to their weekly meeting and activities. For James, this involved playing pitch and putt, whilst Darragh preferred photography and music. Through the men's group Darragh mentioned attending photography-related outings, as well as attending a choir and music group. He mentioned visiting one of the members who had moved into a nursing home to perform music with this music group. Both mentioned the trips away they were about to go on, as well as the weekly Games Nights held at the local school.

Some participants mentioned that attending destinations such as social groups also led to increased interactions when moving through their neighbourhoods (Phase 2). James explained how through retirement he had got to know more people. There were over 40 people in his men's group that he knew and most of them lived a short distance to him. He explained that because of this, when he went out, there was a strong chance he would meet one of the members of his men's group, or even members from another of the groups within the community centre, such as the women's group. He also knew people in his community through being involved with associations and school boards over the years:

And that could be quite often and even more so in recent years as I am retired and you get to know more people, I know forty-seven, forty-odd people of our own ... most of them would be within half a mile of here you know what I mean? When you go out, you're likely to meet one of them or you're likely to meet one of the women, we don't have a women's ...we just have a men's club but there is a women's group beside us there in the same place, I know a lot of them, and I have known a lot of them for years. I was involved ten years ago in the association; there are still some of those people around. Also, the school board for years ago I was on that for the ... there is still some of them around. If I was out, I would be out and about a bit, I would regularly s peak to someone. (James)

As a result, he spoke regularly to people when he was "out and about" in Phase 2, but this was in large part attributed to attending social groups.

6.6.4 Getting Out and About with Purpose

Participants tended to attend destinations in a much more planned way than with Phase 1. As shown in the previous section, travelling through Phase 2 was more likely by having something an excuse to do so, for example by having a destination in mind to attend. Having a purpose or a reason to get out was one way to ensure that getting out happened. One of the reasons why social groups and Mass (for those that attended regularly) was so important, was that having something scheduled created a reason to get up and get out on a given day. Some participants were able to create these schedules or routines related to getting out and about without the need for external accountability, for example David and Áine. Yet, these participants were more the exception than the rule. Many participants such as Eamon benefitted from external accountability, and I discuss his routines in more detail in Chapter 7. Noelle spoke about the importance of one of her social groups, which was her craft social group that she attended on a Tuesday, because it helped her get into the routine of getting up and then getting out:

It's just to get into the routine, you just get up in the morning and then out. (Noelle)

Having something scheduled provided participants with a *reason* or the excuse to get out and leave the house. This created a bigger pull and motivation to get out and about than leaving the house for an activity that was unscheduled and could be carried out at any time, for example going to the shops. This makes sense, because if someone didn't have to be somewhere, then they may just stay at home. Some participants found it much easier than others to push themselves to create these types of habits. For example Mairead admitted during her interview that sometimes she needed a bit of "a little bit of a push to do things" and used the example of attending her local social group to illustrate this. She described herself as the type of person that would try something once but if she did not like it she would not continue participating. However, she also recognised that there had been occasions at her social group when she had tried something new and enjoyed it.

I need to have a little bit of a push to do things but if I don't do them, it doesn't bother me too much. In saying that when I have done it, I say "I enjoyed that". I wouldn't do anything I wouldn't enjoy, I would try something once anyway and then that is not for me. (Mairead)

Having a scheduled and regular group to attend made it much easier to leave the house and engage in activities beyond the home. The act of scheduling something and repeating the activity made it more likely to happen and more likely that the person got out. This was best articulated by one of the participants from Focus Group 4. During the focus group, she revealed that on the day that she had her social group to attend, she got out "with a difference":

Female Participant: And time out from the house because I could be in the house all day and I would be very happy there in the garden and all that, but Monday is a day when I am going, and I am going out.

Hannah (Interviewer): Do you think it is important to get out of the house?

Female Participant: It is yes, not that I don't get out, but I get out *with a difference* [emphasis added].

The fact that she had something booked in, meant that the way that she thought about getting out changed because she had purpose in what she was doing. As mentioned in the previous chapter, some participants expressed concerns about being "stuck" in the house. Scheduled and regular activities created reassurance to participants that this would not happen (see Clodagh in the previous chapter). As established in the previous chapter, it was important for older people to get out and about, to have time out of the house, so that they could feel better when they came back. Having a reason to do this was very important, but so was the regularity of this reason, because the process of carrying out the same activity over time led to the development of valued habits and routines. Many participants talked about the importance of routine in their lives and some of these routines were in relation to engaging with scheduled social groups. Noelle in the quote above talks about routine and how her knitting group created the routine of getting out and so does Eamon, whose routines I present in

Chapter 7. Brenda talked about the day that she attended her knitting group, which she runs. She described how it "stood out" compared with other days in the week, because it was a routine. Interestingly, she did not think it was her favourite day; this was going to visit her children and grandchildren. Nonetheless, it was still important to her, because it was part of her routine:

Brenda: My knitting day is a routine, it stands out among any other day.

Hannah (Interviewer): As your favourite day or just a routine?

Brenda: It is just a routine. My favourite day is going out to the children, especially this weather.

These findings align with existing research that have emphasised the importance of routines and regular rhythms for older people and this theme continues into Chapter 7 and Chapter 8 (Lager et al., 2016).

6.6.5 Using Destinations to Structure the Week

A common pattern of daily routine that I observed in the vast majority of my participants, and particularly illustrated through Bridie at the beginning of the chapter, was attending multiple social groups and this was a way of adding structure to their weeks. Whilst most participants demonstrated some structure to their weeks, only two participants explicitly reflected and commented on this. The first was Margaret. In her joint interview with Nessa, she described her weekly routine and its similarity to Nessa's and the importance of "structure every day":

There is a structure every day. And Saturday morning we meet up in the village, the six of us, for coffee. And Nessa goes to Mass most mornings, me not so much. Saturday evening, we try and go to six o'clock Mass. Sunday Nessa spends the day with her family, and I spend the day with my family, or not as the case may be. There is something going on on a Sunday, we are occupied. (Nessa)

Meanwhile, when Brenda described her weekly routine, she reflected how planned it was. She said this with a degree of surprise, suggesting it is not something she had actively thought about and created, but had happened more organically over time: So I think my week is really planned, isn't it? I have Wednesday and Thursday morning here and a Friday morning here. But now sometimes I have to go to a doctor on a Friday or the chiropodist or things like that on a Thursday or Friday but that is not regular, it is kind of once a month. (Brenda)

Jack was another participant that was engaged with a number of social groups. He was a member of a fishing club, as well as a gym where he practiced weightlifting. He also volunteered at a local community centre. As a result he had "things to keep him going":

I have things to keep me going... every day I have something. (Jack)

Noelle reflected on the absence that she would feel if she didn't have her social group to attend on Monday, that she "would have nothing". In addition, Darragh described how "occupied" he was because hewas involved in so many activities through his men's group and local community centre where he taught computer classes. The examples above show just how busy and involved participants were, where they not only attend one social group, but had multiple activities and projects they were involved with, keeping them busy to a level that they found satisfying to them. As found by Lager et al. (2016, p.1574), busy was the "preferred rhythm" of most participants (p.1574).

6.6.6 Summary of Phase 3

Within this section, I have shown how "useful" social groups were for participants, by providing the meeting places to (re)connect with people in their communities. Attending Mass in churches was another destination that many participants attended frequently and in a scheduled way, and that was important for connecting with others. However, there was far more variation in frequency of attendance. I have shown how attending a destination such as a social group can lead to additional interactions and engagement with places beyond the meeting itself. Having destinations scheduled helped participants to get out "with a difference" and with more purpose. As a result, many participants attended and scheduled multiple destinations such as social groups and attending Mass. This helped to ensure that social interaction was not left to chance and also provided reassurance that the week would be satisfactorily occupied.

195

6.7 Conclusion

Connecting to others was a key component of why getting out and about was important for most of my participants. This chapter has explored the key places that were important to my participants for social interaction. I grouped these places into three different phases of getting out and about: thresholds; moving through public spaces; and being at destinations, in this instance attending social groups and Mass. Each of these phases provided opportunities for different forms of social interaction and with different people, including neighbours, friends, acquaintances, and strangers. Throughout this chapter I have shown how interactions in Phases 1 and 2 were valued but were not always the most reliable places for interaction and there were marked differences in how or whether participants interacted in these phases. Comfort in these places was strongly linked to past experiences. Personal and environmental factors combined to influence the ease with which someone got out and created variations in perceptions of safety and comfort in these phases. This in turn influenced how physically present participants were, as well as how possible (positive) interactions were. This connects to the themes identified for a good quality of life and to age -as well as you can -in place Chapter 5, emphasising the importance of feeling safe and secure to engage in meaningful activities and interactions.

Some participants lived in neighbourhoods where they could be reassured those exchanges would occur, but in other neighbourhoods there were fears or concerns about who they might meet. Whilst interactions in all these phases were valued, I have shown that social interaction could not always be guaranteed in Phases 1 and 2 for several reasons and that seasonality influenced the amount of time spent in these places. If connecting to others is desired on a more constant basis, relying on interactions in Phase 1 and 2 would therefore not be enough. Almost all participants engaged with others in destinations or meeting places, the two most common were social groups and attending Mass in church. In this chapter I have shown how scheduled and frequent attendance in these places had several advantages for social interaction, both during the event and beyond the event, as well as advantages for getting out and about generally. I have shown also how it was common for participants to attend multiple destinations throughout the week in a planned way

196
and use trip chaining to optimise their time and interactions whilst out and about. I now present my final empirical chapter, which continues these threads and focuses in more detail on four individuals' experiences of getting out and about. This page has been left intentionally blank.

Chapter 7. Navigating ageing - as well as you can - in place with health and mobility challenges

7.1 *Overview of the Chapter*

In Chapter 5, I conceptualised ageing – *as well as you can* – in place, as having 'good enough' health and mobility, to be able to get out and about to participate and engage in meaningful activities or interactions. Within Chapter 6, I showed some of the ways that participants interacted and engaged with others, as well as the places and phases where this happened. I have already shown in the previous two chapters that getting out and about was easier for some participants than others and how this varied across my entire participant sample. For example, in Chapter 5, I demonstrated how the need and ability to get out varied between participants, as well as for the same participant at different times, depending on seasons, weather conditions and times of day. In Chapter 6, I provided insight into how some participants avoided certain forms of getting out and about and the reasons for this.

Within this chapter, I continue this line of enquiry to more fully answer my third research question: What factors influenced the ease with which participants could get out and about to engage in meaningful activities and interactions? I focus upon four participants with the greatest health and mobility challenges: Edith in Section 7.2, Eamon in Section 7.3, Moira in Section 7.4, and Dolores in Section 7.5. I explore their experiences of ageing – *as well as they can* - in place and how they navigate and negotiate getting out and about in ways that are meaningful to them. For each participant, I provide an overview of what was most important to them for a good quality of life and to age well (RQ1), as well as the places, routes, routines and interactions that were most valued (RQ2). I then consider their person-environment fit, exploring the factors that made it easier or more challenging to get out and about in ways that are meaningful to them (RQ3).

Each individual's lifeworld is described in geo-narratives with related annotated maps. Preliminary results relating to Edith, Eamon and Moira from Study Area 1 were presented in an article (Grove, 2021) for *Social Science and Medicine*. This chapter has a fuller discussion of the lifeworlds, particularly relating to RQ3, and it

199

also includes a fourth participant, Dolores. Edith and Eamon were chosen because they exemplified ageing *as well as you can*, in that they are still able to do things that are important to them as they age in place with health and mobility challenges. Moira, on the other hand, was chosen because she was struggling to accomplish her valued functionings as a result of a less supportive environment. Whilst Dolores lived in a more supportive environment (Study Area 2), she was considerably older than Edith, Eamon and Moira, and faced a number of challenges to getting out and about owing to reduced health and mobility. Different health and mobility challenges are presented to offer a breadth of experience and to offer insight into how certain conditions may influence daily experiences.

7.2 *Edith*

I mightn't be able to walk very far but I enjoy my life.

Edith, aged 74 had lived in her neighbourhood for 48 years. Interviewed in her home, she explained that life was "mostly what I can do for other people, not what other people can do for me". The most important things to Edith were her family, neighbours and friends. Reciprocity and helping others was highly valued and her wellbeing was greater when she was able to do things for others. Edith had Parkinson's disease and consequently took the difficult decision to retire early from a job she loved. She was unable to walk far, requiring frequent pauses. She used a walker, deploying it as a seat when she needed to rest. Edith's main hobbies were singing, set-dancing and crafts.

Getting out of the house to meet others each day was very important to Edith, and vital for her quality of life. During her interview she evoked an image of an older person that was something she wished to avoid and which introduced Chapter 5:

I don't want to be the granny sitting in the corner with a shawl around her shoulders and a blanket around her knees, I want to be able to get up and go.

Getting out was a goal that she admitted she didn't always meet due to her health challenges, however, she still tried very hard to engage in activities that were important to her. Edith noted that her version of living might not be considered living to others, however, what she strived for was living to her:

It's very important [to get out and about]. First of all, I don't want to become a vegetable sitting in the corner. Life is for living and for whatever bit I have left I want to live it. It might not be living to other people, but to me it's living. You know, just to get myself up in the morning and get myself showered, washed, whatever I do. Get my breakfast, go to mass, come home, and go back out to anything that's going on.

Edith felt that she had a good quality of life because she could still get out and about in some form:

I'd like not to be sick, and I'd like to be doing the things I used to do when I was 60. But that's not ever going to happen again. But I think I have a good quality of life. Even though I have all these things wrong with me, and I have many things wrong with me, I still enjoy life, and I still think I have a good quality of life... I'm still able to get around... And whether it's a 5 minutes' walk up the road, and 10 minutes to get back [smiles], it's a good day that I can do that.

7.2.1 Navigating personal projects based on existing and changing capacities

Edith had a clear sense of a "good day" versus a "bad day". Going for a short walk at a slow pace constituted a "good day". Several places were integral to her conceptualisation of a good life, and these were identified in her mapping exercise and interview. Figure 7.1 shows that within her immediate local environment, there were five important outdoor "personal projects" (Curl et al., 2016; Little, 1983). Edith's personal projects involved visiting certain places or destinations to undertake activities and carrying out certain routes. One place and activity that was important to Edith, as shown in Figure 7.1, was Mass at her local church, which she tried to attend daily. She walked short distances within her local area, such as walking up the road or to her local park, although she admitted that she found this difficult due to tree roots obstructing the footpaths.

Figure 7.1 Edith's Lifeworld and Map Key



able to do it anymore. But it's nice just to even walk down the end of the road and just look into the field and to see the children enjoying themselves ... "

try to get there, it might take me longer than it used to but sure, life is a challenge... I think that as long as I can do that I'm happy."



The place of *most* importance to Edith at the time of the interview, and which she "lived for", was her community centre in the neighbourhood adjacent to her own. Here she attended a craft social group twice a week. Edith drove locally and usually picked up two of her friends on the way to the centre. Edith had attended the social group for six years. During her engagement with the centre, she explained that she experienced a transformation in her mood and emotions, which she brought home with her, lasting longer than the activity itself. She summarised the positive impact that going to the centre had on her:

The centre is the place you can go to when you're happy and when you're sad. The times I have gone to the centre, and I feel the weight of the world on my shoulders. And the way I say to them is that I bring a basket, an imaginary basket... and it's weighing so heavy to get out there. And when I come back to it at the door, I lift the basket up and it's so light to bring it back in. Then my worries are shared... and I feel better when I'm back in. So it's my way of trying to explain to people the way I feel... while I go out sometimes with a tear in my eye, I come back with a smile on my face.

Edith admitted that when she first went to the centre, she dreaded it. Her daughter suggested she go, and Edith initially said "no" because she felt that she was too young to attend. However, when I spoke to her six years later in the interview, it was her most valued activity and she exclaimed that if she could go five days a week she would. She described how supportive the staff were, how they waited for her when she arrived to help carry her walker out of the car. The centre was one of the key places that Edith connected to others, where she met people, talked, laughed, and drank tea. She was supported so that she could still attend with health and mobility challenges. In addition to the regular meetings, the centre had tours during the summer, where they would hire a bus and take day trips to different locations.

Edith identified some places and projects that were no longer possible for her to attend. This included her workplace, shown in Figure 7.1 in grey. She also identified places beyond her immediate environment that she no longer visited, including pubs, where she used to participate in set dancing and singing. These are not mapped because she did not specify during her interview where these happened. She had to give these activities up because she did not feel comfortable going out at night and stated "those days are over":

I don't go out in the evening much, unless I'm out and I don't come back, but I don't normally go out in the evening. And I certainly wouldn't go out at night on my own.

Time of day was therefore an important factor in whether Edith got out. Seasonal and weather variations were shown in Chapter 5 to be an important component for some participants as to whether they got out, but for Edith this did not appear to be as much of an issue. I interviewed Edith immediately after the unprecedented snowfall and her community centre had closed as a result. There was still snow on the footpaths in her neighbourhood but the roads had been cleared. Earlier on in the morning, prior to her interview, she had been out to Mass. She stated that if the community centre had been open she would have found a way to get there:

Whether it's hot or it's cold I'd still go down to the centre... It's very important for me, for, to get out. I just, could not stay in all the time. I don't want... [pauses] As I said, if I just sat in all the time, I would seize up and, I just... I don't want that.

Edith found it very difficult to sit at home all the time, because she seized up, one of the symptoms of her illness. Due to her fear of seizing up, she had additional motivation to leave the house. As a result, she appeared to be less willing to stay in more in the winter, because she was already limited in other ways. The interaction of a number of factors influenced the extent that weather and in particular, snow was an issue for Edith. One enabling factor was that she drove to the centre rather than walked. The condition of the roads were better than the footpaths as they had been cleared first and some of these roads had been cleared by her neighbours. She also knew that social support was available when she got to the centre.

7.2.2 Summary

Edith, as is common in those with Parkinson's disease had both an increased risk and fear of falling, as well as concerns about seizing up or freezing of gait (Jonasson et al., 2018), leading to reduced time spent walking in her neighbourhood and carrying out more local and familiar driving routines. Yet despite these challenges, Edith was still able to get out and about with her health challenges and had a clear sense of a good life for her. Whilst she would have preferred not to be ill or have reduced

capacities, she appeared to have come to terms with these limitations and had made several adaptations to her routines and projects. Edith had been forced to give up some of her more valued activities, such as work, dancing and singing, while other activities such as going to the local park were now carried out more passively. To compensate for those activities she had lost, she had found new activities through her community centre to engage in. Attending this community centre was very important to Edith's wellbeing and she provided a powerful explanation of her transformation after attending. There is evidence to suggest that individuals with Parkinson's disease face an increased risk of social withdrawal and difficulties participating in activities (Kudlicka et al., 2018; Bramley and Eatough, 2005). However, Edith showed that with the right social supports, it was possible for her to continue attending.

Edith had found a way to conceptualise a good quality of life so that she was still able to attain it and feel good about herself. She felt that she had a good quality of life because she enjoyed life and was still able to get out most of the time to attend valued projects. She had a clear sense of what a good day was for her. Reference to good or bad days is common within literature about participants with chronic health conditions (Charmaz, 1991). This recognises the fluctuations in health that are experienced and the potential disruptions to daily routines that can occur as a result. Whilst her condition sometimes resulted in disruptions to her routine, when she was able to, she showed additional determination to get out. This was also found by Charlton and Barrow (2002), who identified a "fighting spirit" in some of their participants with Parkinson's disease, defined as a strong desire to "maintain as normal a life as possible in spite of the illness" (Charlton and Barrow, 2002, p.476). This was reinforced for Edith by a clear visualisation of someone that she did not want to be and feared she might become if she didn't get out. This determination was so strong, that additional factors such as seasonal or weather variations did not impact her in the same way that other participants described (see Chapter 5). This was highlighted with her assessment that she would have found a way to get to the community centre if it had been open. Edith's lifeworld demonstrated the negotiation process and risk assessments she carried out when making decisions about when to go out, as well as how certain activities were prioritised and held on to over others.

7.3 Eamon

I like to do things myself... You can say, "oh but I've Alzheimer's now I'm afraid I'll get lost", [but] you've got to push yourself a bit, you know? (Goalong Interview)

Eamon lived in the same neighbourhood as Edith and had done so for over 40 years. Eamon had Alzheimer's disease and was recruited through a dementia day-care centre he attended several times a week. During a joint interview with another attendee (and participant) and with his carer, I asked him how he defined a good quality of life. He explained that it was important for him to "keep up with neighbours", to remain independent by doing things for himself, and in turn, to feel like himself. As with Edith, it took very little to draw Eamon into an account of his wider lifeworld.

7.3.1 The importance of routine

On days that Eamon attended the day centre, a minibus arrived outside his house at 9.30am to collect him. During the interview, Eamon described the importance of having an activity or project to be up and ready for. He had a routine where he made his wife breakfast at 9am and was ready waiting outside by 9.30am:

I have to be up every morning, if it's rain or snow, I've got to be up.

Like Edith, variations in seasons or weather conditions did not significantly influence Eamon's routine. This was because he had an activity that he had to be up for, and was also made easier by the mode of transport. He explained that if it was a cold morning or raining and he wanted to stay in bed for a bit longer, he couldn't because his minibus would be there to collect him:

It's great, I can't do that [stay in bed for longer] ... 'cause the bus is coming for me [laughs].

Eamon felt that routines were a very important part of managing his condition and admitted that he had routines for everything, which he believed "protect" him. He had routines about getting dressed, taking tablets, and leaving items in particular places around the home, but also when it came to leaving the house and engaging with the wider neighbourhood. One of his most valued daily routines was walking through his local park to visit the local shop and buy a newspaper. His go-along interview involved driving from the day centre to his local park and walking some of this daily route. Figure 7.2 illustrates Eamon's outdoor personal projects, which included attending the day centre, and walking to the shop every day. It is not known whether he made this route to the shop *every day* or not, but he had the intention and desire to do so. He admitted himself, that although he tried to walk every day, the length of this varied depending on how he felt and the map shows this aspiration, which was met on some days but not others.

Another important activity that Eamon participated in once a month was traveling by public transport to attend an Alzheimer's Society meeting, also shown in Figure 7.2. Here he gave a talk to others with the condition and shared his experiences. He advised other people with Alzheimer's to get out and walk every day and this in turn gave him an extra push to do it himself. He remarked during the interview "why am I telling people to do this if I don't do it myself?". Talking at the centre gave Eamon a sense of purpose and of pride. People listened to him, and he was a role model for others. Eamon explained that an important part of feeling like himself was going for a walk and being independent. He mentioned that occasionally people offered him lifts or taxis to his meetings and offered to go to the shop for him. However, maintaining his independence was very important to him, so he tried to do these activities on his own.

Neighbourly support played a role in helping Eamon to achieve and maintain his independence. His neighbours knew about his diagnosis, and during his go-along interview he told a story of how he went to go for a walk in his local park one day and two of his neighbours interrupted him as he left the house. They asked him where he was going and he said "I'm going for a walk", however his neighbours told him he was wearing the wrong shoes and escorted him back to the house. He was grateful for this, and rather than saying "you've Alzheimer's you shouldn't be going walking", his neighbours said "you're wearing the wrong shoes, you can't go for a walk in them". Eamon felt this was very nice of them and a "lovely way to do it". In Eamon's own words, "the main thing is for people to have patience".



Public Transport

Activity: Travels by bus alone into city centre every month to attend Alzheimer's meeting, where he gives advice to others with condition (additional place of importance beyond scope of map). It is very important to him that he does this journey to prove that he can do things independently in order to feel like himself.



"... I give enough talks to people that have been diagnosed [with Alzheimer's disease] and I tell them... get out and walk every day... and because I always say that... I say, always go for a walk today... Why am I telling people to do this if I don't do it myself..."

"You can say, 'oh but I've Alzheimer's now I'm afraid I'll get lost', [but] you've got to push yourself a bit, you know... "



Activity: Buy a newspaper

7.3.2 Summary

Eamon told very similar stories on both occasions and some of the time references were inconsistent. However, the *meaning* and *value* he placed upon these stories was evident and consistent. Like Edith, he underlined the practicalities and negotiations involved in sustaining a sense of a good life with cognitive health challenges. This included creating a series of routines to manage his Alzheimer's when out and about. These daily routines were important for his sense of self and feeling independent and this has been shown elsewhere as important for people with dementia and in particular, Alzheimer's disease (Olsson et al., 2013; Öhman and Nygård, 2005).

Eamon negotiated with others when they suggested adaptations to his routines to protect him. Whilst this was done out of concern for him because he might get lost, this would have had a negative impact on his wellbeing and sense of self, as he would have had to concede that these routines were no longer possible. Eamon's personal projects gave him purpose, an extra push, and a reason to get up in the morning. Additional provisions such as a minibus to collect him meant that he was able to travel safely beyond his immediate local environment, extending his activity space. As a result, Eamon dwelt in a lifeworld and with functionings that went well beyond the home and, like Edith, stressed the importance of altruism. Alongside this evident interdependence and social support, Eamon was still able to insist that he had significant independence.

7.4 Moira

Anywhere I'm going I really need a lift.

Moira was aged 66 and my youngest participant and had lived in her neighbourhood for 40 years. Moira lived in a different area to Edith and Eamon. Her neighbourhood was built several years later and was located on the periphery of Study Area 1, a suburban new town. Moira had Chronic Obstructive Pulmonary Disease (COPD) and found it difficult to walk long distances as she got out of breath easily. When Moira was asked how she would define a good quality of life she emphasised being able to get up, get out, and do her own shopping and cleaning:

Well, I think once you're able to get out, get up... you know, yourself, and take your own shower... look after your own, you know? Able to do your own cleaning and washing... once you're able to do things like that and do your own shopping.

7.4.1 *Challenges with maintaining independence*

Like Eamon, being independent was a highly valued functioning for Moira; this meant being able to do activities within her home, as well as in her broader local environment. The places Moira identified as important to her during the mapping exercise were predominantly related to carrying out errands and utilitarian projects, such as going shopping, going to the post office to collect her pension, and attending

doctor's appointments. During her mapping exercise, Moira pointed out in the town centre the shopping centre, which she attended once a week to do her grocery shopping and collect her pension, as shown in Figure 7.3. She relied on a lift to get to the centre, usually from a family member. This was reflected in the format of her go-along, which was a driving go-along. It took 30 minutes to walk to the centre or town from Moira's house, and travelling on public transport took 25 minutes, which included 15 minutes of walking. This was too far for Moira to walk. She mentioned a local bus that used to go into her estate, but it was no longer running. She suspected it was because of an assumption that people drove, but she did not learn to drive, and this was something she regretted.

In this more geographically isolated and car-dependent neighbourhood, activities like shopping were a challenge, because it was not easy to get to the town centre where most services were concentrated. Moira identified a small local shop located in the estate, but its supplies were extremely limited. She occasionally used it if she ran out of milk or bread. Moira expressed frustration that some of her friends in the adjacent neighbourhood had more convenient services, including a post office and better local shops. When describing the last week and her routine, she mentioned two days when she did not leave the house.

Two additional places that Moira identified as important were local community centres in an adjacent neighbourhood (see Figure 7.3), where she attended an active ageing group and social group once a week, one of which she was recruited from. In order to participate, Moira obtained a lift from a friend who also attended. During her go-along, I drove us 3.8km in just under 27 minutes, from the community centre, to her doctor's surgery in the town centre and then finally to her house. She stopped off at the shop on the way to pick up some milk and bread. During her mapping exercise, she explained that it looked as though it was a short distance to the community centre from her house on the map, but in reality it was much further. It took approximately twenty minutes to walk to the community centre via the main roads, which was too far for Moira. A much shorter route to the centre was walking through her local park. However, as identified in Chapter 6, she did not feel safe walking through it, because it was very open and she was fearful about whom she might encounter.

Figure 7.3 Moira's Lifeworld

"Well to be honest with you, I didn't think I'd be living here this long, but I am. But how or never, that's neither here nor there, you know?"

<u>Route:</u> Getting a lift to the community centre

Local Shop

<u>Activity:</u> Occasionally walks to shop but there are limited items to buy

"...there's no shops, there is only one little shop up in [name of neighbourhood], everything is based over here [adjacent neighbourhood where community centre is located]. So... you're talking about a good 15 minutes' walk that... years ago you would have done it, but not now."

"...it's only a local shop. That you'd only really buy bread and milk in it, you know, the paper..."

Nearest Bus Stop

<u>Barrier:</u> This is over a 10 minutes' walk through a park she is fearful of and is too far for her to walk to.

"So, anywhere I'm going I really need a lift. Now, you mightn't think it's that bad up the bus stop, but when there's something wrong with you, it is..."



"You don't know who you'd meet... especially in the nighttime, there's no lights, and you'd never go through, I wouldn't."

Doctors

Activity: Attending Appointments

Shopping Centre

<u>Activity:</u> Getting weekly shop and collecting pension. Moira is dependent on lifts as the bus stop is too far from her house to walk to.

To get to the centre from participant's house it is either a 30 minute walk, 25 minutes to get the bus (including 15 minutes of walking), or a 6 minute drive.

"...they used to run a bus that'd go into the estate, and I could get it at the top of the road, and potter down... It went into the [name of shopping centre], but that's gone. And I think they just think "Ah, everybody drives", but I never drove..."

When Moira explained that her fear of walking through the park was related to not knowing who she would meet, she alluded to distrust of some of her neighbours. Whilst Edith and Eamon both demonstrated strong place attachment to their neighbourhood and neighbourly ties, this was not the case for Moira. She knew some of her neighbours that had moved to the area at the same time as she did, but many of these people had since moved out of the area. This was connected to policy initiatives in the 1980s in Ireland, also mentioned in Chapter 6, which financially incentivised people to move out of their social housing and relocate elsewhere. Moira's neighbourhood had been disproportionately impacted by this. Moira expressed sadness about those that had moved and revealed concerns about new people moving in. She expressed regret that she was still living in the area and missed the way that her neighbours and friends used to call into houses but explained that this no longer happens. During her interview, she expressed a desire for more positive interactions with her neighbours than she currently had. Whilst Moira did not have strong connections with her neighbours, she did visit family members regularly and they visited her. Her daughter had been living with her but had recently moved country for work, so she was living alone at the time of her interview. She looked after her grandchildren most days and visited family members on the weekend.

7.4.2 Summary

Moira described challenges achieving her desired personal projects, and her reliance on lifts conflicted with her valued functioning of being independent. Her neighbourhood served to exclude her from valued projects in many ways. She was spatially excluded, living in a geographically isolated and car-dependent neighbourhood. Research elsewhere has shown the disadvantage and challenges to participation that older people, particularly women, face when living in suburban environments and do not drive (Zeitler and Buys, 2014; Giesel and Rahn, 2015; Stjernborg et al., 2015). In addition, to difficulties travelling, Moira had reduced services and destinations to engage with in her immediate environment. Her neighbourly social environment had been impacted negatively by policy initiatives several decades ago, the impact of which she still felt today. Moira's lifeworld demonstrates the importance of recognising structural or macro factors that can influence and change, not just the physical environment, but also the social environment (Buffel et al., 2018; Lager et al., 2013). This displacement of others, impacted negatively on her place attachment, and caused to her to feel that she no longer belonged (Fullilove, 1996). Certain characteristics of her social and physical environment were therefore not supporting her to age as well as she could.

212

7.5 Dolores

Look, I have come to terms with the fact that you can't expect to be 21 when you are nearly 90. So if I walk as far as [name of local church] and back, I am delighted with myself.

My final participant for this chapter is Dolores. Dolores was aged 89 at the time of the interview and was one of the oldest participants in this study. When I asked how she defined a good quality of life, she emphasised being able to move putting "one foot in front of the other", being happy with what she had and staying out of the "obituary column":

To be happy with what you have got, don't be yearning after things that were, and don't be sorry they are gone, be sorry [sic] they lived. I mean there is no point in killing yourself, we all grow old, if you are lucky, you will grow old and you are still able to move. So if you can put one foot in front of the other and you don't put your name in the obituary column, you are okay [laughs].

Dolores lived alone in a semi-detached home in an inner-suburban neighbourhood. When at home, she thoroughly enjoyed reading. She had a tablet that she played solitaire on and emailed her friends. As mentioned in Chapter 6 when I described an interaction she had with a person and her dog during her go-along interview, Dolores was passionate about dogs. Whilst she kept dogs as pets for many years, she was no longer able to manage looking after them and so no longer had her own. In terms of activities outside the home, Dolores loved walking, particularly in green and blue spaces and used to spend many hours walking her dogs in the past. She enjoyed meeting friends or family members for coffee or lunch, as well as visiting art galleries exhibitions and gardens. Dolores described certain routines that she carried out within the home and when she left her home:

Well, every day is different, thank God. I get up ... I would have my breakfast, which consists of porridge and a mug of tea. Then I would get ready to go to Mass, ten o'clock Mass ... it depends whether I would meet friends and have coffee or whether I would come straight home, whether [name of younger sister] was coming over, whether I was meeting friends afterwards. Like last Tuesday I met [name of friend] down in the [name of hotel], I walked down the [name of river]. But I try to get out every day anyway. Dolores was the only participant to have been recruited through a Public Health Nurse. When I asked if she was limited at all by health challenges, she said quite matter of factly:

Of course I am for God's sake. There are days I can sweep the floor but there are days I can't.

Dolores referred to various health challenges throughout the interview, although the exact nature of her health conditions was not clear. One condition that she did reveal was vertigo, resulting in spells of dizziness, although she managed this to an extent with medication. Dolores' mobility was very restricted, and she used a walker. She talked about walking predominantly in the past tense, recognising that it was an activity she was not able to carry out as much as she had previously. She was not able to climb stairs and had a stair lift at her home, as well as a "claw grabber" to pick items up from the floor.

Out of all my participants, Dolores identified the most challenges getting out and about. As a result of these difficulties, there were many instances where the activities she valued were no longer possible. At the time of the interview, she was negotiating and navigating additional challenges that had further restricted her movement as she had been in hospital the previous week. Despite these restrictions and the risks she faced as a result, Dolores demonstrated a great deal of determination to get out and tried to get out every day if she could. Dolores conceptualised a good day as being able to get out to Mass first thing in the morning. She tried to go to Mass six days a week and to return to the quotation at the beginning, if she managed to walk to and from her local church (approximately 800m away and shown in Figure 7.4 below), she was "delighted" with herself. Dolores had a sense of what she felt able to walk based on her capacities at the time of the interview; a distance manageable to her on a "good day" with her walker was 20 minutes away. Dolores walked to Mass using her walker most days but on Sundays she got a lift from a neighbour.





7.5.1 Trip chaining and negotiating finite energy budgets

Whilst she declared that "every day is different", the beginning of each day was quite structured and similar. However, what she did after Mass varied depending on whether she had arranged to meet a friend or family member. A common practice for Dolores therefore, as identified as a pattern across many of my participants in Chapter 6, was to "anchor" herself to an activity that was scheduled and was a consistent routine, to get out in the first instance (Lager et al., 2016, p.1574). She then carried out more flexible and varied arrangements afterwards in the form of trip chaining. This was particularly important for Dolores because of her restricted mobility and energy budget. During her mapping exercise, she described an ambitious outing that she carried out about six weeks ago prior to the interview. This outing involved attending Mass, going to a restaurant for lunch, and then onto a local park much further away:

And one day I went to 12:40 Mass [at local Church], this is about six weeks' ago, I couldn't do it now, and came out, had a bit of lunch in the [name of Restaurant] and then I walked up to [name of Local Park]. That was a big, big deal for me at that time, but I made it. It took about half an hour, and I went in, and I was so chuffed. It was a Friday, I texted [name of friend] and said, "Where are you?" "I am down at [name of post office] getting my pension." "I am up here in the park." "Right, I'll be up". So up she came anyway, and we sat at the pond for a bit.

She described this as a "big, big deal for her", due to the distance, and was really pleased, or "chuffed", with herself when she managed it. Since this walk, Dolores' mobility had become additionally restricted and at the time of the interview, she felt that this routine would be beyond her capabilities; when she explained she "couldn't do it now". For this reason, it is shown in grey in Figure 7.4.

Dolores spoke about another routine that she commonly carried out, also shown in Figure 7.4. This involved walking along the river to a hotel where she then had a cup of coffee. Sometimes she did this on her own and sometimes she met a friend. Both the hotel and riverside walk were located a short distance from her house, and she described carrying out this trip the previous week. This riverside walk was a place and route mentioned several times. She particularly enjoyed seeing the different types of birds, including the "ubiquitous seagull" and occasional heron. The last time she had carried out this route she had been delighted when she saw a cormorant for the first time. However, on the way home she felt dizzy and had not completed the route since.

The local library was another place important to Dolores because she was an avid reader. Dolores had a Kindle but preferred not to read this late at night because it affected her sleep. Recently, she had "run out of reading" at home, however, she explained that she "can't get to the library very well". This was because the library was in a different part of her local neighbourhood, in the opposite direction to her usual routines (see Figure 7.4). Dolores' use of tense when talking about her routines, routines and places of importance were sometimes inconsistent, as if she was unclear herself of what she could currently do and what was no longer possible. For example, in the mapping exercise, she described a route she would take "I go down to [name of library], which is by the [name of building it is next to]." I asked how often she visited the library and she replied, "Whenever I need a book, whenever I feel like it". However, she then immediately reflected:

....it depends on (a) how I feel, (b) what the weather is doing, and we will have to wait and see. I just can't, I have to wait and see how I am feeling.

She wasn't sure if she could still get to the library but was hopeful, she still could:

I could go around to the library. I am not sure whether I can do it now or not, I hope I will. Anyway, the library is just the same length as the church, 20 minutes.

Dolores reasoned that because it was the same distance as the church, she could potentially do this trip, although it was unlikely she could do both unless it was a particularly good day. Based on an understanding of Dolores' regular routines, capacities, and priorities, it became apparent how other directions then became unavailable. This highlights that when an individual has to negotiate smaller "energy budgets" to manage reduced capabilities, some instances this means only being able to move in one direction at a time, emphasising the "fragmented" nature of reduced mobility, as other options that are lower priorities and in different directions become unavailable (see Lord et al., 2011).

In many ways, Dolores' immediate physical and social environment were more supportive than for Edith, Eamon, and Moira. Within her inner-suburban neighbourhood, she lived on a main road, with a local shopping parade a short distance away (this began less than 100m away and ended 250m away from her

217

house). The shopping parade included restaurants, coffee shops, her doctor's surgery, supermarkets, and a post office. One of the main benefits of her immediate local environment was that "everything is on tap". In addition to local shops, there were many places nearby where she could walk, including local parks and her favourite riverside walk (this began approximately 250m from her house). However, Dolores identified several barriers that made getting out and about particularly challenging for her. Some of these barriers meant that she was no longer able to carry out certain activities. For example, prior to a fall in 2008, she had attended a choir group, but was no longer able to participate because she could not climb the stairs in the building. At the time of the interview, Dolores was not attending any social groups. She attended a friendship club associated with her local church, but they were on a break over the summer when she was interviewed. The group were intending to meet for a meal in a few weeks' time and Dolores was hopeful she would be able to attend.

7.5.2 Variability across personal and environmental factors

Dolores admitted that on any given day, she had to consider how she felt, as well as the weather conditions. Like Gardner's (2014) participants, weather was a "constant consideration" (p.1252). Dolores struggled with windy conditions and mentioned how the wind stopped her from going out and how she had been "knocked down by the wind years ago". Dolores was particularly fearful of windy conditions because she did not weigh very much. She worried she would be knocked off her feet. During the mapping exercise, she pointed out a place that she avoided in windy conditions. This was the entrance to her local train station, the design of which created a wind tunnel effect.

It is a bad day if I can't get out to walk. It is usually because, even if I don't feel great, I still go out, but it is the wind. What started that was I couldn't get a taxi, I rang but couldn't get a taxi, I forgot it was Father's Day. So [name of younger sister] said there's always taxis at [close to the train station] ... when I got to [the train station], my God, it was like being in a vacuum, the wind, and I was holding onto a post and a man came along and I said, "Please would you help me down...?" There was only the one taxi, being a Sunday, so anyway I was in bits, really terrified. The rollator, which is a heavy thing, was taking off as well.

Time of day was another important factor and she mentioned that she would "never go out at night". During Dolores' go-along interview, I was able to get a sense of how she engaged with her immediate environment. During the go-along, Dolores and I walked along the main road where her house was located. Dolores was very keen to carry out the go-along interview and we agreed a distance that would be manageable for her on the day. We carried out a loop that involved passing through her local shopping parade, which had several shops, pubs, and restaurants. The interview took just under 20 minutes and was a round trip distance of approximately 540m. This route is shown in Figure 7.5. Dolores pointed out several places where she had fallen in the past. Whilst she admired the trees in her local area and found them "beautiful", she also described them as a "nuisance", due to the way they caused the tarmac to become uneven, highlighting how environmental features can simultaneously serve as assets and hindrances.

Throughout the interview, Dolores referred to a few "incidents", which had impacted negatively on her health and mobility. One of these was the "awful fall" she experienced in 2008, where she broke her hip and cut her knee quite badly, resulting in her no longer being able to attend her choir. She described another fall where she had hurt her legs and they had bruised. Dolores pointed out some of the footpaths that were cracked and uneven (see Figure 7.5). She showed me how difficult it was for her to lift the walker up and over these footpaths. This had become an issue for her recently in particular, because she had lost some of her strength:

Lately either I've lost my strength, or... the thing has got stuck, because now I find I've got no strength to put it down.

Dolores described how she had to watch out for other people using the footpaths and would look up and down her road to see if someone was coming. She was particularly cautious of cyclists and people running. When I asked her how her environment could be more supportive, she replied:

Honest to God, I can't think of anything except the bloody paths.

Figure 7.5 Dolores' Go-Along Route



7.5.3 When challenges become insurmountable: making mistakes and mustering through

Well everywhere was important when I could get there but now I can't.

Dolores' interview provided insight into how she made decisions about whether to engage with certain places beyond her immediate environment. In many instances, environmental variability and uncertainty meant that she no longer attended certain places or used certain modes of travel. As a result of the interaction between her mobility and her physical and social environment, there were several places important to Dolores that she could no longer attend. The main challenge with visiting places beyond her immediate environment was that she had to use public transport because she no longer drove. Dolores had several public transport options available to her in reasonably close to her home, including a bus stop (approximately 600m away) and a train station (650m away). However, past experiences had knocked her confidence and created so much fear, uncertainty, and risk, that she now avoided using them. She described several examples which highlight the challenges she faced, and these are now discussed.

Dolores faced several challenges when getting on a bus. She described the inconvenience of having to get past a lot of people when it was busy, as well as not wanting to inconvenience them by them having to get off to let her on. She described the variability of the bus drivers and buses, how some would be happy to lower the ramp and wait for her to get settled, while others wouldn't. The implications of this variability was the difference between her potentially going to Accident and Emergency or not:

And then when you are getting off... you couldn't go from town to get off at [name of her bus stop] because there would be all people standing and how are you going to get your rollator past all these legs? They just don't want to move and they really all have to get off. So that really is a no-no... Some of them [bus drivers] are very good, they will lower the ramp, some of them won't bother... They don't say I don't want to, but they say I can't or something. The rollator is very heavy, trying to get that up, then trying to put my card up to the thing, then trying to get up to the thing before he starts... Some are very good, they wait until I am sitting down, some of them start straight away and I am holding onto the rollator with one hand and the pole with the other and I am afraid of my life of falling because I am not steady on my feet. So it is not worth [saving] €10 to end up in A&E [sic].

Dolores described a time when she had to travel to hospital for an appointment. On this day, she decided to "chance" the bus, stating the high costs associated with getting taxis all the time. She got onto the bus, however when she was sitting on it, she hit and injured her elbow:

I had to go out to [name of hospital] to get a [name of a type of scan] and I decided I would get the bus out, chance it, because it is very expensive to get taxis everywhere. So anyway, it was okay, I got on, I have a seat on the rollator, and I sat on the rollator but whatever way the bus jarred I hit my elbow and when I got off the bus my blouse was all blood. I am on aspirin. I had plasters, so that was okay.

She managed to make it to her appointment and afterwards a friend came out to meet her. It was not clear whether this meeting with her friend was pre-planned or because of her accident. Her friend had intended to get the train home, but Dolores decided to get a taxi instead because her confidence had been shaken by her experience of using the bus. Furthermore, Dolores struggled to use trains because of the large gap between the train and the platform, as it was too difficult to lift her walker onto the train:

So, we got a taxi home, it was €25, it was worth it because I was terrified. You see the DART [name of Dublin rail network], mind the gap, and the rollator is very heavy and trying to get on and trying to anchor it. And as I said the thing about the DART would be fine except for the gap, the gap is terrible.

Dolores spoke about a "very, very brave" trip she made approximately six weeks ago, when she spontaneously visited a cousin that she had not seen for months. Her cousin lived in a seaside town 50 minutes away and she used the train to visit her. She described the trip to me and how she had created a plan to carry out the least amount of walking for both of them, as well as ensuring that the weather conditions were suitable:

So I decided, this day was a beautiful day, so I rang [name of cousin], are you at home? Yes... So I went up to [name of nearest train station], thank God, well of course I wouldn't go on a windy day. And it leaves me right into [name of seaside town] opposite the [name of restaurant]. So she met me at the [name of restaurant] and we had lunch. We didn't go anywhere else, just back to the station. It was just to meet her.

When she returned home, she made a deliberate strategy to try to get into the last carriage. This was to avoid having to walk the whole length of the platform when she reached her station. When she arrived at the platform, Dolores and a man intended to get off the train, however, he pressed the button, and it didn't work. The man ran up to a carriage where the door was working to get out. He tried to hold the door open for Dolores but the train conductor at the top of the train did not see what was happening and tried to close the doors. The man was able to catch the door and eventually helped her out. She concluded her story stating: "That was the only incident". This story shows how an unanticipated disruption to public transport created an additional challenge for her. In this instance, someone had been there to help her. However as far as I could tell, she had not used the train since.

These "incidents" or bad experiences on public transport left her feeling afraid of using them and resulted in Dolores having no alternative but to use taxis instead if she wanted to travel beyond her immediate environment. Using taxis, Dolores visited some places, but did so far less frequently, owing to the cost. An example of this was going into town, which she described doing "occasionally". In this way, she was able to overcome some of her barriers to an extent. However, she also described how there were certain places that she thought were worth using a taxi and others that were not. The places that she no longer visited beyond her immediate environment were typically green and blue spaces, such as gardens or beaches, even though she highly valued these places. Dolores prioritised the use of certain activities over others and even though engaging with nature was important to her, it was possibly seen as too indulgent to spend money on taxis. She was more likely to use a taxi for what she considered necessary trips. This included meeting a friend, or attending appointments, but she did not "feel" like using them for visiting gardens:

I would love to go to the Botanic Gardens, but I don't feel like paying a taxi there and back. My neighbours are great, but I am not going to ask them to bring me to the Botanic Gardens, they are busy working or whatever.

Dolores described in many instances during her interview how "truly blessed" she felt to have the support of her neighbours, family, and friends, as well as her home help. Like Edith, she described the additional support she received during the snow:

It is a very friendly neighbourhood and people are so kind, can I open the gate for you, can I get you a message? And during the snow there I was ... There was a queue at my door [laughs].

She did not do her own "main shopping", her next-door neighbour did this for her and she had meals delivered to her. She mentioned that sometimes she was not in, so she had an arrangement with the delivery person to leave them in a bag inside her bin. In this way she was dependent on others, who supported her to remain independent enough to stay at home with modifications. However, there was a limit to this and while they were very supportive in many ways, she would not ask them to bring her to gardens. She described how she has come to terms with this and what she has: Where else would I like to go? Howth, Dun Laoghaire, Bray [seaside towns] but I am satisfied now with what I have. I have had to come to terms with what I have, and I don't expect people to be bringing me out.

As explained already, Dolores usually had to see what the weather was doing and how she felt on the day to decide to what extent she would get out and about. However, she didn't always get this assessment right. The previous day she had been into town, however she described it as a "big mistake" because she didn't feel very well. She also talked about times when she regretted going and it was a "big mistake", but she managed to "muster through":

I went into town, a big mistake, I really didn't feel great, and I shouldn't have gone into town, I got a taxi of course, I can't go on the buses anymore, but my God, big mistake. Anyway, I managed to muster through, and I met my sisterin-law.

Despite doing her best to manage risks and make strategic decisions, both her capacity and environment could still be unexpectedly diminished.

7.5.4 Summary

Dolores highly valued getting out and about in her local neighbourhood, and like Edith, conceptualised a good day in terms of being able to get out to walk in it. Dolores engaged with a number of places of importance, including attending Mass at church. She felt a real sense of accomplishment and wellbeing when she engaged with these places. The amount she was able to walk had changed over time and certain valued places were no longer available to her. The implications of this was that she had to limit her activity space to closer to home, or use taxis, in case she felt unwell whilst she was out. This meant that she was not able to experience as much variation within her local environment. This was one of the few instances during my research where I could get a sense of capacities changing even within the past few weeks. She demonstrated how she continually managed her expectations and negotiated what she was capable of, based on her past experience.

Dolores identified a number of challenges and risks that she faced on a daily basis and that had injured her in the past. For example, some of the footpaths created many additional risks for her, and she was reminded of these when she carried out her daily walking routines. Her mobility restricted her movement and there were routes that she would have liked to carry out and places she would like to visit, but felt they were now too far for her. The main reason for this was that public transport had become too challenging and uncertain for her to use. Unlike Eamon and Edith, she was much more impacted by weather conditions and how she felt on the day, highlighting how weather has both the:

potential to comfort, invigorate and connect, but also to disorientate, threaten and isolate, at times supporting moments of wellbeing, at others exacerbating experiences of impairment and disability (Bell et al., 2019, p.270).

There was a clear sense of her activity space reducing over time. However, it was not in a uniform way, as certain activities were prioritised over others, using taxis to occasionally extend beyond her immediate environment. Like Edith, Dolores demonstrated remarkable determination and spirit to get out and about because it was so important to her. However, sometimes this resulted in her pushing herself into situations where she had to be particularly "brave" and confront risks that might easily have been minimised through considerate and age appropriate design. Getting out and about therefore, led to a sense of regret, a lucky escape, or a story to tell. Sometimes it created a dent to her confidence and served as a reminder to make adjustments. And unfortunately, sometimes it led to serious incidents and injuries that forever changed the way she got out.

7.6 Conclusion

The ease with which Edith, Eamon, Moira and Dolores were able to get out and engage in meaningful projects was influenced by the interaction of a number of personal and environmental factors. To obtain a true sense of the factors that influenced getting out and about, it was necessary to consider the interaction between the individual's existing capacity, what was important to them, as well as the extent to which the physical and social environment supported their attainment of personal projects and meaningful activities. As has been shown through these lifeworlds, both personal and environmental characteristics were dynamic, and interacted in complex ways. They were also highly varied; indeed what was a barrier for one person served as a motivation to get out for someone else. This has important

225

implications for how we conceptualise and define age friendly environments and support older people to age well in place, which I will discuss further in the next chapter. Whilst objective characteristics of the physical and social environment are important to consider, we must also take into consideration the way that older people view and navigate their local environment through the lens of their existing capacities and priorities, which has in turn been shaped by the environments they have lived over a lifetime.

Of particular note here, is that for individuals who found it challenging to get out, it was no less valued. As demonstrated by Dolores, even when there were additional challenges, risk or stress that came with getting out, she still sought to get out in some instances and had decided on certain activities that were still important enough to get out, whilst others had been let go. This demonstrates both remarkable determination, resilience and strategic decision making abilities (also shown by Ewart and Luck, 2013; Gardner 2014; Holland et al., 2005). This was because the activities, interactions and places that lay beyond her home gave her wellbeing benefits that outweighed the risks. Whilst engaging with *some* risk is a natural part of everyday life and can even contribute to a "sense of dignity" (Marsh and Kelly, 2018, p.297), I can't help but wonder whether there might have been some relatively simple interventions that would have meant that she did not have to engage with what mattered most. I now consider what approaches and supports might help older people in Ireland to age better in place.

Chapter 8. Discussion and Conclusion

8.1 Introduction

The overall aim of this thesis was to explore older adults' lived experiences of ageing well in place; specifically, how individuals engaged and interacted with their local physical and social environments. The research design was a detailed multistage qualitative and geo-spatial approach, involving the use of focus groups, interviews, mapping exercises and go-along interviews (in-situ methods). In the first empirical chapter (Chapter 5), I examined how participants defined a good quality of life (RQ1), focusing in on how important getting out and about was for this. This led me to conceptualise and define ageing well in place through a new conceptual model of ageing – *as well as you can* – in place, which focused on the importance of being able to get out and about to carry out meaningful or important activities.

Whilst getting out and about was valued, the way that individuals engaged with their local physical and social environments, as well as what was important to them, was extremely diverse. In Chapter 6, I provided an overview of the key places, routes, routines, activities, and interactions of importance to my participants whilst out and about (RQ2). Overall, shops, banks and post offices (aspects of a typical high street), green and blue spaces, as well as the places where social groups were held, were the most highly referenced place categories and social interaction was the most highly referenced activity. Owing to the importance of social interaction whilst out and about, in Chapter 6 I presented my participants' experiences of social interaction (or lack of it) in different phases of getting out and about: when they left their homes, whilst travelling, and when they arrived at destinations to carry out regular and scheduled activities. The activities and destinations I focused on were attending social groups within community centres (or related buildings) and attending Mass at their respective local churches. In Chapter 6, I also explored individual barriers or enablers that influenced their ability to carry out personally meaningful projects; in this chapter the focus was on social interaction in particular (RQ3). Results demonstrated the importance of scheduled and regular activities to provide reasons to leave the house, give a sense of purpose and to provide guaranteed forms of social interaction. This led to deepening relationships and often provided additional reasons

227

to get out through trip chaining, either immediately after the activity or later. Results highlighted the importance of the social environment and the role of social support from neighbours, family members or friends and how this influenced place attachment, belonging and perceptions of safety whilst out and about. They clearly showed that this was not a uniform experience and there were marked differences in the experience of social interaction across participants.

In Chapter 7, instead of focusing on the collective experience of my participants, I focused on four individuals who had identified health and mobility challenges, Edith, Eamon, Moira, and Dolores. I presented a spatially referenced account of their lifeworld as an annotated map and story, identifying what was of most importance to them for a good quality of life (RQ1), the places and activities that were of most importance to them (RQ2) and the extent that they were able to age well in place by engaging meaningfully with their local physical and social environments. I demonstrated how valued functionings and independence were negotiated over time in response to health or mobility challenges (RQ3).

In this final chapter, I situate my findings in relation to existing literature and consider the implications of what I have found. In Section 8.2, I present and further develop the theoretical model I have developed for this thesis: ageing – *as well as you can* – in place. I provide an overview of its four components and situate these components in relation to existing literature. In Section 8.3, I outline two key recommendations that geographers and planners could implement to support older people to age – *as well as they can* – in place. Finally, in Section 8.4 I reflect on some of the challenges and limitations of this research, including a reflection on the Covid-19 pandemic.

8.2 Ageing – as well as you can – in Place

The main conclusion reached through my empirical findings was that participants defined quality of life and ageing well in place subjectively, according to what was of most importance to them and based on their existing person-environment context. From this, I have developed a lay-informed theory of ageing – *as well as you can* – in place. This is a relationship-centred model that integrates the concepts ageing well

and ageing in place through a Capability Approach lens. It has four interconnected components (see Figure 8.1). The first component is a pragmatic and subjective definition of *ageing well* (used interchangeably with quality of life). This is individually defined, with an emphasis on doing the best you can, for as long as you can, based on individual capabilities and circumstances at a given point in time. It continually meets individuals where they are at, meaning it is inclusive, dynamic, and attainable for all. The second component of this model is an emphasis on achieving *valued functionings*, i.e. the "beings" or "doings" of everyday life that are important or meaningful to an individual (Sen, 1993). The third component of this model recognises the importance of participating and engaging beyond the home to feel and be *in place*. Finally, the fourth component builds and connects the ideas of enacting relative independence on the one hand, with the importance of the social environment on the other, to highlight how *interdependence* plays a vital role in shaping the experience of ageing well in place.



Figure 8.1 Ageing – as well as you can – in Place

The "as well as you can" part of this theoretical model was underpinned by existing literature critiquing objective definitions of ageing well (see Section 2.3 in Chapter 2). The particular phrase was presented by one of my participants from Focus Group 1. When I asked them what they thought terms such as healthy or successful ageing meant to them, they stated:

I suppose as we are getting older our quality of life depends on how our health is, you know, it is very important to keep yourself *as well as you can*

[emphasis added] and as positive in your thinking. (Participant from Focus Group 1)

I felt this statement represented what my participants overall were telling me: the importance of having realistic and attainable definitions of ageing well and a good quality of life. This quote also highlighted that in lay definitions, there is considerable overlap between these terms. Of all the participants, Edith was the most influential in developing this theory, when she stated:

It might not be living to other people, but to me it's living. (Edith)

Existing research has argued that objective definitions of ageing well can leave the majority of older people feeling like failures, putting pressure on those with health and mobility challenges to reach a goal that is not possible for them (Stephens et al., 2015; Stephens, 2016; von Faber et al., 2001; Jeste et al., 2010). Edith alludes to these high standards and that she may be falling short in some way, yet all I could see was her strength and courage after receiving a life-changing health diagnosis. I was particularly struck by her attitude, her resilience, her caring nature towards others, her determination to continue carrying out her most valued activities and her appreciation of a "good day". She was ageing as well as she could.

8.2.1 A pragmatic and subjective definition of health and ageing well

When asked to define a good quality of life, many participants identified that "having their health" was very important to them. However, health was very differently defined by participants and often this was in accordance with existing abilities. There was little evidence of expectations similar to the traditional World Health Organisation definition of health (World Health Organization, 1946, p.1) as a "a state of complete physical, mental and social well-being and not merely the absence of disease or infirmity". Instead, the way that participants defined health was pragmatic, rather than perfect. For those living with health and mobility challenges, they wanted to age as well as they could for as long as they could, recognising and accepting that the ageing process was likely to result in some reduced capacities, slowing down, or re-negotiation of expectations. But what did matter to participants, was that their health was good enough to still be able to do what was important to them. The importance of more pragmatic and nuanced definitions of health have been found elsewhere in research examining ageing well and quality of life of older people through a Capability Approach lens (Gilroy, 2006; Stephens, 2016).

These findings align with Huber's (2011) alternative definition of health, which is more appropriate for individuals living with health or mobility challenges and is defined as the "ability to adapt and self manage in the face of social, physical, and emotional challenges" (p.1). Whilst there were some common themes identified, which aligned with existing literature on quality of life (discussed in Chapters 2 and 5), overall, my results demonstrated that lay definitions of a good quality of life are extremely diverse, reflecting the heterogeneity of the older adult population. Furthermore, they do not necessarily align with clinical or validated measures or ideals of quality of life, health or ageing well. This empirical work adds to existing research that has argued for the need to also consider and define ageing well, in particular healthy ageing and quality of life, through a subjective and lay-person lens (Cosco et al., 2013; Bowling and Gabriel, 2007; Bowling and Iliffe, 2006; Fernández-Ballesteros et al., 2008; Richard et al., 2005). Examining perceptions of a good quality of life, it becomes apparent that quality of life is more a "process", than a "fixed state" (Scott et al., 2009, p.134).

8.2.2 A Capability Approach lens: emphasis on achieving valued functionings

Health and mobility were important to older people because they enabled functioning and flourishing. However, for those constrained participants, they still felt it was possible to have a good quality of life, providing there were adequate and individually defined supports in place to achieve their valued functionings. This was illustrated by Edith, Eamon, Moira and Dolores in Chapter 7, who all had clear and realistic ideas about what a good quality of life or a good day meant for them. These findings align with existing research that has examined and defined health and ageing well using a Capability Approach. Such research has defined health, not as the requirement to achieve good physical health, but "the capability to continue to do valued things in later life" (Stephens et al., 2015, p.729), or the "abilities to be and do things that make up a minimally good, flourishing and non-humiliating life" (Venkatapuram, 2011, p.20). Such an approach recognises that "a person's wellbeing or quality of life is located in the opportunities that they have to lead the life they value" (Stephens, 2016, p.4). This definition of health and wellbeing is far more attainable and importantly, can still be achieved with physical decline. This work therefore adds weight to existing research that has argued for the need to consider ageing well, wellbeing and quality of life through a Capability Approach lens. However, the model of ageing -as well as you can - in place adds an important contribution to existing work, which is the consideration of *where* these valued functionings occur.

8.2.3 The importance of being and feeling - in place: getting out and about

The third component of the model recognises that getting out and about and being in place matters to older people. I define this as both physically being in their most valued place(s) (i.e. getting out and about) but also feeling in place, which connects more to intangible feelings or emotions about a neighbourhood, including sense of belonging, place attachment or feeling safe and secure (Rowles, 2018). The research findings and theoretical model align with existing work that has argued for the need to consider the broader than home environment of older people when examining experiences of ageing in place (Wiles et al., 2012). Participants identified and valued a number of places, activities, interactions, routes and routines that they carried out (functionings), which involved leaving their homes. As a result, being able to get out and about in meaningful or important ways, was an important component of ageing *well* in place. Furthermore, the attainment of this was important to feel in place. The importance of place for a good quality of life in old age is by no means a new idea; it is one of the key principles within health geography and geographical gerontology. Gilroy (2008) has argued that older adult "well-being is determined to a significant degree by quality of place" (p.146) and that the quality of an older person's home, neighbourhood, as well as their mobility between the two, is vital for a good old age.

Examining the importance of place for a good quality of life through a Capability Approach lens, all participants identified valued functionings which required getting out and about in some form. Getting out and about was a valued functioning in and of itself for most participants. As a result, getting out and about was either directly or indirectly important to all participants within this research project. Both Gilroy

232
(2006) and Meijering et al. (2019) have identified specific capabilities and functionings important to older people and have emphasised the role of the physical and social environment in supporting older people to achieve these functionings and capabilities. Broadly, there was considerable agreement of the valued functionings identified by (Gilroy, 2006) and (Meijering et al., 2019), and those identified by my participants. For example, the importance of independence, being mobile, engaging with others and the importance of a safe neighbourhood in which to enact independence through mobility were shared findings. Where this work differs from existing work, is that I have spatially grounded valued functionings with the use of geo-spatial approaches and provided annotated maps to illustrate *individual* experiences of this. In particular, I have shown how a valued functioning such as independence, may be defined very differently by an older person, depending on their person and environmental context. This work therefore adds to existing research through a geographical application of the Capability Approach. In particular, I have shown where these beings and doings take place, how they are negotiated by older people throughout their daily lives, as well as the individual barriers or enablers (or resources) that may support people to achieve their particular set of capabilities.

8.2.4 *Connecting with others: Enacting relational and relative interdependence*

This thesis has contributed to the empirical understanding of why getting out and about was important for a quality of life and ageing well. Existing research has stated that it is generally accepted that getting out and about is important to older people (Holland et al., 2005). The empirical findings within this thesis show the nuances within this assumption and provide insight into why this is the case. I argue that the overarching reason that getting out and about was important to my participants is firstly, it was the physical enactment of independence and secondly, because it allowed my participants to connect with others. I have combined these two ideas within this section through the concept of interdependence.

The idea of independence is often identified as a policy priority, particularly within more individualistic societies. As shown by this research, it was valued as a vital component of a good quality of life and ageing well in place by many of my participants. Within a Capability Approach to older adult wellbeing, freedom or independence is regularly cited as the overarching valued functioning (Meijering et al., 2019; Sen, 1993). Within Chapter 5, I revised my theme relating to independence to label it *relative* independence, recognising the different ways it was conceptualised by participants with differing capacities. The idea of relative independence connects to Hillcoat-Nallétamby's (2014) work, who has argued that policymakers need to shift away from an "oversimplified dichotomy between independence and dependence" and instead recognise the more "dynamic" nature of relative independence (Hillcoat-Nallétamby, 2014). Furthermore, Ball et al. (2004) have identified strategies older adults have employed to maximise their independence. An important component of this was redefining independence over time to align with what they could currently attain. This suggests that older people have more pragmatic and flexible definitions of independence, as well as ageing well and health. In particular, what constitutes independence for one individual may be perceived as dependence by someone else (Wiles et al., 2012).

This research has confirmed the importance of independence but has also highlighted the diversity in how it was conceptualised and negotiated by individuals over time. Many of the ways that it was conceptualised by my participants has been found in existing literature. In a study examining the *meaning* of independence for older adults in assisted living, residents emphasised a strong desire to remain autonomous and physically independent (Ball et al., 2004). Furthermore, independence had multiple dimensions and emphasised certain beings and doings, such as: being self-reliant (being able to do some things for yourself), preserving individual identity by carrying out certain tasks, having valued roles and passing time in meaningful ways. Whilst my participants identified *independence* as being important to them, when this was examined further through their actions and everyday activities, what they described more closely aligned with the idea of *interdependence*. This has been found elsewhere in the literature (White and Groves, 1997).

The idea of independence versus dependence align with more objective and biomedical definitions of successful ageing independence is an "ideal state" and dependence by its definition, insinuates a deficit or a failing in some way (Beeber, 2008; Bell and Menec, 2013; Breheny and Stephens, 2009; White and Groves, 1997;

Smith et al., 2007). Yet in practice, complete independence and autonomy, like successful ageing, are unrealistic and unattainable for most people (White and Groves, 1997). Interdependence, on the other hand, recognises that most people are dependent on others in some way, with most people relying on some form of social network or support (Beeber, 2008; White and Groves, 1997). Furthermore, independence in old age usually requires some support from humans, such as neighbours, friends or family members. It may also involve support from nonhumans including pets, with a growing literature recognising the importance of pet ownership on mental health and wellbeing of older people (see McNicholas, 2014; McNicholas and Collis, 2000; Gee and Mueller, 2019; Hui Gan et al., 2020; Young et al., 2020). Furthermore, support may be obtained from assistive technologies and walkers (Goins et al., 2015; Schwanen and Ziegler, 2011), such that for the older people, as for so many others, the living body can be thought of as a "cyborg" or more-than-human, a composite of both living organism and machine technologies (Andrews and Duff, 2019; Joyce and Mamo, 2006). Interdependence recognises the "contribution and assistance of others" and the vital role that this plays in supporting older people to be independent (White and Groves, 1997, p.86). As Wiles (2018, p.36) argues:

for governments to encourage healthy ageing in place requires much more than simplistic rhetoric around independence and autonomy. Such rhetoric is often framed in highly individualistic, middle-class and middle-aged terms and fails to recognise that in fact all of us are interdependent and that concepts like 'autonomy' can mean quite different things in varying social and cultural contexts.

Flexible or "fuzzy" conceptualisations of independence that aligned more with interdependence were commonly demonstrated by my participants (Schwanen et al., 2012, p.1313). For instance, Nuala, at aged 89, wanted "a little bit" of independence and her family provided her with social supports so that she could get out and act in a relatively independent way. Meanwhile, Eamon was happy to attend his dementia day care social group and take the minibus to this, as well as give talks about his experience of Alzheimer's. However, he preferred to attend these via public transport himself. In this way, both Eamon and Nuala were interdependent but still able to express choice and agency in how they spent their time outside of the home;

for example, by attending valued social groups several times a week. They were also able to decide which areas of their lives they would like support and which they wished to maintain independently. In this way, they were still able to assert some autonomy and claim that they were relatively independent. This confirms existing literature that has found being able to have *some* choice and agency, or the "right to mediate and control their decision making" is a vital component of independence whilst ageing in place (Wiles et al., 2012; Ball et al., 2004; White and Groves, 1997 p.87). Ball et al.'s (2004) participants emphasised the importance of having some choice and autonomy and avoiding dependency by accepting some help but importantly, refusing it where they could.

This research has confirmed that for most participants, independence was relational and often negotiated with others. Many participants expressed that they did not not wish to be a *burden* on others. This has also been found elsewhere on literature on ageing and independence (Ball et al., 2004; Bell and Menec, 2013; Hillcoat-Nallétamby, 2014; Schwanen et al., 2012). It was important to many of my participants that they were still able to contribute towards helping others, in addition to receiving supports. For example, Edith was grateful for the social supports at her community centre to assist with getting her walker out of the car, yet she also liked to give to others in other ways by providing friends with a lift and looking out for her neighbours. Both Edith and Eamon demonstrated how important reciprocity and altruism was to them, because it helped them to be involved and connected. Reciprocity was an important part of their identity and helped them to feel good about themselves and their communities. The importance of altruism, reciprocity and helping others has also been identified as important in research on older adults and independence and is a key component of interdependence (Ball et al., 2004; Bell and Menec, 2013; Hillcoat-Nallétamby, 2014). In particular, research has shown that participants are more willing to accept help and not perceive themselves to be dependent, if they are able to reciprocate in some way (White and Groves, 1997).

This research has demonstrated the important connection between an individuals' social environment (including social supports) and the ability to maintain independence through forms of interdependence. These findings connect with a large body of literature that has examined the importance of the social environment for

older people. An important finding was that my participants were able to compensate for more objectively unsupportive environments or declining mobility through social supports. Existing research has found that older people themselves recognise the need for social supports to remain independent and that a lack of social supports can lead to social exclusion (Bell and Menec, 2013). This recognises the importance of the social environment as an important component of ageing well in place, in particular literature that emphasises the *relational* nature of this (Lager et al., 2013; Lager et al., 2015). Owing to the importance of the social environment, in particular social supports and social interaction found within this thesis, I revised the model of ageing – *as well as you can* – in place from what was initially a person-centred approach, to a relationship-centred approach. This is based on Nolan et al.'s (2004) critique of person-centred care, who argue that such approaches can ignore the importance of interactions, relationships and interdependence.

8.2.5 Summary

The research for this thesis has allowed me to conceptualise what ageing – *as well as you can* – in place meant to older people themselves. A key thread throughout this model is to recognise the diversity of the older adult experience and how they define ageing well in place, based on their differing personal-environmental contexts. I have demonstrated the fluidity of lay conceptualisations, in terms of how complex concepts such as health, quality of life, ageing well and independence are understood. Yet what shone through this data was the importance my participants placed on *connection*:

So the bottom line is that we need connection, connection to people, connection to facilities, connection to what we need. (Participant from Focus Group 4)

My participants valued being connected with themselves, they valued relationships and being connected to others, such as friends, family, neighbours, acquaintances, and animals. They valued being connected to something bigger than themselves and having a higher sense of purpose and belonging through their activities. This included connection to place, community, to nature, and in some instances religion. As a result, they demonstrated what Weathers et al. (2016) would define as spirituality:

Spirituality is a way of being in the world in which a person feels a sense of connectedness to self, others, and/or a higher power or nature; a sense of meaning in life; and transcendence beyond self, everyday living, and suffering (p.93).

In the next section, I consider the following question: how can geographers and planners support older people to age -as well as they can - in place?

8.3 Recommendations: how can geographers and planners support older people to age – as well as they can – in place?

Supporting older people to age – *as well as they can* – in place, as opposed to ageing in place, requires a different response because of the emphasis on subjective experiences of what matters most to people. As a result, there is no "one size fits all" approach that will support older people to age *well* in place (Hammond and Saunders, 2021, p.10). As my own background is in planning and geography, within this section I consider what these disciplines can do to support older people to age – *as well as they can* - in place. I have two overarching recommendations, informed by my findings, existing literature on ageing, wellbeing and environment and drawing from my own experience working as a Planning Policy Officer. The first recommendation focuses on how geographers or planners could support individuals to age – *as well as they can* – in place, building on Finlay and Rowles (2021) ideas about *clinical geography*. The second recommendation looks at more of a neighbourhood and structural level and considers how to prioritise and incorporate more wellbeing into the design of age friendly environments.

8.3.1 'Person-in-Environment' Interventions for Ageing Well in Place

Recommendation 1: Support individuals to age – as well as they can – in place through person-in-environment interventions

A key component of supporting individuals to age -as well as they can - in place, is to ensure that they continue to feel and be *in place* by recognising the importance of getting out and about to older people. A promising new sub-discipline within geography, that could be utilised to prioritise this, is the idea of *clinical geography*. This involves applying geographical or spatial knowledge and working directly with individuals, in order to co-produce and apply "person-in-environment" interventions (Finlay and Rowles, 2021, p.2). A key goal of this is to support people to physically be and feel *in-place* (Rowles, 2018), which can in turn enhance health and wellbeing. Clinical geography operates at the scale of the individual within the context of their immediate environment. This aligns well with the principles of ageing -as well as you can – in place because it is inclusive and involves "meeting people where they are at" both physically and psychologically through its person-centred approach (Registered Nurses' Association of Ontario, 2010). However, for the reasons explained earlier, I would extend the ideas of person-centred to more of a "relationship-centred" approach. As a result of its person-centred approach, it is built on a recognition of the subjectivity and heterogeneity of individual experience and the uniqueness of a persons' "history, abilities, needs, and preferences", as well as an awareness that this will change over time (Finlay and Rowles, 2021, p.2). Clinical geography goes beyond and complements traditional person-centred care, which would typically focus on home supports and be delivered by health practitioners such as nurses or occupational therapists. Instead, it broadens the scope, allowing for greater consideration to be given of the neighbourhood elements of ageing in place.

On a practical level, this involves working with individuals to identify their "personin-environment challenges and opportunities" as well as determining their "priorities and preferences" (p.3). Of relevance for this thesis, is their suggestion of developing "strategies to get outside the home" (p.6). I see great potential in clinical geography and think it could be applied to support and empower older people to navigate what ageing – *as well as they can* – in place means to them and to then help them attain it. For example, geographers can help older people identify what matters most to them; where they have resources and may need support; and co-produce strategies that will help them to stay connected and engaged. It can help individuals be proactive about potential challenges that may arise as part of the ageing experience and develop a plan that works for them, receiving support when needed to remain inter-dependent, autonomous, and well within their communities. This will help to ensure that individuals can continue to make decisions and remain in control of their lives, as well as feel comforted and reassured, knowing they are doing the best they can. An important area of future research will be the trialling of clinical geography approaches. This could involve working alongside health and social care practitioners, developing and evaluating the success of "in-environment" interventions to improve older adult wellbeing. However, it will be vital that measures of success reflect what is important to older people themselves.

This section has provided recommendations at the individual level to provide them with increased agency, but alongside this, it is vital that we consider the broader structural context and the unequal nature of the ageing experience. In particular, there is a need to think about the way we plan and design our built environments to ensure that wellbeing is prioritised.

8.3.2 Prioritising Health and Wellbeing in Urban Planning and Design

Recommendation 2: Prioritise mental health and wellbeing in urban planning by focusing on reducing health inequalities and social exclusion.

Whilst my research confirmed that getting out and about was vital to my participants sense of ageing *well* in place, there were marked differences in the extent that their local environments could be considered supportive. When opportunities were available and it was safe to do so, my participants highly valued social interactions with others, engaging with nature and moving through their immediate local environments. However, my research also identified instances where it was not safe for my participants to get out and about and how this produced risk, fear and in some instances, avoidant behaviour. This impacted negatively on my participants' ability to engage with their local environments in meaningful and restorative ways, feel a sense of belonging and place attachment within their communities, and feel independent. This in turn, negatively influenced their wellbeing, quality of life and their overall experiences of ageing in place.

To provide supportive environments for ageing *well* in place, we need to prioritise aspects of the physical and social environment that are restorative and provide more *quality* of experiences for older people. There is a large body of literature that has examined the ways that local environments can influence health and wellbeing. This

includes extensive research and theory within the disciplines of health geography, including the concept of therapeutic landscapes, as well as within healthy urban planning. Returning back to the idea of the *Restorative City* framework developed by Roe and McCay (2021) and introduced in Chapter 3, there are two pillars that have relevance for this recommendation on reducing health inequalities and social exclusion. The neighbourly city recognises the importance of the social environment for quality of life and wellbeing. If we want to provide supportive environments for ageing well in place, planners must value and prioritise the social environment. This includes providing greater opportunities for older people to participate and engage with others, which in turn will foster deeper social connections. This requires a need to pay attention to the subtle and softer details of local environments and is a much "quieter" approach to urban design, which "assigns importance to the smaller signatures - the settings for 'episodic' moments of city life" (Roe and McCay, 2021, p.2). This includes moments such as spontaneous social interactions, which were so important to my participants throughout this research and which have also been identified in existing research (Gardner, 2011; Fingerman, 2009; van Eck and Pijpers, 2017; Finlay and Kobayashi, 2018). The best way to design a neighbourly city is to build a variety of forms of social infrastructure, providing opportunities for people to engage and interact within the various phases of getting out and about.

A key recommendation of my thesis therefore is that we need to invest more in the *quality* of our physical and social environments. As Gilroy (2006) notes:

If our policy goal is to improve quality of life by supporting independence and involvement in the community then meeting a friend for lunch or going to look at the spring flowers needs to be taken as seriously as getting to a hospital appointment (p.351).

Connecting back to the findings from Chapter 6, providing supportive environments for older people to age well in place involves paying attention to features of the built environment that influence whether social interaction happens. This includes the orientation of gardens, the provision of safe, accessible walkable communities where people can pause along the way and with a variety of destinations, activities, and reasons to get out for all ages and interests so that lifelong habits can develop. This also connects to elements of the playable city, another pillar of the *Restorative City*

framework. Supportive environments require community centres and third places to host a range of social groups and provide people with reasons to leave their homes. They require safe transport provision to support older people getting to and from these valued places with reduced risks. They also requires paying attention to the location and grouping of destinations, so that they provide opportunities for trip chaining. This elevates the importance of our high streets and how they are so much more than a place just for shopping (see Fullilove, 2020). Finally, supportive environments require green and blue spaces and public realm that are invested in and are well-maintained, so that people feel safe and comfortable talking to others. This will build social capital and deepen social connections.

The importance of social participation is already accepted within policy, and is one of the eight domains of an Age Friendly Environment (World Health Organization, 2007a). However, planning policy has been slow to respond to the age friendly and healthy environment agendas. This is partially due to the lack of integration of health and planning within policy and practice, despite calls for them to be "re-united" (see Chang and Ross, 2015). Whilst there has been some progress, significant challenges remain. A key one is that there remains a lack of *prioritisation* placed on healthy urban planning within governments, compared with economic growth (Chang and Ross, 2015). This has led to "tokenistic" references within planning policy, rather than significant and much needed change (Carmichael et al., 2016).

As demonstrated throughout this thesis, critiques of the Age Friendly Environment model have highlighted that the movement has not addressed the unequal and diverse experiences of ageing in place (Hammond et al., 2020; Buffel et al., 2018; Finlay and Finn, 2021), with those living in "difficult" or "unsuitable" places at increased risk of social isolation and spatial exclusion (Scharf et al., 2007; Severinsen et al., 2016). A strength of a Capability Approach to ageing well in place, is that it allows for the consideration of the role of inequalities, by exploring how different resources can influence the ability to achieve capabilities (Meijering et al., 2019). Meijering et al. (2019) argue that independence is intrinsically related to an individual being able to exert their agency and convert a resource into a functioning and that this differs considerably between individuals in differing contexts. For example, someone living in an unsupportive environment may be able to overcome the specific barriers of this, by ordering a taxi instead of taking public transport when this becomes too difficult. However, this requires money, which may or may not be available.

Through examination of individual lifeworlds, it was possible to identify the specific challenges that individuals face as they get out and about, and the ways that they negotiate and manage this, drawing on a variety of resources if they have them. This research complements Meijering et al.'s (2019) work and contributes to it by showing the way that these negotiations vary depending on the activity in question. Even when an individual potentially has the resources available, they may not choose to use it for some activities and may prioritise some activities or projects over others. As shown by Dolores, the activities that may get dropped are those that may be perceived as more indulgent (such as visiting a beloved garden) rather than necessary (such as a hospital appointment). I would question why one needs to be chosen over the other, if we value the quality of life of older people then we need to support both. Findings raise the importance of interventions that incorporate elements of "social prescribing", which could potentially provide benefits for health and wellbeing for those individuals that may need additional support with engaging (see Hamilton-West et al., 2020; Woodall et al., 2018). However, the success of such interventions would be influenced by whether there is adequate provision of services available to begin with, whether they match with older adults interests and whether there are adequate supports getting to and from these places, recognising the specific needs of the individual.

Overall, older people living in the most unsupportive environments with the fewest resources will find it most challenging to age well in place. This was demonstrated by Moira; she was the youngest of all my participants, lived in the most unsupportive neighbourhood and had few resources to compensate for this, resulting in her not feeling like she could be independent and was instead reliant on others. Her lack of social supports, her declining mobility, combined with perceived environmental barriers connected to distrust of her neighbours, meant that she struggled to carry out valued functionings and was not ageing as well as she would have liked. The quality of the local environment and the extent that it can be restorative is an underrecognised and under-appreciated *resource* for wellbeing and for ageing well in place. It is important that we recognise this within planning policy so that we can

identify "thin" places and intervene to provide "thick" places where necessary (Duff, 2010).

Recognising the subjectivity of experience of ageing well in place, raises the importance of including older people within the planning process, as they are the experts within their local communities and lives. In order to provide more restorative environments for ageing well in place, first we have to determine what that looks like in a particular local context. This cannot be done without engaging older people. It is therefore vital that we conduct more participatory research and community engagement within planning policy to ensure that age friendly interventions are locally appropriate. There is already excellent research that has led the way on this and provided excellent guidance on co-designing with older people (Handler, 2014; Buffel and Phillipson, 2018; Hammond and Saunders, 2021; Hammond et al., 2020; Doran and Buffel, 2018; Buffel, 2018). To understand the different experiences and needs of diverse population groups, requires alternative research designs (Handler, 2014) and methodologies that are more qualitative, creative, spatial and participatory, such as those included within this thesis. For those groups who are more disengaged and disempowered from the planning process, additional efforts will be required to ensure that their perspectives are heard.

8.4 Reflecting on the Challenges and Limitations of this Research

Within this section I reflect on some of the main challenges and limitations of this research. I focus on go-along interviews as a method, recruitment and sampling limitations, as well as Covid-19 implications.

8.4.1 Carrying out go-along interviews

Go-along interviews provided valuable insight into my participants' experiences. They captured aspects of lived experience and context that would not have been possible with the use of traditional interviews. This was the first time I had used goalong interviews as a research method and so I want to offer here an honest and critical appraisal of the method, which I often feel is missing in journal articles and books. Carrying out go-along interviews with older participants with diverse needs was not easy. It required empathetic and sensitive caring skills to match the participants' pace and needs. Adopting a more care-full approach was more emotionally draining than traditional approaches. I had not anticipated the level of responsibility and concern I would feel towards my participants during the go-along interviews and sometimes this lasted well after the interview, particularly when they experienced challenges.

I tried to balance the potential risk of taking part in go-along interviews with the right to participate and to carry out inclusive research and the right of older people to engage with some risks as part of their daily lives (Marsh and Kelly, 2018). I managed this by making the go-along part optional and asking participants to do what was normal or manageable to them at that moment. As a result, I would not expose them to any additional risks than they might experience as part of their everyday life whilst out and about. However, I had not anticipated how I would react when I carried out go-along interviews where I perceived the risks of their usual behaviour to be greater than they did. Sometimes it was only after the go-along interview that I reflected on the journey and questioned whether they were pushing themselves for my benefit or using me as a 'crutch' themselves for their own benefit. Would they have gone out in the same way had I not been present? And did I make them feel safer than was usually the case? It was not always easy to obtain a sense of what was normal to a particular participant during the interview and to navigate these "ethical tensions" (Marsh and Kelly, 2018, p.297).

Go-along interviews are far more time-consuming than traditional interviews and the interview transcripts are much harder to analyse and transcribe because they are conducted in the 'real world'. In my three-hour cycling interview, I lost a significant proportion of the conversation because we cycled along a very windy path and had to rely on my field notes after the event. Reflecting on the experience of conducting go-along interviews, I feel that more care-full approaches are needed for both researcher and participant whilst carrying out these methods, particularly as they are growing in popularity. As researchers we need to have more honest and critical discussions about the challenges of carrying out these methods. I think that better ethical and safety protocols, training, and support could be developed for researchers to navigate ethical dilemmas whilst in the field, as well as recognise the emotional challenges

that this type of fieldwork can bring (Punch, 2012). The RGS-IGB Geographies of Health and Wellbeing Group ran a 'Hack Day' in July 2018 led by Sarah Bell, Ronan Foley, and Andrew Powers on the topic of In-Situ Methodologies. This was a valuable space for researchers to come together and discuss some of these issues and led to a co-authored paper on the subject (see Foley et al., 2020) and I hope these important conversations will continue.

8.4.2 Sampling and Recruitment

My recruitment process involved trying to obtain maximum variation (Tashakkori and Teddlie, 2010). I took several steps to diversify my older adult population (discussed in Chapter 4), yet some of these were unsuccessful, reflecting the challenges of engaging with harder to reach population groups. As a result, there are noticeable absences within my sample. This includes participants who are homebound. As a result, I cannot comment on how getting out and about was perceived by those who could no longer attain this and how they felt about this. All but two of my participants (David and Dolores) attended a social group of some kind and Dolores was on a break with her group for the summer. The importance of social groups comes through because my participants engaged with these. However, I do not have perspectives from older people who do not use them and why this might be. Most of the participants attended social groups with an older adult focus, apart from (Michael), who stated that he avoided these groups, because he did not like engaging with groups where everyone was older; instead preferring groups based on activities for all ages. This is important to note, because the growth in older adult social groups within Ireland has been a relatively recent development. Whilst my participants enjoyed attending these groups, they will not work for everyone. Some people may actively avoid them, feel excluded from these groups, or might wish to attend but are prevented from doing so for several reasons.

The diversity of my sample was limited, in that all participants were Irish. Future research could therefore seek to engage with population groups who have not been represented within this thesis, to examine what ageing -as well as you can - in place means to them within their specific contextual and the challenges to achieving it. This includes older people who are: homebound, not attending social groups,

members of the travelling community, homeless, LGBTQi, migrants, and living in a variety of assisted living situations or institutional settings. Finally, my fieldwork was not longitudinal, it took a snapshot of my participants' lives at one moment in time. Owing to the importance of flexible and changing needs within the ageing – *as well as you can* – in place model, future research could follow the same participant over different seasons or throughout several years, to explore how their definitions change and how they negotiate different beings and doings that were of importance to them in response to lifecourse transitions.

8.4.3 Covid-19

This fieldwork was completed before the Covid-19 outbreak. As a result, I was not able to present empirical work about how this impacted the everyday routines of getting out and about. However, there has been research that has examined the lived experience of Covid-19. Results from this research have indicated that being locked down at home negatively affected mental health, producing "psychological parallels" with imprisonment and "unintended adverse consequences" (Dhami et al., 2020, pp.11-12). This was particularly the case for older people, who were advised to 'cocoon' or 'shield', owing to increased rates of hospitalisation and death for this age group. This approach was highly criticised as being ageist and homogenising the older adult population (Leahy, 2020). Findings revealed various coping strategies and adaptations made by older people (Finlay et al., 2021). The importance of routine "as a strategy to feel normal or purposeful in everyday life" (p.4) was identified, as well as getting out and about to get outdoors, engage with nature and exercise (Finlay et al., 2021; Guzman et al., 2021). This suggests that getting out and about remained very important, if not more important to older people throughout the pandemic.

Alongside evidence of resilience, coping and adaptation, there was also evidence of widening inequalities throughout the pandemic, with those older adults living in more deprived urban areas being disproportionately affected, as they were already experiencing disadvantage due to both social and spatial inequalities (Buffel et al., 2021). With restrictions of 2km and 5km throughout lockdowns in Ireland, the quality of the immediate physical and social environment would have significantly

impacted the ability to stay well during this time. It remains to be seen how older people re-engage with their local enviroinments 'post-Covid' and whether they are able to regain what might have been lost, or whether some activities may have changed for the better.

Buffel et al. (2021) have identified six age friendly principles to guide cities in their Covid-19 recovery. Of key importance is prioritising resources in the most deprived neighbourhoods, investing community-based services, such as social infrastructure and green spaces, and engaging older people in the design of these cities. Throughout the pandemic there have been many challenges, but it has also provided a chance to reflect on what is really important. We have witnessed unprecedented urban changes throughout this time, including the incorporation of many characteristics of healthy, active and restorative urban design, for example cycling lanes, widening footpaths and outdoor seating areas (Roe and McCay, 2021). I am therefore cautiously optimistic, that with so many global environmental and societal challenges currently facing us, that these incite the urgency we need to re-prioritise and focus on what matters most to us for a good quality of life. I believe, as my participants did, that this revolves around human connection in daily life. If planners, policymakers, researchers, as well as older people themselves were to prioritise this and work towards the goal of ageing or living -as well as we can -in place, it could lead to dramatic improvements in both our local environments and our quality of life. We know how to provide restorative environments, but it requires all of us to demand and value it for it to happen.

8.5 Conclusion

Increasing numbers of Ireland's ageing population will be cared for within their homes in the future, as policymakers adopt an ageing in place framework. The aim of this is for older adults to remain living within their homes and communities for as long as possible (Department of Health, 2013). Whilst this is often deemed to be what older people themselves prefer, it is important to recognise that the experience of ageing in place will vary, depending on the congruence between personal characteristics and values, and the social and physical environments that people age (Lawton and Nahemow, 1976). Rather than a blanket goal of ageing in place, I argue that a far better policy goal would be to prioritise ageing *well* in place. Such a goal recognises that the experience and quality of ageing in place matters and might allow Ireland to achieve the goals set out within the National Positive Ageing Strategy, which were discussed at the beginning of this thesis. Rather than leaving individuals to their personal and contextual fates, it would allow us to consider what type of places promote human flourishing and quality of life and design these to support older people to live meaningful and engaged lives. To do this, it is necessary to meet older people where they are at, and offer person-centred solutions, recognising that this is subjectively defined by the older person themselves. In this research, I have shown that rarely is an older person's definition of ageing well in place perfect or static. Instead, it is instead good enough for them at that current moment and subject to change over time. However, a common goal for most older people, is to get out and about, to benefit from the variety of possible wellbeing benefits that leaving the home can provide. As a result, we must first listen to older people to find out what ageing *well* in place means to them at a particular place and time, and second, we need to bridge the gap where environments are not able to support an older persons' valued functionings. This will require person-centred solutions, to support older people to get out and about safely and enjoyably for as long as they may wish.

This page has been left intentionally blank.

References

- Aberg, A. C., Sidenvall, B., Hepworth, M., O'Reilly, K. and Lithell, H. (2005) On loss of activity and independence, adaptation improves life satisfaction in old age: A qualitative study of patients' perceptions. *Quality of Life Research*, 14(4), 1111–1125.
- Agee, J. (2009) Developing qualitative research questions: A reflective process. *International Journal of Qualitative Studies in Education*, 22(4), 431–447.
- Age Friendly Ireland (2021a) *About Us: Global Response* [Online]. Available at: <u>https://agefriendlyireland.ie/category/about-us/vision-mission-goals/</u> (Accessed 29th April 2022).
- Age Friendly Ireland (2021b) About Us: Vision, Mission, Goals [Online]. Available at: <u>https://agefriendlyireland.ie/category/about-us/vision-mission-goals/</u> (Accessed 29th April 2022).
- Age Friendly Ireland (2021c) *About Us: About the Programme* [Online]. Available at: <u>https://agefriendlyireland.ie/category/about-us/vision-mission-goals/</u> (Accessed 29th April 2022).
- Age Friendly Ireland (2020) *10 Years Delivering for Ireland: A Decade of Progress*, [Online] Available at: <u>https://dev.agefriendlyireland.ie/wp-</u> <u>content/uploads/2020/04/10-Years-Delivering-For-Ireland-A-Decade-Of-</u> <u>Progress.pdf</u> (Accessed 29th April 2022).
- Age UK (2019) *Is 'Elderly' Offensive?* [Online] Available at: <u>https://www.ageuk.org.uk/northtyneside/about-us/news/articles/2019/offensive-description-of-older-people/ https://www.ageuk.org.uk/northtyneside/about-us/news/articles/2019/offensive-description-of-older-people/(Accessed 3rd May 2022).</u>
- Ahrens, A. and Lyons, S. (2019) Changes in land cover and urban sprawl in Ireland from a comparative perspective over 1990–2012. *Land*, 8(1), 1–14.
- AIRO (2016) All-Island Research Observatory [Online]. Available at: <u>https://airo.maynoothuniversity.ie/</u> (Accessed 4th April 2016).

- Alidoust, S., Bosman, C. and Holden, G. (2019) Planning for healthy ageing: How the use of third places contributes to the social health of older populations. *Ageing and Society*, 39(7), 1459–1484.
- Alves, S. and Sugiyama, T. (2006) Inclusive Design for Getting Outdoors: Literature review on the benefits of access to outdoor environments for older people
 [Online] Available at:

http://www.idgo.ac.uk/useful_resources/for_other_researchers.htm (Accessed 12th July 2021).

- Andrews, G. J., Cutchin, M., McCracken, K., Phillips, D. R. and Wiles, J. (2007) Geographical Gerontology: The constitution of a discipline. *Social Science and Medicine*, 65(1), 151–168.
- Andrews, G. J. and Duff, C. (2019) Matter beginning to matter: On posthumanist understandings of the vital emergence of health. *Social Science and Medicine*, 226, 123–134.
- Antonovsky, A. (1987) Unravelling the Mystery of Health: How People Manage Stress and Stay Well. San Francisco: Jossey-Bass.
- An Garda Síochána (2010) *Older People Strategy* [Online] Available at: <u>https://www.garda.ie/en/crime-prevention/community-engagement/community-engagement-offices/older-person-strategy-an-garda-siochana-2010.pdf</u> (Accessed 29th April 2022).
- Baker, P. S., Bodner, E. V. and Allman, R. M. (2003) Measuring life-space mobility in community-dwelling older adults. *Journal of the American Geriatrics Society*, 51(11), 1610–1614.
- Ball, M. M., Perkins, M. M., Whittington, F. J., Hollingsworth, C., King, S. V. and Combs, B. L. (2004) Independence in assisted living. *Journal of Aging Studies*, 18(4), 467–483.
- Baltes, P. B., and Baltes, M. M. (1990) Psychological perspectives on successful aging: The model of selective optimization with compensation. In Baltes, P. B. and Baltes, M. M. (eds.) Successful Aging: Perspectives from the Behavioural Sciences. New York: Cambridge University Press, 1–34.

- Barrett, J. and Kirk, S. (2000) Running focus groups with elderly and disabled elderly participants. *Applied Ergonomics*, 31, 621–629.
- Baum, S. K., Boxley, R. L. and Sokolowski, M. (1984) Time perception and psychological well-being in the elderly. *Psychiatric Quarterly*, 56(1), 54–61.
- Bazeley, P. (2007) Qualitative Data Analysis with NVivo. London: Sage.
- Beard, J. R. and Montawi, B. (2015) Age and the environment: The global movement towards age-friendly cities and communities. *Journal of Social Work Practice*, 29(1), 5–11.
- Beard, R. L., Knauss, J. and Moyer, D. (2009) Managing disability and enjoying life: How we reframe dementia through personal narratives. *Journal of Aging Studies*, 23(4), 227–235.
- Beeber, A. S. (2008) Interdependence: Building partnerships to continue older adults' residence in the community. *Journal of Gerontological Nursing*, 34(1), 19–25.
- Bell, S. L., Leyshon, C. and Phoenix, C. (2019) Negotiating nature's weather worlds in the context of life with sight impairment. *Transactions of the Institute of British Geographers*, 44(2), 270–283.
- Bell, S. L., Phoenix, C., Lovell, R., and Wheeler, B. W. (2015) Using GPS and geonarratives: A methodological approach for understanding and situating everyday green space encounters. *Area*, 47(1), 88–96.
- Bell, S. L., Wheeler, B. W., and Phoenix, C. (2017) Using geonarratives to explore the diverse temporalities of therapeutic landscapes: Perspectives from "green" and "blue" settings. *Annals of the American Association of Geographers*, 107(1), 93–108.
- Bell, S., and Menec, V. (2013) "You don't want to ask for the help". The imperative of independence: Is it related to social exclusion? *Journal of Applied Gerontology*, 34(3), NP1–NP21.
- Bennett, K. M., and Soulsby, L. K. (2012) Wellbeing in bereavement and widowhood. *Illness, Crisis and Loss*, 20(4), 321–337.
- Bhatti, M. (2006) "When I'm in the garden I can create my own paradise": Homes and gardens in later life. *The Sociological Review*, 54(2), 318–341.

- Bhatti, M., and Church, A. (2001) Cultivating matures: Homes and gardens in late modernity. *Sociology*, 35(2), 365–383.
- Bigonnesse, C. and Chaudhury, H. (2021) Ageing in place processes in the neighbourhood environment: A proposed conceptual framework from a capability approach. *European Journal of Ageing*. 19(1), 63–74.
- Black, K., Dobbs, D., and Young, T. L. (2012) Aging in community: Mobilizing a new paradigm of older adults as a core social resource. *Journal of Applied Gerontology*, 34(2), 219–243.
- Böcker, L., Van Amen, P., and Helbich, M. (2017) Elderly travel frequencies and transport mode choices in Greater Rotterdam, the Netherlands. *Transportation*, 44(4), 831–852.
- Borglin, G., Edberg, A.-K., and Rahm Hallberg, I. (2005) The experience of quality of life among older people. *Journal of Aging Studies*, 19(2), 201–220.
- Bowling, A. (2014) *Research Methods in Health: Investigating Health and Health Services.* Berkshire: Open University Press.
- Bowling, A., and Dieppe, P. (2005) What is successful ageing and who should define it? *British Medical Journal*, 331(7531), 1548–1551.
- Bowling, A., and Gabriel, Z. (2007) Lay theories of quality of life in older age. *Ageing and Society*, 27(6), 827–848.
- Bowling, A., and Iliffe, S. (2006) Which model of successful ageing should be used?
 Baseline findings from a British longitudinal survey of ageing. *Age and Ageing*, 35(6), 607–614.
- Bramley, N., and Eatough, V. (2005) The experience of living with Parkinson's disease: An interpretative phenomenological analysis case study. *Psychology and Health*, 20(2), 223–235.
- Braun, V., and Clarke, V. (2019) Reflecting on reflexive thematic analysis. *Qualitative Research in Sport, Exercise and Health*, 11(4), pp.589–597.
- Braun, V., and Clarke, V. (2013) Successful Qualitative Research: A Practical Guide for Beginners. London, UK: Sage.

- Braun, V., and Clarke, V. (2006) Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77–101.
- Breheny, M., and Stephens, C. (2010) Ageing in a material world. New Zealand Journal of Psychology, 39(2), 41–48.
- Breheny, M., and Stephens, C. (2009) "I sort of pay back in my own little way": Managing independence and social connectedness through reciprocity. *Ageing and Society*, 29(8), 1295–1313.
- Brittain, K., Corner, L., Robinson, L., and Bond, J. (2010) Ageing in place and technologies of place: The lived experience of people with dementia in changing social, physical and technological environments. *Sociology of Health and Illness*, 32(2), 272–287.
- Bruggencate, T. T., Luijkx, K. G., and Sturm, J. (2017) Social needs of older people: A systematic literature review. *Ageing and Society*, 38(9), 1745–1770.
- Bryman, A. (2016) Social Research Methods. Oxford: Oxford University Press.
- Buffel, T., Yarker, S., Phillipson, C., Lang, L., Lewis, C., Doran, P., and Goff, M. (2021) Locked down by inequality: Older people and the COVID-19 pandemic. *Urban Studies*.
- Buffel, T. (2018) Social research and co-production with older people: Developing age-friendly communities. *Journal of Aging Studies*, 44, 52–60.
- Buffel, T., Handler, S., and Phillipson, C. (2018) *Age-Friendly Cities and Communities: A Global Perspective*. Bristol: Policy Press.
- Buffel, T., and Phillipson, C. (2018) A manifesto for the age-friendly movement:
 Developing a new urban agenda. *Journal of Aging and Social Policy*, 30(2), 173–192.
- Buffel, T., Phillipson, C., and Scharf, T. (2012) Ageing in urban environments: Developing 'age-friendly' cities. *Critical Social Policy*, 32(4), 597–617.
- Burton, E., Mitchell, L. and Stride, C. (2015) Bed of roses? The role of garden space in older people's well-being. *Proceedings of the Institution of Civil Engineers: Urban Design and Planning*, 168(4), 164–173.

- Burton, E. J., Mitchell, L. and Stride, C. B. (2011) Good places for ageing in place: Development of objective built environment measures for investigating links with older people's wellbeing. *BMC Public Health*, 11, 839, 1–13.
- Buttimer, A. (1976) Grasping the dynamism of lifeworld. *Annals of the Association* of American Geographers, 66(2), 277–292.
- Carmichael, L., Lock, K., Sweeting, D., Townshend, T., and Fischer, T. B. (2016)
 Evidence base for health and planning: Lessons from an ESRC seminar series. *Town and Country Planning*, 85(11), 461–464.
- Carp, F. M. (1988) Significance of mobility for the well-being of the elderly. *Transportation in an Aging Society: Improving Mobility and Safety of Older Persons*, 2, 1–20.
- Carpiano, R. M. (2009) Come take a walk with me: The "Go-Along" interview as a novel method for studying the implications of place for health and well-being. *Health and Place*, 15(1), 263–272.
- Carroll, S., Jespersen, A. P., and Troelsen, J. (2020) Going along with older people: Exploring age-friendly neighbourhood design through their lens. *Journal of Housing and the Built Environment*, 35(2), 555–572.
- CSO [Central Statistics Office] Central Statistics Office (CSO) (2021) Population and Migration Estimates, April 2021, Available at: <u>https://www.cso.ie/en/releasesandpublications/ep/p-</u> pme/populationandmigrationestimatesapril2021/ (Accessed 20th April 2022).
- CSO [Central Statistical Office] (2020a) Census 2016 Profile 3: An Age Profile of Ireland E3009 – Population 2011 to 2016. Available at: <u>https://data.cso.ie/</u>, (Accessed 10th August 2021).
- CSO [Central Statistical Office] (2020b) Census 2016 Profile 4: Households and Families E4029 – Percentage of Population Living Alone in Private Households 2011 to 2016. Available at: <u>https://data.cso.ie/</u> (Accessed 10 August 2021).
- CSO [Central Statistical Office] (2019) *National Travel Survey 2019*. Available at: <u>https://www.cso.ie/en/releasesandpublications/ep/p-</u> <u>nts/nationaltravelsurvey2019/cycling/</u> (Accessed 3rd May 2022).

- CSO [Central Statistics Office] (2018) *Population and Labour Force Projections* 2017–2051, Available at: <u>https://www.cso.ie/en/releasesandpublications/ep/p-</u> <u>plfp/populationandlabourforceprojections2017–</u> 2051/populationprojectionsresults/ (Accessed 3rd May 2022).
- CSO [Central Statistics Office] (2017) Census of Population 2016: Profile 3 An Age Profile of Ireland: Age Groups. Available at: <u>https://www.cso.ie/en/releasesandpublications/ep/p-cp3oy/cp3/agr/</u> (Accessed 3rd May 2022).
- Centre for Health Geoinformatics (2016) *An Atlas of Health Inequalities in Ireland from 2006 to 2011* [Online]. Available at: <u>http://www.chg.ie/index.php?action=healthinequalities</u> (Accessed 12th April 2017).
- Chang, M., and Ross, A. (2015) Briefing: Reuniting health and planning in England:
 Reflections from the front line. *Proceedings of the Institution of Civil Engineers: Urban Design and Planning*, 168(2), 69–74.
- Charlton, G. S., and Barrow, C. J. (2002) Coping and self-help group membership in Parkinson's disease: An exploratory qualitative study. *Health and Social Care in the Community*, 10(6), 472–478.
- Charmaz, K. (1991) Good Days, Bad Days: The Self in Chronic Illness and Time. New Brunswick, New Jersey: Rutgers University Press.
- Cheng, J. M., Batten, G. P., Cornwell, T., and Yao, N. (2020) A qualitative study of health-care experiences and challenges faced by ageing homebound adults. *Health Expectations*, 23(4), 934–942.
- Choi, N. G., and McDougall, G. J. (2007) Comparison of depressive symptoms between homebound older adults and ambulatory older adults. *Aging and Mental Health*, 11(3), 310–322.
- Christensen, P., Mikkelsen, M. R., Nielsen, T. A. S., and Harder, H. (2011) Children, mobility, and space: Using GPS and mobile phone technologies in ethnographic research. *Journal of Mixed Methods Research*, 5(3), 227–246.

- Coleman, T., and Kearns, R. (2015) The role of bluespaces in experiencing place, aging and wellbeing: Insights from Waiheke Island, New Zealand. *Health and Place*, 35, 206–217.
- Conway, R., Morrow, J., Brennan, R., Mulvey, C., and ÓhAiseadha, C. (2019) The current state of cycling infrastructure in Dublin and Copenhagen: A comparison of cycling infrastructure in eight radial routes into the city centre of Dublin and Copenhagen. *Irish Medical Journal*, 112(1), 856.
- Cope, M., and Elwood, S. (2009) *Qualitative GIS: A Mixed Methods Approach*. Thousand Oaks, California: Sage.
- Cosco, T. D., Prina, A. M., Perales, J., Stephan, B. C. M., and Brayne, C. (2013) Lay perspectives of successful ageing: A systematic review and meta-ethnography. *British Medical Journal Open*, 3(6), e002710, 1–9.
- Cresswell, T. (1996) In Place/Out of Place: Geography, Ideology, and Transgression. Minneapolis, Minnesota: University of Minnesota Press.
- Criado-Perez, C. (2019) *Invisible Women: Exposing Data Bias in a World Designed for Men.* London: Random House.
- Csikszentmihalyi, M. (1975) *Beyond Boredom and Anxiety: The Experience of Play in Work and Leisure.* San Francisco, California: Jossey-Bass.
- Curl, A., and Musselwhite, C. (eds.) (2018) *Geographies of Transport and Ageing*. Cham, Switzerland: Springer.
- Curl, A., Thompson, C. W., Alves, S., and Aspinall, P. (2016) Outdoor environmental supportiveness and older people's quality of life: A personal projects approach. *Journal of Housing for the Elderly*, 30(1), 1–17.
- Currie, G., and Stanley, J. (2008) Investigating links between social capital and public transport. *Transport Reviews*, 28(4), 529–547.
- Daniels, N. (1985) Just Health Care. Cambridge: Cambridge University Press.
- Davey, J. A. (2007) Older people and transport: Coping without a car. *Ageing and Society*, 27, 49–65.

- Davidson, B., Kendig, H., Stephens, F., and Merryl, V. (1993) It's My Place: Older People Talk About Their Homes. Canberra: Australian Government Publishing Service.
- Davies, N., Crowe, M., and Whitehead, L. (2016) Establishing routines to cope with the loneliness associated with widowhood: A narrative analysis. *Journal of Psychiatric and Mental Health Nursing*, 23(8), 532–539.
- De Lyser, D., Aitken, S. C., Herbert, S., Crang, M., and McDowell, L. (2010) *The SAGE Handbook of Qualitative Geography*. London: Sage.
- Denzin, N. K. (2008) Strategies of Qualitative Inquiry. London: Sage.
- Denzin, N. K., and Lincoln, Y. S. (2018) The SAGE Handbook of Qualitative Research. Thousand Oaks, California: Sage.
- Department of Health (2015) *Healthy and Positive Ageing for All: Research Strategy* 2015–2019. Dublin: Department of Health [Online]. Available at: <u>https://www.lenus.ie/bitstream/handle/10147/621490/Healthy.pdf?sequence=1</u> (Accessed 3rd May 2021).
- Department of Health. (2013) *Positive Ageing Starts Now!: The National Positive Ageing Strategy*. Dublin: Department of Health.
- Dhami, M. K., Weiss-Cohen, L., and Ayton, P. (2020) Are people experiencing the "pains of imprisonment" during the COVID-19 lockdown? *Frontiers in Psychology*, 11(3179).
- Dodgshon, R. A. (2008) Geography's place in time. *Geografiska Annaler. Series B*, *Human Geography*, 90(1), 1–15.
- Donoghue, O. A., McGarrigle, C. A. and Kenny, R. A. (2019) Who's in the driver's seat? Impact on social participation and psychosocial wellbeing in adults aged 50 and over. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 522–531.
- Donoghue, O., Orr, J., Leahy, S., and Kenny, R. A. (2017). Transport Patterns in Community-Dwelling Adults Aged 50 Years and Over in Ireland. Dublin: The Irish Longitudinal Study on Ageing [Online]. Available at:

https://tilda.tcd.ie/publications/reports/pdf/Report_TransportPatterns.pdf?msclkid =f9c99deccf6511ecb81fa76c0df3e53f (Accessed 2nd April 2022).

- Doran, P., and Buffel, T. (2018) Translating research into action: Involving older people in co-producing knowledge about age-friendly neighbourhood interventions. *Working with Older People*, 22(1), 39–47.
- Drilling, M., Grove, H., Ioannou, B. and Moulaert, T. (2021) Towards a structural embeddedness of space in the framework of the social exclusion of older people.
 In: Walsh, K., Scharf, T., Van Regenmortel, S. and Wanka, A. (eds.), *Social Exclusion in Later Life: Interdisciplinary and Policy Perspectives*. Cham: Springer. Chapter 15.
- Duff, C. (2010) On the role of affect and practice in the production of place. *Environment and Planning D: Society and Space*, 28(5), 881–895.
- Elsadek, M., Liu, B., and Xie, J. (2020) Window view and relaxation: Viewing green space from a high-rise estate improves urban dwellers' wellbeing. *Urban Forestry and Urban Greening*, 55, 126846.
- Elwood, S. (2010) Mixed methods: Thinking, doing, and asking in multiple ways. In: De Lyser, D., Aitken, S. C., Herbert, S., Crang, M., and McDowell, L. (eds.), *The SAGE Handbook of Qualitative Geography*. London: Sage, 94–113.
- Elwood, S. (2006) Negotiating knowledge production: The everyday inclusions, exclusions, and contradictions of participatory GIS research. *The Professional Geographer*, 58(2), 197–208.
- European Union Agency for Fundamental Rights (2020) *Travellers in Ireland: Key Results from the Roma and Travellers Survey 2019* [Online]. Available at: <u>https://op.europa.eu/en/publication_detail/_/publication/1a77a06f-00e9-11ec-</u> <u>8f47-01aa75ed71a1/language-</u>

en?msclkid=2164460ccf6611ecb7800ea06fd605f5 (Accessed 2nd May 2022).

Eurostat (2017) Self-perceived long-standing limitations in usual activities due to health problem by sex, age and income quintile. European Union Statistics on Income and Living Conditions (EU-SILC) [Online]. Available at: <u>https://ec.europa.eu/eurostat/web/health/data/database</u> (Accessed 3rd March 2022). Ewart, I., and Luck, R. (2013) Living from home. Home Cultures, 10(1), 25–42.

- Ewing, R., Schmid, T., Killingsworth, R., Zlot, A., and Raudenbush, S. (2003)
 Relationship between urban sprawl and physical activity, obesity, and morbidity. *American Journal of Health Promotion*, 18(1), 47–57.
- Fahey, T., Maitre, B., Nolan, B., and Whelan, C.T. (2007) A Social Portrait of Older People in Ireland. Dublin: Economic and Social Research Institute [Online]. Available at: <u>https://www.lenus.ie/bitstream/handle/10147/44964/6659.pdf?sequence=1</u> (Accessed 12th February 2021).
- Fenton, N. E., Baxter, J., Elliott, S. J., and Williams, D. A. (2016) *Practicing Qualitative Methods in Health Geographies*. Abingdon: Taylor and Francis.
- Fernández–Ballesteros, R., García, L. F., Abarca, D., Blanc, L., Efklides, A., Kornfeld, R., Lerma, A. J., Mendoza-Nuñez, V. M., Mendoza-Ruvalcaba, N. M., Orosa, T., Paúl, C., and Patricia, S. (2008) Lay concept of aging well: Crosscultural comparisons. *Journal of the American Geriatrics Society*, 56(5), 950– 952.
- Fingerman, K. L. (2009) Consequential strangers and peripheral ties: The importance of unimportant relationships. *Journal of Family Theory and Review*, 1(2), 69–86.
- Finlay, J. M. (2018) "Walk like a penguin": Older Minnesotans' experiences of (non)therapeutic white space. *Social Science and Medicine*, 198, 77–84.
- Finlay, J. M. and Bowman, J. A. (2017) Geographies on the move: A practical and theoretical approach to the mobile interview. *The Professional Geographer*, 69(2), 263–274.
- Finlay, J.M., Esposito, M., Kim, M. H., Gomez-Lopez, I., and Clarke, P. (2019) Closure of "third places"? Exploring potential consequences for collective health and wellbeing. *Health and Place*, 60, 102225.
- Finlay, J. M., and Finn, B. M. (2021) Geography's blind spot: The age-old urban question. *Urban Geography*, 42(8), 1061–1078.

- Finlay, J. M., Gaugler, J. E., and Kane, R. L. (2018) Ageing in the margins: Expectations of and struggles for "a good place to grow old" among low-income older Minnesotans. *Ageing and Society*, 40(4), 759–783.
- Finlay, J. M., Kler, J. S., O'Shea, B. Q., Eastman, M. R., Vinson, Y. R., and Kobayashi, L. C. (2021) Coping during the COVID-19 pandemic: A qualitative study of older adults across the United States. *Frontiers in Public Health*, 9, 643807.
- Finlay, J. M., and Kobayashi, L. C. (2018) Social isolation and loneliness in later life: A parallel convergent mixed-methods case study of older adults and their residential contexts in the Minneapolis metropolitan area, USA. *Social Science and Medicine*, 208, 25–33.
- Finlay, J. M., and Rowles, G. D. (2021) Clinical geography: A proposal to embrace space, place and wellbeing through person-centered practice. *Wellbeing, Space* and Society, 2, 100035.
- Foley, R., Bell, S. L., Gittens, H., Grove, H., Kaley, A., McLauchlan, A., Osborne, T., Power, A., Roberts, E., and Thomas, M. (2020) "Disciplined research in undisciplined settings": Critical explorations of in-situ and mobile methodologies in geographies of health and wellbeing. *Area*, 52(3), 514–522.
- Franke, T., Sims–Gould, J., Chaudhury, H., Winters, M., and Mckay, H. (2019) "It makes your life worthwhile. It gives you a purpose in living": Mobility experiences among active older adults with low income. *Ageing and Society*, 39(8), 1639–1666.
- Franke, T., Tong, C., Ashe, M. C., McKay, H., and Sims-Gould, J. (2013) The secrets of highly active older adults. *Journal of Aging Studies*, 27(4), 398–409.
- Frumkin, H., Frank, L., and Jackson, R. (2004) Urban Sprawl and Public Health: Designing, Planning, and Building for Healthy Communities, Washington, DC: Island Press.
- Fullilove, M. T. (2020) Main Street: How a City's Heart Connects Us All. New York: New York University Press.
- Fullilove, M. T. (1996) Psychiatric implications of displacement: Contributions from the psychology of place. *American Journal of Psychiatry*, 153(12), 1516–1523.

- Gallagher, C. (2012) Connectedness in the lives of older people in Ireland: A study of the communal participation of older people in two geographic localities. *Irish Journal of Sociology*, 20(1), 84–102.
- Gardner, P. (2014) The role of social engagement and identity in community mobility among older adults aging in place. *Disability and Rehabilitation*, 36(15), 1249–1257.
- Gardner, P. J. (2011) Natural neighborhood networks: Important social networks in the lives of older adults aging in place. *Journal of Aging Studies*, 25(3), 263–271.
- Garrard, J., Rissel, C., and Bauman, A. (2012) Health benefits of cycling. In: Pucher, J. and Buehler, R. (eds.), *City Cycling*. Cambridge Massachusetts: MIT Press, chapter 3.
- Gatrell, A. C. (2013) Therapeutic mobilities: Walking and "steps" to wellbeing and health. *Health and Place*, 22, 98–106.
- Gatrell, A. C., and Elliott, S. J. (2015) *Geographies of Health: An Introduction*, third edition. Chichester: John Wiley.
- Gee, N. R., and Mueller, M. K. (2019) A systematic review of research on pet ownership and animal interactions among older adults. *Anthrozoös*, 32(2), 183– 207.
- Gibney, S., Ward, M., Shannon, S., Moore, T., and Moran, N. (2018) *Positive Ageing in Age Friendly Cities and Counties: Local Indicators for Ireland*. Dublin: Department of Health.
- Giesel, F., and Rahn, C. (2015) Everyday life in the suburbs of Berlin: Consequences for the social participation of aged men and women. *Journal of Women and Aging*, 27(4), 330–351.
- Gilroy, R. (2021) Planning for an Ageing Society. London, UK: Lund Humphries.
- Gilroy, R. (2008) Places that support human flourishing: Lessons from later life. *Planning Theory and Practice*, 9(2), 145–163.
- Gilroy, R. (2006) Taking a capabilities approach to evaluating supportive environments for older people. *Applied Research in Quality of Life*, 1(3), 343– 356.

- Glass, T. A. (2003) Assessing the success of successful aging. Annals of Internal Medicine, 139(5 (Part 1)), 382–383.
- Glass, T.A., Mendes de Leon, C., Marotolli, R.A., and Berkman, L.F. (1999)
 Population based study of social and productive activities as predictors of survival among elderly Americans. *British Medical Journal*, 319, 478–483.
- Goins, R. T., Jones, J., Schure, M., Rosenberg, D. E., Phelan, E. A., Dodson, S., and Jones, D. L. (2015) Older adults' perceptions of mobility: A metasynthesis of qualitative studies. *The Gerontologist*, 55(6), 929–942.
- Golant, S.M. (2018) Explaining the ageing in place realities of older adults. In:
 Skinner, M., Andrews, G. and Cutchin, M. (eds.) *Geographical Gerontology: Concepts and Approaches*. London: Routledge, chapter 15.
- Golant, S. M. (2015) *Aging in the Right Place*. Baltimore, Maryland: Health Professions Press.
- Goodson, L. and Phillimore, J. (2004) *Qualitative Research in Tourism: Ontologies, Epistemologies and Methodologies*. London: Taylor and Francis.
- Green, J., Jones, A., and Roberts, H. (2014) More than A to B: The role of free bus travel for the mobility and wellbeing of older citizens in London. *Ageing and Society*, 34(3), 472–494.
- Greenwood, N., Pound, C., Brearley, S., and Smith, R. (2019). A qualitative study of older informal carers' experiences and perceptions of their caring role. *Maturitas*, 124, 1–7.
- Grove, H. (2021) Ageing as well as you can in place: Applying a geographical lens to the capability approach. *Social Science and Medicine*, 288, 113525.
- Guzman, V., Garrido–Cumbrera, M., Braçe, O., Hewlett, D., and Foley, R. (2021)
 Health and wellbeing under COVID-19: The GreenCOVID survey. *Irish Geography*, 53(2), 157–162.
- Hagan, R. J. (2019) Getting out of the house: The use of community transport as a third place for rural-dwelling older adults. *Ageing and Society*, 40(11), 2519– 2539.

- Hamilton-West, K., Milne, A. and Hotham, S. (2020) New horizons in supporting older people's health and wellbeing: Is social prescribing a way forward? *Age and Ageing*, 49(3), 319–326.
- Hammond, M., and Saunders, N. (2021) *A Design for Life: Urban Practices for an Age–Friendly city.* Manchester: MMU Press.
- Hammond, M., White, S., and Phillipson, C. (2020) From precarity to interdependence: The role of age-friendly communities in promoting wellbeing in excluded communities. In: Boyko, C., Cooper, R., and Dunn, N. (eds.), *Designing Future Cities for Wellbeing*. London: Routledge, 122–138.
- Hand, C., Huot, S., Laliberte Rudman, D., and Wijekoon, S. (2017) Qualitativegeospatial methods of exploring person-place transactions in aging adults: A scoping review. *The Gerontologist*, 57(3), e47–e61.
- Handler, S. (2014) *An Alternative Age Friendly Handbook*. Manchester: University of Manchester Library.
- Hart, J. T. (1971) The inverse care law. The Lancet, 297(7696), 405-412.
- Heggestad, A. K. T., Nortvedt, P., and Slettebø, Å. (2013) "Like a prison without bars": Dementia and experiences of dignity. *Nursing Ethics*, 20(8), 881–892.
- Heo, J., Lee, Y., Pedersen, P. M., and McCormick, B. P. (2010) Flow experience in the daily lives of older adults: An analysis of the interaction between flow, individual differences, serious leisure, location, and social context. *Canadian Journal on Aging / La Revue canadienne du vieillissement*, 29(3), 411–423.
- Hockey, A., Phillips, J., and Walford, N. (2013) Planning for an Ageing Society:
 Voices from the Planning Profession. *Planning Practice and Research*, 28(5), 527–543.
- Holland, C., Kellaher, L., Peace, S., Scharf, T., Breeze, E., Gow, J., and Gilhooly, M. (2005) Getting out and about. In: Walker, A. (ed.), *Understanding Quality of Life in Old Age*. Berkshire: Open University Press, chapter 4.
- Home, R., Hunziker, M., and Bauer, N. (2012) Psychosocial outcomes as motivations for visiting nearby urban green spaces. *Leisure Sciences*, 34(4), 350– 365.

- Hillcoat-Nallétamby, S. (2014) The meaning of "independence" for older people in different residential settings. *The Journals of Gerontology: Series B*, 69(3), 419–430.
- Hillcoat-Nalletamby, S., and Ogg, J. I. M. (2013) Moving beyond "ageing in place": Older people's dislikes about their home and neighbourhood environments as a motive for wishing to move. *Ageing and Society*, 34(10), 1771–1796.
- Hirsch, J. A., Winters, M., Clarke, P., and McKay, H. (2014) Generating GPS activity spaces that shed light upon the mobility habits of older adults: A descriptive analysis. *International Journal of Health Geographics*, 13(1), 51.
- Hirsch, J. A., Winters, M., Ashe, M. C., Clarke, P. J., and McKay, H. A. (2016)
 Destinations that older adults experience within their GPS activity spaces:
 Relation to objectively measured physical activity. *Environment and Behavior*, 48(1), 55–77.
- Hjorthol, R. (2013) Winter weather: An obstacle to older people's activities? *Journal* of *Transport Geography*, 28, 186–191.
- Hopkins, P., and Pain, R. (2007) Geographies of age: Thinking relationally. *Area*, 39(3), 287–294.
- Holt-Lunstad, J., Smith, T. B., Baker, M., Harris, T., and Stephenson, D. (2015) Loneliness and social isolation as risk factors for mortality: A meta-analytic review. *Perspectives on Psychological Science*, 10(2), 227–37.
- Holt-Lunstad, J., Smith, T. B., and Layton, J. B. (2010) Social relationships and mortality risk: A meta-analytic review. *PLOS Medicine*, 7(7), e1000316.
- Hong, A., Sallis, J. F., King, A. C., Conway, T. L., Saelens, B., Cain, K. L., Fox, E.
 H., and Frank, L. D. (2018) Linking green space to neighborhood social capital in older adults: The role of perceived safety. *Social Science and Medicine*, 207, 38–45.
- Hubbard, G., Downs, M. G., and Tester, S. (2003) Including older people with dementia in research: Challenges and strategies. *Aging and Mental Health*, 7(5), 351–362.

- Huber, M., Knottnerus, J. A., Green, L., van Der Horst, H., Jadad, A. R., Kromhout, D., Leonard, B., Lorig, K., Loureiro, M. I., van der Meer, J. W., Schnabel, P., Smith, R., van Weel, C., and Smid, H. (2011) How should we define health? *British Medical Journal*, 343, d4163, 1–3.
- Hui Gan, G. Z., Hill, A.-M., Yeung, P., Keesing, S., and Netto, J. A. (2020) Pet ownership and its influence on mental health in older adults. *Aging and Mental Health*, 24(10), 1605–1612.
- Jeste, D. V., Depp, C. A., and Vahia, I. V. (2010) Successful cognitive and emotional aging. *World Psychiatry*, 9(2), 78–84.
- Jonasson, S. B., Nilsson, M. H., Lexell, J., and Carlsson, G. (2018) Experiences of fear of falling in persons with Parkinson's disease: A qualitative study. *BMC Geriatrics*, 18(1), 44, 1–10.
- Jones, A., Goodman, A., Roberts, H., Steinbach, R., and Green, J. (2013) Entitlement to concessionary public transport and wellbeing: A qualitative study of young people and older citizens in London, UK. *Social Science and Medicine*, 91, 202– 209.
- Joyce, K., and Mamo, L. (2016) Graying the cyborg: New directions in feminist analyses of aging, science, and technology. In: Calasanti, T. M., and Slevin, K. F. (eds.), *Age Matters: Realigning Feminist Thinking*. London: Routledge, 99–121.
- Kamiya, Y. and Sofroniou, N. (2011) Socio-demographic characteristics of older people in Ireland. In: Barrett, A., Savva, G., Timonen, V., and Kenny, R. A. (eds.), *Fifty Plus in Ireland 2011: First Results from The Irish Longitudinal Study on Ageing (TILDA)* [Online]. Dublin: Trinity College Dublin, chapter 2. Available at: <u>https://tilda.tcd.ie/publications/reports/pdf/w1-key-findings-report/Tilda_Master_First_Findings_Report.pdf</u> (Accessed 10th March 2022).
- Kearns, A., and Tannahill, C. (2015) Briefing Paper 22: Loneliness in Glasgow's Deprived Communities [Online]. Glasgow: GoWell. Available at: <u>https://www.gowellonline.com/assets/0000/3722/GoWell_Briefing_Paper_BP_2</u> <u>2_Loneliness.pdf</u> (Accessed 22nd April 2021).

- Kemperman, A., and Timmermans, H. (2014) Green spaces in the direct living environment and social contacts of the aging population. *Landscape and Urban Planning*, 129, 44–54.
- Kidder, L. H. and Fine, M. (1987) Qualitative and quantitative methods: When stories converge. *New Directions for Program Evaluation*, 1987(35), 57–75.
- Kitchin, R. and Tate, N. J. (2013) *Conducting Research in Human Geography: Theory, Methodology and Practice*. Abingdon: Routledge.
- Klinenberg, E. (2018) *Palaces for the People: How to Build a More Equal and United Society*. London: Random House.
- Klinenberg, E. (2016) Social isolation, loneliness, and living alone: Identifying the risks for public health. *American Journal of Public Health*, 106(5), 786–787.
- Knigge, L., and Cope, M. (2009) Grounded visualization and scale: A recursive analysis of community spaces. In: Cope, M., and Elwood, S. (eds.), *Qualitative GIS: A Mixed Methods Approach*. London: Sage.
- Knigge, L., and Cope, M. (2006) Grounded visualization: Integrating the analysis of qualitative and quantitative data through grounded theory and visualization. *Environment and Planning A: Economy and Space*, 38(11), 2021–2037.
- Kudlicka, A., Hindle, J. V., Spencer, L. E., and Clare, L. (2018) Everyday functioning of people with Parkinson's disease and impairments in executive function: A qualitative investigation. *Disability and Rehabilitation*, 40(20), 2351–2363.
- Kusenbach, M. (2003) Street phenomenology: The go-along as ethnographic research tool. *Ethnography*, 4(3), 455–485.
- Kwan, M.-P. (2021) *Geo-Narrative or Geo-Narratives* [Online]. Available at: <u>http://meipokwan.org/Geonarrative.html</u> (Accessed 10th February 2022).
- Kwan, M.-P. (2002) Feminist visualization: Re-envisioning GIS as a method in feminist geographic research. Annals of the Association of American Geographers, 92(4), 645–661.
- Kwan, M.-P. and Ding, G. (2008) Geo-narrative: Extending Geographic Information Systems for narrative analysis in qualitative and mixed-method research. *The Professional Geographer*, 60(4), 443–465.
- Kweon, B.-S., Sullivan, W. C., and Wiley, A. R. (1998) Green common spaces and the social integration of inner-city older adults. *Environment and Behavior*, 30(6), 832–858.
- Lager, D. R. (2015) Perspectives on ageing in place: Older adults' experiences of everyday life in urban neighbourhoods. PhD thesis, University of Groningen.
- Lager, D., and van Hoven, B. (2019) Exploring the experienced impact of studentification on ageing-in-place. *Urban Planning*, 4(2), 96–105.
- Lager, D., van Hoven, B., and Huigen, P. P. P. (2019) Neighbourhood walks as place-making in later life. *Social and Cultural Geography*, 22(8), 1080–1098.
- Lager, D., van Hoven, B., and Huigen, P. P. P. (2016) Rhythms, ageing and neighbourhoods. *Environment and Planning A: Economy and Space*, 48(8), 1565–1580.
- Lager, D., Van Hoven, B., and Huigen, P. P. P. (2015) Exploring the subjective dimensions of older adults' use of neighbourhood space through walking interviews. In: Lager, D. (ed.), *Perspectives on Ageing in Place. Older Adults' Experiences of Everyday Life in Urban Neighbourhoods*. Enschede: Gildeprint, 71–92.
- Lager, D., van Hoven, B. and Huigen, P. P. P. (2013) Dealing with change in old age: Negotiating working-class belonging in a neighbourhood in the process of urban renewal in the Netherlands. *Geoforum*, 50, 54–61.
- Lang, F. R. (2001) Regulation of social relationships in later adulthood. *The Journals* of Gerontology: Series B, 56(6), P321–P326.
- Lavery, I., Davey, S., Woodside, A., and Ewart, K. (1996) The vital role of street design and management in reducing barriers to older people's mobility. *Landscape and Urban Planning*, 35, 181–192.

- Lawton, M. P., and Nahemow, L. (1973) Ecology and the aging process. In: Eisdorfer, C. and Lawton, M.P. (eds.), *Psychology of Adult Development and Aging*. Washington, DC: American Psychological Association, 619–674.
- Leahy, A. (2020) Time to confront the portrait in the attic? Reflections on theories and discourses of ageing sparked by the COVID-19 crisis. *British Society of Gerontology* [Online]. Available at: <u>https://ageingissues.wordpress.com/2020/04/09/time-to-confront-the-portraitin-the-attic-reflections-on-theories-and-discourses-of-ageing-sparked-bycovid-19/ (Accessed 20th April 2021).</u>
- Lefebvre, H. (1991) The Production of Space. Oxford: Basil Blackwell.
- Lefebvre, H., Moore, G., and Elden, S. (2004) *Rhythmanalysis: Space, Time and Everyday Life*. London: Continuum.
- Leyden K. (2003) Social capital and the built environment: The importance of walkable neighborhoods. *American Journal of Public Health*, 93, 1546–1551.
- Limb, M. and Dwyer, C. (2001) *Qualitative Methodologies for Geographers: Issues and Debates*. Oxford: Oxford University Press.
- Little, B. R. (1983) Personal projects: A rationale and method for investigation. *Environment and Behavior*, 15(3), 273–309.
- Ludwig, F. M. (1997) How routine facilitates wellbeing in older women. Occupational Therapy International, 4(3), 215–230.
- Maher, C., Hadfield, M., Hutchings, M., and de Eyto, A. (2018) Ensuring rigor in qualitative data analysis: A design research approach to coding combining NVivo with traditional material methods. *International Journal of Qualitative Methods*, 17(1), 1–13.
- Mahmood, A., Chaudhury, H., Michael, Y. L., Campo, M., Hay, K. and Sarte, A. (2012) A photovoice documentation of the role of neighborhood physical and social environments in older adults' physical activity in two metropolitan areas in North America. *Social Science and Medicine*, 74(8), 1180–1192.
- Marmot, M. (2018) An inverse care law for our time. *British Medical Journal*, 362, k3216, 1–2.

- Marmot, M. (2015) *The Health Gap: The Challenge of an Unequal World*. London: Bloomsbury.
- Marsh, P. and Kelly, L. (2018) Dignity of risk in the community: a review of and reflections on the literature. *Health, Risk and Society*, 20(5-6), 297–311.
- Martin, P., Kelly, N., Kahana, B., Kahana, E., Willcox, B. J., Willcox, D. C., and Poon, L. W. (2015) Defining successful aging: A tangible or elusive concept? *Gerontologist*, 55(1), 14–25.
- Matthews, S. A., Detwiler, J. E., and Burton, L. M. (2005) Geo-ethnography: Coupling Geographic Information Analysis techniques with ethnographic methods in urban research. *Cartographica: The International Journal for Geographic Information and Geovisualization*, 40(4), 75–90.
- McGee, H., Morgan, K., Hickey, A., Burke, H., and Savva, G. (2011) Quality of Life and Beliefs about Ageing. In: Barrett, A., Savva, G., Timonen, V., and Kenny, R.A. (eds.), *Fifty Plus in Ireland 2011: First Results from The Irish Longitudinal Study on Ageing (TILDA)* [Online]. Dublin: Trinity College Dublin, chapter 10. Available at: <u>https://tilda.tcd.ie/publications/reports/pdf/w1-key-findings-</u> <u>report/Tilda_Master_First_Findings_Report.pdf</u> (Accessed 10th March 2022).
- McManus, R. (2019) Suburbanization in Europe: A focus on Dublin. In: Hanlon, B., and Vicino, T. J. (eds.), *The Routledge Companion to the Suburbs*. London: Routledge, chapter 7.
- McNicholas, J., and Collis, G. M. (2000) Dogs as catalysts for social interactions: Robustness of the effect. *British Journal of Psychology*, 91(1), 61–70.
- McNicholas, J. (2014) The role of pets in the lives of older people: A review. *Working with Older People*, 18(3), 128–133.
- Meijering, L., van Hoven, B., and Yousefzadeh, S. (2019) "I think I'm better at it myself": The capability approach and being independent in later life. *Research on Ageing and Social Policy*, 7(1), 229–259.
- Melis, G., Gelormino, E., Marra, G., Ferracin, E., and Costa, G. (2015). The effects of the urban built environment on mental health: A cohort study in a large Northern Italian city. *International Journal of Environmental Research and Public Health*, 12(11), 14898–14915.

- Mennis, J., Mason, M. J., and Cao, Y. (2013) Qualitative GIS and the visualization of narrative activity space data. *International Journal of Geographical Information Science*, 27(2), 267–291.
- Metz, D. H. (2000) Mobility of older people and their quality of life. *Transport Policy*, 7(2), 149–152.
- Mezuk, B., and Rebok, G. W. (2008) Social integration and social support among older adults following driving cessation. *Journal of Gerontology Social Science*, 63B, 298–303.
- Miles, D. A. (2017) A taxonomy of research gaps: Identifying and defining the seven research gaps. *Journal of Research Methods and Strategies*, 1–15.
- Milligan, C., Gatrell, A., and Bingley, A. (2004) 'Cultivating health': Therapeutic landscapes and older people in northern England. *Social Science and Medicine*, 58(9), 1781–1793.
- Milton, S., Pliakas, T., Hawkesworth, S., Nanchahal, K., Grundy, C., Amuzu, A., Casas, J. P., and Lock, K. (2015) A qualitative geographical information systems approach to explore how older people over 70 years interact with and define their neighbourhood environment. *Health and Place*, 36, 127–33.
- Moeyersons, M., De Vliegher, K., Huyghe, B., De Groof, S., Milisen, K., and De Casterlé, B. D. (2022) "Living in a shrinking world": The experience of loneliness among community-dwelling older people with reduced mobility: a qualitative grounded theory approach. *BMC Geriatrics*, 22, 1–13.
- Mollenkopf, H., Hieber, A. and Wahl, H.-W. (2011) Continuity and change in older adults' perceptions of out-of-home mobility over ten years: A qualitativequantitative approach. *Ageing and Society*, 31(5), 782–802.
- Morgan, D. L. (2007) Paradigms lost and pragmatism regained: Methodological implications of combining qualitative and quantitative methods. *Journal of Mixed Methods Research*, 1(1), 48–76.
- Morgan, T., Wiles, J., Park, H.-J., Moeke-Maxwell, T., Dewes, O., Black, S., Williams, L. and Gott, M. (2019). Social connectedness: What matters to older people? *Ageing and Society*, 41(5), 1126–1144.

- Nathan, A., Pereira, G., Foster, S., Hooper, P. Saarloos, D., and Giles-Corti, B. (2012) Access to commercial destinations within the neighbourhood and walking among Australian older adults. *International Journal of Behavioral Nutrition and Physical Activity*, 9(133), 1–8.
- National Transport Authority (2015) *Transport Strategy for the Greater Dublin Area* 2016–2035. Dublin: National Transport Authority [Online]. Available at: <u>https://www.nationaltransport.ie/wp-</u> <u>content/uploads/2016/08/Transport_Strategy_for_the_Greater_Dublin_Area_201</u> <u>6–2035.pdf</u> (Accessed 12th April 2022).
- Nedovic-Budic, Z., Knaap, G. J., Shahumyan, H., Williams, B., and Slaev, A. (2016) Measuring urban form at community scale: Case study of Dublin, Ireland. *Cities*, 55, 148–164.
- Newton, R., Ormerod, M., Burton, E., Mitchell, L., and Ward-Thompson, C. (2010) Increasing independence for older people through good street design: Managing community care. *Journal of Integrated Care*, 18(3), 24–29.
- Nolan, M. R., Davies, S., Brown, J., Keady, J., and Nolan, J. (2004) Beyond "person-centred" care: a new vision for gerontological nursing. *Journal of Clinical Nursing*, 13(3a), 45–53.
- Nordbakke, S. (2013) Capabilities for mobility among urban older women: Barriers, strategies and options. *Journal of Transport Geography*, 26, 166–174.
- Nordbakke, S., and Schwanen, T. (2014) Well-being and mobility: A theoretical framework and literature review focusing on older people. *Mobilities*, 9(1), 104–129.
- Nussbaum, M. (2003) Capabilities as fundamental entitlements: Sen and social justice. *Feminist Economics*, 9(2–3), 33–59.
- Oana, P. L., Harutyun, S., Brendan, W., and Sheila, C. (2011) Scenarios and indicators supporting urban regional planning. *Procedia: Social and Behavioral Sciences*, 21, 243–252.
- Ohman, A., and Nygård, L. (2005) Meanings and motives for engagement in selfchosen daily life occupations among individuals with Alzheimer's Disease. *OTJR: Occupation, Participation and Health*, 25(3), 89–97.

- Oldenburg, R. (1997) Our vanishing "third places". *Planning Commissioners Journal*. 25(4), 6–10.
- Oldenburg, R. (1989) The Great Good Place: Cafes, Coffee Shops, Bookstores, Bars, Hair Salons and Other Hangouts at the Heart of the Community. New York: Marlowe and Company.
- Oldenburg, R., and Brissett, D. (1982) The third place. *Qualitative Sociology*, 5(4), 265–284.
- Olsson, A., Lampic, C., Skovdahl, K., and Engström, M. (2013) Persons with earlystage dementia reflect on being outdoors: A repeated interview study. *Aging and Mental Health*, 17(7), 793–800.
- Orr, N., Wagstaffe, A., Briscoe, S., and Garside, R. (2016) How do older people describe their sensory experiences of the natural world? A systematic review of the qualitative evidence. *BMC Geriatrics*, 16(1), 116, 1–16.
- Odzakovic, E., Hellström, I., Ward, R., and Kullberg, A. (2018) "Overjoyed that I can go outside": Using walking interviews to learn about the lived experience and meaning of neighbourhood for people living with dementia. *Dementia*, 19(7), 2199–2219.
- Pani-Harreman, K. E., Bours, G. J. J. W., Zander, I., Kempen, G. I. J. M., and Van Duren, J. M. A. (2020) Definitions, key themes and aspects of "ageing in place": a scoping review. *Ageing and Society*, 41, 2026–2059.
- Parkhurst, G., Galvin, K., Musselwhite, C., Phillips, J., Shergold, I., and Todres, L. (2014) Beyond transport: Understanding the role of mobilities in connecting rural elders in civic society. In: Hennesey, C., Means, R., and Burholt, V. (eds.), *Countryside Connections: Older People, Community and Place in Rural Britain*. Bristol: Policy Press, 125–157.
- Pavlovskaya, M. (2017) Qualitative GIS. In: Richardson, D., Castree, N., Goodchild, M. F., Kobayashi, A. L., Liu, W., and Marston, R. A. (eds.) *The International Encyclopedia of Geography: People, the Earth, Environment, and Technology* [e–book]. Chichester, UK: John Wiley. Available at: Wiley Online Library https://doi.org/10.1002/9781118786352.wbieg1156 (Accessed 10th March 2022).

- Payne, B. R., Jackson, J. J., Noh, S. R., and Stine–Morrow, E. A. L. (2011) In the zone: Flow state and cognition in older adults. *Psychology and Aging*, 26(3), 738–743.
- Peace, S., Holland, C., and Kellaher, L. (2011) "Option recognition" in later life: Variations in ageing in place. *Ageing and Society*, 31(5), 734–757.
- Peace, S., Holland, C., and Kellaher, L. (2005). Making space for identity. In: Andrews, G., and Phillips, D. R. (eds.), *Ageing and Place: Perspectives, Policy, Practice.* London: Routledge, 188–204.
- Pearson, D. G., and Craig, T. (2014) The great outdoors? Exploring the mental health benefits of natural environments. *Frontiers in Psychology*, 5(1178), 1–4.
- Perchoux, C., Brondeel, R., Wasfi, R., Klein, O., Caruso, G., Vallée, J., Klein, S., Thierry, B., Dijst, M., Chaix, B., Kestens, Y., and Gerber, P. (2019). Walking, trip purpose, and exposure to multiple environments: A case study of older adults in Luxembourg. *Journal of Transport and Health*, 13, 170–184.
- Perchoux, C., Chaix, B., Cummins, S., and Kestens, Y. (2013) Conceptualization and measurement of environmental exposure in epidemiology: Accounting for activity space related to daily mobility. *Health and Place*, 21, 86–93.
- Phelan, E. A., Anderson, L. A., Lacroix, A. Z., and Larson, E. B. (2004) Older adults' views of "successful aging": How do they compare with researchers' definitions? *Journal of the American Geriatrics Society*, 52(2), 211–216.
- Phelan, E. A., and Larson, E. B. (2002) "Successful aging": Where next? *Journal of the American Geriatrics Society*, 50(7), 1306–1308.
- Phillips, J. (2018) Planning and design of ageing communities. In: Skinner, M., Andrews, G., and Cutchin, M. (eds.), *Geographical Gerontology: Concepts and Approaches*. London: Routledge, chapter 6.
- Phillips, J. E., Ajrouch, K. J., and Hillcoat-Nallétamby, S. (2010) Key Concepts in Social Gerontology, London: Sage.
- Phillips, J., Walford, N., Hockey, A., Foreman, N., and Lewis, M. (2013) Older people and outdoor environments: Pedestrian anxieties and barriers in the use of familiar and unfamiliar spaces. *Geoforum*, 47, 113–124.

- Phillips, J., Walford, N., Hockey, A., and Sparks, L. (2021) Older people, town centres and the revival of the "High Street". *Planning Theory and Practice*, 11(21), 21–26.
- Phillipson, C. (2007) The "elected" and the "excluded": Sociological perspectives on the experience of place and community in old age. *Ageing and Society*, 27(3), 321–342.
- Phinney, A., Moody, E. M., and Small, J. A. (2014). The effect of a communityengaged arts program on older adults' well-being. *Canadian Journal on Aging/La Revue canadienne du vieillissement*, 33(3), 336–345.
- Phoenix, C. (2018) Why qualitative research is needed in Gerontology and how we can do it better. *The Journals of Gerontology: Series B*, 73(7), e81–e85.
- Phoenix, C., and Orr, N. (2014) Pleasure: A forgotten dimension of physical activity in older age. *Social Science and Medicine*, 115, 94–102.
- Pierce, J., and Martin, D. G. (2015) Placing Lefebvre. Antipode, 47(5), 1279–1299.
- Pinquart, M., and Sorensen, S. (2001) Influences on loneliness in older adults: A meta-analysis. *Basic and Applied Social Psychology*, 23(4), 245–266.
- Pretty, J., Angus, C., Bain, M., Barton, J., Gladwell, V., Hine, R. & Sellens, M. (2009) *Nature, Childhood, Health and Life Pathways*. Interdisciplinary Centre for Environment and Society Occasional Paper 2009-02. University of Essex, UK.
- Provencher, C., Keating, N., Warburton, J., and Roos, V. (2014) Ageing and community: Introduction to the special issue. *Journal of Community and Applied Social Psychology*, 24(1), 1–11.
- Punch, S. (2012) Hidden struggles of fieldwork: Exploring the role and use of field diaries. *Emotion, Space and Society*, 5(2), 86–93.
- Rantakokko, M., Portegijs, E., Viljanen, A., Iwarsson, S., Kauppinen, M., and Rantanen, T. (2016) Changes in life-space mobility and quality of life among community-dwelling older people: A 2-year follow-up study. *Quality of Life Research*, 25(5), 1189–1197.

- Rantakokko, M., Iwarsson, S., Vahaluoto, S., Portegijs, E., Viljanen, A., and Rantanen, T. (2014) Perceived environmental barriers to outdoor mobility and feelings of loneliness among community-dwelling older people. *The Journals of Gerontology: Series A*, 69(12), 1562–1568.
- Registered Nurses' Association of Ontario (2010) *Strategies to Support Self-Management in Chronic Conditions: Collaboration with Clients*. Toronto, Canada: Registered Nurses' Association of Ontario.
- Reinhard E., Courtin E., van Lenthe F.J., and Avendano, M. (2018) Public transport policy, social engagement and mental health in older age: A quasi-experimental evaluation of free bus passes in England. *Journal of Epidemiology and Community Health*, 72, 361–368.
- Richard, L., Laforest, S., Dufresne, F., and Sapinski, J. P. (2005) The quality of life of older adults living in an urban environment: Professional and lay perspectives. *Canadian Journal on Aging*, 24(1), 19–30.
- Rigby, J., Boyle, M., Brunsdon, C., Charlton, M., Dorling, D., Foley, R., French, W., Noone, S., and Pringle, D. G. (2017) Towards a geography of health inequalities in Ireland. *Irish Geography*, 50(1), 37–58.
- Robeyns, I. (2020) Wellbeing, place and technology. *Wellbeing, Space and Society,* 1, 100013.
- Robeyns, I., and Byskov, M.F. (2021) The Capability Approach. In: Zalta, E.N. (ed.), *The Stanford Encyclopedia of Philosophy* (Fall 2021 Edition). Stanford: Metaphysics Research Lab, Stanford University. Available at: https://plato.stanford.edu/archives/fall2021/entries/capability-approach/ (Accessed 3rd May 2022).

Roe, J., and McCay, L. (2021) Restorative Cities. London: Bloomsbury.

- Rogers, D. L. (2017) Understanding Ageing Well from the Perspective of Homebound Older Adults. Masters Thesis, University of Texas, Arlington. Available at: <u>https://rc.library.uta.edu/uta-ir/handle/10106/26794</u> (Accessed 2nd November 2020).
- Rowe, J. W., and Kahn, R. L. (1997) Successful aging. *Gerontologist*, 37(4), 433–440.

- Rowles, G. (2018) Being in place, identity and place attachment in late life. In: Skinner, M., Andrews, G., and Cutchin, M. (eds.) *Geographical Gerontology: Concepts and Approaches*. London: Routledge, 203–215.
- Rowles, G. D. (1983) Place and personal identity in old age: Observations from Appalachia. *Journal of Environmental Psychology*, 3(4), 299–313.
- Russell, C., Hill, B., and Basser, M. (1998) Older people's lives in the inner city: Hazardous or rewarding? *Australian and New Zealand Journal of Public Health*, 22(1), 98–106.
- Ryan, J., and Wretstrand, A. (2019) What's mode got to do with it? Exploring the links between public transport and car access and opportunities for everyday activities among older people. *Travel Behaviour and Society*, 14, 107–118.
- Saldaña, J., Leavy, P., and Beretvas, N. (2011) *Fundamentals of Qualitative Research*. Oxford: Oxford University Press.
- Savva, G., Hanly, M., McDaid, O., Richardson, K., Kenny, R. A., and Kee, F. (2011) *Multimorbidity in the Older Population*. Dublin: Centre for Ageing Research and Development in Ireland.
- Scharf, T., Phillipson, C., and Smith, A. (2007) Aging in a difficult place: Assessing the impact of urban deprivation on older people. In: Wahl, H. W., Tesch-Roemer, C., and Hof, A. (eds.), *New Dynamics in Old Age: Individual, Environmental and Societal Perspectives*. Amityville, New York: Baywood, chapter 9.
- Scharf, T., Phillipson, C., Kingston, P., and Smith, A. E. (2001) Social exclusion and older people: Exploring the connections. *Education and Ageing*, 16(3), 303–320.
- Scott, J., Nolan, J., and Plagnol, A. C. (2009) Panel data and open–ended questions: Understanding perceptions of quality of life. *Twenty–First Century Society*, 4(2), 123–135.
- Schwanen, T., Banister, D., and Bowling, A. (2012) Independence and mobility in later life. *Geoforum*, 43(6), 1313–1322.
- Schwanen, T., and Páez, A. (2010) The mobility of older people: An introduction. Journal of Transport Geography, 18(5), 591–595.

- Schwanen, T. and Ziegler, F. (2011) Wellbeing, independence and mobility: An introduction. *Ageing and Society*, 31(5), 719–733.
- Seamon, D. (2018) Life Takes Place: Phenomenology, Lifeworlds, and Placemaking. London: Routledge.
- Sen, A. (1993) Capability and well-being. In: Nussbaum, M., and Sen, A. (eds.), *The Quality of Life*. Oxford: Oxford University Press, 30–53.
- Severinsen, C., Breheny, M., and Stephens, C. (2016) Ageing in unsuitable places. *Housing Studies*, 31(6), 714–728.

Sheehan, A., and O'Sullivan, R. (2020) Ageing and Public Health: An Overview of Key Statistics in Ireland and Northern Ireland. Dublin: Institute of Public Health. Available at: <u>https://publichealth.ie/wp-content/uploads/2020/04/20200416-</u> <u>AGEING-PUBLIC-HEALTH-</u> <u>MAIN.pdf#:~:text=In%202019%20one%20in%20seven%20people%20on%20th</u> <u>e,both%20jurisdictions%20will%20be%20aged%2060%2B%20by%202046</u>. (Accessed 7th January 2022).

- Silver, C., and Lewins, A. (2014) Using Software in Qualitative Research: A Stepby-Step Guide, second edition. London: Sage.
- Sixsmith, A. J., and Sixsmith, J. A. (1991) Transitions in home experience in later life. *Journal of Architectural and Planning Research*, 8(3), 181–191.
- Skinner, M. W., Cloutier, D., and Andrews, G. J. (2014) Geographies of ageing: Progress and possibilities after two decades of change. *Progress in Human Geography*, 39(6), 776–799.
- Smith, A. (2009) Ageing in Urban Neighbourhoods: Place Attachment and Social Exclusion. Bristol: Policy Press
- Smith, J. M. (2012) Toward a better understanding of loneliness in communitydwelling older adults. *The Journal of Psychology*, 146(3), 293–311.
- Smith, J. A., Braunack-Mayer, A., Wittert, G., and Warin, M. (2007) "I've been independent for so damn long!": Independence, masculinity and aging in a help seeking context. *Journal of Aging Studies*, 21(4), 325–335.

- Stalvey B. T., Owsley C., Sloane, M. E., and Ball, K. (1999) The life space questionnaire: A measure of the extent of mobility of older adults, *Journal of Applied Gerontology*, 18, 460–478.
- Stephens, C. (2016) From success to capability for healthy ageing: Shifting the lens to include all older people. *Critical Public Health*, 27(4), 490–498.
- Stephens, C., Breheny, M., and Mansvelt, J. (2015) Healthy ageing from the perspective of older people: A capability approach to resilience. *Psychology and Health*, 30(6), 715–731.
- Stjernborg, V., Wretstrand, A., and Tesfahuney, M. (2015) Everyday life mobilities of older persons: A case study of ageing in a suburban landscape in Sweden. *Mobilities*, 10(3), 383–401.
- Stones, D., and Gullifer, J. (2016) "At home it's just so much easier to be yourself": Older adults' perceptions of ageing in place. *Ageing and Society*, 36(3), 449–481.
- Sturge, J., Klaassens, M., Lager, D., Weitkamp, G., Vegter, D., and Meijering, L. (2021) Using the concept of activity space to understand the social health of older adults living with memory problems and dementia at home. *Social Science and Medicine*, 288(1–2):113208.
- Sugiyama, T., Ward-Thompson, C., and Alves, S. (2009) Associations between neighborhood open space attributes and quality of life for older people in Britain. *Environment and Behavior*, 41(1), 3–21.
- Sugiyama, T., and Ward-Thompson, C. (2007) Outdoor environments, activity and the well-being of older people: Conceptualising environmental support. *Environment and Planning A*, 39, 1943–1960.
- Tanner, D. (2010) Managing the Ageing Experience: Learning from Older People.Bristol: Policy Press.
- Tashakkori, A., and Teddlie, C. (2010) *Sage Handbook of Mixed Methods in Social and Behavioral Research*, second edition. Thousand Oaks, California: Sage.
- Tashakkori, A., and Teddlie, C. (1998) *Mixed Methodology: Combining Qualitative and Quantitative Approaches*. Thousand Oaks, California: Sage.

- Tate, R. B., Lah, L., and Cuddy, T. E. (2003) Definition of successful aging by elderly Canadian males: The Manitoba follow-up study. *The Gerontologist*, 43(5), 735–744.
- Taylor, A. F., Kuo, F. E., and Sullivan, W. C. (2002) Views of nature and selfdiscipline: Evidence from inner city children. *Journal of Environmental Psychology*, 22(1–2), 49–63.
- Teddlie, C., and Yu, F. (2007) Mixed methods sampling: A typology with examples. *Journal of Mixed Methods Research*, 1(1), 77–100.
- Thomas, W. and Blanchard, J. (2009) Moving Beyond Place: Aging in Community. *Generations*, 33(2), 12–17.
- Timonen, V., Kamiya, Y., and Maty, S. (2011) Social engagement of older people.
 In: Barrett, A., Savva, G., Timonen, V., and Kenny, R.A. (eds.), *Fifty Plus in Ireland 2011: First Results from The Irish Longitudinal Study on Ageing* (*TILDA*) [Online]. Dublin: Trinity College Dublin, chapter 4. Available at: <u>https://tilda.tcd.ie/publications/reports/pdf/w1–key–findings–</u> <u>report/Tilda_Master_First_Findings_Report.pdf</u> (Accessed 10th March 2022).
- Tsai, L.-T., Portegijs, E., Rantakokko, M., Viljanen, A., Saajanaho, M., Eronen, J., and Rantanen, T. (2015) Objective physical activity and life-space. *Scandinavian Journal of Medicine and Science in Sports*, 25: e368–e373
- Tsai, L.-T., Rantakokko, M., Rantanen, T., Viljanen, A., Kauppinen, M., and Portegijs, E. (2016) Objectively measured physical activity and changes in lifespace mobility among older people. *The Journals of Gerontology: Series A*, 71(11), 1466–1471.
- Tse, T., and Linsey, H. (2005) Adult day groups: Addressing older people's needs for activity and companionship. *Australasian Journal on Ageing*, 24(3), 134– 140.
- Tuan, Y.-F. (1977) Space and Place: The Perspective of Experience. London: Edward Arnold.
- Ulrich, R. S. (1984) View through a window may influence recovery from surgery. *Science*, 224(4647), 420–421.

- United Nations (1991) United Nations Principles for Older Persons. Available at: http://www.ohchr.org/EN/ProfessionalInterest/Pages/OlderPersons.aspx (Accessed 29th April 2022).
- Utz, R. L., Swenson, K. L., Caserta, M., Lund, D., and Devries, B. (2014) Feeling lonely versus being alone: Loneliness and social support among recently bereaved persons. *The Journals of Gerontology, Series B, Psychological Sciences and Social Sciences*, 69(1), 85–94.
- van Cauwenberg, J., Clarys, P., De Bourdeaudhuij, I., Ghekiere, A., De Geus, B., Owen, N., and Deforche, B. (2018) Environmental influences on older adults' transportation cycling experiences: A study using bike-along interviews. *Landscape and Urban Planning*, 169, 37–46.
- van Cauwenberg, J., van Holle, V., Simons, D., Deridder, R., Clarys, P., Goubert, L., Nasar, J., Salmon, J., De Bourdeaudhuij, I., and Deforche, B. (2012)
 Environmental factors influencing older adults' walking for transportation: A study using walk-along interviews. *International Journal of Behavioral Nutrition and Physical Activity*, 9(1), 85.
- van Eck, D. and Pijpers, R. (2017) Encounters in place ballet: A phenomenological perspective on older people's walking routines in an urban park. *Area*, 49(2), 166–173.
- van Hees, S., Horstman, K., Jansen, M., and Ruwaard, D. (2017) Photovoicing the neighbourhood: Understanding the situated meaning of intangible places for ageing-in-place. *Health and Place*, 48(Supplement C), 11–19.
- van Hoven, B., and Meijering, L. (2019) Mundane mobilities in later life: Exploring experiences of everyday trip-making by older adults in a Dutch urban neighbourhood. *Research in Transportation Business and Management*, 30, 100375.
- van Hoven, B., and Poelman, A. (2003) Using computers for qualitative data analysis: An example using NUD. IST. *Journal of Geography in Higher Education*, 27(1), 113–120.
- Venkatapuram, S. (2011) *Health Justice: An Argument from the Capabilities Approach.* Cambridge: Polity Press.

- Victor, C.R., Scrambler, S.J., Bond, J., and Bowling, A. (2004) Loneliness in later life. In: Hagan Hennessy, C., and Walker, A., (eds) *Growing Older: Quality of Life in Old Age: Extending Quality of Life*. Berkshire: Open University Press, chapter 6.
- von Faber, M., Bootsma-van Der Wiel, A., van Exel, E., Gussekloo, J., Lagaay, A. M., van Dongen, E., Knook, D. L., van der Geest, S., and Westendorp, R. G. (2001) Successful aging in the oldest old: Who can be characterized as successfully aged? *Archives of Internal Medicine*, 161(22), 2694–2700.
- Walsh, S. (2014) Salutogenesis 2.0: An Examination of Healthy Ageing-in-place Using a Qualitative Application of the Sense of Coherence. PhD Thesis, Maynooth University. Available at: <u>http://mural.maynoothuniversity.ie/6416/</u> (Accessed: 30th March 2021).
- Walsh, K., Scharf, T., and Keating, N. (2017) Social exclusion of older persons: A scoping review and conceptual framework. *European Journal of Ageing*, 14(1), 81–98.
- Weathers, E., McCarthy, G., and Coffey, A. (2016) Concept analysis of spirituality: An evolutionary approach. *Nursing Forum*, 51(2), 79–96.
- Webber, S. C., Porter, M. M., and Menec, V. H. (2010) Mobility in older adults: A comprehensive framework. *The Gerontologist*, 50(4), 443–450.
- Wenger, G. C., Davies, R., Shahtahmasebi, S., and Scott, A. (1996) Social isolation and loneliness in old age: Review and model refinement. *Ageing and Society*, 16(3), 333–358.
- White, A. M., and Groves, M. A. (1997) Interdependence and the aged stereotype. *Australian Journal on Ageing*, 16(2), 83–89.
- Wiles, J. L., and Jayasinha, R. (2013) Care for place: The contributions older people make to their communities. *Journal of Aging Studies*, 27(2), 93–101.
- Wiles, J. (2018) Health geographies of ageing. In: Skinner, M., Andrews, G., and Cutchin, M. (eds.), *Geographical Gerontology: Concepts and Approaches*. London: Routledge, chapter 3.

- Wiles, J. L., Leibing, A., Guberman, N., Reeve, J., and Allen, R. E. S. (2012) The meaning of "aging in place" to older people. *The Gerontologist*, 52(3), 357–366.
- Wiles, J. L., Allen, R. E. S., Palmer, A. J., Hayman, K. J., Keeling, S., and Kerse, N. (2009) Older people and their social spaces: A study of well-being and attachment to place in Aotearoa New Zealand. *Social Science and Medicine*, 68(4), 664–671.
- Williams, B., and Shiels, P. (2002) The expansion of Dublin and the policy implications of dispersal. *Journal of Irish Urban Studies*, 1(1), 1–20.
- Winston, N. (2007) From boom to bust? An assessment of the impact of sustainable development policies on housing in the Republic of Ireland. *Local Environment*, 12(1), 57–71.
- Woodall, J., Trigwell, J., Bunyan, A.-M., Raine, G., Eaton, V., Davis, J., Hancock,
 L., Cunningham, M. and Wilkinson, S. (2018) Understanding the effectiveness and mechanisms of a social prescribing service: A mixed method analysis. *BMC Health Services Research*, 18(1), 604, 1–12.
- World Health Organization. (2007a) *Global Age–friendly Cities: A Guide*. Geneva: WHO Press.
- World Health Organization. (2007b) Checklist of Essential Features of Age-friendly Cities. Geneva: WHO Press.
- World Health Organization (2002) *Active Ageing: A Policy Framework*. Geneva: World Health Organization.
- World Health Organization (1946) *Constitution of the World Health Organization*, New York: United Nations.
- Yarker, S. (2021) A research agenda for geographies of everyday intergenerational encounter. Area, 53(2), 264–271.
- Yen, I. H., Michael, Y. L., and Perdue, L. (2009) Neighborhood environment in studies of health of older adults: A systematic review. *American Journal of Preventative Medicine*, 37(5), 455–463.

- Yen, I. H., Shim, J. K., Martinez, A. D., and Barker, J. C. (2012) Older people and social connectedness: How place and activities keep people engaged. *Journal of Aging Research*, 2012, 139523.
- Young, J., Bowen-Salter, H., O'Dwyer, L., Stevens, K., Nottle, C., and Baker, A.
 (2020) A qualitative analysis of pets as suicide protection for older people. *Anthrozoös*, 33(2), 191–205.
- Zebhauser, A., Baumert, J., Emeny, R. T., Ronel, J., Peters, A., and Ladwig, K. H. (2015) What prevents old people living alone from feeling lonely? Findings from the KORA-Age-study. *Aging and Mental Health*, 19(9), 773–780.
- Zeitler, E., and Buys, L. (2014) Mobility and out-of-home activities of older people living in suburban environments: "Because I'm a driver, I don't have a problem". *Ageing and Society*, 35(4), 785–808.
- Zenk, S. N., Schulz, A. J., Matthews, S. A., Odoms-Young, A., Wilbur, J., Wegrzyn, L., Gibbs, K., Braunschweig, C., and Stokes, C. (2011) Activity space environment and dietary and physical activity behaviors: A pilot study. *Health* and Place, 17(5), 1150–1161.
- Ziegler, F., and Schwanen, T. I. M. (2011) "I like to go out to be energised by different people": An exploratory analysis of mobility and wellbeing in later life. *Ageing and Society*, 31(5), 758–781.

This page has been left intentionally blank.

Appendices

Appendix 1. Philosophical and Theoretical Underpinnings of the Thesis

Theoretical Underpinnings (from Chapters 1 and 2)

- Capability Approach (CA) lens to ageing well in place and quality of life emphasis on subjective valued beings and doings that only an older person themselves can define
- Recognition of diversity and heterogeneity of older adult population, including whether older person has health or mobility challenges
- Recognition of broader determinants of health including the built and natural environment and how certain environments are more 'supportive' for ageing in place.
- Importance of broader than home environment and 'getting out and about' for a good quality of life
- Focus of inquiry: working class suburban places 'newly aged'

Ontology

- Both objective and subjective ways of knowing (multiple realities)
- Rejection of a purely objective truth, instead recognition of the diversity of experience and interpretations.
- Emphasis on lived experience, lay person definitions, and subjective ways of knowing

Epistemology

- Pragmatism
- Interpretivism
- Recognition that truth is impartial, incomplete, dynamic, and contextually bounded; methods capture a moment in time.

Methodology

- Qualitative Research Paradigm 'Big Q'
- Emphasis on spatial approaches and 'in-situ' methods to capture qualitative details and experiences of place
- Informed by Qualitative GIS, recognising that spatial work can be Qualitative

Study Design

- Multi-staged and multi-method study combining qualitative and geo-spatial methods
- Flexible, 'care-full' and inclusive approach

Data Collection Methods

- Focus Groups
- Semi-structured Interviews

- Mapping Exercises
- 'Go-along' Interviews

Data Analysis Techniques

- Thematic Analysis
- Person-centred 'lifeworld' mapping and use of 'geo-narratives' to combine spatial and qualitative data

Appendix 2. Stage 1: List of Questions

Questions explored in the Focus Groups:

- What activities and interactions outside the home are important to you?
- Why are these activities important to you?
- Does your local area make it easy or difficult to do the things that are important to you?
- What do you think it means to live or to age well?
- How would you define quality of life?
- If you were researching this topic, what questions do you think might be important to ask, and who would you ask?
- Are there particular older adults that may be more or less restricted by particular environments?
- Are there issues that are of particular importance to people in Ireland?
- If you could tell policy makers or decision makers the things that would support you (or people you know), to live and age well, what would they be?

Appendix 3. Stage 2: Interview, Mapping Exercise and Go-Along Interview

Schedule

(*Page 1 of 3*)

	Interview Schedule
	Main Interview
	Main Interview
Intro	duction
	Have you ever done an interview before? Set out what is to be expected, can stop at any time, longer than a survey, interested in your perspectives – so how you really feel about thinks, not what you think I might want to hear. Could you start by telling me a bit about yourself? [Prompts if needed]: • Where are you from originally? • If attending social group, how long have you been attending this group?
Indiv	idual Interests / Hobbies
:	What do you like to do? What are your hobbies and interests? What things are important to you? What are your hopes and aspirations for the future?
Local	Environment / Neighbourhood
:	How long have you lived here? Have you lived anywhere else? Could you tell me a bit about your neighbourhood? Can you identify some things that you like about your neighbourhood? Can you identify some things that you dislike about your neighbourhood? Do you think your neighbourhood has changed over time? If so, how?
Enga	gement with the Local Environment
•	I would like you to think about a typical day in the last week – where did you go and what did you do?
	How important is it to you to get outside the house?
	How motivated do you think you are to leave the house? How much of your time do you think you spent outside of the home?
	How often would you leave the house?
	Are there particular times during the day when this happens?
	How do you think this might have changed from, say, five years ago?
	Have you had to adapt your behaviour at all during this time for any reason?
•	Do you feel that you are ever limited by what you can do by your health?
•	Are there any places that you might like to go but for some reason you are unable to?
•	How often would you talk with people outside the home?
•	Who would this be with? E.g. friends / family / neighbours / strangers?
•	Where would this happen?

(*Page 2 of 3*)

Introd	uce participant to map of their local area
:	I would like you to identify particular places that are important to you I would also like you to identify the places you have visited in the last month? Which of these places are most important to you and why? Are there any additional places that you would like go to but would find it difficult to get to? [<i>If yes</i>] - Why is this? Are there any places in your local environment that you dislike? Are there any places that you might feel uncomfortable and would avoid?
Forea	ach location:
	Why is it important to you? How do you get there? Was it easy or difficult to get there / Why could you not get there? Are there specific reasons for this? What route do you take to get there – please could you draw on the map How long are you there for? How often would you visit? What time of day is this?
Activi	ty Space Boundary
:	Please draw a boundary on the map to identify where you have been in the last month To what extent is this a typical month? Are there particular things that might have influenced the amount you get out and about? • E.g. Weather / Season / Ill-health? How do you think this boundary might have compared to say, five years ago?
Impro	oving the Supportiveness of Local Environment
	Do you think your local environment supports you to do those things that are most important to you? How would you define a good quality of life? If you could tell policy makers or decision makers the things that would support you (or people you know), to live and age well, either now or in the future, what would it be?
Concl	uding Questions
	sn't come up in main interview]: Age? Member of social groups in past or now? Live alone or with people? Would you spend most of time with people or alone? Housing / accommodation type? How long have you lived here? Do you own, or have you ever owned a car? Or does your partner? Does your health ever limit what you would like to do? Green spaces – what do you think of them? Do you use them at all? Weather – does the weather ever influence you getting out and about? Neighbours – how often would you talk to them, what is your relationship with them? Friends / family – do you have people that you can count on if you needed to? Do they live close by?
have b driving	for taking part in interview, ask if they would also be willing to show me some of the areas you een talking about in another short interview 'on the move'. This could be a walking interview or interview depending on your preference. ask about further recruitment – do they know of people who may be interested in taking part.

(*Page 3 of 3*)



Appendix 4. Stage 1: Social Group Details

This table includes details about the community groups that participated in Stage 1: Exploratory Focus Groups.

Focus Groups (n=31)	Details about the Community Group
Focus Groups 1 and 3: Day Centre Group FG1: n=8 FG3: n=8	Run by the Health Service Executive (HSE) in Ireland, older people with health and mobility challenges living in the community were referred to this centre. Members attended at least weekly for a full day of activities, including art projects such as weaving, singing, card games, bingo, exercises, tea and lunch. The centre also had options to book appointments with health specialists and hairdressers on site.
Focus Group 2: Arthritis Walking Group FG2: n=6	Members attended a weekly meet up where they carried out seated class exercises and warm up, before walking (weather permitting) around the scenic grounds of a publicly accessible estate. Afterwards, members had tea and biscuits back at the community space and took part in line dancing.
Focus Group 4: University of the Third Age Group (U3A) FG4: n= 9	Members met weekly with the aim of enhancing social and educational opportunities. They organised and attended a range of activities and trips, such as visits to gardens, museums and historic houses.

INVITATION TO PARTICIPATE IN A GROUP DISCUSSION EXPLORING HOW OUR LOCAL ENVIRONMENT INFLUENCES EVERYDAY LIFE

I am a PhD student interested in how <u>where we live</u> influences <u>how we live</u>. Our local environment can play a big part in how we go about our daily lives and whether we can do those activities that are most meaningful to us, and therefore contribute to our overall quality of life.

In particular, I am interested in finding out environmental features are important to people and how this changes over time. I would then like to explore how local environments can meet every day needs and wants.

This is the first stage of my research and I have been making contact with community groups, as I am looking for local perspectives to help inform the direction my research will take. Your advice and input is highly valued, as you are the experts of your local area. I believe it is very important that I get advice about how to research this topic from people that have experienced this first hand within an Irish context.

I would therefore like to invite you to take part in a group discussion (approx. 6-8 people). The discussion will last no longer than one hour and will be a one off event.

I will ask you a number of questions about what you think it means to "live well" and how our environment can at times help or hinder our ability to do this. I would like to know what characteristics of your local area are most important to you and what a supportive or unsupportive environment might look like. We will then have a discussion about how we can best research this. For example, what questions do we need to ask and what methods could we use to make sure that we understand how different people interact with, and are influenced by their local environment?

If you are interested in finding out more, please email or telephone me. I will be holding several of these group discussions as I am looking for a range of perspectives, so please share this with anyone that you think might also be interested.

Thank you, Hannah Grove | Email: <u>Hannah.Grove@nuim.ie</u> | Tel: 083 865 5063.



(*Page 1 of 2*)



of any of these future outcomes if you would like to provide me with your contact details.

How will information you provide be recorded, stored and protected?

The material and data from the workshop will be stored securely. Group discussions will be digitally recorded and transcribed. False names will be used to make sure that no-one can be identified. Any personal information such as contact details and consent sheets will be stored separately in a locked office. It will be retained for ten years and will then be destroyed by confidential shredding.

Funding Information:

This research was funded by the Health Research Board SPHeRE/2013/1. The Health Research Board (HRB) supports excellent research that improves people's health, patient care and health service delivery. The HRB aim to ensure that new knowledge is created and then used in policy and practice. In doing so, it supports health system innovation and creates new enterprise opportunities.

Who should you contact for further information?

You may contact me at via email at <u>hannah.grove.2016@mumail.ie</u> or via post at Geography Department, Rhetoric House, Maynooth, Co. Kildare, Ireland.

If you experience any concerns for yourself or others following participation in this discussion, the organisation below can offer support: <u>ALONE:</u> working with older people who are in crisis situations and need extra support to age at home. <u>Contact details</u>: Telephone: 01 679 1032 Email: <u>enquiries@alone.ie</u> Address: ALONE, Olympic House, Pleasants Street, Dublin 8

If during your participation in this study you feel the information and guidelines that you were given have been neglected or disregarded in any way, or if you are unhappy about the process, please contact the Secretary of the Maynooth University Ethics Committee at <u>research ethics@nuim.ie</u> or +353 (0)1 708 6019. Please be assured that your concerns will be dealt with in a sensitive manner.

Limits to confidentiality:

It must be recognized that, in some circumstances, confidentiality of research data and records may be overridden by courts in the event of litigation or in the course of investigation by lawful authority. In such circumstances the University will take all reasonable steps within law to ensure that confidentiality is maintained to the greatest possible extent.

Thank you for taking the time to read this information sheet,

Hannah Grove

Appendix 7. Stage 1: Consent Form

г

	provided and agree to participate in the I may withdraw my consent at any time.
Name (printed)	Name (signature)
2) I would like to be kept informed of	the research project: Yes □ No □
Please provide your contact details an if you have selected yes to 2):	nd state your preferred method of conta
Email Address:	
Address:	
Telephone:	
	ail 🗆 Address 🗆 Telephone
Preferred Method of Contact: Ema	
Preferred Method of Contact: Ema <u>My Contact Information:</u> Hannah Grove Email: <u>Hannah.Grove.2016@mumail.i</u> Address: Department of Geography, Rhetoric House, Maynooth University, Co. Kildare, Ireland	Address: Department of Geogr

Appendix 8. Stage 2: Social Group Details

This table includes details about the community centres and social groups that participants were recruited from.

Study Area	Community Centre Type and Details About Specific Group	Participants
SA1	<u>Community Centre</u>Older Adult retirement group involving bingo	3
SA1	 <u>Dementia Day Care Centre</u> Day group for people with dementia including activities, lunch, a garden, and outings 	2
SA1	Community Centre • Women's Craft Group • Men's Group	4 (2 from each group)
SA1	<u>Family Resource Centre</u>Public Speaking Group	1
SA1	 <u>GAA Club (Irish Sports Club)</u> Active Ageing Retirement Group involving a range of activities, trips away and outings 	5
SA2	<u>Community Centre</u>Active Ageing Exercise Group	5
SA2	 <u>Senior Citizens Centre</u> Social group for 'socially isolated' older people including activities such as Yoga and Bingo 	8
SA2	Community Centre • Bingo afternoon • Film Club afternoon	3
SA2	 <u>Resource Centre</u> Senior centre where a variety of activities for older adults are provided 	112
		$Total = 32^{13}$

¹² Participant volunteers at the centre but was recruited through the older adult group.

¹³ Two additional participants were not recruited through community or social group (one was via a Public Health Nurse and the other was through the sheltered housing complex they lived in).



(*Page 1 of 2*)



What will I do with the information?

Your feedback will be used to inform the local 'Age Friendly Environment' evidence base, and will be presented to policymakers and service providers. The results will be published in my PhD thesis, along with scholarly publications and academic presentations.

How will information you provide be recorded, stored and protected?

The material and data from the workshop will be stored securely. Interviews and goalong interviews will be digitally recorded and transcribed. False names will be used to make sure that you cannot be identified. Any personal information such as contact details and consent sheets will be stored in a locked office. It will be retained for ten years and will then be destroyed by confidential shredding.

Funding Information:

This research was funded by the Health Research Board (SPHeRE/2013/1). The Health Research Board (HRB) supports excellent research that improves people's health, patient care and health service delivery. The HRB aim to ensure that new knowledge is created and then used in policy and practice. In doing so, it supports health system innovation and creates new enterprise opportunities.

Who should you contact for further information?

You may contact me at via email at <u>hannah.grove@mu.ie</u> or via post at: Geography Department, Rhetoric House, Maynooth University, Co. Kildare, Ireland.

If you experience any concerns for yourself or others following participation in this discussion, the organisation below can offer support: <u>ALONE:</u> working with older people who are in crisis situations and need extra support to age at home. <u>Contact details:</u> Telephone: 01 679 1032 Email: <u>enquiries@alone.ie</u> Address: ALONE, Olympic House, Pleasants Street, Dublin 8

If during your participation in this study you feel the information and guidelines that you were given have been neglected or disregarded in any way, or if you are unhappy about the process, please contact the Secretary of the Maynooth University Ethics Committee at <u>research.ethics@nuim.ie</u> or +353 (0)1 708 6019. Please be assured that your concerns will be dealt with in a sensitive manner.

Limits to confidentiality:

It must be recognized that, in some circumstances, confidentiality of research data and records may be overridden by courts in the event of litigation or in the course of investigation by lawful authority. In such circumstances the University will take all reasonable steps within law to ensure that confidentiality is maintained to the greatest possible extent.

Thank you for taking the time to read this information sheet.

Appendix 11. Stage 2: List of organisations that did not take part and reasons for this

This table provides a details about the organisations that did not take part and reasons for this.

Organisation and details	Summary of contact and reasons for not participating in research			
Stage 1: Exploratory Focus Groups				
Older Adult Social Group	I contacted the organiser for this social group and met several times with attendees, however they declined to take part in the Focus Groups.			
Men's Shed Group	I called the organiser of this group several times but did not receive any replies.			
Stage 2: Main Study				
ALONE (SA1 and SA2) A national charity that provides services to enable older people to age at home and remain connected to their communities	I sent a letter to the headquarters of the organisation and made several phone calls inviting ALONE and contacts within ALONE to take part in this research project. However, I did not receive any replies.			
Travellers Community Development Project (SA1) This is a branch within SA1 of the broader 'Pavee Point Traveller and Roma Centre', which is a national non- governmental organisation committed to the attainment of human rights for Irish Travellers and Roma.	I spoke with one of the leaders of this community group and met with her to discuss taking part. Whilst she was interested in the project and we discussed how we could go about this, at the time of the research, the community had experienced several young male suicides and were in a state of shock and grief. The leader stated that it was not possible to take part at this time and it would not be possible for me to visit their sites.			
<u>Men's' Shed Group (SA1)</u>	I spoke with the leader of this group and met them and attendees in person and they were very interested in taking part at that time. However, on a later date when I invited them to participate, they stated that they no longer wished to. They mentioned that they had also been contacted several times by other researchers and were feeling over-recruited.			

<u>'Meals on Wheels' Contact</u> (SA1) A service that provides cooked dinners to older adults, particularly those who may be socially isolated.	I sent several emails to the lead contact for this and did not receive any replies.
Older Adult Day Group (SA2) A centre which provides both meals and activities for older people living in the community	I contacted the organiser of this group and agreed to meet on a specific date. However, when I arrived, they had forgotten about our meeting. The group was then due to take a break over the summer when recruitment was taking place.
Public Health Nurses (SA1 and SA2)	I dropped letters into the Public Health Nurses in 5 Primary Care Centres (4 in SA1 and 1 in SA2) and called them multiple times. I heard back from one Public Health Nurse. This resulted in the recruitment of one participant in SA2.

Appendix 12. Stage 1: Ethical Approval Letter


Appendix 13. Stage 2: Ethical Approval Letter



 I have read the information sheet pro group discussion. I understand that I r 	ovided and agree to participate in the nay withdraw my consent at any time.
Name (printed)	Name (signature)
2) I would like to be kept informed of the	e research project: Yes □ No □
Please provide your contact details and if you have selected yes to 2):	state your preferred method of contact
Email Address:	
Address:	
Telephone:	
Preferred Method of Contact: Email	Address D Telephone D
<u>My Contact Information:</u> Hannah Grove Email: <u>Hannah.Grove.2016@mumail.ie</u> Address: Department of Geography, Rhetoric House, Maynooth University, Co. Kildare, Ireland	My Supervisor's Contact Information Professor Jan Rigby Email: <u>Jan.Rigby@nuim.ie</u> Address: Department of Geography, Room 2.20, Iontas Building, Maynoo University, Co. Kildare, Ireland

Appendix 15. Stage 2: Consent Form (Adapted for Cognitive Impairment)

(*Page 1 of 7*)



(*Page 2 of 7*)



(*Page 3 of 7*)



(*Page 4 of 7*)



(*Page 5 of 7*)



(*Page 6 of 7*)





	the research project results D
Please provide your contact details and	select your preferred method of contact
if you have selected yes to 3):	
By post □	
By telephone □	
By telephone □ By email □	
By telephone □ By email □ My Contact Information:	My Supervisor's Contact Information
By telephone □ By email □ <u>My Contact Information:</u> Hannah Grove	My Supervisor's Contact Information Professor Jan Rigby
By telephone □ By email □ My Contact Information:	My Supervisor's Contact Information Professor Jan Rigby Email: Jan.Rigby@mu.ie
By telephone By email My Contact Information: Hannah Grove Email: Hannah.Grove@mu.ie Address: Department of Geography, Rhetoric House, Maynooth University,	My Supervisor's Contact Information Professor Jan Rigby Email: Jan.Rigby@mu.ie Address: Department of Geography, Room 2.20, Iontas Building, Maynoo
By telephone By email My Contact Information: Hannah Grove Email: Hannah.Grove@mu.ie Address: Department of Geography,	My Supervisor's Contact Information Professor Jan Rigby Email: Jan.Rigby@mu.ie Address: Department of Geography,
By telephone By email My Contact Information: Hannah Grove Email: Hannah.Grove@mu.ie Address: Department of Geography, Rhetoric House, Maynooth University, Co. Kildare, Ireland	My Supervisor's Contact Information Professor Jan Rigby Email: Jan.Rigby@mu.ie Address: Department of Geography, Room 2.20, Iontas Building, Maynoo University, Co. Kildare, Ireland

This page has been left intentionally blank.



Appendix 16. Chapter 5 Data Analysis: Thematic Map for Defining a Good Quality of Life

Appendix 17. Chapter 6 Data Analysis: Categories of Social Infrastructure Places by Participant Case

This table shows the different categories of places identified for Chapter 6 and summarises the number of references for each place, as well as how many participants mentioned each place.

(*Page 1 of 6*)

	Category of Place	Main Activity	Code	Study Area	Number of references		Number of participants that mentioned the place
<u>1</u>	Main Community Centre		MCC				
1	Community Centre	Attending Social Group or Class (Active Retirement Group)	MCC-01	SA1	24		3
1	Day Care Centre (for PwD)	Attending Social Group or Class (Dementia Day Care Social Group)	MCC-02	SA1	4		2
1	Community Centre	Attending Social Group or Class (Women's Crochet; Men's Group)	MCC-03	SA1	57	135	4
1	Resource Centre	Attending Social Group or Class (Public Speaking Group)	MCC-04	SA1	13		1
1	Local GAA Club	Attending Social Group or Class (Active Retirement Group)	MCC-05	SA1	37		5
1	Community Hall	Attending Social Group or Class (Active Retirement Group)	MCC-06	SA2	34		5
1	Senior Citizen's Centre	Attending Social Group or Class (Bingo; Yoga)	MCC-07	SA2	11	77	8
1	Community Centre	Attending Social Group or Class (Bingo; Film Nights)	MCC-08	SA2	16	//	3
1	Resource Centre	Attending Social Group or Class (Volunteering)	MCC-09	SA2	16		5
				Total =	212		
<u>2</u>	Other 'Community Centres'		<u>0CC</u>				
2	Contact Centre	Attending Social Group or Class (Attending Church Meetings)	OCC-01	SA1	3		1
2	Community Centre	Now Closed	OCC-02	SA1	2		2
2	Community Centre	Attending Social Group or Class (Attending Computer Classes)	OCC-03	SA1	6		2
2	Shopping Centre	Attending Social Group or Class (Attending Knitting Group)	OCC-04	SA1	5		1
2	Pub	Attending Social Group or Class (Attending Bingo Evenings)	OCC-05	SA1	8	53	2
2	Resource Centre	Attending Social Group or Class (Attending Friendship Club)	OCC-06	SA1	2	55	1
2	Pub	Attending Social Group or Class (Attending Bingo Evenints)	OCC-07	SA1	4		2
2	Arts Centre	Attending Social Group or Class (Attending COPD Prevention Class; Writing Group; Choir)	OCC-08	SA1	10		2
2	Leisure Centre	Attending Social Group or Class (Attending Exercise Classes)	OCC-09	SA1	6		2
2	Church	Attending Social Group or Class (Attending Musical Group)	OCC-10	SA1	7		1
2	Global Company Offices	Attending Social Group or Class (Attending Older Adult Special Events)	OCC-11	SA2	11		7
2	Social Centre	Attending Social Group or Class (Attending Bingo Evenints)	OCC-12	SA2	1		1
2	Leisure Centre	Attending Social Group or Class (Attending Exercise Classes)	OCC-13	SA2	3	23	3
2	Leisure Centre	Attending Social Group or Class (Attending Badminton Class; Swimming)	OCC-14	SA2	6		2
2	Pub	Attending Social Group or Class (Attending Book Club)	OCC-15	SA2	2		1
2	Outdoor Based Group	Attending Social Group or Class (Attending Hiking & Walking Group)	OCC-16	OSA	3		1
2	Gym	Attending Social Group or Class (Attending Exercise Classes)	OCC-17	OSA	2	10	1
2	Gym	Attending Social Group or Class (Doing Weightlifting)	OCC-18	OSA	5		1
				Total =	86		

(Page 2 of 6)

<u>3</u>	Shops		<u>SHO</u>				
3	Supermarket	Shopping	SHO-01	SA1	9		4
3	DIY Store	Shopping	SHO-02	SA1	4		2
3	Supermarket	Shopping	SHO-03	SA1	1		1
3	Charity Shop	Shopping	SHO-04	SA1	1		1
3	Supermarket	Shopping	SHO-05	SA1	12		4
3	Supermarket	Shopping	SHO-06	SA1	1		1
3	Supermarket	Shopping	SHO-07	SA1	21		6
3	Local Shop	Shopping	SHO-08	SA1	6	171	1
3	Local Shopping Parade	Shopping	SHO-09	SA1	6		1
3	Local Shop	Shopping	SHO-10	SA1	10		1
3	Local Shopping Parade	Shopping	SHO-11	SA1	11		4
3	Local Shopping Parade	Shopping	SHO-12	SA1	8		4
3	Supermarket	Shopping	SHO-13	SA1	2		1
3	Large Shopping Centre	Shopping	SHO-14	SA1	77		14
3	Small Shopping Centre	Shopping	SHO-15	SA1	2		2
3	Charity Shop	Shopping	SHO-16	SA2	1		1
3	Local Shop	Shopping	SHO-17	SA2	3	_	1
3	Local Shop	Shopping	SHO-18	SA2	1		1
3	Artisan Supermarket	Shopping	SHO-19	SA2	2		1
3	Local Shop	Shopping	SHO-20	SA2	1		1
3	Local Shop	Shopping	SHO-20	SA2	1		1
3	Local Shop	Shopping	SHO-21	SA2	1	40	1
-	Local Shop		SHO-22 SHO-23	SA2			1
3		Shopping	SHO-23	SA2	1 7		5
-	Supermarket	Shopping					
3	Supermarket	Shopping	SHO-25	SA2	8		3
3	Local Shopping Parade	Shopping	SHO-26	SA2	1		1
3	Local Shopping Parade	Shopping	SHO-27	SA2	13		5
3	Fishing Tackle Shop	Shopping	 SHO-28	OSA	3		1
3	Supermarket	Shopping	SHO-29	OSA	3		3
3	Department Store	Shopping	SHO-30	OSA	4		3
3	Large Shopping Centre	Shopping	SHO-31	OSA	1		1
3	City Centre Shopping	Shopping	SHO-32	OSA	43		19
3	Large Shopping Centre	Shopping	SHO-33	OSA	4		2
3	Bookshop	Shopping	SHO-34	OSA	3		3
3	Shopping Street	Shopping	SHO-35	OSA	3		3
3	Shopping Centre	Shopping	SHO-36	OSA	2	90	2
3	Shopping Centre	Shopping	SHO-37	OSA	2	50	1
3	Supermarket	Shopping	SHO-38	OSA	3		2
3	Large Shopping Centre	Shopping	SHO-39	OSA	7		6
3	Market	Shopping	SHO-40	OSA	3		1
3	Shopping Centre	Shopping	SHO-41	OSA	2		2
3	Shopping Street	Shopping	SHO-42	OSA	3		3
3	Clothing Shop	Shopping	SHO-43	OSA	2	_	2
3	Fishing Tackle Shop	Shopping	SHO-44	OSA	1	_	1
3	Wool Shop	Now Closed	SHO-45	OSA	1		1
	Is			Total =	301		

(Page 3 of 6)

<u>4</u>	Post Offices		POF				
4	Post Office	Collecting Pension	POF-01	SA1	7		1
4	Post Office	Collecting Pension	POF-02	SA1	3	14	1
4	Post Office	Collecting Pension	POF-03	SA1	2	14	1
4	Post Office	Collecting Pension	POF-04	SA1	2		1
4	Post Office	Collecting Pension	POF-05	SA2	2		2
4	Post Office	Collecting Pension	POF-06	SA2	1	9	1
4	Post Office	Now Closed	POF-07	SA2	3		2
4	Post Office	Now Closed	POF-08	SA2	3		3
				Total =	23		
<u>5</u>	Banks & Credit Unions		<u>BCU</u>				
5	Credit Union	Doing Banking	BCU-01	SA1	4	4	1
5	Bank	Now Closed	BCU-02	SA2	2		1
5	Bank	Now Closed	BCU-03	SA2	1	3	1
5	Bank	Doing Banking	BCU-04	OSA	1	,	1
5	Bank	Doing Banking	BCU-05	OSA	2	3	1
				Total =	10		
<u>6</u>	<u>Hairdressers</u>		HAI				
6	Hairdressers	Getting a Haircut	HAI-01	SA1	1		1
6	Hairdressers	Getting a Haircut	HAI-02	SA1	1	3	1
6	Hairdressers	Getting a Haircut	HAI-03	SA1	1		1
6	Hairdressers	Getting a Haircut	HAI-04	SA2	1	1	1
				Total =	4		
<u>7</u>	Health-Related		HRE				
7	Hospital	Attending Hospital Appointment	HRE-01	SA1	24		9
7	Pharmacy	Collecting Prescription	HRE-02	SA1	1		1
7	Health Centre	Attending Doctor's Appointment	HRE-03	SA1	1		1
7	Pharmacy	Collecting Prescription	HRE-04	SA1	1		1
7	Health Centre	Attending Doctor's Appointment	HRE-05	SA1	1	58	1
7	Doctor's Surgery	Attending Doctor's Appointment	HRE-06	SA1	10	50	2
7	Medical Centre	Attending Doctor's Appointment	HRE-07	SA1	8		2
7	Pharmacy	Collecting Prescription	HRE-08	SA1	1		1
7	Medical Centre	Attending Doctor's Appointment	HRE-09	SA1	7		1
7	Medical Centre	Attending Doctor's Appointment	HRE-10	SA1	4		2
7	Medical Centre	Attending Doctor's Appointment	HRE-11	SA2	1		1
7	Primary Care Centre	Attending Doctor's Appointment	HRE-12	SA2	3	5	3
7	Pharmacy	Collecting Prescription	HRE-13	SA2	1		1
7	Hospital	Attending Hospital Appointment	HRE-14	OSA	4		1
7	Hospital	Attending Hospital Appointment	HRE-15	OSA	1	17	1
7	Hospital	Attending Hospital Appointment	HRE-16	OSA	3	±/	3
7	Hospital	Attending Hospital Appointment	HRE-17	OSA	9		5
				Total =	80		

	-						
<u>8</u>	Green & Blue Spaces		GBS				
8	Local Green Space or Park	Engaging with Nature	GBS-01	SA1	6		1
8	Waterway	Engaging with Nature	GBS-02	SA1	3		2
8	Local Green Space or Park	Engaging with Nature	GBS-03	SA1	30		7
8	Local Woods	Engaging with Nature	GBS-04	SA1	2	74	1
8	Local Green Space or Park	Engaging with Nature	GBS-05	SA1	16	/ 4	3
8	Local Green Space or Park	Engaging with Nature	GBS-06	SA1	6		2
8	Local Green Space or Park	Engaging with Nature	GBS-07	SA1	8		2
8	Fairy Tree	Engaging with Nature	GBS-08	SA1	3		1
8	Allotment	Engaging with Nature	GBS-09	SA2	5		1
8	Waterway	Engaging with Nature	GBS-10	SA2	10		5
8	Waterway Path	Engaging with Nature	GBS-11	SA2	23		5
8	Dublin Bay	Engaging with Nature	GBS-12	SA2	2		2
8	Waterway	Engaging with Nature	GBS-13	SA2	13		6
8	Local Green Space or Park	Engaging with Nature	GBS-14	SA2	7		6
8	Naure Park	Engaging with Nature	GBS-15	SA2	15		6
8	Beach	Engaging with Nature	GBS-16	SA2	4	207	1
8	Local Green Space or Park	Engaging with Nature	GBS-17	SA2	43	207	9
8	Waterway	Engaging with Nature	GBS-18	SA2	13		6
8	Beach	Engaging with Nature	GBS-19	SA2	16		6
8	Coastal Path	Engaging with Nature	GBS-20	SA2	12		8
8	Village Green	Engaging with Nature	GBS-21	SA2	12		5
8	Local Green Space or Park	Engaging with Nature	GBS-22	SA2	12		6
8	Beach	Engaging with Nature	GBS-23	SA2	8		5
8	Coastal Path	Engaging with Nature	GBS-24	SA2	12		4
8	Allotment	Engaging with Nature	GBS-25	OSA	7		1
8	Rural Village	Engaging with Nature	GBS-26	OSA	15		3
8	Pitch & Putt Course	Engaging with Nature	GBS-27	OSA	6		1
8	Harbour	Engaging with Nature	GBS-28	OSA	2		1
8	Mountains	Engaging with Nature	GBS-29	OSA	29	72	8
8	Local Green Space or Park	Engaging with Nature	GBS-30	OSA	6		2
8	Large Green Space or Park	Engaging with Nature	GBS-31	OSA	1		1
8	Large Green Space or Park	Engaging with Nature	GBS-32	OSA	4		3
8	Local Green Space or Park	Engaging with Nature	GBS-33	OSA	2		1
				Total =	353		

(*Page 4 of 6*)

(Page 5 of 6)

<u>9</u>	Trip Destinations		TRI			
9	Athlone	Attending Day Trip or Trips Away	TRI-01	OSA	2	
9	Beara Peninsula	Attending Day Trip or Trips Away	TRI-02	OSA	1	
9	Belfast	Attending Day Trip or Trips Away	TRI-03	OSA	3	
9	Bray	Attending Day Trip or Trips Away	TRI-04	OSA	3	
9	Bundoran	Attending Day Trip or Trips Away	TRI-05	OSA	1	
9	Cyprus	Attending Day Trip or Trips Away	TRI-06	OSA	1	
9	Donabate	Attending Day Trip or Trips Away	TRI-07	OSA	1	
9	Dun Laoghaire	Attending Day Trip or Trips Away	TRI-08	OSA	9	
9	Galway	Attending Day Trip or Trips Away	TRI-09	OSA	1	
9	Giant's Causeway	Attending Day Trip or Trips Away	TRI-10	OSA	1	
9	Howth	Attending Day Trip or Trips Away	TRI-11	OSA	7	6
9	Killarney	Attending Day Trip or Trips Away	TRI-12	OSA	3	
9	Lourdes	Attending Day Trip or Trips Away	TRI-13	OSA	3	
9	Malahide	Attending Day Trip or Trips Away	TRI-14	OSA	4	
9	Portmarnock	Attending Day Trip or Trips Away	TRI-15	OSA	1	
9	Skerries	Attending Day Trip or Trips Away	TRI-16	OSA	1	
9	Spain	Attending Day Trip or Trips Away	TRI-17	OSA	2	
9	Waterford	Attending Day Trip or Trips Away	TRI-18	OSA	3	
9	Westport	Attending Day Trip or Trips Away	TRI-19	OSA	1	
9	Wexford	Attending Day Trip or Trips Away	TRI-20	OSA	10	
9	Wicklow	Attending Day Trip or Trips Away	TRI-21	OSA	3	
				Total =	61	
10	Eating & Drinking Places		<u>EDP</u>			
10	Café	Eating or Drinking	EDP-01	SA1	1	
10	Café	Eating or Drinking	EDP-02	SA1	1	
10	Café	Eating or Drinking	EDP-03	SA1	4	
10	Pub	Eating or Drinking	EDP-04	SA1	8	24
10	Pub	Eating or Drinking	EDP-05	SA1	4	2.
10	Café	Eating or Drinking	EDP-06	SA1	2	
10	Pub	Eating or Drinking	EDP-07	SA1	2	
10	Café	Eating or Drinking	EDP-08	SA1	2	
10	Pub	Eating or Drinking	EDP-09	SA2	1	
10	Hotel with Swimming Pool	Eating or Drinking	EDP-10	SA2	2	
10	Restaurant	Eating or Drinking	EDP-11	SA2	1	
10	Café	Eating or Drinking	EDP-12	SA2	2	
10	Café	Eating or Drinking	EDP-13	SA2	1	
10	Betting Shop	Eating or Drinking	EDP-14	SA2	2	1
10	Restaurant	Eating or Drinking	EDP-15	SA2	1	
10	Café	Eating or Drinking	EDP-16	SA2	3	
10	Pub	Eating or Drinking	EDP-17	SA2	1	
10	Hotel	Eating or Drinking	EDP-18	SA2	4	
10	Bar	Eating or Drinking	EDP-19	SA2	1	
10	Café	Eating or Drinking	EDP-20	OSA	2	
10	Café	Eating or Drinking	EDP-21	OSA	1	5
10	Café	Eating or Drinking	EDP-22	OSA	1	
10	Garden Centre	Eating or Drinking	EDP-23	OSA	1	
				Total =	48	

	2
	1
	3
	3
	1
	1
	1
	7
	1
	1
61	6
	3
	2
	3
	1
	1
	2
	2
	1
	7
	2
	1
	1
	1
24	2
	2
	1
	2
	1
	1
	1
	1
	1
	1
19	1
	1
	1
	1
	1
	1
	2
5	1
	1
	1
	1

(Page	6	of	6)	
-------	---	----	----	--

<u>11</u>	Cultural Buildings		CUB			_	
11	Theatre	Going to the Theatre	CUB-01	SA1	8		4
11	Cinema	Going to the Cinema	CUB-02	SA1	3	22	1
11	Bingo Hall	Going to a Bingo Night (mentions but dislikes and does not attend)	CUB-03	SA1	1	22	1
11	Arts Centre	Various Activities / Events	CUB-04	SA1	10		3
11	Cinema - Closed	Going to the Cinema	CUB-05	SA2	2		1
11	Cinema - Closed	Going to the Cinema	CUB-06	SA2	2	4	2
11	Concert Venue	Going to a Concert	CUB-07	OSA	3		1
11	Theatre	Going to the Theatre	CUB-08	OSA	9		7
11	Tourist Attraction	Visiting a Tourist Attraction	CUB-09	OSA	2		1
11	Theatre	Going to the Theatre	CUB-10	OSA	2	39	1
11	Cinema	Going to the Cinema	CUB-11	OSA	2		2
11	Tourist Attraction	Visiting a Tourist Attraction	CUB-12	OSA	3		2
11	Concert Venue	Going to a Concert	CUB-13	OSA	3		3
11	Art Gallery	Visiting an Art Gallery	CUB-14	OSA	1		1
11	Arena	Going to a Concert	CUB-15	OSA	5		3
11	College	Various Activities / Events	CUB-16	OSA	9		5
				Total =	65		
<u>12</u>	Churches		<u>CHU</u>				
12	Church	Attending Mass	CHU-01	SA1	2		1
12	Church	Attending Mass	CHU-02	SA1	1		1
12	Church	Attending Mass	CHU-03	SA1	2		1
12	Church	Attending Mass	CHU-04	SA1	15	45	5
12	Church	Attending Mass	CHU-05	SA1	4	45	1
12	Church	Attending Mass	CHU-06	SA1	2		2
12	Church	Attending Mass	CHU-07	SA1	1		1
12	Church	Attending Mass	CHU-08	SA1	18		5
12	Church	Attending Mass	CHU-09	SA2	16		7
12	Church	Attending Mass	CHU-10	SA2	2		2
12	Church	Attending Mass	CHU-11	SA2	4	25	2
12	Church	Attending Mass	CHU-12	SA2	3		3
12	Hospice	Attending Mass; Volunteering	CHU-13	OSA	12	10	4
12	Church	Attending Mass	CHU-14	OSA	7	19	2
				Total =	89		
<u>13</u>	Libraries		LIB		·		
13	Library	Visiting Library; Borrowing Books	LIB-01	SA1	4	10	2
13	Library	Visiting Library; Borrowing Books	LIB-02	SA1	9	13	4
13	Library	Visiting Library; Borrowing Books	LIB-03	SA2	13	13	8
				Total =	26		
<u>14</u>	Miscellaneous		REC				
14	Recycling Point	Doing Recycling	REC-01	SA2	2	2	1
	, ,			Total =	2		

This page has been left intentionally blank.